Control of Unknown Dynamical Systems: Robustness and Online Learning of Feedback Control

Thesis by Dimitar Ho

In Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy

Caltech

CALIFORNIA INSTITUTE OF TECHNOLOGY Pasadena, California

> 2024 Defended August 7, 2023

© 2024

Dimitar Ho ORCID: 0000-0002-7856-985X

All rights reserved

ACKNOWLEDGEMENTS

I extend my gratitude first and foremost to my defense committee members, who so patiently and helpfully guided this thesis through many rounds of feedback to its long-awaited end. Of course, this section cannot begin without first acknowledging the role of my official and original academic advisor, John Doyle. The extreme level of trust and independence he has afforded me over the course of my time at Caltech has enabled me to explore and pursue my most driving curiosities with unbridled freedom, and as such he has played a significant role in shaping my academic and personal growth.

I would next like to express my deepest gratitude to my unofficial advisor Adam Wierman, for his mentorship, guidance and encouragement during the last phase of my PhD. Without his unwavering support and kind advice, this dissertation would not have been possible. Adam has a superhuman level of efficiency in everything he does, yet at the same time is also very patient and warm; his empathetic excellence continues to inspire me professionally and personally.

Yisong Yue, I thank for his support, guidance and invaluable insight. As someone who entered grad school as a specialized control theorist, I could not have hoped for a more efficient, broad and yet cutting-edge exposure to the field of machine learning through Yisong's lectures, reading groups, and collaborative discourse.

To Joel Burdick, I express my gratitude for his welcoming warmth, support, encouragement and guidance. Joel's group always felt like a second home to me, and I always looked forward to his lunch meetings for the chance to learn more about robotics and make connections to my research. This time spent brainstorming and concepting research projects together has been a great source of inspiration.

Next, I extend my gratitude to the staff and administration of the CMS/CDS department at Caltech, in particular Maria Lopez, Monica Nolesco, and Diana Bohler for their patience, assistance, and resources throughout graduate school.

To Laura Flower Kim from the International Student Office, I would like to express my deepest gratitude for making Caltech, for us international students, feel like a second home. Laura is incredibly kind and caring, and has a contagiously positive and warm personality that makes anyone around her, especially us international students when we first arrived on campus, instantly feel welcomed and taken care of. I also want to thank my fellow Resident Associates and the Office of Residential Experience Team, for giving me the opportunity to be resident associate for the undergrad hovse Ricketts. This work experience has been invaluable and enriching to me for my personal and academic growth.

This thesis and my research journey would be nothing short of incomplete if it were not for amazing collaborators, miscellaneous mentors and dear friends. To Jing Yu, Hoang Le, Alex Pai, Carmen Amo Alonso, Jing Shuang (Lisa) Li and Emily Jensen: working with you all has been the most rewarding, the most fun and most memorable part of my PhD journey. To Jing, my closest collaborator (and former housemate!), I am immensely grateful for her unwavering support as a collaborator, teammate, and close friend; thank you for being you. I am also incredibly grateful to my buddy Alex Pai, to whom I owe my PhD Candidacy, and whose friendship I now mark as a turning point in my grad school journey. Since our first chance meeting over a pickup volleyball game, I've learned a lot from Alex, in particular what a near perfect research experience and collaboration feels like. But more so than his role in my research, I consider Alex one of my closest friends, and I thank him for being a positive and supportive force all throughout grad school. I also would like to thank Prof. W Huang for his mentorship and support. I am so grateful to have had the opportunity to work with and to learn from such amazing people.

I am immensely grateful to all the wonderful friends with which I got to share my time at Caltech and made it such an enriching and fun experience. I want to thank my fellow Doyle Group members, Jiexin Chen, Mandy Huo, Fangzhou Xiao, Jing Yu, Carmen Amo Alonso, Lisa Li, Natalie Bernat, Anish Sarma, Lauren Conger, Soo-Jean Han. I would also like to thank my friends from and outside campus, Kaylyn and Luke Bugbee, Gautam Goel, Andrea Coladangelo, Benson Christalin, Zeynep Turan, Utkan Candogan, Corina Panda, Armeen Taeb and others. Your friendship and camaraderie have made the PhD experience incredibly enjoyable and memorable.

I am eternally grateful to my golden-souled wife Nina, my anchor, adventuring buddy and best friend; I cannot imagine persevering without her by my side. In fact, I likely would not have gotten into Caltech (nor any of the other Phd programs I applied for) in the first place without her helping me day and night to prepare and submit my applications all those years ago - an unwavering and relentless level of support she continues to shower me with to this day. I also want to thank my loving in-laws, Karen and Gervais Tompkin, for their enduring and positive support. I am incredibly grateful to my parents for their immeasurable support and love, who have always strived to provide me with the best opportunities and encouraged me in whatever path I took. My father, My, and my mother, Emilia, are the most intelligent, kind, multi-talented and fascinating people I know and have served as a great source of inspiration and ambition throughout my life. My mother, Emilia, passed away in my first term of grad school, but her curiosity, intelligence, wisdom, and kindness continue to inspire me. Mother, this thesis is dedicated to you; you will always be my powerful and loving life force.

ABSTRACT

Over the past few decades, our physical and digital worlds have become increasingly intertwined and reliant on each other. Advancements in areas such as machine learning, online optimization, and control theory, along with ubiquitous access to computational power, have played a crucial role in this technological evolution. As a result, we are now moving towards a future where complex and intelligent dynamical systems, with humans in the loop, govern our daily lives.

Building advanced control systems is a critical step in this journey, as they enable swift and data-informed decision-making. However, as we aim to create even more sophisticated closed-loop systems, we must proceed with a careful balance of ambition and caution. While the benefits of these interconnected systems are abundant and our dependence on them deepens, ensuring the actual reliability and safety of the systems becomes increasingly challenging due to the growing complexity of their dynamics. This challenge is particularly prominent in safety-critical applications involving physical systems, which often have strict and non-negotiable safety and performance requirements. To establish a harmonious relationship between our physical and digital worlds, it is crucial to develop intelligent closed-loop control systems that are not only fast and efficient, but also reliable and fault-tolerant.

The title of this thesis, "Control of Unknown Dynamical Systems: Robustness and Online Learning of Feedback Control," reflects the central focus of this work on addressing this pressing challenge. The thesis aims to develop theoretical frameworks and tools that provide insights and contribute new approaches to the design of control systems capable of handling the inherent uncertainty in real-world dynamical systems.

The first part of the thesis focuses on the design of closed-loop systems that are robust to dynamic uncertainty, particularly in settings involving nonlinear dynamics and complex control constraints. The second part introduces a general framework for learning-to-control algorithms that provide worst-case guarantees, even in scenarios where the dynamic uncertainty is arbitrarily large. By addressing these key aspects, this work aims to advance our understanding and capabilities in designing control systems that can effectively deal with uncertainty.

PUBLISHED CONTENT AND CONTRIBUTIONS

 Dimitar Ho. A system level approach to discrete-time nonlinear systems. In 2020 American Control Conference (ACC), pages 1625–1630. IEEE, 2020. doi: 10.23919/ACC45564.2020.9147857. URL https://arxiv.org/abs/2004. 08004.

Dimitar Ho conceptualized the project, developed the theoretical results, conducted the experiments and simulations, and wrote the manuscript.

- [2] Dimitar Ho and John C. Doyle. Robust model-free learning and control without prior knowledge. In 2019 IEEE 58th Conference on Decision and Control (CDC), pages 4577–4582. IEEE, 2019. doi: 10.1109/CDC40024.2019.9029986. URL https://arxiv.org/abs/2010.00204.
 Dimitar Ho conceptualized the project, developed the theoretical results, conducted the experiments and simulations, and wrote the manuscript.
- [3] Dimitar Ho and John C. Doyle. Scalable robust adaptive control from the system level perspective. In 2019 American Control Conference (ACC), pages 3683-3688. IEEE, 2019. doi: 10.23919/ACC.2019.8814896. URL https://arxiv.org/abs/1904.00077. Dimitar Ho conceptualized the project, developed the theoretical results, conducted the experiments and simulations, and wrote the manuscript.
- [4] Dimitar Ho, Nikolai Matni, and John C. Doyle. Passive-aggressive learning and control. In 2018 Annual American Control Conference (ACC), pages 1887–1894, June 2018. doi: 10.23919/ACC.2018.8430904.
 Dimitar Ho conceptualized the project, developed the theoretical results, conducted the experiments and simulations, and wrote the manuscript.
- [5] Dimitar Ho, Hoang Le, John C. Doyle, and Yisong Yue. Online robust control of nonlinear systems with large uncertainty. In *Proceedings of The 24th International Conference on Artificial Intelligence and Statistics*, volume 130 of *Proceedings of Machine Learning Research*, pages 3475–3483. PMLR, Apr 13–15 2021. URL https://proceedings.mlr.press/v130/ho21a.html. Extended version on https://arxiv.org/abs/2103.11055,

Dimitar Ho led the conception of the project, developed the algorithm and theoretical results, conducted the experiments and simulations, and led the writing of the manuscript.

[6] Jing Shuang Li and Dimitar Ho. Separating controller design from closed-loop design: A new perspective on system-level controller synthesis. In 2020 American Control Conference (ACC), pages 3529–3534. IEEE, 2020. doi: 10.23919/ACC45564.2020.9147736. URL https://doi.org/10.23919/ACC45564.2020.9147736.

Dimitar Ho led the conception of the project, participated in developing the

theoretical results and algorithms, and helped with the writing and revising of the manuscript.

- [7] Carmen Amo Alonso, Dimitar Ho, and Jose M. Maestre. Distributed linear quadratic regulator robust to communication dropouts. *IFAC-PapersOnLine*, 53(2):3072–3078, 2020. doi: 10.1016/j.ifacol.2020.12.1012. URL \https://arxiv.org/abs/2103.03967.
 Dimitar Ho participated in the conception of the project, in the algorithm design and theoretical analysis, and helped with revising the manuscript.
- [8] Jing Yu and Dimitar Ho. Achieving performance and safety in large scale systems with saturation using a nonlinear system level synthesis approach. In 2020 American Control Conference (ACC), pages 968–973. IEEE, 2020. doi: 10.23919/ACC45564.2020.9147577. URL https://arxiv.org/abs/2006. 12766.

Dimitar Ho led the conception of the project, developed the theoretical results, participated in conducting the experiments and simulations, and in the writing of the manuscript.

[9] Jing Yu, Dimitar Ho, and Adam Wierman. Online adversarial stabilization of unknown networked systems. *Proceedings of the ACM on Measurement and Analysis of Computing Systems*, 7(1):1–43, 2023. doi: 10.1145/3579452. URL https://arxiv.org/abs/2203.02630.

Dimitar Ho participated in the conception of the project, in the design of the algorithm, in the development of the theoretical results, and in the writing of the manuscript.

TABLE OF CONTENTS

Acknowledgements		
Abstract		
Published Content and Contributions		
Table of Contents		
List of I	llustrations	xi
List of 7	Tables	xiii
Nomeno	clature	xiv
Chapter	I: Introduction	1
1.1	Definitions and Principles of Control System Design	2
1.2	Major Challenges and Mission Statement	7
1.3	Main Contributions of the Thesis	11
I No	nlineer Closed Leens and System Level Control	20
I INUI Chantan	Innear Closed Loop Mane from the Inside and Outside	20
	In: Nonlinear Closed Loop Maps from the Inside and Outside	23
2.1	Fundamentals of Operator Theory	23
2.2	Fundamentals of Operator Theory	21
2.5	Banaganting Dynamical Systems of Sata Delations and Mana	34 40
2.4	Representing Dynamical Systems as Sets, Relations, and Maps	40
2.5	Closed Lean Mana and Dealizing Controllers	43
2.0	Closed-Loop Maps and Realizing Controllers	45
2.7	Robust and Internal Stability of Closed-Loop	52
2.8	Nonlinear Closed Loop Maps of Linear Systems	61
2.9	Blending SL Controllers	65
2.10		68
2.A	Proofs	68
Chapter	III: Nonlinear Blending Approach to Control of Large-Scale Systems	/0
3.1		70
3.2	Discrete-Time Trajectory-Tracking Control for Nonlinear Continuous-	
		72
3.3	The Constrained LQR Problem	77
3.5	Distributed Anti-Windup Controller for Saturated Systems	83
3.7	Example	91
3.8	Conclusion	94
Chapter	IV: Robustness and Sensitivity of Certainty-Equivalent Adaptation .	96
4.1	Closed-Loop Dynamics in the Presence of CE-Adaptation	97
4.3	Certainty Equivalent Adaptation with LQ-Optimal SL Controllers	108
4.4	Digression on Linear Algebra and Control Theory Basics	115
4.5	Controllability and Observability at a Fixed Time	116

4.6 Perturbation Analysis of Open-Loop Maps over Finite Time-Horiz	zon	119
4.7 From \mathcal{H}_2 -Optimal Control to Least Squares		. 125
4.8 Representation as a Least-Squares Problem		. 126
4.9 Perturbation of Least Squares		. 127
4.10 Lipshitzness of \mathcal{H}_2 -Optimal CLM Parameterization		133

II Learning to Control Unknown Systems 130	6
Chapter V: Robust Model Free Learning and Control of Linear Systems 139	9
5.1 Introduction	9
5.2 Problem Setup	2
5.3 Preview of Main Result	2
5.4 Preliminaries of Convex Geometry	4
5.5 The Model-Free Adaptive Controller and Closed-Loop Equations 14	8
5.6 Main Results	0
5.7 Proving Closed Loop Stability	5
5.8 Proving Finite Occurrence of Unstable Transitions	7
5.9 A Connection Between Metric Entropy Bounds and Model-Free	
Stability Analysis	7
5.10 Simulation	9
5.11 Conclusion	2
5.A Proofs	3
Chapter VI: Robust Model-Based Learning and Control of Unknown Systems 17	6
6.1 Introduction	7
6.2 Overview of Approach and Main Results	2
6.3 Control Oracle: An Abstraction for Nominal Control Design 18	7
6.4 Robust Oracle Guarantees	9
6.5 Consistent Model Chasing	2
6.7 Main Results	1
6.8 Discussion and Extension of Main Results	9
6.9 Design Example: Control of Uncertain Scalar Linear System 21	4
6.10 Application to Uncertain Robotic Systems	7
6.11 Empirical Validation: Cart-Pole Swing-Up on a Constrained Track . 22	0
6.A Proofs	1
6.B Oracle for Cartpole-Swing up with Constraints	6
6.C Mistake Guarantees vs Sublinear Regret	9
Bibliography	2

LIST OF ILLUSTRATIONS

Number		Page
1.1	Deterministic problem formulation: Specifying a model class and	
	stating a hypothesis set of possible models	3
1.2	Overview of problem definition and different control design approaches	. 5
1.3	Overview of major challenges	7
1.4	Thesis overview	11
1.5	Part 1: Nonlinear system level approach	12
1.6	Chapter 6: PixSel Framework for adaptive systems analysis and	
	control design	17
3.1	Swing-up control for cart-pole system with system level controller at	
	30Hz sampling rate. $(x, \dot{x}, \theta, \dot{\theta}, f)$ stand for cart position / velocity,	
	pole angle / velocity, and cart force with units $(m, m/s, deg, deg/s, $	
	N). $\theta = 180^{\circ}$ stands for the upward pole position. The weight	
	of the cart and the pole is chosen as 1 kg and 0.1 kg, the length	
	of the pole is chosen as 0.5 m. Consult [122] for detailed system	
	description and equations. Left: Desired trajectory (red) vs. closed-	
	loop performance under initial condition error $\psi(0) = 45^\circ$ and two	
	scenarios of small (orange) and large (blue) system perturbations	
	w, v, d . <i>Middle</i> : Evolution of internal state \hat{w} for both scenarios	
	and normalized disturbance due to trajectory error \bar{e}_i . Right: i.i.d	
	Gaussian perturbations w, v, d	73
3.2	Performance improvement of nonlinear controller $\mathrm{SL}[\Psi^{*\mathrm{x}},\Psi^{*\mathrm{u}}]$ over	
	optimal linear controller $\mathrm{SL}[\mathbf{\Psi}^{\mathrm{x,lin}*},\mathbf{\Psi}^{\mathrm{u,lin}*}]$ for different variances σ^2	
	of the non-truncated disturbance. The nonlinear blending controller	
	synthesizes over 4 linear controllers w.r.t. to the projection parameters	
	$\eta_1 = 0.05, \eta_2 = 0.1, \eta_3 = 0.2, \eta_4 = \eta_{max} = 1 \dots \dots \dots \dots \dots$	84
3.3	Worst-case Response: The heatmaps show how a worst-case distur-	
	bance is propagated through space-time for the saturated chain system.	
	The integral controller becomes unstable due to saturation and the	
	naive blending controller possesses has the anti-windup property of	
	the non-integral controller.	94

3.4	Step disturbance rejection: Response to small step disturbances at
	node 8,10,12 entering at time 2,6,10, respectively. As in the scalar case,
	the proposed blending controller not only stabilizes under saturation
	but also recovers the performance objective of rejecting small step
	disturbances. This contrasts against the non-integral controller, which
	sacrifices small-signal performance for stability
5.1	Examples in \mathbb{R}^2 . Left: $c(W)$ and $ x _W$ for $W = \{\hat{w}_1, \hat{w}_2\} \cup \{w_t t \in \mathbb{N}\}$
	and $(w_t) = (w_0, w_1, 0, 0,)$. <i>Right</i> : $c(X_1)$ and $\ \cdot\ _{X_1}$ for $X_1 =$
	$[x_1, x_0, \hat{x}_2, \hat{x}_1]$. <i>Middle</i> : $\kappa_2 = \ W\ _{X_1}$ is the smallest factor r such that
	$c(W) \subset rc(X_1). . \ . \ . \ . \ . \ . \ . \ . \ . \ $
5.2	Geometry of the distance function $d(\cdot, \cdot; B)$: The points x and y
	satisfy the inequality $d(x, y; B) > r$ in $n = 2$, where B is taken as the
	two dimensional cube $B := \{(x_1, x_2) x_i \leq 1\}$ and $r = 1.3. \ldots 162$
5.3	$N(x;\varepsilon,B)$ is the intersection of $\{p \mid x _{B\cup p} \leq \varepsilon\}$ and $\{p \mid p _{B\cup x} \leq \varepsilon\}$
	and contains two translates of the set $(\varepsilon - 1)B$. In the picture, B
	is taken as the two dimensional cube $B := \{(x_1, x_2) x_i \leq 1\}$ and
	$\varepsilon = 1.6.$
5.4	The geometry of a $(\delta, c(X_{\tau-1}))$ -separated set $P = \{p_1, \ldots, p_5\}$. All
	sets are geometrically accurate, assuming $c(X_{\tau-1})$ is the box $[-1,1] \times$
	$[-1, 1]$ and $\delta = 1.6.$
5.5	Results of 1000 closed-loop simulations with random A_0^k , x_0^k and
	disturbances w_t^k drawn from $[-1,1]^3$. The plots on the left show
	the largest 1%, 10%, 50% percentile values of $\ x_t^k\ _{W^k}$, $\ u_t^k\ _{W^k}$, κ_{τ}^k ,
	$m(\kappa_{\tau})$. The right plot shows the same percentile values for the state
	x_t^k and input u_t^k measured in 2-norm
5.6	Results of 1000 closed-loop simulations with random A_0^k , x_0^k and
	$w_t^k = 0$. The plots on the left show the largest 1%, 10%, 50%*
	percentile values of $\ x_t^k\ _{W^k}$, $\ u_t^k\ _{W^k}$, κ_{τ}^k , $m(\kappa_{\tau})$. (* this percentile is
	too small to visualize for $ x_t^k _{W^k}$ and $ u_t^k _{W^k}$) The right plot shows
	the same percentile values for the state and input measured in 2-norm. 171
5.7	$ x_t _W$ and $ u_t _W$ trajectories for closed loop with uniform disturbance
	$w_{i,t}$ in $[-1, 1]$ and $x_0 = [0.2, 0, 0.1]^T$
6.1	PixSel Framework for adaptive control

LIST OF TABLES

Number		Page
6.1	Notions of oracle-robustness	. 190
6.2	Notions of oracle-robustness adjusted for finite-time guarantees	. 210
6.3	Fraction of experiments completing the swing up before time T : ideal	
	policy $\pi[\theta^*]$ vs. $\mathcal{A}_{\pi}(SEL)$. 221

NOMENCLATURE

Abbreviations

ℓ_p - fgs	ℓ_p -finite gain stable.
ℓ_p - ifgs	incrementally ℓ_p -finite gain stable.
M^{\dagger}	Moore-Penrose Inverse of matrix M .
cCLM	candidate Closed-Loop Map, see Definition 2.8.
CE	Certainty-Equivalence Principle.
CLM	Closed-Loop Map.
CLMEq	The Closed-Loop Map Equation $\Psi^{\mathrm{x}} = F^{+}(\Psi) + I$, see Theorem 8.
СМС	Consistent Model Chasing.
LQ	Linear Quadratic.
NCBC	Nested Convex Body Chasing.
ОСР	Optimal Control Problem.
SLS	System Level Synthesis.
Notation an	d Definitions for Part 1
$ imes_{i\in\mathcal{I}}\mathcal{S}_i$	The Cartesian product of a family of sets $\{S_i\}_{i \in \mathcal{I}}$ indexed over a countable index set \mathcal{I} .
$oldsymbol{\Delta}[oldsymbol{F}, oldsymbol{\Psi}]$	The Residual $\Delta[F, \Psi] := F^+(\Psi) + I - \Psi^x$, see Definition 2.10.
$\Phi_{ ext{CL}}^{ ext{w}\mapsto ext{xu}}[m{F}]$	The space of CLMs of F or the CLM-space of F , see Definition 2.6.
$oldsymbol{F}^+$	The operator $F^+ := S^+[F]$ resulting from right-shifting F .
F_K	Closed-Loop Dynamics operator, representing the mapping $x \mapsto F(x, K(x))$ for fixed operators F and K .
$P^{\mathcal{I}}$	Truncation operator mapping a sequence x to its restriction $x_{\mathcal{I}}$.
$oldsymbol{P}^t$	Truncation operator $\boldsymbol{P}^{\mathcal{I}}$, where $\mathcal{I} = [0, t] \subset \mathbb{N}$.
$x_\mathcal{I}$	Restriction of a sequence x to the interval $\mathcal{I} \subset \mathbb{N}$.
$CL_{[F,K]}$	Closed-Loop Model, defined in equation (2.31), representing the set of all closed-loop trajectories for dynamics F and controller K .

$\mathcal{X}^{\mathbb{N}}, \mathcal{X}^{\mathcal{I}}$	The space of all sequences $x : \mathbb{N} \mapsto \mathcal{X}$ and $x : \mathcal{I} \mapsto \mathcal{X}$.
$\mathcal{C}_s(\ell^{\mathcal{X}},\ell^{\mathcal{Y}})$	Space of all strictly causal operators $\ell^{\mathcal{X}} \mapsto \ell^{\mathcal{Y}}$.
$\mathcal{C}(\ell^{\mathcal{X}},\ell^{\mathcal{Y}})$	Space of all causal operators $\ell^{\mathcal{X}} \mapsto \ell^{\mathcal{Y}}$.
$\mathcal{LC}(\ell^{\mathcal{X}},\ell^{\mathcal{Y}})$	The subspace of linear operators in $\mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$.
$\ell^{\mathcal{X}}$	The vector space consisting of the set $\mathcal{X}^{\mathbb{N}}$ equipped with the standard operations of vector-addition and scalar-multiplication over \mathbb{R} .
$\ell_p^{\mathcal{X}}$	The subspace of ℓ_p -bounded sequences in $\ell^{\mathcal{X}}$.
$\ \! \! oldsymbol{A} \ \! _p^\star$	ℓ_p -induced operator pseudo-norm of A , as defined in (2.47).
$\overline{P^{\mathcal{I}}}$	Truncation operator $\overline{P^{\mathcal{I}}} := I - P^{\mathcal{I}}$.
$\operatorname{SL}[{oldsymbol{A}},{oldsymbol{B}}]$	System Level Controller, the dynamical system corresponding to a system level implementation of the operator $B(A)^{-1}$, see Definition 2.7.

Notation and Definitions for Part 2

$oldsymbol{K}^{ ext{cc}}$	Causal Cancellation Controller, defined as in (5.4).
$\mathrm{c}(S)$	The absolute convex hull of a set S, see Definition 5.4.
$\ \cdot\ _{S}$	The unique norm whose unit normball is $\operatorname{c}(S);$ see Definition 5.5 .
π	A control-policy oracle.
SEL	A consistent model selector.

INTRODUCTION

Over the past few decades, the integration of our physical and digital worlds has deepened, with advances in machine learning, online optimization, and control theory playing pivotal roles. This progress, coupled with widespread access to computational power, is propelling us towards a future where interconnected smart systems, with humans in the loop, play an integral part in our lives. Understanding this complex closed-loop system and ensuring its reliability and safety, especially in safety-critical settings involving physical systems, is a paramount contemporary challenge. From a system engineering standpoint, designing control systems capable of operating in complex and unpredictable dynamic environments is an especially difficult problem. This naturally leads to a pertinent question regarding robust system design:

How do we design control systems for real world systems in one-shot, which can handle large model uncertainty online, and with worst-case guarantees?

Indeed, the central focus of this thesis seeks to address this pressing question, and contributes new approaches to the design of control systems that are capable of handling the inherent uncertainties that come with real-world dynamical systems. Part 1 of this thesis focuses on outlining a novel method of designing closed-loop systems that are robust to dynamic uncertainty, especially in settings involving nonlinear dynamics and complex control constraints. Part 2 introduces a general framework for learning-to-control algorithms that provide worst-case guarantees, even in scenarios with arbitrarily large dynamic uncertainty. By addressing these dually-important aspects, this work aims to advance our understanding and capabilities in designing control systems that can autonomously and effectively deal with uncertainty.

If you would be a real seeker after truth, it is necessary that at least once in your life you doubt, as far as possible, all things.

- René Descartes: Principles of Philosophy

1.1 Definitions and Principles of Control System Design

In this section, we lay the groundwork by outlining fundamental concepts and terminologies, and offer a review of current control design methods. We differentiate these methods into two categories, namely "episodic" and "one-shot" strategies.

Every real-world control design problem begins merely with an unknown dynamical system and the aspiration to control it. As we venture into this, an almost philosophical question arises:

"How do we rationalize our certainty and uncertainty about the unknown?"

On the one hand, if we believe that the nature of the unknown system and the environment is predominantly random, then it would make sense to proceed with a probabilistic problem setup, i.e., viewing system and environment behavior as the realizations of random processes. That being said, it is oftentimes warranted to be more skeptical and prefer a more falsifiable, deterministic problem setup. In this dissertation, we advocate for a deterministic view. Such perspective characterizes a control problem using three fundamental definitions: 1. Closed-Loop Complexity: The level of detail and class of models with which we want to describe the closed-loop system. 2. Model Hypothesis: Formulating a set of possible models. 3. Control Objective: Paraphrasing our objectives and requirements in control-theoretic terms. We next discuss these terms in detail.

Deterministic Formulation of a Control Problem

A deterministic setup consists of constructing a set of candidate models and phrasing our objectives as worst-case guarantees. Step one involves defining observation and action spaces, then choosing a representation to describe the closed-loop system. For example, one needs to choose whether to model the system as a linear or nonlinear dynamical system, how best to represent interactions with the environment, and how to impose constraints on the structure of the closed-loop system (i.e., constraints that may arise due to technological limitations: restricted and delayed communication, limited and distributed computational resources, limited sensing and actuation capabilities, etc.).

This first step in our decision-making process determines the level of *model complexity*. Essentially, we're choosing how detailed we want our representation to be of the system's behavior, its interaction with the controller, and the environment. This choice will set the complexity level for the entire closed-loop system dynamics.

Concurrently, or subsequently, we postulate our *model hypothesis*. This hypothesis comprises a collection of models, within which we expect at least one model to accurately depicting the dynamics of the unknown system. The size of this set mirrors the level of uncertainty (or confidence level) concerning our model hypothesis.



Figure 1.1: Deterministic problem formulation: Specifying a model class and stating a hypothesis set of possible models.

These two steps, choosing the model complexity and making a hypothesis, are illustrated in Figure 1.1. The box Ω represents a set of models forming our model hypothesis. The vertical axis indicates the level of complexity intrinsic to each of the models in the set, whereas the width of the box depicts the size of our set Ω , i.e., the breadth of uncertainty of our model hypothesis.

Note that determining whether or not our uncertainty is large or small can only be achieved once our control objectives are fully articulated. Flying a plane stably through turbulence, operating a manufacturing plant at peak efficiency, managing power distribution in a smart grid; many real world objectives and requirements can be phrased in terms of closed-loop stability conditions which should remain true even in worst-case scenarios. This type of control objective is often referred to as *worst-case closed-loop guarantees*. Some of the most common examples, which we will focus on in this work, are listed below:

- ℓ_p -(*internal*)-stability: A closed-loop system is ℓ_p -internally stable [49, 77, 139, 142] if its trajectories change continuously¹ under ℓ_p -bounded perturbations; that is, by adding small ℓ_p -bounded noise to every observed, controlled or computed signal.
- <u>Worst-Case Safety-Guarantees</u>: Most practical safety requirements are stated as verifiable and measurable conditions that should either always or never hold. Very often, this can be formally translated into some ℓ_{∞} -stability related condition [8, 45, 53], which can be expressed qualitatively as the following informal inequality

 $\max_{t} \{ \text{Risk Score } \mathcal{R}_t \text{ at time } t \} \leq \text{Safety-Threshold},$

describing a worst-case ℓ_{∞} -bound on some scalar sequence $\{\mathcal{R}_t\}$ quantifying the risk or "unsafety" of the system at each time.

• <u>Worst-Case Cost-Performance Guarantees</u>: Many meaningful performance metrics can be phrased in terms of some cumulative sum of cost functions over time. Qualitatively, performance guarantees can be represented as an inequality of the type:

Total Cost = $\sum_{t} \{ \text{Cost } C_t \text{ incurred at time } t \} \leq \text{Worst-Case Bound} .$

Guarantees of the above form can often be related to notions of ℓ_p -stability [49, 68, 85, 142]² by expressing performance guarantees as worst-case ℓ_p -norm bounds on some suitable scalar cost sequence $\{C_t\}$.

In summary, our ground zero for control system design is a deterministic problem formulation, which entails choosing a model representation of our closed-loop, stating our model hypothesis Ω , and translating our control objectives into suitable desired worst-case guarantees. Next, we review existing schools of thought and methods from literature. As illustrated in Figure 1.2, our overview will categorize literature by whether an episodic, "learning-then-control"-type, problem setting is assumed, or a one-shot problem setting is allowed.

¹The stronger Lipshitz-continuity property corresponds with the definition of ℓ_p -finite-gain stability, and the Lipshitz constant is called ℓ_p -gain.

²Particularly common are parallels to ℓ_1 - and ℓ_2 -stability.



Figure 1.2: Overview of problem definition and different control design approaches.

One-Shot Approach to Control System Design

In the one-shot approach, we use the model hypothesis Ω and specified control objectives as our basis, and, without further input, task ourselves with designing a control algorithm that can be directly deployed on the unknown system; in other words, our goal is to design an algorithm capable of handling model uncertainty and feedback control decisions autonomously. In the classical control literature, this problem setting is the core subject of the fields commonly referred to as robust control theory (with the caveat that model uncertainty must be relatively small) [49, 141, 142] and adaptive control theory [15].

If the level of model uncertainty is *small* relative to our control objectives, we can apply methods of robust control theory literature. In robust control, synthesis processes primarily use the same internal structure or architecture of the feedback controller as in nominal control design. However, the feedback control law is developed to meet our control objective concurrently on all potential models of our assumption. Of course, these robust control design methods are only viable when the level of uncertainty is sufficiently small.

After a certain size of uncertainty, i.e, a certain set size of our model hypothesis, we have to drastically change the internal structure and architecture of our control algorithm, for it has to allow for incorporating online-learning in the closed loop. This is the canonical problem setting of the domain known as adaptive control [14, 69, 112].

Episodic Approach to Control System Design

If we have access to a controlled environment in which we can perform experiments and measurements of the unknown system (such as for the purpose of recording system trajectories, etc.), we can resort to an alternative approach to control design

that leverages a two step process rather than just "one shot": a learning phase, and a controller synthesis phase. In the learning phase, we follow a process commonly known as system identification [16, 88], in which we perform repeated experiments, collect data, and apply learning algorithms to reduce the uncertainty of our initial model hypothesis Ω . Sometimes experiments are performed in closed-loop with a stabilizing feedback controller (assumed given as part of the problem formulation); this is referred to as closed-loop system identification [89] and is in general more difficult than open-loop system identification due to the distributional shift caused by the coupling of state and input signal. Particularly in the case of closed-loop system identification, it can also make sense to repeat the data collection and estimation process several times, wherein each round (also called *episode*), the experiment and stabilizing controller are redesigned based on the most recent data. The learning phase concludes once the system identification has made sufficient progress such that a new model hypothesis $\Omega' \subset \Omega$ can be formulated (refer to Figure 1.2) with minimal uncertainty. At this stage, we can synthesize the desired controller by choosing an arbitrary model within Ω' and designing a nominal controller specifically for it.

The episodic control design approach also commonly arises in model-based reinforcement learning [24, 57, 108]. Many modern learning and control algorithms also follow similar learning-then-control design approaches [3, 4, 62, 115].

Fundamental Prerequisites and Limitations to Episodic Design Optimally, the episodic control design is favored or paired with a single-shot method. However, the conditions for system identification may be too time-consuming or unattainable in reality. This includes the prerequisite of a strictly governed lab setting, essential for conducting safe experiments and affording the flexibility to pause or terminate the system when required. Particularly in real world applications, characterized by complex dynamical systems and uncertain environments, the former renders an episodic design approach impractical. Two main factors contribute to the impracticality of an episodic design approach in such settings. Firstly, system identification becomes quickly intractable for complex systems, particularly ones with nonlinear dynamics. Additionally, the potential for large model uncertainty in real-time operations necessitates the flexibility to adapt, further complicating the application of an episodic design approach.

An additional challenge is a fundamental theoretical obstacle inherent to the deterministic framework. Within episodic control design, worst-case error bounds on the complete system identification process are mandatory to ensure worst-case guarantees for the final control algorithm. However, achieving this is an intractable issue even for simpler linear systems [41]. This difficulty is evident in existing literature: probabilistic approaches offering closed-loop guarantees (stated with high probability) are plentiful [1, 28, 43, 50, 51]. Yet, deterministic approaches providing worst-case guarantees are scarce [35].

1.2 Major Challenges and Mission Statement

The main mission motivating this work is studying control design that can reliably and autonomously handle the complexity and large uncertainty inherent to real world dynamical systems. Tackling such a problem naturally calls for a one-shot design approach, and specifically, finding answers to the following central question:

How do we design control systems in one-shot, which provide worst-case guarantees in presence of large model uncertainty and complex closed-loop dynamics?

At its core there are three intertwined aspects to the problem: Complex Constraints and Dynamics, Worst-Case Guarantees, and Large Model Uncertainty. These issues present two critical challenges existing literature has yet to adequately address.



Figure 1.3: Overview of major challenges.

Challenge 1: Robust Control Theory for Complex Closed-Loop Dynamics

As alluded to previously, stability analysis and control design for nonlinear and complex systems is a core problem to consider.

Nonlinear dynamics in the system or controller result in overall nonlinear closed loops and require analysis and synthesis from the nonlinear control theory literature. Nonlinear stability analysis is based on some fundamental results in the classical literature [77, 101, 111, 117, 124], such as Lyapunov functions, passivity, the small-gain theorem, barrier functions [6], Poincaré maps, contraction analysis [90], and describing functions. Among these, the Lyapunov methods are by far the most widely used technique.

Methods for nonlinear control design can be separated into two groups: analytical and computational. Lyapunov stability analysis and the optimal control formalism [77, 85] have inspired most classical analytical methods such as feedback linearization, backstepping, sliding-mode control, and gain scheduling [77, 111, 117], among many others. With the technological breakthrough of computers in the 90s and the rapid advancement in computational capabilities and numerical optimization since then, computation-based control design approaches have also gained dominance. Prominent examples of such methods are Model-Predictive Control [30, 95] and reachability/viability-based approaches [20]. Another important line of work started with the sum-of-squares method (SOS) developed in [99, 103], which made it possible to compute Lyapunov functions through convex optimization and inspired new synthesis methods such as [104, 105, 133].

However, a common limitation in almost all approaches is that they do not scale well with system and controller complexity. We consider the following three aspects to be particularly important:

- 1. High-Dimensional Systems: Most optimal control or SOS-based synthesis procedures [19, 30, 67, 84, 85, 99, 103, 104] require solving (sometimes repeatedly [30]) optimization problems that are only tractable for small system dimensions.
- 2. Complexity of Controller Constraints: Many of the above synthesis methods cannot incorporate additional constraints on the controller. As mentioned above, technical limitations in practice can require us to impose structural constraints on the implementation of the realizing controller. These constraints are particularly common in large-scale systems, and some important ones include communication constraints between components used for sensing or actuation, limited actuation caused by saturation, and available computational resources that are possibly distributed.

3. Complexity of Controller Dynamics: Both stability analysis and control synthesis in a nonlinear setting primarily assume that the chosen controller is static or has a simple internal structure, such as an added state observer. Aside from a few exceptions (often analytical methods such as backstepping), many of the mentioned synthesis procedures, particularly computation-based ones, do not extend past static controllers. This is partially due to the fact that stability analysis of nonlinear closed-loop dynamics and controllers with rich internal structures is not well-studied. However, analysis and synthesis methods that are compatible with more complex and nonlinear control structures are increasingly needed in many modern control applications, especially in the rapidly growing area of learning and control [3, 5, 9, 37, 42, 59, 115]. In such problem settings, it is not unusual for controllers to have high-dimensional and complex internal dynamics. A common cause for this is that controller implementations require continuous and iterative solving of multiple (possibly interconnected and layered) tasks in closed loop, which potentially generate high-dimensional internal states with dynamics; such tasks include planning, machine learning, and online optimization.

Until recently [126], even for linear time-invariant systems, control design was challenging for the above problem settings. However, the system-level approach, as introduced in [126], enabled new efficient controller synthesis methods [66, 127] that allowed for localized, distributed, and scalable control design in large-scale linear systems. This was achieved by transforming constrained optimal linear control problems into convex optimization problems over achievable closed-loop maps that can be solved efficiently. A key component of the system-level synthesis (SLS) procedure is that once we have solved for the desired closed-loop map, there is a simple way to construct a controller that stably realizes this on the system.

We endorse the central idea of the system-level approach and use it in Part 1 of this thesis as our starting point, then go further to introduce a new framework for nonlinear control design and system analysis dedicated to addressing the aforementioned additional challenges.

Challenge 2: One-Shot Worst-Case Guarantees vs. Large Model Uncertainty

One-shot control design, in the case of arbitrarily large model uncertainty, naturally has to embed some kind of online learning process into the closed-loop system, i.e., data collection and inference is performed in closed loop, at the same time as the system is being controlled. This design approach is commonly taken in modern "learning and control" literature [1, 3, 4, 37, 43, 58, 62, 115] and in the more classical "adaptive control" literature [14, 15, 52, 69, 71, 72, 83, 112, 121, 132]. Exactly how these two fields integrate online interference into the closed-loop, however, remains quite different.

One-shot control design approaches in modern learning and control literature focus exclusively on problem settings wherein the system is either *perfectly* known [3, 4, 37, 58, 62, 115], or the model hypothesis enjoys small uncertainty. Moreover, said literature primarily focuses on the task of regret-optimal (with respect to some cost function) online control [1, 3, 4, 37, 42, 42, 43, 51, 62]. Furthermore, aside from a few exceptions ([28, 76] for example), the online learning for control literature mostly focuses on linear systems [35, 43, 62, 116] and convex cost functions. Linear-quadratic (LQ) costs have been of particular interest: [1, 2, 42, 43, 62] study algorithms with sublinear regret bounds, while [58, 115] present online control algorithms with competitiveness guarantees. However, compared to the adaptive control literature, even for linear systems, modern learning and control approaches that provide worst-case guarantees in the large model uncertainty setting are almost non-existent ([35]³).

In contrast, adaptive control algorithms are primarily focused on providing stability guarantees in the presence of large model uncertainty, and less on cost performance guarantees. Despite a long history dating back to the 1960s [23], the field of adaptive control has not matured as much as other areas of control literature. From a conceptual point of view, the central problem of adaptive control, that is robustness to large model uncertainty, is clearly relevant and of interest in almost any control application, and one would naturally expect that methods and principles of adaptive control can be applied in a way that builds on top of existing control methods of other areas of control; however, this is not the case. Despite some efforts to address this [8–10], there is still no unifying framework that allows us to bridge this gap. As an example, theoretical tools, particularly concerning robustness and stability analysis, which are considered standard in other areas of control [6, 90, 139, 141], cannot be applied (or have no analog) in the adaptive control literature. In particular, theory and design methods which come with safety and cost performance under adversarial disturbances and noise, are almost non-existent ([68]⁴ is a rare exception).

³Proves sublinear regret and stability guarantees for a linear system with large model uncertainty.

 $^{{}^4\}mathcal{L}_1$ -adaptive control provides ℓ_∞ -stability guarantees for linear systems and a small class of nonlinear systems

Reviewing the current literature, old and new, we are still very far from a unifying theory for designing control systems with worst-case guarantees in the presence of large model uncertainty. Part 2 of this thesis is dedicated to closing this gap (at least partially) by introducing new theory and frameworks for control design.

1.3 Main Contributions of the Thesis

In this thesis, we aim to address the previous challenges by investigating the problem of one-shot control design from two complementary perspectives.



Figure 1.4: Thesis overview.

In Part 1, we restrict ourselves to the small uncertainty setting and focus on developing new theory and methods for robust control design and system analysis of closed-loop systems with nonlinear dynamics, complex constraints and complex internal structure. In the Part 2, we develop new theory and algorithms for the large model uncertainty setting. In particular, we introduce a first general and modular one-shot control design framework ("PixSel") with worst-case cost- and safety performance guarantees in the presence of arbitrarily large uncertainty in our model hypothesis. The following sections will delve into these topics, shedding light on our novel contributions to this emerging interdisciplinary intersection.

Synopsis of Part I: Nonlinear Closed Loops and System Level Control

In Part 1 of the thesis we introduce a new framework for nonlinear control design and system analysis dedicated to addressing the first set of challenges of Section 1.2. In Chapter 2, we show that for a very general class of nonlinear systems, there is a universal connection between the closed-loop map, the external representation of a closed-loop as a map between input and output sequences (see Figure 1.5), and the operators corresponding to the feedback controller and system dynamics, which describe the behavior of the closed-loop from an internal point of view. We discuss important equivalence relations between different representations of the closed-loop system via a set of characterizing operator equations. These equations and their solution space provide useful insights for system analysis and control design, which form the foundation of the *nonlinear system level approach*.



Figure 1.5: Part 1: Nonlinear system level approach

The solutions, called closed-loop maps (CLM) of the characterizing operator equations, lead to concrete implementations of the realizing controller. This enables us to reformulate control design problems equivalently in terms of CLM operators, and realize feedback control as a structured dynamic system parameterized by the desired CLM operator. We refer to this special control structure as the "system-level" implementation, which turns out to offer more benefits than its intended original purpose. In fact, parameterizing a system-level implementation with approximate solutions of the operator equation provides a stable closed-loop system, as long as the approximation error is small enough in an appropriate sense. This leads to a result for robust stability analysis of nonlinear closed-loop systems, which is both instructive and can be leveraged for robust controller synthesis.

We then proceed to investigate closed-loop maps in the special setting where we have a linear system and nonlinear dynamic control. We define a particular class of "blended" closed-loop maps, which are particularly suited for this problem setting, and offer new insight into the dynamics of such closed-loops. The corresponding system-level implementations of these blended closed loops lead to a promising synthesis approach, which is capable of "blending" multiple linear controllers into one nonlinear controller, thus allowing us to combine desirable properties of each individual component.

In Chapter 3, we explore some first implications of these results, such as robust discrete-

time trajectory tracking controllers for continuous-time nonlinear systems. Second, we investigate some first application scenarios, namely, distributed constrained LQR and distributed anti-windup control, where this technique naturally provides significant benefits in the large-scale system setting over existing methods. We describe a synthesis procedure for blended SLS controllers that outperforms any optimal linear controller for the constrained LQR problem [36, 86, 91, 140]. We then discuss how the blended SL approach provides a natural remedy for controller windup in a way that is easily scalable for use in large-scale control systems. We discuss the efficacy of the methods with simulations and show that synthesis and implementation enjoy the same benefits as previous SLS synthesis methods: both are distributed, handle delays, sparse actuation, and allow for localized disturbance rejection.

In Chapter 4, we begin to shift our focus towards the problem of learning-to-control in closed-loop; the core topic of the second part of the thesis. We investigate a common feature found in many learning and control algorithms across various problem settings: At each time step, we estimate a model of the system dynamics and then switch to a control law designed for that model; i.e., we pretend to have found a model that will remain accurate going forward. This design approach is often referred to as "certainty equivalent" control.

We show that this general design principle imposes a certain structure on the closed-loop maps, which we can leverage for stability analysis and algorithm design. This leads to a sufficient condition for closed-loop stability, which admits a natural interpretation within our framework. Decomposing the dynamics of the lumped disturbance reveals that three factors are important for robust and stable adaptation in closed-loop:

- 1. Model Sensitivity of Nominal Control: Small changes in models should cause small changes in nominal controllers (the control law selected for each model).
- 2. Consistency of Models: Models should be consistent (or consistent up to some bounded error) with our observations.
- 3. Select Models Efficiently: The posited model should change as little as possible.

This observation is a manifestation of a deeper concept, which we explore in Part 2, Chapter 6 of the thesis. Learning-to-control can be decomposed into control

via *Robust Oracle Design* (addressing factor 1) and learning via *Consistent Model Chasing* (addressing factors 2 and 3). This idea serves as the foundation for many of the results in Part 2, where we develop the theory and framework for learning to control largely uncertain dynamical systems with guarantees.

In the second part of the Chapter 4, we explore applications of this result for problem settings concerned with online learning of optimal controllers. To this end, we focus on the setting of linear time-invariant systems and linear-quadratic costs; a problem setting that has received immense recent attention in the learning and control literature.

Guided by the theoretical findings in Part 1, we follow the principle of certainty equivalence to design a learning-to-control scheme with LQ-optimal system-level controllers as our basis of nominal control laws. We perform perturbation analysis of the solutions for the LQ-optimal control problem and show that the solutions are Lipschitz-continuous with respect to changes in the system matrices of the linear system (over parameter sets of equal degree of controllability). This partial result in itself is new and characterizes the sensitivity of LQ-optimal closed-loop maps for linear time-invariant (LTI) systems in terms of system-theoretic properties such as controllability and observability. With this result, we analyze the closed-loop stability of the learning-to-control scheme and provide conditions for model selections that are sufficient for closed-loop stability. We revisit these conditions again in Part 2 in the context of consistent model chasing.

In summary, Part 1 of this thesis discusses new fundamental connections between closed-loop maps, realizing controllers, and system-level implementations that open up new possibilities for the analysis and design of closed-loop control systems, allowing for a broad range of nonlinear dynamics in the system and controller. The strength of this approach, compared to existing ones, is that it is particularly well-suited for the handling of complex systems, such as large-scale systems, as well as complex control structures caused by high-dimensional internal dynamics (such as learning and optimization in closed-loop) and/or restrictive implementation constraints (spatially and temporally structured sensing, actuation, communication, and computation constraints). This has proven to be a catalyst for new control approaches, some of which we outline in the next section.

Impact, Current Work and Outlook. The results presented in Part 1 of the thesis were inspired by our earlier work [66], which for the first time, extended the

SLS-theory to the linear time-varying case and noticed its importance for online learning of feedback control: [66] introduced a framework for distributed adaptive control via system-level controllers, which allowed for incorporating online learning in large-scale closed-loop systems with delay and communication constraints. In later work, we leverage some of the ideas in [7] to develop sub-optimal adaptive system-level controllers that are robust to communication dropouts. With [1], we were the first to extend the system level analysis approach to general nonlinear systems, and then demonstrated its use-case for nonlinear control design in the large-scale system setting in [8]. [8] uses the CLM blending approach to formulate a nonlinear control synthesis method that outperforms linear controllers for the constrained LQR problem; notably, CLM blending is also shown to provide a solution for distributed and localized anti-windup. In [6] we introduced a new way for robust control design through a decomposition into a problem of CLM design and robust controller implementation. The theory developed in [1] and presented in Chapter 2 has served as a foundation for new results on nonlinear controller synthesis: [39] presents synthesis methods for nonlinear system-level control for dynamical systems with polynomial dynamics. [56] applies the theory to provide a characterization of the space of nonlinear stabilizing controllers and introduces a new framework, called neural system-level synthesis, to learn stabilizing nonlinear system-level controllers via neural networks.

Synopsis of Part II: Learning to Control Unknown Systems

Part 2 of the thesis focuses on addressing the second set of challenges, a conundrum we referred to as "One-Shot Worst-Case Guarantees vs. Large Uncertainty" earlier in Section 1.2.

In contrast to Part 1, which focuses on the "small uncertainty" setting, Part 2 focuses on the more general "large uncertainty" scenario, where the set of possible models is bounded (in an appropriate sense), but is allowed to be arbitrarily large. Thus, in a practical sense, we assume that the dynamics of the system are almost entirely "unknown".

Chapter 5 explores possible trade-offs and difficulties associated with our overall objective of handling completely unknown systems. It is clear that with larger model uncertainty comes degradation in robustness; hence, there has to be a limit at which model uncertainty renders possibly desired guarantees impossible to achieve. Surprisingly, there are no clear and general answers to this problem, even for simple

linear systems. In Chapter 5, we focus on a simple class of linear systems and explore the following questions:

At what size of model uncertainty does "learning-to-stabilize" become an impossible task? Is there a minimal model assumption?

The results presented in Chapter 5 are based on the work published in [65] and develop new methods for stability analysis and control design without the need for a model. In particular, we demonstrate the first instance of an all-model-free formulation of controller, closed-loop dynamics, and robust stability analysis. We end up with a positive answer for the considered class of systems. We present a simple model-free control algorithm that is able to robustly learn and stabilize an unknown discrete-time linear system with full control and state feedback subject to arbitrary bounded disturbance and noise sequences. The controller does not require any prior knowledge of the system dynamics, disturbances, or noise, yet it can guarantee robust stability and provide asymptotic and worst-case bounds on the state and input trajectories. To the best of our knowledge, this is the first model-free algorithm that comes with such robust stability guarantees without the need to make any prior assumptions about the system. Moreover, the theory and results developed provide a first set of tools that allow for an entirely model-free formulation of controller, closed-loop dynamics, and robust stability analysis. The simulation results also show that despite the generality and simplicity, the controller demonstrates good closed-loop performance: fast convergence, small learning transients, and almost optimal asymptotic gain.

Investigating this problem led us to a new convex geometry-based approach towards robust stability analysis, which served as a key enabler in our results. A distinguishing feature of the approach is that stability conditions can be phrased entirely in terms of data and there is an intuitive geometric way of quantifying and analyzing model uncertainty, which can be phrased in terms of metric entropy, the absolute convex hull of the observed data, and the disturbance. This perspective allows us to conduct stability analysis independent of the system matrix and the size of the disturbance and noise. The key idea that led to these results was to describe the reduction of uncertainty obtained from new observations in a set-theoretic way and study the geometric relation between the data and the uncertainty. The insight gathered from this approach inspired us to rethink the general problem of learning to control as a whole from a new angle, which led us to a framework introduced in Chapter 6.



Figure 1.6: Chapter 6: PixSel Framework for adaptive systems analysis and control design

In Chapter 6, we introduce a new modular framework for one-shot control design that is particularly well suited for learning-to-control problem settings that require robust safety and cost performance guarantees in the presence of potentially large model uncertainty. Our approach is based on decomposing the problem into two sub-problems: online learning, named "consistent model chasing", and the underlying control problem in the absence of model uncertainty, called "oracle design". Each of these sub-problems can be addressed separately, and their solutions (a control oracle and a model chaser) are used to instantiate a certainty-equivalent learningto-control scheme. This scheme, in a symbiotic way, inherits both control- and learning-theoretic guarantees, certifying the robustness of the closed-loop, even for large model uncertainty in the system dynamics. The range of closed-loop guarantees that we can obtain through this inheritance is fairly broad. In this chapter, which is based on the work in [5], we present worst-case performance guarantees in terms of $\{0, 1\}$ costs (commonly referred to as finite mistake or mistake bound guarantees in online learning) and represent worst-case safety guarantees as bounds on the ℓ_{∞} -norm of the trajectories of the closed-loop. As we show later, this way of phrasing closed-loop "safety" and "performance" is expressive enough to represent many other guarantees of stability, convergence, or even set invariance.

A defining feature of this new approach is the inheritance relation between the learning-to-control algorithm ("PixSel") and its instantiating sub-procedures, which builds a new bridge between the fields of online learning and robust control theory. This allows us to rigorously merge online learning with traditional control algorithms

for the purpose of learning to control uncertain dynamical systems.Moreover, it opens up an important connection between two, so far mostly separated, problem settings in control theory: small model uncertainty and large model uncertainty. As discussed earlier, the classical (adaptive control) and modern (learning and control) literature do not provide a simple way to scale robust controllers designed for the setting of small uncertainty to the more general setting of large model uncertainty. Within the "PixSel" framework, we accomplish this by merging the controller with a consistent model chaser, thereby extending the robustness guarantees of the original controller to the large-model uncertainty setting. As far as we are aware, there are no other existing approaches that enable this in such a general and straightforward way.

As an instructive example and to demonstrate the practicality of our approach, we show how to instantiate this framework for general robotic systems for common tasks such as stabilization or trajectory tracking. In addition to providing theoretical guarantees, empirical results show that our framework is a promising approach to designing efficient algorithms for learning and control in practice. We apply our approach to the problem of swinging up a cartpole with large parametric uncertainty in a realistic and highly challenging setting and show that it consistently achieves good performance over 900 experiments with different parameter settings. Despite its popularity, the cart-pole swing-up problem presents many fundamental challenges (underactuated, non-minimum phase, nonlinear dynamics) for control design, and the majority of existing design approaches in the adaptive control literature [73, 83, 117] are not applicable. There does not seem to be any empirical evidence of other design methods capable of tackling this learning-to-control problem in a large-model uncertainty setting.

In conclusion, the main motivation behind the research presented in this part was to establish a theoretical foundation for a systematic approach to online learning of feedback control in the presence of uncertainty and constraints. The objective was to develop a theory for "Online Learning of Feedback Control with Robustness to Large Uncertainty." Chapter 5 introduced new theoretical tools for a model-free approach to the problem, formulating controller synthesis and analysis in a model-free and model-agnostic way.

Chapter 6 addressed the core problem of learning to control unknown systems from a model-based perspective, introducing the "PixSel" framework. This framework modularly combines online learning and control design methods to provide robust safety and cost performance guarantees in the presence of large model uncertainty. The framework establishes an inheritance relation between online learning and control procedures, merging the fields of online learning and robust control theory. It also bridges the gap between small and large model uncertainty settings, providing a straightforward way to extend robustness guarantees to the more general setting.

Impact, Current Work and Outlook. The theory developed in Part 2 has served as a foundation for recent progress in this problem space. The work of [135] applies the PixSel framework to provide an efficient solution to the problems of robust voltage control under uncertain grid topology. In [9], we adapt the PixSel framework for the distributed case and combine it with the system-level approach to provide the first distributed, localized adaptive control approach, both in learning and control, that provides worst-case safety and stability guarantees for arbitrarily large bounded uncertainties. The algorithms which we presented in [9], form a symbiotic marriage of the theoretical results developed in Chapter 6 and Chapter 4 and provide a tremendous improvement over our early adaptive system-level control framework [66] in terms of scalability (more efficient distributed learning), computational cost ([66] relied on solving a robust optimization problem, which, compared to [9], scales very poorly with local model complexity), generality, and guarantees ([9] provides worst-case stability bounds for arbitrarily large uncertainty, while in [66] we can only provide guarantees for the small uncertainty setting).

More recently, the work presented in [138] applies the PixSel framework to the special case of LTV systems and develops an approach to online stability of unknown linear time-varying systems by reformulating the corresponding problem of consistent model chasing into a convex body chasing problem.

Part I

Nonlinear Closed-Loops and System Level Control

OVERVIEW

In this part of the thesis, we introduce a new operator-theoretic framework for the design and analysis of complex closed-loop systems. This framework allows for general nonlinear dynamics in the plant and controller, robust control synthesis and stability analysis, and complex constraints on the controller implementation. We refer to this framework as the *nonlinear system-level approach*. Chapter 2 introduces this theory, while the following two chapters demonstrate how this theory opens up new methods for control synthesis and system analysis.

Chapter 2 is based on the work presented in [64] and begins with an introduction to the required mathematical concepts from operator theory. It also reviews old and new lemmas for operator stability analysis. The first core result of this chapter characterizes the space of achievable closed-loop systems once feedback control is introduced to a given nonlinear system F. By representing closed-loop systems as nonlinear causal operators mapping between inputs and outputs, we identify the space as solutions to an operator equation, the Closed-Loop-Map-Equation, parameterized by F. Moreover, any operator Ψ that solves the CLM equation, called a closed-loop map (CLM) of the system F, can be realized by means of a particular controller structure, the system level controller $SL(\Psi)$. The internal stability of this realization is the topic of the second main theorem, which discusses sufficient and necessary conditions for stability. The final part of this chapter investigates closed-loop systems consisting of nonlinear controllers and linear plant dynamics. We show that in this case, the set of nonlinear CLMs is closed under a special type of combination, which we call CLM-blending, and this allows us to construct more expressive CLMs from simpler ones in a hierarchical way.

The ramifications of this result are explored in Chapter 3, which is based on the work presented in [8, 64, 66]. As a first application, we use blending to develop a scalable method for robust control design of linear time-varying dynamic controllers for trajectory tracking in nonlinear continuous-time systems. We then discuss how CLM-blending allows for control design with improved safety and cost-performance trade-offs in control applications considering large-scale linear systems subjected to controller constraints and actuator saturation. We discuss this in the context of two control problems, constrained LQR and distributed anti-windup.

The first half of Chapter 4 applies the nonlinear system level framework to derive
new results for stability analysis of closed-loop systems with online model estimation/selection and certainty-equivalent control in the loop. We introduce an operator-theoretic characterization of the space of closed-loop systems such as tuples $(\Omega, \hat{F}, \Psi, S)$ consisting of a compact parameter space Ω of possible system models ω with dynamics $\hat{F}[\omega]$, a parameterization of nominal CLMs $\Psi : \omega \mapsto \Psi[\omega]$ underlying the certainty-equivalent control policies, and an operator S representing an algorithm for online model estimation. Our main theorem states that if the parameterizations of dynamics $\hat{F} : \omega \mapsto \hat{F}[\omega]$ and desired closed-loop behavior $\Psi : \omega \mapsto \Psi[\omega]$ are continuous and the algorithm S guarantees convergent and consistent model selection, then the overall closed-loop dynamics are stable.

Guided by these theoretical findings, in the second half of the chapter, we follow the certainty equivalence principle to design a learning-to-control scheme with LQ-optimal system-level controllers as the basis of nominal control laws. Most of the results are based on work published in [9, 66]. We perform perturbation analysis of the solutions for the LQ-optimal control problem and show that the solutions are Lipschitz-continuous with respect to changes in the system matrices of the linear system (over parameter sets of equal degree of controllability). This partial result is novel and characterizes the sensitivity (i.e., analytic bounds on the Lipschitz constant) of LQ-optimal closed-loop maps for LTI systems in terms of system-theoretic properties such as controllability and observability. With this result, we analyze the closed-loop stability of the proposed learning-to-control scheme and provide conditions for model selection that are sufficient for closed-loop stability, which we revisit in Part 2 in the context of consistent model chasing.

Chapter 2

NONLINEAR CLOSED LOOP MAPS FROM THE INSIDE AND OUTSIDE

There is a universal connection between the achievable closed-loop dynamics and the corresponding feedback controller that produces it, which shows promise of leading to new methods for robust non-linear control in discrete time. In this chapter, we derive that given a causal nonlinear discrete-time system and controller, the resulting closed loop is a solution to a nonlinear operator equation. Conversely, any causal solution to the nonlinear operator equation is a closed loop that can be achieved by some causal controller. Moreover, solutions can be substituted into a simple dynamic controller structure, which we refer to as a *system level* controller, to obtain an implementation of the unique corresponding feedback controller. System-level controllers are a promising approach for robust nonlinear control, as we show that even when they are parameterized with approximate solutions to the operator equation, they can still produce robustly stable closed loops. We provide theoretical results that state how the degree of approximation and robust stability of the closed loop are related and show that this relationship can be leveraged for controller synthesis.

2.1 Introduction

Compared to linear control theory, there are fewer mathematical tools for tackling controller synthesis of general nonlinear systems. Nonlinear stability analysis is based on some fundamental results in the classical literature [77, 101, 111, 117, 124], such as Lyapunov functions, passivity, the small gain theorem, barrier functions [6], Poincare maps, contraction analysis [90] and describing functions. Among these, the Lyapunov methods are by far the most widely used technique.

Methods for non-linear control design can be separated into two groups: analytical and computational. Lyapunov stability analysis and the optimal control formalism [85], [77] have inspired most classical analytical methods such as feedback linearization, backstepping, sliding-mode control, gain scheduling [77, 111, 117], and many others. Nevertheless, with the recent explosion of available computational resources and progress in the optimization and control theory community, significant progress has been made toward achieving a more generalized, data-driven approach to nonlinear control design. With the sum of squares methods (SOS) [103], [99],

it became possible to compute Lyapunov functions for stability analysis through convex optimization. SOS-based controller synthesis methods are presented in [104, 105], and [133] for the continuous-time (CT) and discrete-time (DT) settings, respectively. Examples of computational methods based on approximating solutions of the Hamilton-Jacobi-Bellman type of equations are found in [120], [67], [130]. Other more recent works [84] (CT), [61] (DT) provide alternative formulations of optimal controller synthesis through occupation measures.

Inspired by the recently developed system-level approach to linear control theory [126], we present a new insight into nonlinear discrete-time systems that enables new synthesis methods for nonlinear discrete-time systems. The system-level approach, as introduced in [126], enabled new efficient controller synthesis methods [66, 127] that allow for localized, distributed, and scalable control design in large-scale systems. This is achieved by transforming constrained optimal control problems as convex optimization problems into achievable closed-loop maps that can be solved efficiently. A key component of the system-level synthesis (SLS) procedure is that once we have solved for the desired closed-loop map, there is a simple way to construct a controller that stably realizes this on the system.

In this chapter, we show that this connection between closed-loop maps and their corresponding realizing controller is not merely a phenomenon of linear systems, but rather a surprisingly universal control principle that extends to general nonlinear discrete-time systems. We demonstrate that given any feasible nonlinear closed-loop map from disturbance to state and input, we can construct an internally stable dynamic controller that realizes it. Specifically, we characterize the space of all feasible closed-loop maps as solutions to a nonlinear operator equation and define a dynamic controller that realizes them. This controller structure, which we refer to as a system-level (SL) controller, is parameterized by the solutions of the operator equation. We further show that even approximate solutions of this equation, with a small enough error, can still yield stabilizing controllers when parameterized into the SL controller structure. We characterize the internal stability of the nonlinear closed loop and discuss a simple sufficient closed-loop stability condition based on the small-gain theorem.

The presented approach motivates new paths towards nonlinear control synthesis: 1) finding approximate solutions to the closed-loop operator equation and 2) obtaining a stabilizing controller by parameterizing an SL controller with the approximate solutions. In the latter part of the chapter, we also discuss a method for synthesizing

nonlinear dynamic controllers based on *blending* of closed-loop maps. These ideas and their importance for control applications will be explored in Chapter 3. To formulate the framework rigorously, we first review some background on operator theory and stability and introduce necessary preliminaries such as notation and basic definitions.

Preliminary Definitions and Notation

Binary Relations. A subset $\mathbb{R} \subset \mathcal{X} \times \mathcal{Y}$ of the Cartesian product of two sets \mathcal{X} and \mathcal{Y} is called a binary relation. $\operatorname{do}(\mathbb{R}) = \{x \in \mathcal{X} \mid \exists y \in \mathcal{Y} : (x, y) \in \mathbb{R}\}$ is called the domain and \mathcal{Y} is called the codomain, denoted $\operatorname{co}(\mathbb{R})$ of the relation \mathbb{R} . One says x relates to y to state $(x, y) \in \mathbb{R}$. The image $\mathbb{R}(\mathcal{X}')$ of a subset $\mathcal{X}' \subset \operatorname{do}(\mathbb{R})$ refers to the set $\{y \in \mathcal{Y} \mid \exists x \in \mathcal{X}' : (x, y) \in \mathbb{R}\}$ and similarly $\mathbb{R}^{-1}(\mathcal{Y}') := \{x \in \mathcal{X} \mid \exists y \in \mathcal{Y}' : (x, y) \in \mathbb{R}\}$ is the preimage of a subset $\mathcal{Y}' \subset \mathcal{Y}$. A composition $\mathbb{R}_1 \circ \mathbb{R}_2$ or $\mathbb{R}_1\mathbb{R}_2$ of two relations $\mathbb{R}_1 \subset \mathcal{X} \times \mathcal{Y}$ and $\mathbb{R}_2 \subset \mathcal{Y} \times \mathcal{Z}$ refers to the relation $\{(x, z) \mid \exists y \in \mathcal{Y} : (x, y) \in \mathbb{R}_1, (y, z) \in \mathbb{R}_2\} \subset \mathcal{X} \times \mathcal{Z}$ and, similarly, $z = \mathbb{R}_1\mathbb{R}_2(x)$ expresses that there exists a $y \in \mathcal{Y}$ such that $y = \mathbb{R}_2(x)$ and $z = \mathbb{R}_1(y)$.

Relations and Functions. A relation R is said to be functional if for any two pairs $(x, y) \in \mathsf{R}, (x', y) \in \mathsf{R}$ holds $y = y' \implies x = x'$. We write $f : \mathcal{X} \to \mathcal{Y}$ as a shorthand to say that f is a functional relation with domain \mathcal{X} and codomain \mathcal{Y} and write y = f(x) or $f : x \mapsto y$ to express $(x, y) \in f$. A *restriction* of a function $f : \mathcal{X} \to \mathcal{Y}$ to a subset $\mathcal{X}' \subset$ is a function $g : \mathcal{X}' \to \mathcal{Y}$ such that $g(x) = f(x), \forall x \in \mathcal{X}'$ and we refer to the function g with the notation $f|_{\mathcal{X}'}$ or $f|_{\mathcal{X}'}$. On the other hand, an *extension* of a function $f : \mathcal{X} \to \mathcal{Y}$ to $\mathcal{X}_e \supset \mathcal{X}$ refers to the function $g : \mathcal{X}_e \to \mathcal{Y}$ such that $g(x) = f(x), \forall x \in \mathcal{X}$; *co-restriction* and *co-extension* of $f : \mathcal{X} \to \mathcal{Y}$ refer to functions $g_1 : \mathcal{X} \to \mathcal{Y}'$ for $\mathcal{Y}' \supset f(\mathcal{X})$ and $g_2 : \mathcal{X} \to \mathcal{Y}_e$, $\mathcal{Y}_e \supset \mathcal{Y}$ such that $g_1(x) = g_2(x) = f(x), \forall x \in \mathcal{X}$.

Sequences. A sequence *s* of elements in *S* over the time horizon *T* is a function $s: T \to S$ where the domain $T \subset \mathbb{N}$ is a subset of natural numbers $\mathbb{N} = \{0, 1, ...\}$ and the codomain is *S*; the space of all such sequences is denoted by S^T . If we say *s* is a sequence of *S* and do not specify *T* or just write $s \in S^{\mathbb{N}}$, it is implied that $T = \mathbb{N}$. We will often use s_t to refer to the value s(t) of a sequence *s* in *t*. As defined for general functions above, we will write, for example, $s_{[t_1,t_2]}$ or s_T for some $T \subset \mathbb{N}$, to refer to the *restrictions* of *s* to an interval $[t_1, t_2] \subset \mathbb{N}$ or subset *T*; *extensions* of a sequence $s \in S^T$ to time horizon $T_e \subset \mathbb{N}$ are defined analogously

as in the context of general functions. A sequence $s \in S^T$ over some finite horizon $T \subset \mathbb{N}$, i.e., $|T| < \infty$, is called a *finite sequence*.

Remark. It is important to note that a finite sequence $\mathbf{s}_{[\tau,\bar{\tau}]}$ entails more information than merely the vector $[s_{\tau}, s_{\tau+1}, \ldots, s_{\bar{\tau}}]$, because in the latter all information about the domain is lost. To point out this difference formally: the space of all finite sequences $\mathbf{s} : \mathcal{T} \to S$ with domain of size n (that is: $|\mathcal{T}| = n$) is infinite dimensional and <u>not</u> simply an n-dimensional vector space of S.

Normed Vector Spaces and their Sequence Spaces We will use caligraphic font variables - most commonly just \mathcal{X} and \mathcal{U} - to denote two fixed finite-dimensional vector spaces equipped with norms $|\cdot|_{\mathcal{X}}$ and $|\cdot|_{\mathcal{U}}$. For convenience, we assume that there is some norm $|\cdot| : \mathcal{X} \times \mathcal{U} \mapsto \mathbb{R}_0^+$ such that $|\cdot|_{\mathcal{X}} : x \mapsto |(x,0)|$ and $|\cdot|_{\mathcal{U}} : u \mapsto |(0, u)|$, and therefore we drop the dependence on the space and let $|\cdot|$ represent a default norm for any finite-dimensional vector spaces. If not otherwise specified, the reader can assume that $\mathcal{X} = \mathbb{R}^n$, $\mathcal{U} = \mathbb{R}^m$ for some n and m < n, and $|\cdot|$ represent the standard Euclidean norm. We will use $\ell^{\mathcal{X}}$, $\ell^{\mathcal{U}}$, and $\ell^{\mathcal{X} \times \mathcal{U}}$ to denote the vector space of sequences $\mathcal{X}^{\mathbb{N}}$, $\mathcal{U}^{\mathbb{N}}$, and $(\mathcal{X} \times \mathcal{U})^{\mathbb{N}}$. Similarly, we define the ℓ_p norms in these spaces as

$$\|\boldsymbol{x}\|_p := \left(\sum_{k=0}^{\infty} |x_k|^p\right)^{1/p} \qquad \|\boldsymbol{x}\|_{\infty} := \sup_{k \ge 0} |x_k|$$

where $|\cdot|$ denotes the corresponding norm in finite-dimensional vector spaces. Consequently, define the subspaces of ℓ_p bounded sequences in $\ell^{\mathcal{X}}$ as $\ell_p^{\mathcal{X}} \subset \ell^{\mathcal{X}}$, $\ell_p^{\mathcal{X}} := \{ \boldsymbol{x} \in \ell^{\mathcal{X}} | \| \boldsymbol{x} \|_p < \infty \}$ and define $\ell_p^{\mathcal{U}}, \ell_p^{\mathcal{X} \times \mathcal{U}}$ analogously. Sequences that have finitely many nonzero elements span a linear subspace of $\ell^{\mathcal{X}}$ which we refer to as $\ell_0^{\mathcal{X}}$. Furthermore, the subset of sequences $\boldsymbol{x} \in \ell_0^{\mathcal{X}}$ for which $\boldsymbol{x}(k) = 0, \forall k \ge t$, i.e., the sequence takes on zero values after time t, form a $(t + 1) \dim(\mathcal{X})$ -dim. subspace of $\ell_0^{\mathcal{X}}$, which we refer to as $\ell_{[0,t]}^{\mathcal{X}}$.

Remark. Unless it is crucial for the discussion, we will not distinguish between the two spaces $\ell_{[0,t]}^{\mathcal{X}}$ and $\mathcal{X}^{[0,t]}$. However, it should be noted that strictly speaking, $\mathcal{X}^{[0,t]}$ and $\ell_{[0,t]}^{\mathcal{X}}$ are only isomorphic to each other, since $\ell_{[0,t]}^{\mathcal{X}}$ is an embedding of $\mathcal{X}^{[0,t]}$ into space $\ell^{\mathcal{X}}$.

A truncation x' of a sequence $x \in \ell^{\mathcal{X}}$ is a sequence $x' \in \ell^{\mathcal{X}}_{[0,t]}$ such that $x'_k := x_k$ for $k \in \{0, \ldots, t\}$ and some t, that is, a sequence formed by truncating all but the first of the elements t of x.

Tuples and Concatenations of Sequences We will use small tt-font styled variables {x, u, w, \hat{w} , y, z} to form index sets for sequence tuples or partial operators. For example, if we write $\tau = {\tau_i}_{i \in \mathcal{I}} \in \times_{i \in \mathcal{I}} (\mathcal{V}_i)^{\mathbb{N}}$, $\mathcal{I} = x, u, w$, then we mean that τ is a tuple (τ_x, τ_u, τ_w) of sequences $\tau_x \in (\mathcal{V}_x)^{\mathbb{N}}$, $\tau_u \in (\mathcal{V}_w)^{\mathbb{N}}$, and $\tau_w \in (\mathcal{V}_w)^{\mathbb{N}}$; $\times_{j \in \mathcal{J}} S_j$ generally denotes the Cartesian product of a family of sets ${S_j}_{j \in \mathcal{J}}$. Occasionally, we match the variable names of the sequences with the labels and will write $\tau_{xuw} = (x, u, w)$ to suggest assignment $(\tau_x, \tau_u, \tau_w) = (x, u, w)$. The concatenation of a sequence $x \in \ell^{\mathcal{X}}$ and $u \in \ell^{\mathcal{U}}$ is a new sequence $z \in \ell^{\mathcal{X} \times \mathcal{U}}$, such that $z_t = (x_t, u_t)$, $\forall t \in \mathbb{N}$; we refer to sequence z by writing $\begin{bmatrix} x \\ u \end{bmatrix}$. With slight abuse of notation, we use \top to identify $(x, u)^{\top}$ with $\begin{bmatrix} x \\ u \end{bmatrix}$ and $\begin{bmatrix} x \\ u \end{bmatrix}^{\top}$ with (x, u), i.e., $(\cdot)^{\top} : (x, u) \mapsto \begin{bmatrix} x \\ u \end{bmatrix}$ and $[\cdot]^{\top} : \begin{bmatrix} x \\ u \end{bmatrix} \mapsto (x, u)$, and define this correspondence similarly for any tuple of sequences; hence, $(\cdot)^{\top}$ is the canonical isomorphism between $\times_{i \in \mathcal{I}} (\mathcal{V}_i)^{\mathbb{N}}$ and $(\times_{i \in \mathcal{I}} \mathcal{V}_i)^{\mathbb{N}}$.

2.2 Fundamentals of Operator Theory

Operators will be denoted in bold capital letters A and will represent maps between vector sequence spaces $\ell^{\mathcal{X}} \to \ell^{\mathcal{U}}$. An operator will be called causal if for any pair of input $x \in \ell^{\mathcal{X}}$ and the corresponding output y = A(x), the values of y_t do not depend on future input values x_{t+k} , $k \ge 1$. More precisely, we define $A : \ell^{\mathcal{X}} \to \ell^{\mathcal{U}}$ to be a *causal* operator if there are functions, $A_t : \mathcal{X}^{[0,t]} \to \mathcal{U}$ that allow A to be equivalently represented as

$$\boldsymbol{A}(\boldsymbol{x}) = (A_0(x_0), A_1(x_1, x_0), \dots, A_t(x_{t:0}), \dots).$$
(2.1)

If in addition, the functions A_t satisfy $A_t(x_{t:0}) = A_t(0, x_{t-1:0})$, i.e., are constant in their first parameter, then A will be called *strictly causal*. The functions $\{A_t\}$ fully characterize a causal operator and will also be called *component functions* or just *components* of A. Notice that every component function A_t has t + 1arguments which are populated in reverse-chronological order in Definition (2.1). For notational convenience, for an interval $\mathcal{I} \subset \mathbb{N}$, $\mathcal{I} = [i, j], i \leq j$ we will use $A_{\mathcal{I}} : \mathcal{X}^{[0:j]} \to \mathcal{U}^{\mathcal{I}}$ (or $A_{i:j}$) to refer to the mapping defined below: $x_{j:0} \ni \mathcal{X}^{[0,j]} \mapsto$ $(A_i(x_{i:0}), A_{i+1}(x_{i+1:0}), \ldots, A_j(x_{j:0})) \in \mathcal{U}^{[i,j]}$ for $j \geq i$ as

$$A_{i:j}(x_{j:0}) := (A_i(x_{i:0}), A_{i+1}(x_{i+1:0}), \dots, A_j(x_{j:0})).$$

An operator with the same domain and co-domain, for example: $\mathbf{A}: \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$, is called square. Define the space of all causal and strictly causal operators $\ell^{\mathcal{X}} \mapsto \ell^{\mathcal{Y}}$ as $\mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$ and $\mathcal{C}_s(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$, respectively. Similarly, define $\mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}}) \subset \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$

and $\mathcal{LC}_s(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}}) \subset \mathcal{C}_s(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$ the spaces of all linear causal and strictly causal operators. The corresponding spaces of square (linear)-(strictly)-causal operators on $\ell^{\mathcal{X}}$ will be denoted by $\mathcal{C}(\ell^{\mathcal{X}}), \mathcal{C}_s(\ell^{\mathcal{X}}), \mathcal{LC}(\ell^{\mathcal{X}}), \mathcal{LC}_s(\ell^{\mathcal{X}})$. The right-shift operation $\mathcal{S}^+ : \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}}) \mapsto \mathcal{C}_s(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$ is defined to map any causal operator $\mathbf{A} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$, to a strictly causal one $\mathcal{S}^+[\mathbf{A}] := \mathbf{A}^+ \in \mathcal{C}_s(\ell^{\mathcal{X}}, \ell^{\mathcal{Y}})$ by shifting all component functions to the right, that is, $T_0^+(x_0) := 0, \quad T_t^+(x_{t:0}) := T_{t-1}(x_{t-1:0}).$

Addition and Multiplication of Operators

Sums and products of operators are defined as binary operations on the space of causal operators where

$$oldsymbol{A} + oldsymbol{B} : oldsymbol{x} \mapsto oldsymbol{A}(oldsymbol{x}) + oldsymbol{B}(oldsymbol{x})$$

 $oldsymbol{A} oldsymbol{B} \ or oldsymbol{A} \circ oldsymbol{B} : oldsymbol{x} \mapsto oldsymbol{A}(oldsymbol{B}(oldsymbol{x})).$

It is crucial to remember that for general operators, the above defined multiplication is not commutative and is **only** right-distributed over the summation but **not** leftdistributed, i.e.,

$$(A+B)C = AC + BC$$
 but $C(A+B) \neq CA + CB$.

Moreover, for two operators $A \in C(\ell^{\mathcal{X}})$, $B \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$ with matching domain, $\begin{bmatrix} A \\ B \end{bmatrix}$ refers to the operator $C \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ with components C_t defined each t and sequence $x \in \ell^{\mathcal{X}}$ as

$$C_t(x_{t:0}) := (A_t(x_{t:0}), B_t(x_{t:0})).$$

Truncation Operator

Sequences that have finitely many nonzero elements span a linear subspace of $\ell^{\mathcal{X}}$ which we refer to as $\ell_0^{\mathcal{X}}$. Furthermore, for a fixed subset $\mathcal{I} \subset \mathbb{N}$, the subset of sequences $\boldsymbol{x} \in \ell_0^{\mathcal{X}}$ such that $x_k = 0$ for all $k \notin \mathcal{I}$ forms a subspace of $\ell_0^{\mathcal{X}}$, which we refer to as $\ell_{\mathcal{I}}^{\mathcal{X}}$; for example, $\ell_{[0,t]}^{\mathcal{X}}$ represents the subspace of sequences with elements equal to 0 after the first t + 1 terms. A truncation \boldsymbol{x}' of a sequence $\boldsymbol{x} \in \ell^{\mathcal{X}}$ is a sequence $\boldsymbol{x}' \in \ell_{[0,t]}^{\mathcal{X}}$ such that $x'_k := x_k$ for $k \in \{0, \ldots, t\}$ and some t, that is, a sequence formed by truncating all but the first t elements of \boldsymbol{x} . For each $t \in \mathbb{N}$, the correspondence $\boldsymbol{x} \mapsto \boldsymbol{x}'$ between sequences $\boldsymbol{x} \in \ell^{\mathcal{X}}$ and their truncations $\boldsymbol{x}' \in \ell_{[0,t]}^{\mathcal{X}}$ defines a unique linear projection map $\boldsymbol{P}_n^t : \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$ onto $\ell_{[0,t]}^{\mathcal{X}}$, commonly called a truncation operator. Below, we define the truncation operators $\{\boldsymbol{P}_n^t \mid t \in \mathbb{N}\}$ in terms of the broader – however less standard – family of operators $\{\boldsymbol{P}_n^{\mathcal{I}} \mid \mathcal{I} \subset \mathbb{N}\}$:

Definition 2.1. [*Truncation Operators*] For a subset $\mathcal{I} \subset \mathbb{N}$ and its complement $\mathcal{I}^c := \mathbb{N} \setminus \mathcal{I}$, define the operator $P_n^{\mathcal{I}} : \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$ and $\overline{P_n^{\mathcal{I}}} : \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$ for each $x \in \ell^{\mathcal{X}}$ as:

$$\boldsymbol{P}_{n}^{\mathcal{I}}(\boldsymbol{x}) := \begin{cases} x_{k} & \text{for } k \in \mathcal{I} \\ 0 & \text{else} \end{cases} \qquad \quad \boldsymbol{\overline{P}_{n}^{\mathcal{I}}}(\boldsymbol{x}) := \begin{cases} x_{k} & \text{for } k \in \mathcal{I}^{c} \\ 0 & \text{else.} \end{cases}$$

Moreover, we use the following shorthand notation for $P_n^{\mathcal{I}}$ and $\overline{P_n^{\mathcal{I}}}$ if \mathcal{I} is an interval or a singleton:

• For
$$t \in \mathbb{N}$$
, $P_n^t := P_n^{\mathcal{I}}$ and $\overline{P_n^t} := \overline{P_n^{\mathcal{I}}}$ where $\mathcal{I} = [0, t] \subset \mathbb{N}$.

- For $t, t' \in \mathbb{N}$: t < t', $\mathbf{P}_n^{[t,t']} := \mathbf{P}_n^{\mathcal{I}}$ and $\mathbf{P}_n^{[t,t']} := \mathbf{\overline{P}_n^{\mathcal{I}}}$ where $\mathcal{I} = [t,t'] \subset \mathbb{N}$. For $t \in \mathbb{N}$, $\mathbf{P}_n^{[t]} := \mathbf{P}_n^{\mathcal{I}}$ and $\mathbf{\overline{P}_n^{[t]}} := \mathbf{\overline{P}_n^{\mathcal{I}}}$ where $\mathcal{I} = \{t\}$.

The truncation operators are linear projection maps in sequence space and have several important properties listed in Cor. 1, all of which are trivial consequences of the previous definition:

Corollary 1. For any $\mathcal{I} \subset \mathbb{N}$, the truncation operator $P^{\mathcal{I}}$ satisfies the following identities:

(i) $P_n^{\mathcal{I}}(\boldsymbol{x}) \in \ell_p^{\mathcal{X}}$ for any \boldsymbol{x} and finite set \mathcal{I} . (ii) $P_n^{\mathcal{I}}(\boldsymbol{x} + \boldsymbol{y}) = P_n^{\mathcal{I}}(\boldsymbol{x}) + P_n^{\mathcal{I}}(\boldsymbol{y})$.

(iii)
$$P_n^{\mathcal{I}} P_n^{\mathcal{I}} = P_n^{\mathcal{I}}$$
 (iv) If $x \notin \ell_p^{\mathcal{X}}, (p \in [1, \infty])$ then $P_n^{\mathcal{I}}(x) \notin \ell_p^{\mathcal{X}}$ or $\overline{P_n^{\mathcal{I}}}(x) \notin \ell_p^{\mathcal{X}}$

- (v) $\|\boldsymbol{x}\|_p^p = \|(\boldsymbol{I} \boldsymbol{P}_n^{\mathcal{I}})(\boldsymbol{x})\|_p^p + \|\boldsymbol{P}_n^{\mathcal{I}}(\boldsymbol{x})\|_p^p = \|\overline{\boldsymbol{P}_n^{\mathcal{I}}}(\boldsymbol{x})\|_p^p + \|\boldsymbol{P}_n^{\mathcal{I}}(\boldsymbol{x})\|_p^p$ holds for all $p < \infty$ and $\boldsymbol{x} \in \ell_n^{\mathcal{X}}$.
- (vi) $\|\boldsymbol{x}\|_{\infty} = \|(\boldsymbol{I} \boldsymbol{P}_n^{\mathcal{I}})(\boldsymbol{x})\|_{\infty} \vee \|\boldsymbol{P}_n^{\mathcal{I}}(\boldsymbol{x})\|_{\infty} = \|\overline{\boldsymbol{P}_n^{\mathcal{I}}}(\boldsymbol{x})\|_{\infty} \vee \|\boldsymbol{P}_n^{\mathcal{I}}(\boldsymbol{x})\|_{\infty}$ holds for all $\boldsymbol{x} \in \ell_{\infty}^{\mathcal{X}}$. $(a \lor b := \max\{a, b\})$.

In light of this more general definition, the standard truncations P_n^t we discussed in the beginning are identified with the intervals $\mathcal{I} = [0, t]$. Furthermore, the operation $P_n^{[t]}(x)$ truncates all terms of x_k , except k = t. To familiarize with the introduced notation, let $t, t' \in \mathbb{N}$, t < t' and notice the following relationships:

$$\begin{aligned} \boldsymbol{P}_n^t + \overline{\boldsymbol{P}_n^t} &= \boldsymbol{I}_n & \boldsymbol{P}_n^{[t]} &= \boldsymbol{P}_n^t - \boldsymbol{P}_n^{t-1} \\ \boldsymbol{P}_n^t &= \sum_{k=0}^t \boldsymbol{P}_n^{[k]} & \boldsymbol{P}_n^{[t,t']} &= \boldsymbol{P}_n^{t'} - \boldsymbol{P}_n^{t-1} \end{aligned}$$

Given that we live in the space $\mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}})$, as long as \mathcal{I} is a finite subset of \mathbb{N} , it is clear that $P_n^{\mathcal{I}}$ is a finite-rank linear projection and rank $(P_n^{\mathcal{I}}) = n|\mathcal{I}|$. For finite sets \mathcal{I} , we can factor $P_n^{\mathcal{I}}$ into the familiar form $P_n^{\mathcal{I}} = U^{\mathcal{I}}U^{*\mathcal{I}}$, by letting the linear maps $U^{\mathcal{I}} : \mathbb{R}^N \to \ell^{\mathcal{X}}$ and $U^{*\mathcal{I}} : \ell^{\mathcal{X}} \to \mathbb{R}^N$ be extensions of the canonical isomorphisms between \mathbb{R}^N , where $N = n|\mathcal{I}|$, and the $\ell_0^{\mathcal{X}}$ -subspace $\{x \in \ell_0^{\mathcal{X}} \mid x_k = 0, \forall k \notin \mathcal{I}\}$. To clarify, consider $\mathcal{I} = [1]$ as an example. Then, for any $x \in \mathbb{R}^n, y = U^{[1]}(x)$ is the sequence $y = (0, x, 0, ...) \in \ell_{0|1}^{\mathcal{X}}$ and, vice versa, for any sequence $x = (0, y, 0, ...) \in \ell_{0|1}^{\mathcal{X}}$ with $y \in \mathbb{R}^n$, we have $U^{*[1]}(x) = y$. To match our previous notation, we also abbreviate $U_n^{[0,t]}$ and $U_n^{*[0,t]}$ as U_n^t and U_n^{*t} , respectively, which allows us to factor the standard truncation operators as $P_n^t = U_n^t U_n^{*t}$.

Causal Operators and Truncations

There are several important relationships between truncations and causal operators, one being that the causality of an operator Q can be equivalently defined in terms of its interaction with truncations.

Lemma 2. For each $t \in \mathbb{N}$, let P_n^t and P_m^t denote the t-truncation on $\ell^{\mathcal{U}}$ and $\ell^{\mathcal{U}}$. For any operator $Q : \ell^{\mathcal{X}} \mapsto \ell^{\mathcal{U}}$, the following hold:

$$\begin{aligned} \boldsymbol{Q} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}}) & \Leftrightarrow & \forall t \in \mathbb{N} : \ \boldsymbol{P}_{m}^{t} \boldsymbol{Q} = \boldsymbol{P}_{m}^{t} \boldsymbol{Q} \boldsymbol{P}_{n}^{t} \\ \boldsymbol{Q} \in \mathcal{C}_{s}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}}) & \Leftrightarrow & \forall t \in \mathbb{N} : \ \boldsymbol{P}_{m}^{t} \boldsymbol{Q} = \boldsymbol{P}_{m}^{t} \boldsymbol{Q} \boldsymbol{P}_{n}^{t-1} \end{aligned}$$

Using the above definition and right-distributiveness, we can decompose a causal operator $Q : \ell^{\chi} \mapsto \ell^{U}$ as follows:

$$egin{aligned} oldsymbol{Q} &= (oldsymbol{I}_m - oldsymbol{P}_m^t)oldsymbol{Q} + oldsymbol{P}_m^toldsymbol{Q} + oldsymbol{P$$

As mentioned in Cor. 1, a truncation operator exhibits some properties that resemble orthogonal projections, but without requiring a specific inner product. In particular, in any normed space $\ell_p^{\mathcal{U}}/\ell_{\infty}^{\mathcal{U}}$, from the above decomposition and the properties mentioned in Cor. 1, we obtain that for any \boldsymbol{x} and $\boldsymbol{Q} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$, the following holds:

$$egin{aligned} \|oldsymbol{Q}(oldsymbol{x})\|_p^p &= \|\overline{oldsymbol{P}_m^t}oldsymbol{Q}(oldsymbol{x})\|_p^p + \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_p^p \ \|oldsymbol{Q}(oldsymbol{x})\|_\infty &= \|\overline{oldsymbol{P}_m^t}oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{x}\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{x}\|_\infty ee \|oldsymbol{P}_m^toldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{P}_m^toldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x})\|_\infty ee \|oldsymbol{P}_m^toldsymbol{P}_m^toldsymbol{P}_m^toldsymbol{Q}(oldsymbol{x}$$

Remark. The above equations are consistent in the case $Q(x) \notin \ell_p^{\mathcal{X}}$ or $Q(x) \notin \ell_p^{\mathcal{X}}$, if we define $\|Q(x)\|_p$ and $\|Q(x)\|_{\infty}$ to take on the value $+\infty$.

Using the definitions of $U^{*[t]}$ and U^t , we can also concisely define the component functions of an operator Q as $Q_t = U_m^{*[t]} Q U_n^t$. Similarly, we can factor $P_m^t Q$ as

$$\boldsymbol{P}_m^t \boldsymbol{Q} = \sum_{k=0}^t \boldsymbol{U}_m^{[k]} Q_k \boldsymbol{U}_n^{*k}.$$

Another important consequence of this decomposition is that appending a truncation P_m^t to a causal operator Q always results in smaller $\ell_p^{\mathcal{U}}$ norms than prepending a truncation P_n^t . To see this, we prepend P_n^t to Q and use right-distributiveness and causality to obtain the equivalence:

$$oldsymbol{Q} oldsymbol{P}_n^t = ((oldsymbol{I}_m - oldsymbol{P}_m^t)oldsymbol{Q} + oldsymbol{P}_m^toldsymbol{Q})oldsymbol{P}_n^t = \overline{oldsymbol{P}_m^t}oldsymbol{Q}oldsymbol{P}_n^t + oldsymbol{P}_m^toldsymbol{Q}oldsymbol{P}_n^t = \overline{oldsymbol{P}_m^t}oldsymbol{Q}oldsymbol{P}_n^t + oldsymbol{P}_m^toldsymbol{Q}oldsymbol{P}_n^t = \overline{oldsymbol{P}_m^t}oldsymbol{Q}oldsymbol{P}_n^t + oldsymbol{P}_m^toldsymbol{Q}oldsymbol{P}_n^t$$

Then, for any $x \in \ell^{\mathcal{X}}$, $Q \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$, and $t \in \mathbb{N}$, the following hold:

$$egin{aligned} \|oldsymbol{Q}oldsymbol{P}_n^t(oldsymbol{x})\|_p^p &= \|\overline{oldsymbol{P}_m^t}oldsymbol{Q}oldsymbol{P}_n^t(oldsymbol{x})\|_p^p &= \|\overline{oldsymbol{P}_m^t}oldsymbol{Q}oldsymbol{P}_n^t(oldsymbol{x})\|_\infty &= \|\overline{oldsymbol{P}_m^t}oldsymbol{Q}oldsymbol{P}_n^t(oldsymbol{x})\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}oldsymbol{P}_n^t(oldsymbol{x})\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}\|_\infty ⅇ \|oldsymbol{Q}\|_\infty ⅇ \|oldsymbol{Q}\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{Q}\|_\infty ⅇ \|oldsymbol{P}_m^toldsymbol{P}_m$$

We summarize our findings in the lemma below:

Lemma 3. Let $\{\mathcal{T}_k = [\underline{t}_k, \overline{t}_k]\}$, $(\underline{t}_k \leq \overline{t}_k)$, be a collection of pairwise disjoint intervals of \mathbb{N} over some index set $k \in \mathcal{I}$, and let $\mathbf{Q} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$ be a causal operator. Then, for any $t \in \mathbb{N}$, $p \in \{1, \ldots, \infty\}$, and $\mathbf{x} \in \ell^{\mathcal{U}}$, the following inequality holds:

$$\|(\sum_{k\in\mathcal{I}}\boldsymbol{P}_{n}^{\mathcal{T}_{k}}\boldsymbol{Q})\boldsymbol{x}\|_{p} \leqslant \sum_{k\in\mathcal{I}} \|\boldsymbol{Q}\boldsymbol{P}_{m}^{\bar{t_{k}}}\boldsymbol{x}\|_{p}.$$
(2.2)

The above lemma will be used in the derivation of the small gain theorem, which we derive in the next section.

Causally Invertible Operators

A square operator $A : \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$ is invertible if there exists another operator $B : \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$ such that $AB = BA = I_n$. If such a B exists, it is unique and is therefore called the inverse of A and is denoted as A^{-1} . We call $A \in \mathcal{C}(\ell^{\mathcal{X}})$ causally invertible if it is invertible and its inverse is causal, that is, $A^{-1} \in \mathcal{C}(\ell^{\mathcal{X}})$.

Definition 2.2. $A \in C(\ell^{\mathcal{X}})$ is causally invertible if there exists $A^{-1} \in C(\ell^{\mathcal{X}})$ such that $AA^{-1} = A^{-1}A = I_n$.

In general, the inverse of a causal operator is not necessarily a causal operator [110]. In fact, it is possible to find counterexamples even for operators $\mathbf{A} : \ell_{[0,t]} \to \ell_{[0,t]}$ that map between finite-dimensional spaces. However, if causality of the inverse can be proven, it also provides instructions on how to implement the inverse operator.

If we are given a causally invertible operator $A \in C(\ell^{\mathcal{X}})$, then by definition, for any $t \in \mathbb{N}$, $P^t A^{-1} = P^t A^{-1} P^t$, $\forall t \in \mathbb{N}$. Combining this with causality of A shows that we obtain an invertible function \underline{A}_t by concatenating the first t component functions:

$$\underline{A}_t: (x_t, \dots, x_0) \mapsto (A_t(x_t, \dots, x_0), \dots, A_0(x_0)).$$
(2.3)

The map \underline{A}_t can be written more compactly as $\underline{A}_t = U^{*t}AU^t$. For the next derivation, let $B = A^{-1} \in C(\ell^{\mathcal{X}})$ denote the inverse of A and similarly, let the concatenation of the components of B be denoted as $\underline{B}_t = U^{*t}BU^t$. Below, we derive that \underline{B}_t is in fact the inverse of \underline{A}_t and therefore invertibility of $\underline{A}_t, \forall t \in \mathbb{N}$ is a necessity for causal invertibility:

$$P^{t} = P^{t}AB = P^{t}BA$$

$$\implies \qquad U^{t}U^{*t} = P^{t}AP^{t}BP^{t} = P^{t}BP^{t}AP^{t}$$

$$\stackrel{X \mapsto U^{*t}XU^{t}}{\implies} \qquad I_{N_{t}} = \underbrace{(U^{*t}BU^{t})}_{\underline{B_{t}}}\underbrace{(U^{*t}AU^{*t})}_{\underline{A_{t}}} = (U^{*t}AU^{t})(U^{*t}BU^{*t})$$

$$\implies \qquad \underline{B_{t}} = (\underline{A_{t}})^{-1}.$$

Having established the equivalence $\underline{B}_t = (\underline{A}_t)^{-1}$, going forward, we can refer to \underline{B}_t as \underline{A}_t^{-1} without causing ambiguity, since $U^{*t}A^{-1}U^t$ and $(\underline{A}_t)^{-1}$ are indeed the same function. Furthermore, the above equivalence aids us in constructing a realization of the inverse operator.

Setting t = 0, the above implies that A_0 is invertible. Moving to t = 1, it is easy to see that $\underline{A}_1 : (u_0, u_1) \mapsto (A_0(u_0), A_1(u_1, u_0))$ is invertible if and only if the map $u' \mapsto A_1(u', u_0)$ is invertible for any $u_0 \in \mathbb{R}^n$. This observation motivates us to investigate whether this condition extends to all \mathbb{N} . To this end, let **b** and **a** be such that $\mathbf{b} = \mathbf{A}\mathbf{a}$, and let $\underline{a}_t = (a_t, \dots, a_0)$ and $\underline{b}_t = (b_t, \dots, b_0)$. Then, for some $t \in \mathbb{N}$, we can express a_t as

$$a_{t} = \boldsymbol{U}^{[*t]}\boldsymbol{A}^{-1}(\boldsymbol{b}) = \boldsymbol{U}^{[*t]}\boldsymbol{A}^{-1}\boldsymbol{P}^{t}(\boldsymbol{b}) = A_{t}^{-1}(b_{t},\underline{b}_{t-1}) = A_{t}^{-1}(b_{t},\underline{A}_{t-1}(\underline{a}_{t-1}))$$
$$= A_{t}^{-1}(\underline{A}_{t}(a_{t},\underline{a}_{t-1}),\underline{A}_{t-1}(\underline{a}_{t-1}))$$
(2.4)

where $A_t^{-1} = U^{[*t]} A^{-1} U^t$ denotes the *t*-th component function of A^{-1} . For a fixed choice of \underline{a}_{t-1} , denote $A_{t|t-1}[\underline{a}_{t-1}] : \mathbb{R}^n \to \mathbb{R}^n$ as the restriction of the component function A_t such that $A_{t|t-1}[\underline{a}_{t-1}](a) = A_t(a, \underline{a}_{t-1})$. Similarly, let $B_{t|t-1}[\underline{a}_{t-1}] : \mathbb{R}^n \to \mathbb{R}^n$ be a function defined for each $b \in \mathbb{R}^n$ as $B_{t|t-1}[\underline{a}_{t-1}](b) = A_t^{-1}(b, \underline{A}_{t-1}(a_{t-1}))$. With these two function definitions, (2.4) can be written as $a_t = B_{t|t-1}[\underline{a}_{t-1}] \circ A_{t|t-1}[\underline{a}_{t-1}](a_t)$. Since this equation has to hold for all \underline{a}_{t-1} and $a_t \in \mathbb{R}^n$, we can conclude that for any fixed \underline{a}_{t-1} , the function $B_{t|t-1}[\underline{a}_{t-1}]$ is the left-inverse of $A_{t|t-1}[\underline{a}_{t-1}]$. Hence, equation (2.4) can be written as $a_t = B_{t|t-1}[\underline{a}_{t-1}](b_t)$ and provides us with a realization of A^{-1} . For a given b such that b = A(a) with causally invertible A, the sequence a is a trajectory of the dynamic system described by the equations

$$a_t = B_t[\underline{a}_{t-1}](b_t). \tag{2.5}$$

Our discussion has so far shown that if A is causally invertible, then the functions $\{\underline{A}_t\}_{t\in\mathbb{N}}$ have to be invertible. Moreover, the latter implies that any function in the set $\mathcal{A} = \{A_{t|t-1}[z_{t-1}] : \mathbb{R}^n \to \mathbb{R}^n \mid t \in \mathbb{N}, z_{t-1} \in \mathbb{R}^{N_t}\}$ has to be invertible as well and that using the inverse functions $B_t[\underline{a}_{t-1}]$ we can realize the operator A^{-1} as the dynamical system (2.5). Clearly, this realization is also evidence that invertibility of the functions in \mathcal{A} is sufficient for the existence of A^{-1} . This conclusion closes a chain of implications and leads to the following Lemma, which characterizes causally invertible operators in terms of their component functions:

Lemma 4. For a causal operator $\mathbf{A} : \ell^{\mathcal{X}} \to \ell^{\mathcal{X}}$, the following statements are all equivalent:

- (*i*) *A* is causally invertible.
- (ii) $\{\underline{A}_t : z \mapsto U^{*t} A U^t(z) \mid t \in \mathbb{N}\}$ is a family of invertible functions.
- (iii) $\{A_{t|t-1}[z_{t-1}]: a \mapsto A_t(a, z_{t-1}) \mid t \in \mathbb{N}, z_{t-1} \in \underline{A}_{t-1}^{-1}(\mathbb{R}^{N_{t-1}})\}$ is a family of invertible functions.

Proof. We recap the ring implications. $(i) \implies (ii)$: The causal invertibility of A means that there is some causal B such that AB = I, which leads to $P^t A P^t B P^t = P^t, \forall t \in \mathbb{N}$ and proves that $\underline{B}_t : U^{*t} B U^t$ is the right inverse of \underline{A}_t . $(ii) \implies (iii)$: Pick an arbitrary $t \in \mathbb{N}$ and apply the operator U^t to both sides (multiplication from the left) of the identity $\underline{A}_t \circ \underline{B}_t = I_{N_t}$. The new equation can be restated as:

 $\forall y \in \mathbb{R}^n, \forall y_{t-1} \in \mathbb{R}^{N_{t-1}}: \quad A_t(B_t(y, y_{t-1}), \underline{B}_{t-1}(y_{t-1})) = y.$

This states that for a fixed $z_{t-1} = \underline{B}_{t-1}(y_{t-1})$, $B_{t|t-1}[y_{t-1}] : y \mapsto B_t(y, y_{t-1})$ is the right inverse of $A_{t|t-1}[z_{t-1}]$. Now, since \underline{B}_{t-1} is the inverse of \underline{A}_{t-1} , z_{t-1} ranges

over all \mathbb{R}^{N_t} . We also showed previously left-invertibility in the same way, which establishes (*iii*). Lastly, (*iii*) \implies (*i*) follows through the realization (2.5).

A corollary of this lemma is that operators $A = I_n + A^-$, which are sums of some strictly causal $A^- \in C_s(\ell^{\chi})$ and the identity operator I, are always causally invertible, and the inverse operator A^{-1} can be easily realized. This result is derived below:

Corollary 5. If $(A - I) \in C_s(\ell^{\chi})$, then $A^{-1} \in C(\ell^{\chi})$ exists and $b = A^{-1}(a)$ satisfies

$$b_t = a_t - A_t(0, b_{t-1:0}).$$

Proof. Assume given a, we want to find b such that A(b) = a. Equivalently, we can write b = a - (A - I)(b) Now, since A - I is strictly causal, the component function A_t satisfies $A_t(x_t, x_{t-1:0}) = A_t(0, x_{t-1:0}) + x_t$. Using this factorization, the component form of b = a - (A - I)(b) becomes

$$b_t = a_t - A_t(0, b_{t-1:0}), (2.6)$$

which proves the existence and uniqueness of b as it describes a concrete recursive procedure for its computation.

2.3 Global and Local Stability of Operators

We review some standard stability results from the control literature, which are needed for later analysis. In the following, we derive various versions of the small-gain theorem, which are used in our main results.

If we view a dynamical system as a causal map T between input sequences w and output sequences y = Tw, then many common notions of system stability, such as ℓ_p stability, can be mapped to notions of boundedness and continuity of the operator T. Sometimes it is more natural to define the relationship between input and output in an implicit fashion, where for an input w, the output y is a solution to equation Qy = w, for some fixed operator Q. In that scenario, input-output stability means that for a bounded w, any corresponding solution y is also bounded.

Approaching stability analysis from this operator-theoretic perspective was first studied in the late 1960s, most notably pioneered by George Zames [139], and led to a large class of important stability analysis tools often grouped together under the name of *small gain* theorems.

In this section, we derive some small-gain theorem conditions that match our problem setup and that we use frequently in our analysis.

We use the following notions of ℓ_p -stability for operators in the statement of our results:

Definition 2.3. An operator $A \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$ is called:

- ℓ_p -stable if $\mathbf{A}(\mathbf{a}) \in \ell_p^{\mathcal{U}}$ for all $\mathbf{a} \in \ell_p^{\mathcal{X}}$.
- finite gain (f.g.) ℓ_p -stable ¹ at $\mathbf{a}_0 \in \ell_p^{\mathcal{X}}$, if there exists $\gamma, \beta \ge 0$ such that for all $\mathbf{a} \in \ell_p^{\mathcal{X}}$:

$$\|\boldsymbol{A}(\boldsymbol{a}) - \boldsymbol{A}(\boldsymbol{a}_0)\|_p \leq \gamma \|\boldsymbol{a} - \boldsymbol{a}_0\|_p + \beta.$$

incrementally finite gain² (i.f.g.) ℓ_p-stable if there exists γ, β ≥ 0 such that for all a, a' ∈ ℓ_p^X, it holds:

$$\left\|\boldsymbol{A}(\boldsymbol{a}) - \boldsymbol{A}(\boldsymbol{a'})\right\|_{p} \leq \gamma \left\|\boldsymbol{a} - \boldsymbol{a'}\right\|_{p} + \beta.$$

Remark 1. Finite gain ℓ_p stability and operator continuity in ℓ_p -norm are closely related. In fact, if we set $\beta = 0$ in the above definitions, then, i.f.g. ℓ_p -stability of A with gain γ is equivalent to Lipschitz continuity in ℓ_p -norm of A over ℓ_p^{χ} . On the other hand, f.g.- ℓ_p -stability at \mathbf{a}_0 implies that A is ℓ_p -continuous at \mathbf{a}_0 and that $\delta(\varepsilon)$ can be chosen, in the context of the standard continuity definition [109], as a linear function $\delta : \varepsilon \mapsto \gamma^{-1} \varepsilon$. Allowing β to be nonzero can be thought of as "almost" (Lipschitz) continuity, as possible discontinuities are deemed irrelevant for the purpose of stability analysis.

Notice that while the above definition might not be standard, it allows for the stability analysis of equilibria, trajectories, and limit cycles all within the same definition.

In the following discussion, we consider a fixed causal operator $Q \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{X}})$ and investigate pairs (w, y) of sequences $w \in \ell^{\mathcal{X}}$ and $y \in \ell^{\mathcal{X}}$ that are solutions to the nonlinear operator equation:

$$Q(y) = w$$
 \Leftrightarrow $y = (I - Q)(y) + w.$ (2.7)

¹If we say A is f.g. ℓ_p -stable without specifying a_0 , it is assumed that a_0 is to be taken as 0.

²We write (γ, β) -f.g. and (γ, β) -i.f.g. if we want to specify the constants of the ℓ_p -stability property.

For the purpose of stability analysis, we would ideally like to characterize the space of solutions (w, y) that satisfy relations of the following types:

$$(\boldsymbol{w} \to \boldsymbol{y}) \ \ell_p - stab.: \quad \boldsymbol{Q}(\boldsymbol{y}) = \boldsymbol{w}, \ \boldsymbol{w} \in \ell_p^{\mathcal{X}} \implies \boldsymbol{y} \in \ell_p^{\mathcal{X}},$$
 (2.8a)

$$(\boldsymbol{y} \to \boldsymbol{w}) \ \ell_p - stab.: \qquad \boldsymbol{Q}(\boldsymbol{y}) = \boldsymbol{w}, \ \boldsymbol{y} \in \ell_p^{\mathcal{X}} \implies \boldsymbol{w} \in \ell_p^{\mathcal{X}},$$
 (2.8b)

bi-stab.
$$\ell_p$$
: $\boldsymbol{Q}(\boldsymbol{y}) = \boldsymbol{w} \implies \boldsymbol{y}, \boldsymbol{w} \in \ell_p^{\mathcal{X}}.$ (2.8c)

If we identify w and y with some inputs and outputs of a dynamical system that has dynamic equations of the form (2.7), then condition (2.8a) represents the standard notion of *bounded-input-bounded-output* (BIBO) [75] with respect to the norm ℓ_p . The statement (2.8b) represents the reverse condition, which is more commonly seen as a notion of system observability. Similarly to the definitions in (2.8), one can examine the space of solutions (y, w) for more refined notions of stability, such as (γ, β) -finite gain stability (fgs) and (γ, β) -incremental finite gain stability (ifgs). With respect to the input-output mapping $(w \to y)$, these are defined as follows:

$$\begin{aligned} (\gamma,\beta) - \ell_p - fgs: & \boldsymbol{Q}(\boldsymbol{y}) = \boldsymbol{w}, \ \boldsymbol{w} \in \ell_p^{\mathcal{X}} \implies \|\boldsymbol{y}\|_p \leqslant \gamma \|\boldsymbol{w}\|_p + \beta, \\ (\gamma,\beta) - \ell_p - ifgs \text{ at } \boldsymbol{w}_0: & \boldsymbol{Q}(\boldsymbol{y}) = \boldsymbol{w}, \ \boldsymbol{w} \in \ell_p^{\mathcal{X}} \implies \|\boldsymbol{\delta}\boldsymbol{y}\|_p \leqslant \gamma \|\boldsymbol{\delta}\boldsymbol{w}\|_p + \beta. \\ & \boldsymbol{Q}(\boldsymbol{y}_0) = \boldsymbol{w}_0, \ \boldsymbol{w}_0 \in \ell_p^{\mathcal{X}} \\ & \boldsymbol{\delta}\boldsymbol{y} = \boldsymbol{y} - \boldsymbol{y}_0 \\ & \boldsymbol{\delta}\boldsymbol{w} = \boldsymbol{w} - \boldsymbol{w}_0. \end{aligned}$$

Next, we focus on deriving sufficient conditions, in the form of small-gain theorems, for finite gain ℓ_p -stability as defined above.

Conditions for Finite-Gain Stability

For the following derivations, we investigate pairs $(\boldsymbol{y}, \boldsymbol{w})$ which are solutions to (2.7) and from \boldsymbol{y} we define the scalar sequence $\boldsymbol{s} \in \mathbb{R}^{\infty}$ such that $s_t := \|\boldsymbol{P}^t(\boldsymbol{y})\|_p$, i.e.,

$$s_t := \begin{cases} \sqrt[p]{\sum_{k=0}^t |y_k|^p} & \text{for } p < \infty \\ \sup_{k \le t} |y_k| & \text{for } p = \infty. \end{cases}$$
(2.10)

Furthermore, for convenience we abbreviate the causal operator (I - Q) as $\Delta := I - Q$ to rewrite the equation (2.8a) as

$$oldsymbol{y} = oldsymbol{\Delta}(oldsymbol{y}) + oldsymbol{w}$$

Next, we bound s_t from above, which results in a key inequality used to prove the later small-gain stability conditions. To this end, substitute $y = \Delta(y) + w$, into the

definition (2.10) and by causality of Δ we get an equivalent expression for s_t :

$$s_t = \|\boldsymbol{P}^t(\boldsymbol{y})\|_p = \|\boldsymbol{P}^t(\boldsymbol{\Delta}(\boldsymbol{y}) + \boldsymbol{w})\|_p.$$
(2.11)

From (2.11) and with the help of Lem. 3 we obtain the inequalities below:

Lemma 6. For each term s_t of the sequence s holds:

$$s_t \leq \left\| \boldsymbol{\Delta}(\boldsymbol{P}^t(\boldsymbol{y})) \right\|_p + \left\| \boldsymbol{w} \right\|_p$$
(2.12)

and if $\Delta := (I - Q)$ is strictly causal, then the bound below holds:

$$s_t \leq \left\| \boldsymbol{\Delta}(\boldsymbol{P}^{t-1}(\boldsymbol{y})) \right\|_p + \left\| \boldsymbol{w} \right\|_p.$$
(2.13)

Proof. The first bound follows by applying Lem. 3 to (2.11) directly:

$$s_t = \left\| \boldsymbol{P}^t(\boldsymbol{\Delta}(\boldsymbol{y}) + \boldsymbol{w}) \right\|_p \leq \left\| \boldsymbol{\Delta}(\boldsymbol{P}^t(\boldsymbol{y})) \right\|_p + \left\| \boldsymbol{w} \right\|_p.$$

For the second inequality, we have to do a little more work:

$$s_{t} = \left\| (\boldsymbol{P}^{t} \boldsymbol{\Delta})(\boldsymbol{y}) + \boldsymbol{w} \right\|_{p} = \left\| (\boldsymbol{P}^{t} \boldsymbol{\Delta} \boldsymbol{P}^{t-1})(\boldsymbol{y}) + \boldsymbol{w} \right\|_{p}$$
(2.14a)

$$\leq \left\| (\boldsymbol{P}^{t} \boldsymbol{\Delta})(\boldsymbol{P}^{t-1} \boldsymbol{y}) \right\|_{p} + \left\| \boldsymbol{w} \right\|_{p}$$
(2.14a)

$$\leq \left\| (\boldsymbol{\Delta} \boldsymbol{P}^{t})(\boldsymbol{P}^{t-1} \boldsymbol{y}) \right\|_{p} + \left\| \boldsymbol{w} \right\|_{p} = \left\| (\boldsymbol{\Delta} \boldsymbol{P}^{t} \boldsymbol{P}^{t-1}) \boldsymbol{y} \right\|_{p} + \left\| \boldsymbol{w} \right\|_{p}$$
(2.14b)

$$= \left\| \boldsymbol{\Delta} (\boldsymbol{P}^{t-1}(\boldsymbol{y})) \right\|_{p} + \left\| \boldsymbol{w} \right\|_{p}.$$
(2.14b)

With the above Lemma 6, we present sufficient conditions for the stability (2.8a) with their local/global finite-gain versions.

Theorem 2 (Small Gain Theorem). Assume that the operator Δ satisfies $\|\Delta(x)\|_p \leq \gamma \|x\|_p + \beta$ for all $x \in \ell_p^{\chi}$ and some small gain $\gamma < 1$. Then, correspondingly for all $w \in \ell_p^{\chi}$, the system response y satisfies the bound

$$\|\boldsymbol{y}\|_{p} \leq \frac{1}{1-\gamma}(\|\boldsymbol{w}\|_{p}+\beta).$$

Proof. Applying our assumption to the bound (2.12), we obtain:

$$s_t \leq \left\| \boldsymbol{\Delta}(\boldsymbol{P}^t(\boldsymbol{y})) \right\|_p + \left\| \boldsymbol{w} \right\|_p \leq \gamma \boldsymbol{P}^t(\boldsymbol{y}) + \beta + \left\| \boldsymbol{w} \right\|_p$$
(2.15)

$$\leq \gamma s_t + \beta + \|\boldsymbol{w}\|_p \tag{2.16}$$

$$\Leftrightarrow \qquad s_t \leq \frac{1}{1-\gamma} (\beta + \|\boldsymbol{w}\|_p). \tag{2.17}$$

This establishes the boundedness of s. Since s is nondecreasing by construction, we know that $\lim_{t\to\infty} s_t = s^* = \|\boldsymbol{y}\|_p$ exists and satisfies $\|\boldsymbol{y}\|_p \leq \frac{1}{1-\gamma}(\|\boldsymbol{w}\|_p + \beta)$. \Box

We can also obtain a local version of that result, provided Δ is strictly causal and satisfies the small-gain property locally:

Lemma 7 (Local Small Gain Theorem). Assume that for some $\rho > 0$ and $0 \le \gamma < 1$, $\beta \ge 0$ the operator $\mathbf{I} - \mathbf{Q}$ is strictly causal and satisfies $\|(\mathbf{I} - \mathbf{Q})(\mathbf{x})\|_p \le \gamma \|\mathbf{x}\|_p + \beta$ for all $\|\mathbf{x}\|_p < \rho$ and some $p \in \{1, 2, ..., \infty\}$. Then, for any (\mathbf{y}, \mathbf{w}) such that $\mathbf{Q}(\mathbf{y}) = \mathbf{w}$ holds:

$$\|\boldsymbol{w}\|_{p} < (1-\gamma)\rho - \beta \qquad \Longrightarrow \qquad \|\boldsymbol{y}\|_{p} \leq \frac{1}{1-\gamma}(\|\boldsymbol{w}\|_{p} + \beta).$$

Proof. Our assumption $\|\boldsymbol{w}\|_p < (1 - \gamma)\rho - \beta$ can be equivalently stated as:

$$\|\boldsymbol{w}\|_{p} < (1-\gamma)\rho - \beta \Leftrightarrow \frac{\|\boldsymbol{w}\|_{p} + \beta}{(1-\gamma)} < \rho.$$
(2.18)

We proceed to show $s_t \leq (\|\boldsymbol{w}\|_p + \beta)/(1 - \gamma)$ for all t per induction: $\underline{t} = 0$: $s_0 = |w_0| \leq \|\boldsymbol{w}\|_p \leq (\|\boldsymbol{w}\|_p + \beta)/(1 - \gamma) < \rho$, since according to assumption $\|\boldsymbol{w}\|_p < (1 - \gamma)\rho - \beta$ which is equivalent to the last inequality. $\underline{t} \rightarrow \underline{t} + 1$: Assume s_t satisfies $s_t \leq (\|\boldsymbol{w}\|_p + \beta)/(1 - \gamma)$. Then, due to (2.18), we have $s_t = \|\boldsymbol{P}^t(\boldsymbol{y})\|_p < \rho$ and using the small gain property we know:

$$\left\|\boldsymbol{\Delta}(\boldsymbol{P}^{t}(\boldsymbol{y}))\right\|_{p} \leq \gamma \left\|\boldsymbol{P}^{t}(\boldsymbol{y})\right\|_{p} + \beta.$$
(2.19)

Substituting the above into (2.13) and using our induction assumption $s_t \leq (\|\boldsymbol{w}\|_p + \beta)/(1-\gamma)$, we obtain:

$$s_{t+1} \leqslant \gamma \left\| \boldsymbol{P}^{t}(\boldsymbol{y}) \right\|_{p} + \beta + \left\| \boldsymbol{w} \right\|_{p} = \gamma s_{t} + (1 - \gamma) \frac{\left\| \boldsymbol{w} \right\|_{p} + \beta}{1 - \gamma}$$
(2.20a)

$$\leq \gamma \frac{\|\boldsymbol{w}\|_{p} + \beta}{1 - \gamma} + (1 - \gamma) \frac{\|\boldsymbol{w}\|_{p} + \beta}{1 - \gamma} = \frac{\|\boldsymbol{w}\|_{p} + \beta}{1 - \gamma},$$
(2.20b)

hence, (2.20b) completes the induction step and we can conclude that s_t is bounded above by $(\|\boldsymbol{w}\|_p + \beta)/(1 - \gamma) < \rho$ for all t.

Finally, since s_t is non-decreasing per construction, we know that $\lim_{t\to\infty} s_t = s^* = \|\boldsymbol{y}\|_p$ exists and satisfies

$$\|\boldsymbol{y}\|_{p} \leq \frac{1}{1-\gamma}(\|\boldsymbol{w}\|_{p}+\beta)$$

If we associate w_0 as the initial condition y_0 and we can rewrite the bounds of Theorem 7 and Theorem 2 as:

$$\|\boldsymbol{y}\|_{p} \leq \frac{1}{1-\gamma} \left(\sqrt[p]{|y_{0}|_{p}^{p} + \|\boldsymbol{w}\|_{p}^{p}} + \beta \right) \qquad 1 \leq p < \infty \qquad (2.21a)$$

$$\|\boldsymbol{y}\|_{\infty} \leq \frac{1}{1-\gamma} \left(\max\left\{ |y_0|, \|\boldsymbol{w}\|_{\infty} \right\} + \beta \right) \qquad p = \infty.$$
 (2.21b)

Eventually Small Gain Condition

As so far discussed, the small gain condition $(I - Q)(y) < y + \beta$ is sufficient to prove stability of systems governed by the equation Q(y) = w. However, it is not a necessary requirement for stability. The next Lemma shows that we can relax the small gain condition and instead demand, that the gain of the operator $\Delta := I - Q$ is *eventually* smaller than 1: There exists some t_0 such that the operators $\overline{P^{t_0}x} \mapsto \overline{P^{t_0}\Delta}(U^{t_0}z + \overline{P^{t_0}}(x))$ satisfy the small gain property for any fixed bounded $z \in \mathbb{R}^{N_{t_0}}$.

Lemma 8. Let $(\boldsymbol{y}, \boldsymbol{w})$ be solutions to the operator equation $\boldsymbol{Q}\boldsymbol{y} = \boldsymbol{w}$ for some causal $\boldsymbol{Q} \in \mathcal{C}(\ell^{\mathcal{X}})$. Then, the implication $\boldsymbol{w} \in \ell_p^{\mathcal{X}} \implies \boldsymbol{y} \in \ell_p^{\mathcal{X}}$ holds true, if there exists $t_0 \in \mathbb{N}$ for which the operator $\boldsymbol{\Delta} = \boldsymbol{I} - \boldsymbol{Q}$ meets the following conditions:

(i) There exist $C_1, C_2 > 0$ such that for all solutions $(\boldsymbol{y}, \boldsymbol{w})$, the following ℓ_p -f.g.s condition holds true:

$$\boldsymbol{P}^{t_0}\boldsymbol{w} \in \ell_p^{\mathcal{X}} \implies \|\boldsymbol{P}^{t_0}\boldsymbol{y}\|_p \leq C_1 \|\boldsymbol{P}^{t_0}\boldsymbol{w}\|_p + C_2.$$
 (2.22)

(ii) There exist some $\gamma < 1$ and non-decreasing function $\beta : \mathbb{R}^+ \to \mathbb{R}^+$ such that for all $\mathbf{x} \in \ell_n^{\mathcal{X}}$:

$$\|\overline{\boldsymbol{P}^{t_0}}\boldsymbol{\Delta}(\boldsymbol{x})\|_p \leqslant \gamma \|\overline{\boldsymbol{P}^{t_0}}(\boldsymbol{x})\|_p + \beta (\|\boldsymbol{P}^{t_0}(\boldsymbol{x})\|_p).$$

Proof. Let $(\boldsymbol{y}, \boldsymbol{w})$ be an arbitrary pair such that $\boldsymbol{Q}\boldsymbol{y} = \boldsymbol{w}$ or equivalently $\boldsymbol{y} = \Delta \boldsymbol{y} + \boldsymbol{w}$, and assume that $\boldsymbol{w} \in \ell_p^{\mathcal{X}}$. Split \boldsymbol{y} into the sum $\boldsymbol{P}^{t_0}\boldsymbol{y} + \overline{\boldsymbol{P}^{t_0}}\boldsymbol{y}$ and recall from (2.22) that $\|\boldsymbol{P}^{t_0}\boldsymbol{y}\|_p \leq C_1 \|\boldsymbol{P}^{t_0}\boldsymbol{w}\|_p + C_2$. Hence, to show $\boldsymbol{y} \in \ell_p^{\mathcal{X}}$, we need to show that $\overline{\boldsymbol{P}^{t_0}}\boldsymbol{y} \in \ell_p^{\mathcal{X}}$. Let $s_{t_0} = \|\boldsymbol{P}^{t_0}\boldsymbol{y}\|$, $s_{k|t_0} = \|\boldsymbol{P}_n^{\mathcal{I}_k}(\boldsymbol{y})\|_p$, where \mathcal{I}_k denote the intervals $[t_0, t_0 + k]$ for $k \in \mathbb{N}$, and notice the following chain of inequalities:

$$s_{k|t_0} = \|\boldsymbol{P}^{\mathcal{I}_k}\boldsymbol{y}\|_p \leqslant \|\boldsymbol{P}^{\mathcal{I}_k}\boldsymbol{\Delta}\boldsymbol{P}^{t_0+k}\boldsymbol{y}\|_p + \|\boldsymbol{P}^{\mathcal{I}_k}\boldsymbol{w}\|_p \leqslant \|\boldsymbol{\Delta}\boldsymbol{P}^{t_0+k}\boldsymbol{y}\|_p + \|\boldsymbol{P}^{\mathcal{I}_k}\boldsymbol{w}\|_p$$
$$\leqslant \gamma \|\boldsymbol{P}^{\mathcal{I}_k}(\boldsymbol{y})\|_p + \beta (\|\boldsymbol{P}^{t_0}\boldsymbol{y}\|_p) + \|\boldsymbol{P}^{\mathcal{I}_k}\boldsymbol{w}\|_p$$
$$\Leftrightarrow \qquad (1-\gamma)s_{k|t_0} \leqslant \beta(s_{t_0}) + \|\boldsymbol{P}^{\mathcal{I}_k}\boldsymbol{w}\|_p$$
$$\Leftrightarrow \qquad s_{k|t_0} \leqslant \frac{1}{1-\gamma} \left(\beta(s_{t_0}) + \|\overline{\boldsymbol{P}^{t_0}}\boldsymbol{w}\|_p\right).$$

This shows that $s_{k|t_0}$ is bounded for any $k \in \mathbb{N}$ and together with the fact that $s_{k|t_0}$ is non-decreasing in k, it also proves convergence of the sequence $(s_{[0,t]_0}, s_{1|t_0}, ...)$ to $\lim_{k\to\infty} s_{k|t_0} = s_{\infty|t_0} < \infty$. Hence, we showed $\overline{P^{t_0}} y \in \ell_p^{\mathcal{X}}$ and $\|y\|_p \leq s_{t_0} + s_{\infty|t_0}$ is bounded above as

$$\|\boldsymbol{y}\|_{p} \leq \frac{1}{1-\gamma}\eta(\|\boldsymbol{w}\|_{p}) + C_{1}\|\boldsymbol{w}\|_{p} + C_{2},$$

where $\eta(x) := \beta(C_{1}x + C_{2}) + x.$

2.4 Representing Dynamical Systems as Sets, Relations, and Maps

A discrete-time dynamical system S is a system whose behavior over time can be described by a function $S : \mathbb{N} \to S$ over the index set \mathbb{N} , where each S(t) represents the state of the system S at time-step t. Correspondingly, S is usually called the state space of the dynamical system and represents the granularity of our system description. Depending on the application, S can be a finite set, a finite- or infinitedimensional space of vectors, or even functions. A sequence $s = (s_0, s_1, \ldots) \in S^{\mathbb{N}}$, $S(t) = s_t$, describing a specific realization of the system's behavior is called a state trajectory of S.

Models are used to characterize the behavior of a dynamical system and ideally provide a minimal, yet complete representation of the behavior of S over time, capable of describing all possible state trajectories s of S. There are many ways to formulate such a representation. A natural approach, often easiest when S is known to obey certain laws (e.g., the laws of physics), is to describe the system behavior implicitly as a set of solutions to a set of difference equations. If the behavior of S is random, another suitable representation is to model S as a stochastic process over \mathbb{N} , where the system state S_t at time t is represented as a random variable with support S.

However, as our starting point, we choose a more explicit (and deterministic) formulation by identifying the behavior of a dynamical system S with its set $M_S \subset S^{\mathbb{N}}$ of all possible state trajectories τ^x . We call the set M_S the *dynamic model* or model of the system S.

Remark 3. Many fundamental concepts of systems theory such as controllability, observability, stability, etc., have an equivalent formulation in this representation. This alternative framework of control and systems theory was first developed by Jan C. Willems [131] and is known as the Behavioral Approach to Control.

Input-Output Maps of Dynamical Models Often the state space S is represented by a family $\{Z_i\}_i, i \in \mathcal{I}$, of smaller spaces indexed over some index set \mathcal{I} ; often \mathcal{I} is used to further specify the internal structure of the system S (see example below).

Example 1. Assume that S represents the behavior of a large-scale system composed of a family of interconnected dynamical systems $\{X_i\}_{i \in \mathcal{I}_1}$ driven by a distributed family of actuators $\{U_j\}_{j \in \mathcal{I}_2}$ and a collection of external disturbance sources $\{W_j\}_{j \in \mathcal{I}_3}$. Then a natural choice to define S is as the Cartesian product $\mathcal{X} \times \mathcal{U} \times \mathcal{W}$ of the spaces $\mathcal{X} = \bigotimes_{i \in \mathcal{I}_1} \mathcal{X}_i$, $\mathcal{U} = \bigotimes_{j \in \mathcal{I}_2} \mathcal{U}_j$, and $\mathcal{Z} = \bigotimes_{k \in \mathcal{I}_3} \mathcal{Z}_k$ corresponding to the families of subsystems, actuators, and disturbance sources.

Sometimes a dynamical system offers a natural way to divide \mathcal{I} into a union $\mathcal{I}^{in} \cup \mathcal{I}^{out}$ of inputs \mathcal{I}^{in} representing independent variables $Z_i, i \in \mathcal{I}^{in}$ and outputs \mathcal{I}^{out} such that the variables $Z_i, i \in \mathcal{I}^{out}$ depend causally on the variables $Z_i, i \in \mathcal{I}$. A dynamical model has a viable assignment of inputs \mathcal{I}^{in} and outputs \mathcal{I}^{out} if and only if the binary relation $\{(\{z_i\}_{i\in\mathcal{I}^{in}}, \{z_j\}_{j\in\mathcal{I}^{out}}) | \{z_i\}_{i\in\mathcal{I}} \in M_S\}$ is causal according to the following definition:

Definition 2.4. A binary relation $\mathcal{R} \subset \mathcal{X}^{\mathbb{N}} \times \mathcal{Y}^{\mathbb{N}}$ is said to be causal if for any $(\boldsymbol{x}, \boldsymbol{y}) \in \mathcal{R}$ and $(\boldsymbol{x}', \boldsymbol{y}') \in \mathcal{R}$, and any $t \in \mathbb{N}$, the following implication holds true:

$$oldsymbol{P}^t oldsymbol{x} = oldsymbol{P}^t oldsymbol{x}' \quad \Longrightarrow \quad oldsymbol{P}^t oldsymbol{y} = oldsymbol{P}^t oldsymbol{y}'$$
 .

Remark 4. The above condition is equivalent to requiring that for any t, the relation $\{(\mathbf{P}^t \mathbf{x}, \mathbf{P}^t \mathbf{y}) \mid (\mathbf{x}, \mathbf{y}) \in \mathcal{R}\}$ is functional. Hence, if true, it defines a unique causal operator $\Psi : \operatorname{do}(\mathcal{R}) \mapsto \mathcal{Y}^{\mathbb{N}}$ such that $\Psi(\mathbf{x}) = \mathbf{y} \Leftrightarrow (\mathbf{x}, \mathbf{y}) \in \mathcal{R}$.

From the above definition, we see that a viable assignment of inputs \mathcal{I}^{in} and outputs \mathcal{I}^{out} implicitly defines a unique causal mapping between inputs $s^{in} = \{z_i\}_{i \in \mathcal{I}^{in}}$ and outputs $s^{out} = \{z_i\}_{i \in \mathcal{I}^{out}}$ of trajectories $s = \{z\}_{i \in \mathcal{I}} \in M_S$. We call the operator representing the mapping $\Psi : s^{in} \mapsto s^{out}$ the $\mathcal{I}^{in} \mapsto \mathcal{I}^{out}$ -map of the dynamical model M_S . We formulate this definition and summarize our discussion below:

Definition 2.5 (Input-Output Maps of Dynamical Models). Let $M_S \subset S^{\mathbb{N}}$ be a dynamical model, where $S = \{Z_i\}_{i \in \mathcal{I}}$. Consider two subsets $\mathcal{I}^{in}, \mathcal{I}^{out} \subset \mathcal{I}$ such that $\mathcal{I}^{in} \cup \mathcal{I}^{out} = \mathcal{I}$, and for a trajectory $s = \{z\}_{i \in \mathcal{I}} \in M_S$, let $s^{in} = \{z\}_{i \in \mathcal{I}^{out}}$.

We say that the pair \mathcal{I}^{in} and \mathcal{I}^{out} is an input-output assignment for M_S if the binary relation $\mathcal{R}^{\mathcal{I}^{in} \mapsto \mathcal{I}^{out}}$ defined below is causal:

$$\mathcal{R}^{\mathcal{I}^{in}\mapsto\mathcal{I}^{out}} \triangleq \{(\boldsymbol{s}^{in}, \boldsymbol{s}^{out}) \mid \boldsymbol{s} \in \mathsf{M}_S\}.$$

Furthermore, under this condition, there exists a unique causal operator Ψ : $\operatorname{do}(\mathcal{R}^{\mathcal{I}^{in}\mapsto\mathcal{I}^{out}})\to\mathcal{S}^{out}$ such that

$$s^{out} = \Psi(s^{in}) \Leftrightarrow s \in \mathsf{M}_S$$

This causal operator is referred to as the $\mathcal{I}^{in} \mapsto \mathcal{I}^{out}$ -map of M_S .

If an input-output map exists, it provides a complete description of the dynamical system S, similar to M_S . However, in general, a viable input-output assignment is not always possible, and if one exists, it is not always unique. Hence, while M_S is uniquely tied to the dynamical system S, there might be different input-output maps of S depending on how many viable assignments of input-output pairs there are. Therefore, input-output maps of dynamic model M_S have to be distinguished by the respective choice of inputs \mathcal{I}^{in} and outputs \mathcal{I}^{out} .

Closed Loop Maps of Dynamical Models In a standard control problem setup, we are given a dynamical system, called the plant, with state X, which we control with some input U using measurements Y, and which is subjected to external (non-measurable) disturbances W. We represent the time behavior of this control system as a dynamical system with state S = (X, U, W, Y) and state space $S = (\mathcal{X}, \mathcal{U}, \mathcal{W}, \mathcal{Y})$ and denote trajectories of S as a tuple $\tau = (\tau^x, \tau^u, \tau^y, \tau^w)$ composed of the individual state, input, output, and disturbance sequences $\tau^x \in \mathcal{X}^{\mathbb{N}}, \tau^u \in \mathcal{U}^{\mathbb{N}}, \tau^y \in \mathcal{Y}^{\mathbb{N}}$, and $\tau^w \in \mathcal{W}^{\mathbb{N}}$ corresponding to a realization of the dynamics of the control system S. Correspondingly, the set of all such realizations forms the dynamical model $M_S \subset S^{\mathbb{N}}$, which describes the behavior of the control system S in open-loop.

On the other hand, introducing a feedback controller $\mathbf{K} \in \mathcal{C}(\mathcal{Y}^{\mathbb{N}}, \mathcal{U}^{\mathbb{N}})$ leads to a different dynamical system $S_{\mathbf{K}}$, describing the closed-loop dynamics of the interconnection of the open-loop system S and controller \mathbf{K} . We say that the controller \mathbf{K} realizes the closed-loop system $S_{\mathbf{K}}$ or equivalently, realizes the closedloop model $M_{S_{\mathbf{K}}}$. The relation between open and closed-loop system can be simply stated in terms of their dynamic models:

$$\mathsf{M}_{S_{\boldsymbol{K}}} = \{\boldsymbol{\tau} \in \mathsf{M}_{S} \mid \boldsymbol{\tau}^{u} = \boldsymbol{K}(\boldsymbol{\tau}^{y})\} = \mathsf{M}_{S} \cap \mathsf{M}_{\boldsymbol{K}}$$

Thus, the model of the closed-loop system S_K is a subset of the model of the open-loop system S, and more specifically, it is obtained from M_S by imposing the additional constraint $\tau^u = K(\tau^y)$ on the trajectories of S. If we represent the controller K as a dynamical model $M_K = \{\tau \in S^{\mathbb{N}} \mid \tau^u = K(\tau^y)\}$ embedded in the space $S^{\mathbb{N}}$, then we can also simply view the closed-loop model as an intersection of the open-loop and controller models. From this point of view, we can characterize the space of closed-loop systems realizable by some causal operator $K : \mathcal{Y}^{\mathbb{N}} \to \mathcal{U}^{\mathbb{N}}$ as the following set of dynamical models, which can be referred to as the space of realizable closed-loop models of the system S:

$$\mathbb{M}_{S} = \left\{ \mathsf{M}_{S} \cap \mathsf{M}_{K} \mid K \in \mathcal{C}(\mathcal{Y}^{\mathbb{N}}, \mathcal{U}^{\mathbb{N}}) \right\}.$$

Maps of closed-loop models $M_{S_{K^*}} \in \mathbb{M}_S$ are usually called closed-loop maps of S, which will be the focus of the discussion in the next section.

Remark 5. In later discussion, we consider dynamical systems where for all $M^* \in M_S$, the pair $\mathcal{I}_{in} = \{w\}$ and $\mathcal{I}_{out} = \{xuy\}$ is a valid input-output pairing. This permits us to represent the space of realizable closed-loop system behavior as a space of operators, which we later denote by $\Phi_S^{w \mapsto \{uxy\}}$ and refer to as the space of closed-loop maps of S or M_S .

In this section, we explored the concept of general exterior representations of dynamical systems, which can be understood as a collection of sequences that encompass all possible manifestations of system behavior. This discussion led us to examine dynamic models and the causality conditions that enable us to establish correspondences between dynamic models and maps linking inputs to outputs. Additionally, we delved into the relationship between the model of an open-loop system and the space of realizable closed-loop models through feedback control.

Building upon these ideas, in the upcoming section, we will shift our focus to an interior representation of dynamical systems. This representation takes the form of nonlinear difference and operator equations, allowing us to establish connections and equivalences with the exterior representations discussed earlier, and leads to the main result of this chapter: the characterization and realization of closed-loop maps via solutions of the CLM operator equation.

2.5 Dynamic Equations of the Open-Loop System

Here we discuss the class of discrete-time systems of interest and define our setup of the open-loop system S and the governing dynamic equations.

The open-loop system S is a dynamical system with state space $S = S_x \times S_u \times S_w = X_{i \in \mathcal{I}} S_i, \mathcal{I} = \{x, u, w\}$, where S_x, S_w, S_u denote the spaces corresponding to the state of the plant, control action, and external disturbance. Moreover, let $\mathcal{X} = S_x = S_w$ and $\mathcal{U} = S_u$, where \mathcal{X} and \mathcal{U} are some fixed finite-dimensional vector spaces, e.g., $\mathcal{X} = \mathbb{R}^n, \mathcal{U} = \mathbb{R}^m$. We define the open-loop system S in terms of its trajectories $\boldsymbol{\tau} = (\boldsymbol{x}, \boldsymbol{u}, \boldsymbol{w})^{\mathsf{T}}$ as follows:

Let $x \in \ell^{\mathcal{X}}$ be the state trajectory of a nonlinear discrete-time dynamical system, and let $u \in \ell^{\mathcal{U}}$ and $w \in \ell^{\mathcal{X}}$ be the corresponding sequences of control inputs and disturbances that generate the trajectory according to the following set of equations:

$$\forall t \ge 1: \quad x_t = f_{t-1}(x_{t-1:0}, u_{t-1:0}) + w_t, \quad x_0 = w_0, \tag{2.23}$$

where $f := \{f_t\}_{t=0}^{\infty}, f_t : \mathcal{X}^{[0,t]} \times \mathcal{U}^{[0,t]} \to \mathcal{X}^{[0,t]}$ is a fixed sequence of functions representing the (open-loop) dynamics of the system. We make *no* further assumptions on f, i.e., each f_t can be an arbitrary nonlinear function. We can obtain a more compact description of the dynamics by embedding the function sequence f as the component functions of a causal operator $\mathbf{F} : \ell^{\mathcal{X} \times \mathcal{U}} \to \ell^{\mathcal{X}}$. Thus, let $\mathbf{F} \in \mathcal{C}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$, and its strictly causal right-shifted version $\mathbf{F}^+ := \mathcal{S}^+ \mathbf{F} \in \mathcal{C}_s(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ be defined in terms of f as:

$$\boldsymbol{F}(\boldsymbol{x}, \boldsymbol{u}) := (f_0(x_0, u_0), f_1(x_1, x_0, u_1, u_0), \dots, f_t(x_{t-1:0}, u_{t-1:0}), \dots) \quad (2.24)$$

$$F^{+}(x, u) := S^{+}F(x, u) = (0, f_{0}(x_{0}, u_{0}), f_{1}(x_{1:0}, u_{1:0}), \dots).$$
(2.25)

We refer to the operator F as the open-loop *dynamics* of the system, since the dynamic equations (2.23) can be equivalently defined in sequence space as

$$\boldsymbol{x} = \mathcal{S}^+ \boldsymbol{F}(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w} =: \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w}. \tag{2.26}$$

The above equation is a compact description of the state transitions of the dynamical system; hence, one can view this as an *internal* characterization of the system dynamics. On the other hand, the corresponding external representation of the open-loop system S is given by the dynamical model $M_F \subset \ell^X \times \ell^U \times \ell^X$, which is defined for a fixed dynamics operator F as:

$$\mathbf{M}_{\boldsymbol{F}} := \left\{ \boldsymbol{\tau} = (\boldsymbol{\tau}^{\mathrm{x}}, \boldsymbol{\tau}^{\mathrm{u}}, \boldsymbol{\tau}^{\mathrm{w}})^{\top} \mid \boldsymbol{\tau}^{\mathrm{x}} = \boldsymbol{F}^{+}(\boldsymbol{\tau}^{\mathrm{x}}, \boldsymbol{\tau}^{\mathrm{u}}) + \boldsymbol{\tau}^{\mathrm{w}} \right\}.$$
 (2.27)

Hence, as discussed in the previous section, M_F is the set of all possible trajectories $\tau = (x, u, w)^{\top}$ of S, represented as a tuple of state x, input u, and disturbance w sequences corresponding to realizations of the open-loop system.

From the dynamic equations, it is also easy to see that w, u can be treated as input signals of the open-loop system and the resulting state trajectory x as the corresponding output. In other words, $\{u, w\}$ and $\{x\}$ is a valid input-output assignment of the open-loop model M_F , and it has, therefore, a $\{u, w\} \mapsto x$ -map, which we refer to as the open-loop map G_F . We derive the closed-form of G_F next.

Remark. Aside from G_F , M_F also has other valid input-output map representations. For example, $\mathcal{I}_{in} = \{x, u\}$, $\mathcal{I}_{out} = \{w\}$ is a valid input-output assignment as well, and therefore M_F can be equivalently represented as a $\{x, u\} \mapsto w$ -map.

Open-Loop Maps

Since a system state trajectory x is always a causal function of the input u and the disturbance w, it is clear that $G_F : (u, w) \mapsto x$, st.: $(x, u, w)^\top \in M_F$, represents a well-defined mapping. We call $G_F \in C(\ell^{X \times U}, \ell^X)$ the open-loop map of dynamics F, and notice that the model set M_F is the graph of G_F . For a fixed u, denote $F^+|_u : \ell^X \mapsto \ell^X$ as the restriction of F^+ such that $F^+|_u(x) = F^+(x, u)$. Then, for each trajectory $(x, u, w)^\top \in M_F$, holds $x = (I - F^+|_u)^{-1}w$ (inverse exists due to Proposition 5), and therefore G_F is the mapping

$$G_F: (\boldsymbol{u}, \boldsymbol{w}) \mapsto (\boldsymbol{I} - \boldsymbol{F}^+|_{\boldsymbol{u}})^{-1} \boldsymbol{w}.$$

In this section, we discussed the relationships between different representations of the open-loop system S: as a set of dynamic equations $\boldsymbol{x} = \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w}$, as a dynamic model M_F , and as a causal input-output map $\boldsymbol{G}_F : (\boldsymbol{u}, \boldsymbol{w}) \mapsto (\boldsymbol{I} - \boldsymbol{F}^+|_{\boldsymbol{u}})^{-1} \boldsymbol{w}$. In the next section, we discuss this trinity of representations for the closed-loop system.

2.6 Closed-Loop Maps and Realizing Controllers

In this section, we introduce feedback control and characterize the dynamics of the closed-loop, by establishing a correspondence between the closed-loop system's exterior representation, as closed-loop maps, and the internal representation, as dynamics operators and system equations. In short, the key theoretical results discuss a characterization of the space of realizable CLMs as the solution space of an operator equation and the one-to-one correspondence between CLMs and their realizing controllers.

The Closed-Loop System

Let $K \in C(\ell^{\chi}, \ell^{\mathcal{U}})$ represent a causal feedback control law, and let the component functions K_t denote the decision rule for u_t based on the state observations x_0, \ldots, x_t :

$$\boldsymbol{K}(\boldsymbol{x}) := (K_0(x_0), \dots, K_t(x_{t:0}), \dots).$$
(2.28)

Correspondingly, let $M_{K} = \{ \tau^{xu} = (\tau^{x}, \tau^{u}) \mid \tau^{u} = K\tau^{x} \}$ denote the dynamic model of K. The interconnection of (2.23) with the open-loop system S defines a dynamical system CL, which we refer to as the closed-loop, defined by the following dynamic equations:

CL:
$$x_t = f_{t-1}(x_{t-1:0}, u_{t-1:0}) + w_t, \quad x_0 = w_0$$
 (2.29a)

$$u_t = K_t(x_{t:0}). (2.29b)$$

Equivalently, the dynamic system equations of the closed-loop CL can be written as:

CL:
$$\boldsymbol{x} = \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{K}(\boldsymbol{x})) + \boldsymbol{w}, \quad \boldsymbol{u} = \boldsymbol{K}(\boldsymbol{x})$$
 (2.30)
 $= \boldsymbol{F}^+_{\boldsymbol{K}}(\boldsymbol{x}) + \boldsymbol{w},$

where F_K denotes the mapping $x \mapsto F(x, K(x))$, also referred to as the closed-loop dynamics operator.

Definition (Operator of Closed-Loop Dynamics). For $F \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and $K \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$, define the operator $F_K \in C(\ell^{\mathcal{X}})$ for all $x \in \ell^{\mathcal{X}}$ as $F_K(x) := F(x, K(x))$.

As formulated in Section 2.4, intersecting the open-loop model M_F with the controller model M_K yields $CL_{[F,K]} \triangleq M_F \cap M_K$, the dynamic model of the closed-loop system CL. $CL_{[F,K]}$ is the set of all trajectories $\tau_{xuw} = (x, u, w)$ that satisfy the equations 2.30 for some fixed pair of F and K and hence admits the equivalent definition: and define the set of all trajectories $\tau = (x, u, w)^{\top}, x \in \ell^{\mathcal{X}}, u \in \ell^{\mathcal{U}}$, and $w \in \ell^{\mathcal{X}}$ which are solutions to the above equation as the *closed-loop* model $CL_{[F,K]}$:

$$\mathsf{CL}_{[F,K]} = \left\{ (\boldsymbol{x}, \boldsymbol{u}, \boldsymbol{w})^\top \mid \boldsymbol{x} = F^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w}, \quad \boldsymbol{u} = K(\boldsymbol{x}) \right\}.$$
(2.31)

Recall that $F^+ \in C_s(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and therefore the map $F_K^+ \in C_s(\ell^{\mathcal{X}})$ is also strictly causal. As shown in (5), this observation certifies that the operator $I - F_K^+$ has a causal inverse $(I - F_K^+)^{-1} \in C(\ell^{\mathcal{X}})$. Hence, we can equivalently rewrite the equation $x = F_K(x) + w$ as $x = (I - F_K^+)^{-1}w$ and see that $(x, u, w)^{\top}$ is a trajectory of $CL_{[F,K]}$ if and only if it satisfies the equations:

$$x = (I - F_K^+)^{-1} w$$
 (2.32a)

$$\boldsymbol{u} = \boldsymbol{K} (\boldsymbol{I} - \boldsymbol{F}_{\boldsymbol{K}}^{+})^{-1} \boldsymbol{w}. \tag{2.32b}$$

In other words, for any F and K, the operator $\begin{bmatrix} I \\ K \end{bmatrix} (I - F_K^+)^{-1} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ represents the $w \mapsto \begin{bmatrix} x \\ u \end{bmatrix}$ -map of the closed-loop model $\mathsf{CL}_{[F,K]}$.

Remark. In terms of the discussion in Section 2.4, 2.32 proves the causality of the relation $\mathcal{R}^{w\mapsto x,u} = \{(w, \begin{bmatrix} x \\ u \end{bmatrix}) \mid (x, u, w)^{\top} \in \mathsf{CL}_{[F,K]}\}$ and provides a formula for the $w \mapsto x, u$ -map of the closed-loop model $\mathsf{CL}_{[F,K]}$.

The input-output representation of a closed loop provides a natural way to define closed-loop stability [139]: The closed-loop trajectories $\tau \in CL_{[F,K]}$ have ℓ_p bounded state and input trajectory τ^{xu} for any ℓ_p bounded disturbances τ^w if and only if the $w \mapsto \{x, u\}$ -map of $CL_{[F,K]}$ is ℓ_p -stable, and correspondingly, a controller Kis ℓ_p -stabilizing if and only if the corresponding $w \mapsto \{x, u\}$ -map, i.e., the operator $\begin{bmatrix} I \\ K \end{bmatrix} (I - F_K^+)^{-1}$ is ℓ_p -stable. The next theorem summarizes our discussion so far, and states important properties of the $w \mapsto \{x, u\}$ -map representation of $CL_{[F,K]}$.

Theorem 6 (Input-Output Maps of closed loops). For any fixed pair $\mathbf{F} \in \mathcal{C}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and $\mathbf{K} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$, the closed-loop model $\mathsf{CL}_{[\mathbf{F}, \mathbf{K}]}$ has a $w \mapsto \{x, u\}$ -map $\Psi \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$, and it holds:

i) $\Psi = \begin{bmatrix} \Psi^x \\ \Psi^u \end{bmatrix}$, where $\Psi^x = (I - F_K^+)^{-1}$ and $\Psi^u = K(I - F_K^+)^{-1}$.

ii)
$$\tau \in \mathsf{CL}_{[F,K]}$$
 if and only if $\tau^{\mathrm{xu}} = \Psi(\tau^{\mathrm{w}})$ *.*

iii)
$$\boldsymbol{K} = \boldsymbol{\Psi}^{\mathrm{u}}(\boldsymbol{\Psi}^{\mathrm{x}})^{-1}$$
.

Proof. We have already established that $CL_{[F,K]}$ always has a $w \mapsto \{x, u\}$ -map; according to the setup of the theorem we denote this map Ψ .

i): As discussed previously, i) follows since for any $(\boldsymbol{x}, \boldsymbol{u}, \boldsymbol{w})^{\top} \in \mathsf{CL}_{[\boldsymbol{F},\boldsymbol{K}]}$ holds 2.32, which provides a closed-form expression for Ψ . ii): This is just the restatement of the general definition Def. 2.5. (iii): From i), which is already proven, we have $\Psi^{x} = (\boldsymbol{I} - \boldsymbol{F}_{K}^{+})^{-1}$ and $(\Psi^{x})^{-1} = (\boldsymbol{I} - \boldsymbol{F}_{K}^{+})$, which implicitly states that Ψ^{x} is always causally invertible. Now since $\Psi^{u} = \boldsymbol{K}(\boldsymbol{I} - \boldsymbol{F}_{K}^{+})^{-1}$, multiplying both sides from the right by $(\Psi^{x})^{-1}$ yields $\boldsymbol{K} = \Psi^{u}(\Psi^{x})^{-1}$.

The above result leads us to a very important conclusion: For fixed F, there is a one-to-one correspondence, via the transformation $\mathbb{H} : K \mapsto \begin{bmatrix} I \\ K \end{bmatrix} (I - F_K^+)^{-1}$ and its inverse $\mathbb{H}^{-1} : \begin{bmatrix} \Psi^x \\ \Psi^u \end{bmatrix} \mapsto \Psi^u(\Psi^x)^{-1}$, between a controller K, later called the *realizing controller*, and the corresponding $w \mapsto \{x, u\}$ -map of $\mathsf{CL}_{[F,K]}$. **Remark 7.** In fact, there is a one-to-one correspondence between $w \mapsto \{x, u\}$ -maps of $CL_{[F,K]}$ and the pairs (K, F_K) of controller K and closed-loop dynamics F_K : If Ψ is the $w \mapsto \{x, u\}$ -map of $CL_{[F,K]}$, then $I - (\Psi^x)^{-1} = F_K^+$ and $\Psi^u(\Psi^x)^{-1} = K$; the reverse direction is clear, since $w \mapsto \{x, u\}$ -maps of $CL_{[F,K]}$ always have the form $\Psi = \begin{bmatrix} I \\ K \end{bmatrix} (I - F_K^+)^{-1}$.

This observation motivates us to characterize the space of all $w \mapsto \{x, u\}$ -maps of $\mathsf{CL}_{[F,K]}$. We define this as the space $\Phi_{\mathrm{CL}}^{w \mapsto \{x,u\}}[F]$ of *closed-loop maps* (CLMs) of F and call an operator Ψ in the set $\Phi_{\mathrm{CL}}^{w \mapsto xu}[F]$ a closed-loop map of F:

Definition 2.6 (Closed Loop Maps). An operator $\Psi \in C(\ell^{\chi}, \ell^{\chi \times U})$ is called a closed-loop map (CLM) of F, if for some $\tilde{K} \in C(\ell^{\chi}, \ell^{U})$, called realizing controller of Ψ , the operator Ψ coincides with the $w \mapsto (x, u)$ -map of the closed-loop model $CL_{[F,\tilde{K}]}$. We denote the set of all such Ψ , with $\Phi_{CL}^{w \mapsto xu}[F]$ and call it the space of CLMs of F or the CLM-space of F.

Remark. In terms of the discussion of Section 2.4, the above considers the special case where y = x, i.e., we can measure the state x and used as measurement for feedback control.

The next theorem presents different characterizations of the CLM-space $\Phi_{CL}^{w \mapsto xu}[F]$ and represents a main result of this chapter.

Theorem 8 (Characterization of CLMs). For a fixed causal operator $\mathbf{F} \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$, the following are all equivalent definitions of $\Phi_{CL}^{w \mapsto u}[\mathbf{F}]$:

 $i) \ \Phi_{\mathrm{CL}}^{\mathrm{w}\mapsto\mathrm{xu}}[\boldsymbol{F}] = \{ \boldsymbol{\Psi} \in \mathcal{C}(\ell^{\mathcal{X}},\ell^{\mathcal{X}\times\mathcal{U}}) \mid \boldsymbol{\Psi} \text{ is a CLM of } \boldsymbol{F} \}.$

$$ii) \ \Phi_{\mathrm{CL}}^{\mathrm{w}\mapsto\mathrm{xu}}[\boldsymbol{F}] = \{ \Psi \mid \forall \begin{bmatrix} \boldsymbol{\tau}^{\mathrm{xu}} \\ \boldsymbol{\tau}^{\mathrm{w}} \end{bmatrix} \in \mathrm{CL}_{[\boldsymbol{F}, \Psi^{\mathrm{u}}(\Psi^{\mathrm{x}})^{-1}]} \ s.t. \ \boldsymbol{\tau}^{\mathrm{xu}} = \Psi(\boldsymbol{\tau}^{\mathrm{w}}) \}.$$

$$iii) \ \Phi_{\mathrm{CL}}^{\mathrm{w}\mapsto\mathrm{xu}}[\boldsymbol{F}] = \{ \boldsymbol{\Psi} = \begin{bmatrix} \boldsymbol{\Psi}^{\mathrm{x}} \\ \boldsymbol{\Psi}^{\mathrm{u}} \end{bmatrix} \in \mathcal{C}(\ell^{\mathcal{X}},\ell^{\mathcal{X}\times\mathcal{U}}) \mid \boldsymbol{\Psi}^{\mathrm{x}} = \boldsymbol{F}^{+}(\boldsymbol{\Psi}) + \boldsymbol{I} \}.$$

$$iv) \ \Phi_{\mathrm{CL}}^{\mathrm{w}\mapsto\mathrm{xu}}[\boldsymbol{F}] = \{ \begin{bmatrix} \boldsymbol{I} \\ \boldsymbol{K} \end{bmatrix} (\boldsymbol{I} - \boldsymbol{F}_{\boldsymbol{K}}^{+})^{-1} \mid \boldsymbol{K} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}}) \}.$$

Proof. <u>i)</u> is just reiterating the original definition Def. 2.6. We proceed by proving that the other statements are all equivalent to i). <u>ii)</u>: This statement follows from the fact that there is a one-to-one correspondence between CLMs $\Psi \in \Phi_{CL}^{w \mapsto xu}[F]$ and their realizing controllers $K = \Psi^{u}(\Psi^{x})^{-1}$ and recalling Theorem 6 ii). Part <u>iv</u>): This statement follows directly from ii) of Theorem 6. It remains to establish the equivalence between iii) and i).

<u>i)</u> \Longrightarrow iii): Assume Ψ is a CLM of F. Then, per definition, there exists some $\tilde{K} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$ such that Ψ is an $w \mapsto (x, u)$ -map of $\mathsf{CL}_{[F,\tilde{K}]}$, which is equivalent, by the characterization Theorem 6, to stating that $(x, u, w)^{\top} \in \mathsf{CL}_{[F,\tilde{K}]}$ if and only if $x = \Psi^{x}(w)$ and $u = \Psi^{u}(w)$. Since $(x, u, w)^{\top} \in \mathsf{CL}_{[F,\tilde{K}]} \Leftrightarrow x = F^{+}(x, u) + w$, we conclude by substitution that for all $w \in \ell^{\mathcal{X}}, \Psi^{x}(w) = F^{+}(\Psi(w)) + w$, i.e., the operator Ψ^{x} and $F^{+}(\Psi) + I^{x}$ are the same.

$$(\Psi^{\mathrm{x}} - I)(\Psi^{\mathrm{x}})^{-1}\boldsymbol{x} + (\Psi^{\mathrm{x}})^{-1}\boldsymbol{x} = F^{+}\Psi(\Psi^{\mathrm{x}})^{-1}\boldsymbol{x} + \boldsymbol{w}$$

 $\Leftrightarrow (\Psi^{\mathrm{x}} - I - F^{+}\Psi)(\Psi^{\mathrm{x}})^{-1} + (\Psi^{\mathrm{x}})^{-1}\boldsymbol{x} = \boldsymbol{w},$

which, due to $\Psi^{x} - I = F^{+}(\Psi)$, implies $x = \Psi^{x}w$ and $u = \Psi^{u}(\Psi^{x})^{-1}\Psi^{x}w = \Psi^{u}w$.

As it turns out, the space of realizable CLMs $\Phi_{CL}^{w \mapsto xu}[F]$ can be characterized as solutions to the nonlinear operator equation:

$$\Psi^{\mathrm{x}} = \boldsymbol{F}^{+}(\Psi) + \boldsymbol{I}. \tag{2.33}$$

We therefore also refer to (2.33) as the *CLM equation*. Writing out the CLM equation (2.33) in terms of component functions gives the more explicit condition on the functions Ψ_t^x , Ψ_t^u : The map $\Psi = (\Psi^x, \Psi^u)$ satisfies (2.33) if and only if its component functions satisfy the following infinite set of function equations for all inputs $w_{t:0}$:

$$\Psi_t^x(w_{t:0}) = F_{t-1}(\Psi_{t-1:0}^x(w_{t-1:0}), \Psi_{t-1:0}^u(w_{t-1:0})) + w_t.$$
(2.34)

As proven in Theorem 6, the mapping between CLMs $(\Psi^x, \Psi^u) \in \Phi_{CL}^{w \mapsto xu}[F]$ and the corresponding realizing controllers K' is one-to-one, via the relationship $K' = \Psi^u(\Psi^x)^{-1}$. While the realizing controller K of a CLM Ψ is a unique operator, we have degrees of freedom regarding the implementation – usually in the form of a dynamical system – of K. Next, we discuss a particularly simple approach enabled by the fact that Ψ^x is causally invertible for any $\Psi \in \Phi_{CL}^{w \mapsto xu}[F]$.

System Level Implementations of Realizing Controllers

A crucial step of establishing Theorem 8, is to show that $(\Psi^x)^{-1}$ always exists. We briefly rederive this partial result as a consequence of Lem. 5, restated below, which highlights the main idea behind the system level implementation.

Lemma. The inverse A^{-1} of a square operator $A \in C(\ell^{\mathcal{X}})$ exists if $A - I \in C_s(\ell^{\mathcal{X}})$, and $b = A^{-1}(a)$ is calculated by evaluating equations $b_t = a_t - A_t(0, b_{t-1:0})$ in the order t = 0, 1, 2, ...

Due to Theorem 8, we know that for any CLM $\Psi \in \Phi_{CL}^{w \mapsto xu}[F]$ holds $\Psi^x - I = F^+(\Psi)$. Furthermore, since $F^+ \in C_s(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ holds, we are also ensured that $\Psi^x - I \in C_s(\ell^{\mathcal{X}})$. This allows us to apply the previous lemma and proves that $(\Psi^x)^{-1}$ always exists and that it is a causal operator in $C(\ell^{\mathcal{X}})$.

Furthermore, as shown in Lem. 5, the condition $\Psi^{x} - I \in C_{s}(\ell^{\mathcal{X}})$ allows a particularly simple method of implementing $K' = \Psi^{u}(\Psi^{x})^{-1}$ of Theorem 8: Given an input a, the output b = K'(a) can be computed recursively through the equations

$$c_t = a_t - \Psi_t^x(0, c_{t-1:0}), \tag{2.35a}$$

$$b_t = \Psi_t^u(c_{t:0}). \tag{2.35b}$$

The above implementation represents a dynamical system with input a, output b and internal state c and will be referred to as the system level implementation of K'.

Remark. It is more common in control literature [49, 142] to refer to (2.35) as the "realization" of the controller, however to prevent overloading and potential confusion with the "realizing controller" (see Def. 2.6), we instead choose to call (2.35) the implementation of K.

Moreover, in later sections, we make use of this implementation to define controllers K that are parameterized by operators Ψ^x , Ψ^u that are not necessarily CLMs. In particular, the next section will show that such an implementation can yield closed-loop stability if Ψ approximately satisfies (2.33). Therefore, we define controllers with (or which permit) the above implementation separately as *System Level* (SL)-controllers:

Definition 2.7. Assume given operators $A \in C(\ell^{\chi})$, $B \in C(\ell^{\chi}, \ell^{U})$, where $A - I \in C_{s}(\ell^{\chi})$. The above dynamical system will be referred to as the system level controller

SL[A, B]. Consider the dynamical system with input a, output b, and internal state c according to the equations

$$c_t = a_t - A_t(0, c_{t-1:0}), (2.36)$$

$$b_t = B_t(c_{t:0}). (2.37)$$

The above dynamical system will be referred to as the system level controller SL[A, B].

Remark. A trivial yet important consequence of the definition Def. 2.7 is that both input a and output b can always be expressed through the internal state c as a = Ac, b = Bc.

The only requirements for the above implementation are that the operator $A \in C(\ell^{\chi})$ is square and A - I strictly causal, and that $B \in C(\ell^{\chi}, \ell^{U})$ is of compatible dimensions. We will abbreviate these conditions with the term *candidate closed-loop map* (cCLM):

Definition 2.8 (Candidate Closed-Loop Map). A causal operator $\Psi = \begin{bmatrix} A \\ B \end{bmatrix}$ is called a candidate closed-loop map (cCLM), if A is square and such that A - I is strictly causal. More specifically, we say that Ψ is a candidate CLM of some $F \in C(\ell^{\chi \times U}, \ell^{\chi})$ if $A \in C(\ell^{\chi})$ and $B \in C(\ell^{\chi}, \ell^{U})$, i.e., F and Ψ are of compatible domain and co-domain.

The dynamic model $\operatorname{SL}[\Psi]$ of the implementation can be defined as a subset $\operatorname{SL}[\Psi] \subset \ell^{\mathcal{S}}$, where $\mathcal{S} = \{\mathcal{S}_i\}_{i \in \mathcal{I}}$, $\mathcal{I} = \{x, u, \hat{w}\}$ s.t.: $\mathcal{S}_{\hat{w}} = \mathcal{S}_x = \mathcal{X}$. We can define $\operatorname{SL}[\Psi]$ in terms of its trajectories $\boldsymbol{\tau} = (\boldsymbol{\tau}^{\hat{w}}, \boldsymbol{\tau}^x, \boldsymbol{\tau}^u)^\top \in \operatorname{SL}[\Psi]$ as

$$SL[\Psi] = \{ \boldsymbol{\tau} \in \ell^{\mathcal{X} \times \mathcal{X} \times \mathcal{U}} \mid \boldsymbol{\tau}^{\hat{w}} = (\Psi^{x})^{-1} \boldsymbol{\tau}^{x}, \ \boldsymbol{\tau}^{u} = \Psi^{u} \boldsymbol{\tau}^{\hat{w}} \},$$
(2.38)

and notice that it can be equivalently represented by its $\mathbf{x} \mapsto \{\hat{\mathbf{w}}, \mathbf{u}\}$ -map, the operator $\begin{bmatrix} \mathbf{I} \\ \Psi^u \end{bmatrix} (\Psi^{\mathbf{x}})^{-1}$. We denote the former as $\mathbf{K}_{SL} = \begin{bmatrix} \mathbf{K}_{SL}^{\hat{\mathbf{w}}} \\ \mathbf{K}_{SL}^u \end{bmatrix} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ for a fixed $\Psi \in \Phi_{CL}^{\mathbf{w} \mapsto \mathbf{x}\mathbf{u}}[\mathbf{F}]$. The partial map $\mathbf{K}_{SL}^{\mathbf{u}}$ corresponds to the realizing control law $\Psi^{\mathbf{u}}(\Psi^{\mathbf{x}})^{-1}$, while $\mathbf{K}_{SL}^{\hat{\mathbf{w}}}$ represents the dynamics of the internal state of the implementation used to compute the control action \boldsymbol{u} from the internal state $\hat{\boldsymbol{w}}$ via $\boldsymbol{u} = \Psi^{\mathbf{u}}\hat{\boldsymbol{w}}$.

For a CLM $\Psi \in \Phi_{CL}^{w \mapsto xu}[F]$, the dynamic system $SL[\Psi]$ provides a simple and straightforward implementation, of the realizing controller $K = \Psi^{u}(\Psi^{x})^{-1}$, via the

dynamic equations (2.35). However, the system level implementation of K is not unique, that is there exist cCLMs $\hat{\Psi} \neq \Psi$ such that $K = \Psi^{u}(\Psi^{x})^{-1} = \hat{\Psi}^{u}(\hat{\Psi}^{x})^{-1}$.; a trivial example is $\hat{\Psi} = \begin{bmatrix} I \\ K \end{bmatrix}$. The space of all such cCLM $\hat{\Psi}$ defines an equivalence class which we refer to as the *Implementation Space* of a controller K.

Definition 2.9. The (SL)-Implementation Space \mathbb{I}_{K} of causal controller $K \in C(\ell^{\chi}, \ell^{\mathcal{U}})$ is the set of all cCLMs $\Psi \in C(\ell^{\chi}, \ell^{\chi \times \mathcal{U}})$ such that $\Psi^{u}(\Psi^{x})^{-1} = K$.

Remark. This was first formulated and studied in [6] for the case of LTI systems. The same work shows that the implementation space can be leveraged to formulate novel robust control synthesis procedures for LTI systems.

The following result provides a characterization of the implementation space \mathbb{I}_{K} of a controller K in terms of the CLM it realizes:

Lemma 9 (Space of System Level Implementations). Let $\Phi \in \Phi_{CL}^{w \mapsto xu}[F]$ be a CLM of some fixed F and let $K = \Phi^{u}(\Phi^{x})^{-1}$ be the corresponding unique realizing controller. Then,

$$\mathbb{I}_{\boldsymbol{K}} = \{ \begin{bmatrix} \boldsymbol{\Psi}^{\mathrm{x}} \\ \boldsymbol{\Psi}^{\mathrm{u}} \end{bmatrix} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}}) \mid \boldsymbol{\Phi}(\boldsymbol{\Psi}^{\mathrm{x}} - \boldsymbol{F}^{+}\boldsymbol{\Psi}) = \boldsymbol{\Psi} \}.$$

Despite that for any $\Psi \in \mathbb{I}_K$, the dynamic model $\mathrm{SL}[\Psi]$ realizes the controller K, special care needs to be taken when selecting $\Psi \in \mathbb{I}_K$ in practice, since the choice of the realizing dynamical system $\mathrm{SL}[\Psi]$ can impact the stability of the overal closed-loop: We need to make sure that the dynamical system $\mathrm{SL}[\Psi]$ is stable even if we add small perturbations v to the computation of the internal state as $\hat{w} = (\Psi^x)^{-1}x + v$. This is a crucial concern for control applications since even if our evaluation of $\hat{w} = (\Psi^x)^{-1}x$ is entirely digital, numerical errors can introduce perturbations v that are capable of causing numerical instability in our implementation. In the worst-case scenario, this can jeopardize the stability of the entire closed-loop $\mathrm{CL}_{[F,K]}$. In control literature, this problem is known as the question of closed-loop internal stability. In Section 2.7, we show that the system-level implementation $\mathrm{SL}[\Psi]$ of a realizing controller $K = \Psi^u(\Psi^x)^{-1}$ is internally stable under suitable assumptions and provided Ψ is a stable CLM of F.

2.7 Robust and Internal Stability of Closed-Loop

As shown in the previous section, any closed-loop map $\Psi = (\Psi^x, \Psi^u)^\top \in \Phi_{CL}^{w \mapsto xu}[F]$ can be realized with the corresponding system level controller $SL[\Psi^x, \Psi^u]$ as defined in Def. 2.7. Thus, we know that if we choose $K = SL[\Psi^x, \Psi^u]$, then for any disturbance w, the trajectories (x, u) of the closed-loop CL will be $(x, u)^{\top} = \Psi(w)$. In this section, we show that under mild assumptions, the controller $K = SL[\Psi^x, \Psi^u]$ guarantees internal stability of the closed-loop. Moreover, we show that closed-loop stability is guaranteed even if Ψ is satisfying the CLM equation (2.33) only approximately.

To setup the stability analysis we take the original closed-loop (2.29a) with K chosen to be $SL[\Psi^x, \Psi^u]$ and add additional perturbation signals v and d to the internal state of the system level controller and control input. We call the new perturbed closed-loop δCL :

$$\delta CL: \qquad x_t = F_{t-1}(x_{t-1:0}, u_{t-1:0}) + w_t, \quad x_0 = w_0 \qquad (2.39a)$$

$$\hat{w}_t = x_t + v_t - \Psi_t^x(0, \hat{w}_{t-1:0})$$
(2.39b)

$$u_t = \Psi_t^u(\hat{w}_{t:0}) + d_t.$$
(2.39c)

As before, w, x and u represent system disturbance, state and input, and the added state \hat{w} represents the internal state of the system level controller. In contrast, to the previous section, we now assume that $\Psi = (\Psi^x, \Psi^u)$ are merely *candidate* closed-loop maps (cCLM) of F, and derive conditions sufficient for ℓ_p -stability of δ CL with respect to the *residual* operator $\Delta[F, \Psi]$, which is defined for fixed cCLM Ψ and dynamics F as:

Definition 2.10. Let $\Psi = (\Psi^x, \Psi^u)$ be cCLMs of some dynamics F. Then, the (CLM)-residual of the pair $[F, \Psi]$, denoted by $\Delta[F, \Psi]$, is the operator defined as $\Delta[F, \Psi] := F^+(\Psi) + I - \Psi^x$.

As alluded to by its name, the residual $\Delta[F, \Psi]$ of a cCLM Ψ represents the CLM-equation error with respect to the dynamics F. Hence, candidate closed-loop map Ψ which satisfy $\Delta[F, \Psi] = 0$ define the space of closed-loop maps of F, i.e., we can equivalently define the set $\Phi_{CL}^{w \to xu}[F]$ as

$$\Phi_{\mathrm{CL}}^{\mathrm{w}\mapsto\mathrm{xu}}[F] = \{\Psi \mid \Delta[F,\Psi] = 0\}.$$

In sequence space, we can equivalently describe δCL by the equations:

$$\delta \text{CL}: \qquad \qquad \boldsymbol{x} = \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w} \qquad (2.40a)$$

$$\hat{w} = x + v - (\Psi^{x} - I)(\hat{w})$$
 (2.40b)

$$\boldsymbol{u} = \boldsymbol{\Psi}^{\mathrm{u}}(\boldsymbol{\hat{w}}) + \boldsymbol{d}. \tag{2.40c}$$

We can rewrite the equations (2.40) into a form, more suitable for analysis. Denote $\hat{u} := u - d$, $\hat{x} := x + v$ to rewrite (2.40) as

$$\delta CL': \qquad \hat{\boldsymbol{x}} = \boldsymbol{F}^+(\hat{\boldsymbol{x}} - \boldsymbol{v}, \hat{\boldsymbol{u}} + \boldsymbol{d}) + \boldsymbol{w} + \boldsymbol{v} \qquad (2.41a)$$

$$\hat{\boldsymbol{w}} = \hat{\boldsymbol{x}} - (\boldsymbol{\Psi}^{\mathrm{x}} - \boldsymbol{I})(\hat{\boldsymbol{w}}) = \hat{\boldsymbol{x}} + \overline{\boldsymbol{\Psi}^{\mathrm{x}}}(\hat{\boldsymbol{w}})$$
 (2.41b)

$$\hat{\boldsymbol{u}} = \boldsymbol{\Psi}^{\mathrm{u}}(\hat{\boldsymbol{w}}). \tag{2.41c}$$

For fixed sequences v, d, and denote $T_{-v,d} \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X} \times \mathcal{U}})$ as the affine operator such that $\forall \begin{bmatrix} x' \\ u' \end{bmatrix} \in \ell^{\mathcal{X}} : T_{-v,d} \begin{bmatrix} x' \\ u' \end{bmatrix} = \begin{bmatrix} x'-v \\ u'+d \end{bmatrix}$, i.e., $T_{-v,d}$ represents the identity operator shifted by the constant sequence $\begin{bmatrix} -v \\ d \end{bmatrix}$. With this affine operator, we can formally rewrite (2.41a) as $\hat{x} = F^+T_{-v,d}(\hat{x}, \hat{u}) + w + v$ and substitute it into (2.41b) to obtain an equation describing the dynamics of the internal state \hat{w} :

$$\hat{\boldsymbol{w}} = \boldsymbol{F}^{+} \boldsymbol{T}_{-\boldsymbol{v},\boldsymbol{d}}(\boldsymbol{\Psi})(\hat{\boldsymbol{w}}) - (\boldsymbol{\Psi}^{\mathrm{x}} - \boldsymbol{I}_{n})(\hat{\boldsymbol{w}}) + \boldsymbol{w} + \boldsymbol{v}$$
$$\hat{\boldsymbol{w}} = \boldsymbol{F}^{+} \boldsymbol{T}_{-\boldsymbol{v},\boldsymbol{d}}(\boldsymbol{\Psi})(\hat{\boldsymbol{w}}) - (\boldsymbol{T}_{-\boldsymbol{v},\boldsymbol{d}}^{\mathrm{x}} \boldsymbol{\Psi}^{\mathrm{x}} - \boldsymbol{I}_{n})(\hat{\boldsymbol{w}}) + \boldsymbol{w} = \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v},\boldsymbol{d}} \boldsymbol{\Psi}](\hat{\boldsymbol{w}}).$$
(2.42)

Since the residual is a sum of two strictly causal operators $F^+(\Psi)$ and $I - \Psi^x$ (by definition of cCLMs), the operator $I_n - \Delta[F, T_{-v,d}\Psi]$ is causally invertible, and we can solve for the lumped disturbance \hat{w} as

$$\hat{\boldsymbol{w}} = (\boldsymbol{I}_n - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v},\boldsymbol{d}}\boldsymbol{\Psi}])^{-1} \boldsymbol{w}, \qquad (2.43)$$

and \hat{x}, \hat{u} as

$$\hat{\boldsymbol{x}} = \boldsymbol{\Psi}^{\mathrm{x}} \left(\boldsymbol{I}_n - \boldsymbol{\Delta} [\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}} \boldsymbol{\Psi}]
ight)^{-1} \boldsymbol{w}, \quad \hat{\boldsymbol{u}} = \boldsymbol{\Psi}^{\mathrm{x}} \left(\boldsymbol{I}_n - \boldsymbol{\Delta} [\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}} \boldsymbol{\Psi}]
ight)^{-1} \boldsymbol{w}.$$

For v, d = 0, the operator $(I_n - \Delta[F, \Psi])^{-1}$ represents the map $w \mapsto \hat{w}$, and we see that ℓ_p -boundedness of \hat{w} is entirely determined by ℓ_p -stability of the former operator. Suitably, we will refer to the dynamic system governed by equation (2.43) as the *effective/lumped disturbance dynamics*. In fact, as shown in our next result, the stability of the overall closed-loop is entirely determined by the stability of the dynamics (2.43).

Sufficient and Necessary Conditions for Internal Stability

Summarizing our derivations so far, we can describe the input-output mapping $\Phi_{\delta CL}$: $(\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \mapsto (\hat{\boldsymbol{w}}, \boldsymbol{x}, \boldsymbol{u})$ of the closed-loop δCL by the equations:

$$\hat{\boldsymbol{w}} = (\boldsymbol{I}_n - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v},\boldsymbol{d}}\boldsymbol{\Psi}])^{-1} \boldsymbol{w}$$
(2.44a)

$$\boldsymbol{x} = \boldsymbol{\Psi}^{\mathrm{x}} \left(\boldsymbol{I}_{n} - \boldsymbol{\Delta} [\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}} \boldsymbol{\Psi}] \right)^{-1} \boldsymbol{w} - \boldsymbol{v}$$
(2.44b)

$$\boldsymbol{u} = \boldsymbol{\Psi}^{\mathrm{u}} \left(\boldsymbol{I}_n - \boldsymbol{\Delta} [\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}} \boldsymbol{\Psi}] \right)^{-1} \boldsymbol{w} + \boldsymbol{d}. \tag{2.44c}$$

The closed-loop δCL is internally stable if its input-output map $\Phi_{\delta CL}$ is ℓ_p -stable. The next theorem states sufficient and necessary conditions for that:

Theorem 9. Consider the closed-loop system δ CL described by the equations (2.40) for a fixed $\mathbf{F} \in \mathcal{C}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and some fixed compatible ℓ_p -stable candidate CLM $\Psi \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$. Then, the map Φ_{δ CL : $(\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \mapsto (\hat{\boldsymbol{w}}, \boldsymbol{x}, \boldsymbol{u})$ is ℓ_p -stable if and only if the operator $(\boldsymbol{I}_n - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}} \Psi])^{-1} \in \mathcal{C}(\ell^{\mathcal{X}})$ is ℓ_p -stable for any $\boldsymbol{v} \in \ell_p^{\mathcal{X}}$, $\boldsymbol{d} \in \ell_p^{\mathcal{U}}$.

Proof. Sufficiency: This is clear from equations (2.44). Necessity: Assume $(I_n - \Delta[F, T_{-v',d'}\Psi])^{-1}$ is not ℓ_p -stable for some $v' \in \ell_p^{\mathcal{X}}, d' \in \ell_p^{\mathcal{U}}$, then there exists some $w' \in \ell_p^{\mathcal{X}}$ such that $(I_n - \Delta[F, T_{-v',d'}\Psi])^{-1} w' \notin \ell_p^{\mathcal{X}}$ and therefore $\Phi_{\delta CL}^{\hat{w}}(w', v', d') \notin \ell_p^{\mathcal{X}}$.

Robustness of SL controllers in the LTV case was first discussed in [66]. Here, we discuss the generalization of these results in the context of linear operators.

If we assume that F is linear, we can split F into the linear operators $A \in \mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}})$ and $B \in \mathcal{LC}(\ell^{\mathcal{U}}, \ell^{\mathcal{X}})$ such that $F : (x, u) \mapsto A(x) + B(u)$. In that case, the operator $\Delta[F, T_{-v, d}\Psi]$ represents the mapping

$$\Delta[F, T_{-oldsymbol{v}, d}\Psi] : oldsymbol{z} \mapsto \Delta[F, \Psi](oldsymbol{z}) + (I_n - A^+)oldsymbol{v} + B^+d,$$

and correspondingly, $(I_n - \Delta[F, T_{-v,d}\Psi])^{-1}$ represents the mapping:

$$\boldsymbol{z} \mapsto (\boldsymbol{I}_n - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}])^{-1} (\boldsymbol{z} + (\boldsymbol{I}_n - \boldsymbol{A}^+)\boldsymbol{v} + \boldsymbol{B}^+ \boldsymbol{d}).$$
(2.45)

The dynamic equations (2.44) of δ CL can be equivalently stated in the form:

$$\hat{\boldsymbol{w}} = (\boldsymbol{I}_n - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}])^{-1} (\boldsymbol{w} + (\boldsymbol{I} - \boldsymbol{A}^+)\boldsymbol{v} + \boldsymbol{B}^+ \boldsymbol{d}), \qquad (2.46a)$$

$$\boldsymbol{x} = \boldsymbol{\Psi}^{\mathrm{x}} \left(\boldsymbol{I}_{n} - \boldsymbol{\Delta} [\boldsymbol{F}, \boldsymbol{\Psi}] \right)^{-1} \left(\boldsymbol{w} + (\boldsymbol{I} - \boldsymbol{A}^{+}) \boldsymbol{v} + \boldsymbol{B}^{+} \boldsymbol{d} \right) - \boldsymbol{v}, \quad (2.46b)$$

$$\boldsymbol{u} = \boldsymbol{\Psi}^{\mathrm{u}} \left(\boldsymbol{I}_n - \boldsymbol{\Delta} [\boldsymbol{F}, \boldsymbol{\Psi}] \right)^{-1} \left(\boldsymbol{w} + (\boldsymbol{I} - \boldsymbol{A}^+) \boldsymbol{v} + \boldsymbol{B}^+ \boldsymbol{d} \right) + \boldsymbol{d}.$$
(2.46c)

Remark. Notice that Ψ is still allowed to be nonlinear, and therefore, the dynamics of the closed-loop (2.46) are, in general, nonlinear.

From the above equations, we can see that the stability of $(I_n - \Delta[F, T_{-v,d}\Psi])^{-1}$ is equivalent to the stability of $(I_n - \Delta[F, \Psi])^{-1}$ and the boundedness of A and B. Thus, as a corollary of Theorem 9, we obtain a simpler characterization of internal ℓ_p -stability for the case where F is linear: **Corollary 10.** Consider the closed-loop system δCL described by the equations (2.40) for a fixed $\mathbf{F} \in \mathcal{LC}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and some fixed compatible ℓ_p -stable candidate $CLM \Psi \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$. Then, the map $\Phi_{\delta CL} : (\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \mapsto (\hat{\boldsymbol{w}}, \boldsymbol{x}, \boldsymbol{u})$ is ℓ_p -stable if and only if the operator $(\mathbf{I}_n - \Delta[\mathbf{F}, \Psi])^{-1} \in \mathcal{C}(\ell^{\mathcal{X}})$ is ℓ_p -stable, and \mathbf{F} is ℓ_p -stable.

Proof. Envoke Theorem 9 and recall (2.45), which makes it clear that the operator $(I_n - \Delta[F, T_{-v,d}\Psi])^{-1}$ is ℓ_p -stable for all $v \in \ell_p^{\mathcal{X}}$, $d \in \ell_p^{\mathcal{U}}$, if and only if the map $(I_n - \Delta[F, \Psi])^{-1}$ is ℓ_p -stable and both A and B are ℓ_p -stable, i.e., F is ℓ_p -stable.

Remark. It is important to note that requiring ℓ_p -stability of \mathbf{F} does **not** mean we assume that the dynamics \mathbf{F} is open-loop stable; that requirement would be ℓ_p -stability of $(\mathbf{I}_n - \mathbf{F}^+|_{u=0})^{-1} = (\mathbf{I}_n - \mathbf{A}^+)^{-1}$.

Thus, as long as A and B are ℓ_p -bounded linear operators, we see that (\hat{w}, x, u) is ℓ_p -bounded if and only if the operator $(I_n - \Delta[F, \Psi])^{-1}$ is ℓ_p -bounded. If Ψ is a CLM of F, hence $\Delta[F, \Psi] = 0$, then the former is trivially satisfied. Thus, we see that for the linear dynamics case, any (potentially nonlinear) CLM can be realized via the system level controller $SL[\Psi^x, \Psi^u]$ in an internally stable way.

In contrast to the linear case, verifying the stability of $(I_n - \Delta[F, \Psi])^{-1}$ is not sufficient alone to ensure internal stability for the case of nonlinear F. Instead, one must also verify ℓ_p -stability of the lumped dynamics (2.44a) for all perturbed versions $\tilde{\Psi} = T_{-v,d}\Psi$ of the original CLM Ψ ; this is, in general, not a trivial question.

In the next section, we use small-gain theorem techniques to break down the internal stability criterion of Theorem 9 into simpler ones. However, at the expense of losing necessity.

Remark. It should be noted that for nonlinear systems, it might even be impossible to achieve ℓ_p -internal stability, despite the controllability and observability of the system. For example, consider the simple scalar system $x_{k+1} = x_k^2 + u_k + w_k$ with $\hat{x}_k = x_k + v_k$. It is easy to verify that there does not exist a causal controller that realizes an ℓ_{∞} -internally stable closed loop.

Sufficient Small-Gain Conditions for Internal Stability

We use the small-gain theorems and ideas derived in Section 2.3 to provide sufficient conditions for the stability of the map $\Phi_{\delta CL}$. For the next theorem, recall the following theorems from Section 2.3:

Theorem (Global Small Gain Condition). Assume the operator $\Delta \in C(\ell^{\mathcal{X}})$ satisfies $\|\Delta(\boldsymbol{x})\|_p \leq \gamma \|\boldsymbol{x}\|_p + \beta$ for all $\boldsymbol{x} \in \ell_p^{\mathcal{X}}$ and some small gain $\gamma < 1$. Then, for all $\boldsymbol{w} \in \ell_p^{\mathcal{X}}$ and $\boldsymbol{y} = \Delta \boldsymbol{y} + \boldsymbol{w}$ holds:

$$\|\boldsymbol{y}\|_{p} \leq \frac{1}{1-\gamma}(\|\boldsymbol{w}\|_{p}+\beta).$$

Theorem (Local Small Gain Condition). Assume that for some $\rho > 0$ and $0 \le \gamma < 1$, the operator $\Delta \in C_s(\ell^{\mathcal{X}})$ satisfies $\|\Delta(\mathbf{x})\|_p \le \gamma \|\mathbf{x}\|_p$ for all $\|\mathbf{x}\|_p < \rho$. Then, for any $\mathbf{w} \in \ell_p^{\mathcal{X}}$ and $\mathbf{y} = \Delta \mathbf{y} + \mathbf{w}$ the following statement holds true:

$$\|\boldsymbol{w}\|_p < (1-\gamma)
ho \qquad \Longrightarrow \qquad \|\boldsymbol{y}\|_p \leqslant rac{1}{1-\gamma} \|\boldsymbol{w}\|_p.$$

It suits to measure the gain of the residual in terms of the operator (pseudo)-norm $\||\Delta[F, \tilde{\Psi}]|\|_p^{\star 3}$, where $\||\cdot\||_p^{\star}$ and the related operator norm $\||\cdot\||_p$ for some operator A are defined as:

$$\|\|\boldsymbol{A}\|_{p}^{\star} := \sup_{\boldsymbol{x} \in \mathcal{X}_{p}^{\mathbb{N}}: \, \boldsymbol{x} \neq 0} \, \frac{\|\boldsymbol{A}(\boldsymbol{x}) - \boldsymbol{A}(\boldsymbol{0})\|_{p}}{\|\boldsymbol{x}\|_{p}} \quad \|\|\boldsymbol{A}\|_{p} := \|\|\boldsymbol{A}\|_{p}^{\star} + \|\boldsymbol{A}(\boldsymbol{0})\|_{p}.$$
(2.47)

The residual norm $|||\Delta[F, \Psi]|||_p^*$ measures how well, up to a constant offset, a candidate CLM Ψ fulfills the CLM equation corresponding to F and can be viewed as a degree of approximation of the CLM equation. As shown in the following results, the closer a cCLM Ψ approximates a true CLM of F, the more robust the closed-loop system is. In particular, we require $|||\Delta[F, \Psi]||_p^*$ and the perturbed residual norm $|||\Delta[F, T_{-v,d}\Psi]||_p^*$ to be uniformly smaller than 1; if the condition can only be fulfilled for a subset of perturbations v, d, then we also obtain corresponding local ⁴ stability results.

Theorem 11. Consider the closed-loop system δCL described by the equations (2.40) and let $\Phi_{\delta CL} : (\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \mapsto (\hat{\boldsymbol{w}}, \boldsymbol{x}, \boldsymbol{u})$ be the corresponding input-output mapping of δCL for a fixed $\boldsymbol{F} \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and some fixed compatible ℓ_p -stable candidate CLM $\boldsymbol{\Psi} \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ such that $\|\|\boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}]\|_p^* = \gamma < 1$. Then, the following statements hold:

i) Sufficient Condition for Global ℓ_p -internal-stability: If \mathbf{F} is i.f.g. ℓ_p -stable, then $\Phi_{\delta CL}$ is ℓ_p -stable. If in addition, Ψ is f.g. ℓ_p -stable, then, respectively, $\Phi_{\delta CL}$ is f.g. ℓ_p -stable.

³See Appendix Section 2.A for proof of norm-properties.

⁴Local means for ℓ_p -bounded subsets around 0 of inputs (w, u, d).
ii) Sufficient Condition for Local ℓ_p -internal-stability: Assume $\Delta[F, \Psi](0) = 0$ and that at w = 0, the map Ψ is ℓ_p -continuous. Moreover, assume that there exists a nondecreasing function ${}^5L : \mathbb{R}^+ \to \mathbb{R}^+$ such that:

$$\|\boldsymbol{z}\|_p, \|\boldsymbol{z'}\|_p \leq \eta \implies \|\boldsymbol{F}(\boldsymbol{z}) - \boldsymbol{F}(\boldsymbol{z'})\|_p^{\mathcal{X}} \leq L(\eta) \|\boldsymbol{z} - \boldsymbol{z'}\|_p^{\mathcal{X}}.$$

Then, there exists a $\delta > 0$ such that $\Phi_{\delta CL}(\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \in \ell_p^{\mathcal{X} \times \mathcal{X} \times \mathcal{U}}$ whenever $\|\boldsymbol{w}\|_p^{\mathcal{X}}, \|\boldsymbol{v}\|_p^{\mathcal{X}}, \|\boldsymbol{d}\|_p^{\mathcal{U}} \leq \delta$. If, in addition, $\|\|\boldsymbol{\Psi}\|\|_p^{\star} < \infty$, then for any $\delta_w > 0$, there exists a $\delta_{vd} > 0$ such that $\Phi_{\delta CL}(\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \in \ell_p^{\mathcal{X} \times \mathcal{X} \times \mathcal{U}}$ whenever $\|\boldsymbol{w}\|_p^{\mathcal{X}} \leq \delta_w$ and $\|\boldsymbol{v}\|_p^{\mathcal{X}}, \|\boldsymbol{d}\|_p^{\mathcal{U}} \leq \delta_{vd}$.

Proof. Part i): Fix some $v \in \ell_p^{\mathcal{X}}$, $d \in \ell_p^{\mathcal{U}}$, let $\Phi^{\hat{w}}|_{v,d} := (I_n - \Delta[F, T_{-v',d'}\Psi])^{-1}$ and decompose $\Delta[F, T_{-v,d}\Psi]$ into the sum $\Delta[F, \Psi] + \delta \Delta$ where the operator $\delta \Delta$ denotes the difference $\Delta[F, \Psi] - \Delta[F, T_{-v,d}\Psi]$. Now, notice that for all $w \in \ell_p^{\mathcal{X}}$:

$$\begin{split} \delta \boldsymbol{\Delta}(\boldsymbol{w}) &= \boldsymbol{F} \boldsymbol{\Psi}(\boldsymbol{w}) + \overline{\boldsymbol{\Psi}^{\mathrm{x}}}(\boldsymbol{w}) - \boldsymbol{F}(\boldsymbol{\Psi}(\boldsymbol{w}) + \begin{bmatrix} -\boldsymbol{v} \\ \boldsymbol{d} \end{bmatrix}) - \overline{\boldsymbol{\Psi}^{\mathrm{x}}}(\boldsymbol{w}) - \boldsymbol{v} \\ &= \boldsymbol{F} \boldsymbol{\Psi}(\boldsymbol{w}) - \boldsymbol{F}(\boldsymbol{\Psi}(\boldsymbol{w}) + \begin{bmatrix} -\boldsymbol{v} \\ \boldsymbol{d} \end{bmatrix}) - \boldsymbol{v}. \end{split}$$

Next, we show that the ℓ_p -norm of $\Delta_1(w)$ is bounded by a constant independent of w:

Per assumption, Ψ is ℓ_p-stable, and therefore z = Ψw ∈ ℓ^{X×U}_p. Furthermore, since v and d are assumed to be bounded in the ℓ_p-norm, we have z + [^{-v}_d] ∈ ℓ^{X×U}_p. Invoking i.f.g.-ℓ_p stability of F at z, we know there are fixed constants (γ_f, β_f) such that for any z: ||F(z) - F(z + [^{-v}_d])||_p ≤ γ_f(||v||_p + ||d||_p) + β_f. This implies that ||δΔ(w)||_p is bounded above for all w ∈ ℓ^X as:

$$\|\delta \boldsymbol{\Delta}(\boldsymbol{w})\|_{p} \leq (\gamma_{f}+1) \|\boldsymbol{v}\|_{p} + \gamma_{f} \|\boldsymbol{d}\|_{p} + \beta_{f}.$$
(2.48)

Denote $e_0 = \Delta[F, \Psi](0)$ and note that since F is i.f.g. ℓ_p -stable and Ψ is assumed ℓ_p -stable, the sequence e_0 belongs to $\ell_p^{\mathcal{X}}$. Now, per our assumption $\||\Delta[F, \Psi]||_p^{\star} =: \gamma < 1$, for any $w \in \ell_p^{\mathcal{X}}$ holds:

$$\|\boldsymbol{\Delta}[\boldsymbol{F},\boldsymbol{\Psi}](\boldsymbol{w})\|_{p} \leqslant \|\boldsymbol{\Delta}[\boldsymbol{F},\boldsymbol{\Psi}](\boldsymbol{w}) - \boldsymbol{e}_{0}\|_{p} + \|\boldsymbol{e}_{0}\|_{p} \leqslant \gamma \|\boldsymbol{w}\|_{p} + \|\boldsymbol{e}_{0}\|_{p}$$

From the above inequality and (2.48), we can conclude that $\Delta[\mathbf{F}, \mathbf{T}_{-v,d}\Psi]$ is (γ, β_1) -f.g. ℓ_p -stable for any $v \in \ell_p^{\chi}, d \in \ell_p^{\mathcal{U}}$. In particular, for all $w \in \ell_p^{\chi}$ holds:

$$\|\boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}}\boldsymbol{\Psi}](\boldsymbol{w})\|_{p} \leq \gamma \|\boldsymbol{w}\|_{p} + \underbrace{\|\boldsymbol{e}_{0}\|_{p} + (\gamma_{f} + 1)\|\boldsymbol{v}\|_{p} + \gamma_{f}\|\boldsymbol{d}\|_{p} + \beta_{f}}_{\beta_{1}}.$$
 (2.49)

⁵This can be thought of a local Lipshitz-constant.

According to the above, $\Delta[F, T_{-v,d}\Psi]$ satisfies the ℓ_p small-gain property. Using the small-gain Theorem 2 (presented in Section 2.3), we confirm that $\Phi^{\hat{w}}|_{v,d}$ achieves $(\frac{1}{1-\gamma}, \beta_2) - \ell_p$ fg stability, with $\beta_2 = \frac{1}{1-\gamma}\beta_1$. On substituting the constants defined in (2.49), we establish that the partial input-output map of δ CL, $\Phi^{\hat{w}}_{\delta$ CL : $(w, v, d) \mapsto \hat{w}$, is $(\gamma_3, \beta_3) - \ell_p$ -stable where

$$\gamma_3 = \frac{\gamma_f + 1}{1 - \gamma}, \quad \beta_3 = \frac{\beta_f + \|\boldsymbol{e}_0\|_p}{1 - \gamma}.$$

The other partial input-output map of the closed loop δCL , $\Phi_{\delta \text{CL}}^{\text{xu}} : (\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \mapsto (\boldsymbol{x}, \boldsymbol{u})$, is written as $\Phi_{\delta \text{CL}}^{\text{xu}} = \Psi \Phi_{\delta \text{CL}}^{\hat{w}}$. This implies that, at a minimum, it is ℓ_p -stable, given that Ψ is ℓ_p -stable and $\Phi_{\delta \text{CL}}^{\hat{w}}$ is (γ_3, β_3) -f.g.- ℓ_p -stable. If Ψ is additionally fg.- ℓ_p -stable, $\Phi_{\delta \text{CL}}^{\text{xu}}$ also shares this property. Given that $\Phi_{\delta \text{CL}} = (\Phi_{\delta \text{CL}}^{\hat{w}}, \Phi_{\delta \text{CL}}^{\text{x}}, \Phi_{\delta \text{CL}}^{\text{u}})$, we affirm the desired statement.

Part ii): Aside from a few modifications, the proof is structured as in the previous part. Pick $\varepsilon > 0$, then, by the ℓ_p -continuity assumption, there exists some $\delta(\varepsilon) > 0$ such that $\|\Psi(\boldsymbol{z})\|_p^{\mathcal{X}\times\mathcal{U}} \leq \varepsilon$ for all $\boldsymbol{z} \in \mathsf{B}^x_{\ell_p}[\delta(\varepsilon)]$ where $\mathsf{B}^x_{\ell_p}[r] := \{\boldsymbol{w}' \in \ell_p^{\mathcal{X}\times\mathcal{U}} | \|\boldsymbol{w}'\|_p^{\mathcal{X}} \leq r\}$. Fix some $\boldsymbol{v}, \boldsymbol{d}$ and denote $\delta_{vd} := \|\boldsymbol{v}\|_p^{\mathcal{X}} \vee \|\boldsymbol{d}\|_p^{\mathcal{U}}$, then for all $\boldsymbol{z} \in \mathsf{B}^x_{\ell_p}[\delta(\varepsilon)]$ holds

 $\|\delta \boldsymbol{\Delta}(\boldsymbol{z})\|_{p} \leq \|\boldsymbol{F}(\boldsymbol{\Psi}(\boldsymbol{z})) - \boldsymbol{F}(\boldsymbol{\Psi}(\boldsymbol{z}) + \begin{bmatrix} -\boldsymbol{v} \\ \boldsymbol{d} \end{bmatrix})\|_{p} \leq L(\varepsilon + \delta_{vd})\delta_{vd}.$

From $\||\Delta[F, \Psi]||_p^* = \gamma < 1$ and since we assumed $\Delta[F, \Psi](\mathbf{0}) = \mathbf{0}$, we have $e_0 = 0$ and therefore $\Delta[F, \Psi](z) \leq \gamma \|z\|_p$. We proceed to bound $\|\Delta[F, T_{-v,d}\Psi](z)\|_p^{\mathcal{X}}$ for $z \in \mathsf{B}_{\ell_p}^{xu}[\delta(\varepsilon)]$:

$$\|\boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{T}_{-\boldsymbol{v}, \boldsymbol{d}} \boldsymbol{\Psi}](\boldsymbol{z})\|_{p}^{\mathcal{X}} \leq \gamma \|\boldsymbol{z}\|_{p}^{\mathcal{X}} + L(\varepsilon + \delta_{vd})\delta_{vd}.$$

We can now use the local small gain theorem Lem. 7, letting $\delta \Delta(z)$ play the role of disturbance and setting $\rho = \delta(\varepsilon)$. Notice that since L is not decreasing, the term $L(\varepsilon + \delta_{vd})\delta_{vd}$ is not decreasing and converges to 0 as $\delta_{vd} \to 0$. This implies that there exists $r_{vd} > 0$ such that $L(\varepsilon + r_{vd})r_{vd} \leq (1 - \gamma)\delta(\varepsilon)$. Therefore, if $\delta_{vd} \leq r_{vd}$, then $\|\delta \Delta(z)\|_p \leq (1 - \gamma)\delta(\varepsilon)$ and we can conclude that $\Phi^{\hat{w}}|_{v,d}$ maps to $\ell_p^{\mathcal{X}}$ when restricted to $w \in \mathsf{B}^x_{\ell_p}[\delta(\varepsilon)]$ and provided that $\|v\|_p^{\mathcal{X}} \vee \|d\|_p^{\mathcal{U}} \leq r_{vd}$. With $\delta^* =$ $\min\{\delta(\varepsilon), r_{vd}\}$, this shows that $\Phi^{\mathrm{xu}}_{\delta\mathrm{CL}}(w, v, d) \in \ell_p^{\mathcal{X}}$ whenever $\|w\|_p^{\mathcal{X}}, \|v\|_p^{\mathcal{X}}, \|d\|_p^{\mathcal{U}} \leq$ δ^* and leads to the desired result, since Ψ is ℓ_p -stable, $\Phi^{\mathrm{xu}}_{\delta\mathrm{CL}} = \Psi \Phi^{\hat{w}}_{\delta\mathrm{CL}}$, and $\Phi_{\delta\mathrm{CL}} = (\Phi^{\hat{w}}_{\delta\mathrm{CL}}, \Phi^{\mathrm{x}}_{\delta\mathrm{CL}}, \Phi^{\mathrm{u}}_{\delta\mathrm{CL}})$. If in addition, $\||\Psi||_p^* = L_{\Psi} < \infty$, then $\delta(\varepsilon) = L_{\Psi}^{-1}\varepsilon$ is onto \mathbb{R}_+ and we can replace $\delta(\varepsilon)$ simply by a variable $\delta_w \ge 0$ and construct r_{vd} as a function of δ_w such that $L(L_{\Psi}\delta_w + r_{vd})r_{vd} \leq (1 - \gamma)\delta_w$, which, as argued before, is possible for any $\delta_w \in \mathbb{R}_+$.

In the previous theorem, the global stability condition i) required the dynamics operator F to be i.f.g ℓ_p -stability and the local one ii) required the statement

$$\|\boldsymbol{z}\|_{p}, \|\boldsymbol{z'}\|_{p} \leq \eta \implies \|\boldsymbol{F}(\boldsymbol{z}) - \boldsymbol{F}(\boldsymbol{z'})\|_{p}^{\mathcal{X}} \leq L(\eta) \|\boldsymbol{z} - \boldsymbol{z'}\|_{p}^{\mathcal{X}}$$
(2.50)

to hold for an appropriate choice of non-decreasing⁶ function $L : \mathbb{R}^+ \to \mathbb{R}^+$. It is important to point out that both requirements are continuity-type conditions and are **completely unrelated** to the question of whether the dynamics F are open-loop stable or unstable! As mentioned in our discussion of the different notions of ℓ_p -stability of Def. 2.3, requiring F to be (γ, β) -i.f.g.- ℓ_p -stable, with $\beta = 0$, is the same as requiring F to be Lipschitz continuous over the subspace $\ell_p^{\mathcal{X} \times \mathcal{U}}$ with respect to the corresponding ℓ_p -norms of the domain and codomain. On the other hand, the statement (2.50) is a weaker continuity requirement that is equivalent to uniform continuity of F in the ℓ_p -norm. For most types of dynamical system, these continuity assumptions can be further reduced to the standard continuity properties of functions over finite-dimensional normed spaces. For example, the dynamics operator F corresponding to any of the following dynamical system equations is (γ, β) -i.f.g.- ℓ_p -stable:

- LTI finite-dim.⁷ dynamics: $x_t = \sum_{k=1}^{h_x} A_k x_{t-k} + \sum_{k=1}^{h_u} B_k u_{t-k} + w_t$.
- LTV finite-dim. dynamics: $x_t = \sum_{k=1}^{h_x} A_{t,k} x_{t-k} + \sum_{k=1}^{h_u} B_{t,k} u_{t-k} + w_t, \text{ for a bounded set of matrices}$ $\{(A_{t,k}, B_{t,k}) \mid t \in \mathbb{N}, 1 \leq k \leq h\}.$
- Nonlinear "almost"-Lipshitz TI⁸ finite-dim. dynamics:
 x_t = f(x_{t-1:t-h_x}, u_{t-1:t-h_u}) + w_t where f = f₁ + f₂ and with f₁ being γ-Lipshitz-continuous and f₂ being a (possibly discontinuous) β-bounded function, that is: sup_{x,u} |f₂|(x, u) ≤ β.
- Nonlinear "almost"-equi-Lipshitz TV finite-dim. dynamics:
 x_t = f_t(x_{t-1:t-h_x}, u_{t-1:t-h_u})+w_t where for each t, the function f_t = f_{t,1}+f_{t,2} can be decomposed by a γ-Lipshitz continuous f_{1,t} and β-bounded (possibly discontinuous) function f_{2,t}.

Similarly, F satisfies the condition (2.50) for the above nonlinear systems if we replace Lipschitz continuity by uniform continuity; for the linear system examples,

⁶It is clear that the requirement L to be non-decreasing is without loss of generality.

⁷Finite-dimensional.

⁸TI=time-invariant, TV=time-varying.

the conditions are the same. On the other hand, if a dynamical system cannot be realized with a finite-dimensional internal state, then correspondingly the continuity requirements on the dynamics operator F may need to be verified directly in the corresponding infinite-dimensional ℓ_p -normed spaces.

In summary, Theorem 11 showed to us that for any of the dynamical systems listed above, any ℓ_p -stable CLM Ψ of F, i.e., $\Delta[F, \Psi] = 0$, can be realized with an internally ℓ_p -stable system level controller $SL[\Psi^x, \Psi^u]$; in the context of closed loop of δ CL, Ψ being realized means that the operator $\Phi_{\delta CL}^{xu}|_{v,d=0} : w \mapsto \Phi_{\delta CL}^{xu}(w, 0, 0)$ is the same as Ψ . Moreover, if a candidate CLM Ψ is an accurate enough approximate solution to the CLM equation, i.e., the residual operator $\Delta[F, \Psi]$ has a small enough norm $|||\Delta[F, \Psi]|||_p^* < 1$, then the system level controller $SL[\Psi^x, \Psi^u]$ still stabilizes the system and guarantees ℓ_p -internal stability of the closed loop dynamics. However, in that scenario, the resulting mapping $\Phi_{\delta CL}^{xu}|_{v,d=0}$ is the operator

$$\Phi^{\mathrm{xu}}_{\mathrm{\delta CL}}|_{oldsymbol{v},oldsymbol{d}=0} = \Psi \left(oldsymbol{I}_n - oldsymbol{\Delta} [oldsymbol{F}, \Psi]
ight)^{-1}$$

and no longer matches Ψ . Nevertheless, if Ψ has stronger stability properties such as fg- ℓ_p -stability, then in either case, whether Ψ is an exact or approximate CLM of F, the resulting input-output map $\Phi_{\delta CL} : (w, v, d) \mapsto (\hat{w}, x, u)$ of the closed loop δCL is guaranteed to be fg- ℓ_p -stable. Moreover, if F is only uniformly ℓ_p -continuous but not necessarily i.f.g.- ℓ_p -stable, then as stated in Theorem 11ii), we can ensure ℓ_p -internal closed loop stability of $\Phi_{\delta CL}$ for a bounded set of small enough perturbations (w, v, d).

2.8 Nonlinear Closed Loop Maps of Linear Systems

Linear techniques, such as loop-shaping, \mathcal{H}_{∞} , \mathcal{H}_2 -optimal control, to name a few, provide powerful tools for feedback control design of linear systems. However, there are many applications where, despite the linearity of the system, nonlinear controller design is required or provides better solutions. Two popular application scenarios where nonlinear control design is common are model-predictive control and adaptive control. Due to the nonlinear feedback controller K, the overall closed loop in these settings has nonlinear dynamics, and it is necessary and crucial to study the closed-loop behavior in the context of nonlinear control theory. Understanding the properties of the closed loop maps of F, which are also nonlinear, is also of great interest, however, they are far less studied in control literature so far, partially due to the lack of analytical tools.

The operator-theoretic tools developed in the previous sections open up new ways of studying the closed-loop dynamics in this setting by analyzing the structure of the closed-loop maps. It turns out that the linearity of \boldsymbol{F} imposes structure on the entire space of CLMs $\Phi_{CL}^{w\mapsto xu}[\boldsymbol{F}]$, not just the subspace $\Phi_{CL}^{w\mapsto xu}[\boldsymbol{F}] \cap \mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ of linear ones. In this section, we show that the set of CLMs $\Phi_{CL}^{w\mapsto xu}[\boldsymbol{F}]$ is closed under a particular type of combination, which we refer to as *blending*:

Definition 2.11 (Blend of a Family of Operators). For a fixed family of operators $\mathcal{F} = \{\Psi_i\}_{i \in \mathcal{I}} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}}), \text{ we call an operator } \Psi_{\Sigma} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}}) \text{ a blend of } \mathcal{F}$ if it can be expressed as a sum $\Psi_{\Sigma} = \sum_{i \in \mathcal{I}} \Psi_i G_i$ for some collection of operators $\mathcal{G} = \{G_i\}_{i \in \mathcal{I}} \subset \mathcal{C}(\ell^{\mathcal{X}}), \text{ called weights, which satisfy the identity } \sum_{i \in \mathcal{I}} G_i = I_{\mathcal{X}}.$

Thus, a blend can be thought of as a sum of operators Ψ_i of a family \mathcal{F} , "weighted" by the operators G_i of another family \mathcal{G} , which, as expressed by condition $\sum_{i \in \mathcal{I}} G_i = I_{\mathcal{X}}$, represents a sum decomposition of the identity operator. In this section, we show that, for linear system dynamics, the CLM space $\Phi_{CL}^{W \mapsto xu}[F]$ is closed under the combination of blending described above. In particular, we can combine a family $\mathcal{F} = {\Psi_i}_{i \in \mathcal{I}} \subset \mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ of linear CLMs with a suitable family \mathcal{G} of nonlinear "weight" operators to merge the desirable properties of different linear CLMs into one nonlinear CLM. We will demonstrate the power of this approach in the next chapter, where we consider control applications with large-scale linear systems subjected to state- and input constraints and actuator saturation. The enabling theoretical result underlying these methods is stated and proven below:

Theorem 12. Let F be linear, $\{G_i : \ell^{\mathcal{X}} \mapsto \ell^{\mathcal{X}}\}$ be N causal operators, and $\{\Psi_i\}_{i=1}^N$ be N candidate CLMs. Then,

$$\forall i: \Psi_i \in \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[F] \text{ and } \sum_{i=1}^N G_i = I \implies \sum_{i=1}^N \Psi_i G_i \in \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[F]. \quad (2.51)$$

Proof. Denote Ψ_{Σ} as the cCLM $\Psi_{\Sigma} = \sum_{i=1}^{N} \Psi_i G_i$ and evaluate the residual

 $\Delta[F, \Psi_{\Sigma}]$:

$$\begin{split} \boldsymbol{\Delta}[\boldsymbol{F},\boldsymbol{\Psi}_{\Sigma}] &= \boldsymbol{F}^{+}\left(\boldsymbol{\Psi}_{\Sigma}\right) + \boldsymbol{I} - \boldsymbol{\Psi}_{\Sigma}^{\mathrm{x}} = \boldsymbol{F}^{+}\left(\sum_{i=1}^{N}\boldsymbol{\Psi}_{i}\boldsymbol{G}_{i}\right) + \boldsymbol{I} - \sum_{i=1}^{N}\boldsymbol{\Psi}_{i}^{\mathrm{x}}\boldsymbol{G}_{i} \\ &\stackrel{a)}{=} \sum_{i=1}^{N}\boldsymbol{F}^{+}(\boldsymbol{\Psi}_{i}\boldsymbol{G}_{i}) + \boldsymbol{I} - \sum_{i=1}^{N}\boldsymbol{\Psi}_{i}^{\mathrm{x}}\boldsymbol{G}_{i} \\ &\stackrel{b)}{=} \sum_{i=1}^{N}\boldsymbol{F}^{+}(\boldsymbol{\Psi}_{i}\boldsymbol{G}_{i}) + \sum_{i=1}^{N}\boldsymbol{G}_{i} - \sum_{i=1}^{N}\boldsymbol{\Psi}_{i}^{\mathrm{x}}\boldsymbol{G}_{i} \\ &\stackrel{c)}{=} \sum_{i=1}^{N}\boldsymbol{F}^{+}(\boldsymbol{\Psi}_{i}\boldsymbol{G}_{i}) + \boldsymbol{G}_{i} - \boldsymbol{\Psi}_{i}^{\mathrm{x}}\boldsymbol{G}_{i} = \sum_{i=1}^{N}\left(\boldsymbol{F}^{+}(\boldsymbol{\Psi}_{i}) + \boldsymbol{I} - \boldsymbol{\Psi}_{i}^{\mathrm{x}}\right)\boldsymbol{G}_{i} \\ &\stackrel{d)}{=} \sum_{i=1}^{N}\boldsymbol{\Delta}[\boldsymbol{F},\boldsymbol{\Psi}_{i}]\boldsymbol{G}_{i} = 0. \end{split}$$

The first equality a) follows by linearity of F. Step b) follows by the definition of operators G_i . Step c) uses the right-distributive property of the operator product. In d) we use the fact that each map Ψ_i is a CLM of F. This establishes the desired result, since we showed $\Delta[F, \Psi_{\Sigma}] = 0$ and can invoke Theorem 8.

As an immediate corollary of the above result, we conclude that the space of CLMs of linear dynamics operators F is closed under blending:

Corollary 13. If F is linear, then for any blend $\Psi_{\Sigma} \in C(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ of a family $\mathcal{F} = \{\Psi_i\}_{i \in \mathcal{I}} \subset \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[F]$, holds $\Psi_{\Sigma} \in \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[F]$.

This result provides concrete instructions on how to construct new CLMs Ψ_{Σ} from existing ones Ψ_i . If we interpret the operators G_i as "weighting" operators (because they sum up to "one" in the operator space), then $\Psi_{\Sigma} = \sum_{i=1}^{N} \Psi_i G_i$ resembles a weighted average of the individual CLMs Ψ_i . We refer to the resulting CLM Ψ_{Σ} as a *non-linear blend* of the CLMs Ψ_i and call a finite collection of operators $\{G_i\}$ weight operators / weights if $\sum_i G_i = I$.

It is also important to note that the above result only requires linearity of F, so both Ψ_i and G_i can be nonlinear causal operators. In particular, this allows us to construct blended CLMs composed out of other sets of blended CLMs; one could call that a layered blend:

Corollary 14. Let F be linear, $\{G_i : \ell^{\mathcal{X}} \mapsto \ell^{\mathcal{X}}\}, \{G'_j : \ell^{\mathcal{X}} \mapsto \ell^{\mathcal{X}}\}\$ be N causal operators, and $\{\Psi_{ij}\}_{i,j=1}^N$ be N^2 candidate CLMs. Then,

$$\forall i, j: \Psi_{ij} \in \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[\boldsymbol{F}] \text{ and } \sum_{i=1}^{N} \boldsymbol{G}_{i} = \boldsymbol{I}, \sum_{j=1}^{N} \boldsymbol{G}_{j}' = \boldsymbol{I}$$
$$\implies \sum_{j=1}^{N} \sum_{i=1}^{N} \Psi_{ij} \boldsymbol{G}_{j}' \boldsymbol{G}_{i} \in \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[\boldsymbol{F}].$$
(2.52)

Naturally, the above corollary can be applied arbitrarily many times, and it is an open question whether there is a generating set of CLMs $\{E_1, E_2, \ldots,\} \subset \Phi_{CL}^{w \mapsto xu}[F]$ from which all of $\Phi_{CL}^{w \mapsto xu}$ can be constructed in this way.

For our discussion, we will focus on simple blends using static weighting operators $\{G_i\}$. We refer to static operators as those that act on sequences by applying the same function $f : \mathbb{R}^n \to \mathbb{R}^m$ to each element. For notational convenience, we refer to the corresponding operator via the Kronecker product $I \otimes f$ as defined below:

Definition. For a fixed function $f : \mathbb{R}^n \mapsto \mathbb{R}^m$, let $\mathbf{I} \otimes f : \ell^{\mathcal{X}} \to \ell^{\mathcal{U}}$ be defined as *the map*

$$(x_0, x_1, \dots) \mapsto (f(x_0), f(x_1), \dots).$$

For a matrix $M \in \mathbb{R}^{n \times m}$, the operator $\mathbf{I} \otimes f_M$, where $f_M : x \mapsto Mx$, will be referred to as $\mathbf{I} \otimes M$.

In the later discussion, we make use of the following types of static weighting operator:

Definition 2.12. Any collection of operators $\{G_i\}_{i=1}^N$ defined as below satisfies $\sum_{k=1}^N G_k = I$:

- 1. Linear Orthonormal Projections: $G_i = I \otimes U_i U_i^{\top}$, where $U_i \in \mathbb{R}^{n \times r_i}$, $U_i^{\top} U_i = I_{r_i}$, and $\sum_{i=1}^{N} U_i U_i^{\top} = I_n$.
- 2. *N*-Zone Saturation: Let $W_1 \subset W_2 \cdots \subset W_{N-1}$ be a collection of nested convex sets in some normed vector space $(| \cdot |, \mathbb{R}^n)$ and denote $\Pi_k : x \mapsto \arg\min_{u \in W_k} |x - u|, \forall k \in \{1, \dots, N-1\}$ as the projection maps onto set W_k . With the abbreviations $\Pi_0 := 0$, $\Pi_N := I_n$, define $G_k = I \otimes (\Pi_k - \Pi_{k-1})$ for all $k \in \{1, \dots, N\}$.
- 3. Sum Inversion: For any collection of causal operators $\{G'_i\}_{i=1}^N$ for which the sum $G'_{\Sigma} = \sum_{i=1}^N G'_i$ is causally invertible, define $G_k := G'_k (G'_{\Psi_{\Sigma}})^{-1}$.

Next, we investigate how to implement the realizing controller $K = \Psi_{\Sigma}^{u}(\Psi_{\Sigma}^{x})^{-1}$. This leads to a controller implementation scheme which we call a "blended" system-level controller.

2.9 Blending SL Controllers

We investigate the realizing controller corresponding to candidate CLMs of the type $\Psi_{\Sigma} = \sum_{i=1}^{N} \Psi_i G_i$ formed from a collection of candidate CLMs $\{\Psi_i\}_{i=1}^{N}$ and weight operators $\{G\}_{i=1}^{N}, \sum_{i=1}^{N} G = I$.

As usual, we decompose the computation of the realizing control law $\boldsymbol{u} = \Psi_{\Sigma}^{\mathrm{u}}(\Psi_{\Sigma}^{\mathrm{x}})^{-1}\boldsymbol{x}$ into the computation of an internal state, called the lumped (or effective) disturbance $\hat{\boldsymbol{w}} := (\Psi^{\mathrm{x}})_{\Sigma}^{-1}\boldsymbol{x}$, and the final control action $\boldsymbol{u} = \Psi_{\Sigma}^{\mathrm{u}}\hat{\boldsymbol{w}}$. Since we no longer assume that $\Psi_{\Sigma}^{\mathrm{x}}$ is a CLM of \boldsymbol{F} , we have to at least verify that it is a candidate CLM to ensure that $(\Psi^{\mathrm{x}})_{\Sigma}^{-1}$ exists and is causal. This is easy to verify from the fact that we assumed $\{\Psi_i\}_{i=1}^N$ to be all cCLMs:

Lemma 10. Given some operators $\{G\}_{i=1}^{N}$ such that $\sum_{i=1}^{N} G = I$, if $\{\Psi_i\}_{i=1}^{N}$ are all *cCLMs, then* $\Psi_{\Sigma} = \sum_{i=1}^{N} \Psi_i G_i$ *is a cCLM as well.*

Proof. Since the composition of causal operators is causal, it is clear that Ψ_{Σ} is causal. It is left to verify that $\Psi_{\Sigma}^{x} - I$ is strictly causal. To this end, write the former as

$$\Psi_{\Sigma}^{\mathbf{x}} - \boldsymbol{I} = \sum_{i=1}^{N} \Psi_{i}^{\mathbf{x}} \boldsymbol{G}_{i} - \boldsymbol{I} = \sum_{i=1}^{N} (\Psi_{i}^{\mathbf{x}} \boldsymbol{G}_{i} - \boldsymbol{G}_{i}) = \sum_{i=1}^{N} (\Psi_{i}^{\mathbf{x}} - \boldsymbol{I}) \boldsymbol{G}_{i}$$

and notice that $(\Psi_{i}^{\mathbf{x}} - \boldsymbol{I}) \in \mathcal{C}_{s}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}}) \implies (\Psi_{i}^{\mathbf{x}} - \boldsymbol{I}) \boldsymbol{G}_{i} \in \mathcal{C}_{s}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}}).$

From the above, we can realize K as a system-level implementation $SL[\Psi_{\Sigma}^{x}, \Psi_{\Sigma}^{u}]$ using the components of Ψ_{Σ} :

$$\hat{w}_t = x_t - \Psi_{\Sigma,t}^x(0, \hat{w}_{t-1:0}), \qquad u_t = \Psi_{\Sigma,t}^u(\hat{w}_{t:0}).$$

However, we derive an implementation in terms of the components of Ψ_i .

Rewrite the relation $\Psi^{\mathrm{x}}_{\Sigma}(\hat{\boldsymbol{w}}) = \boldsymbol{x}$ as

$$\Psi_{\Sigma}^{\mathbf{x}}(\hat{\boldsymbol{w}}) = \boldsymbol{x}$$

$$\Leftrightarrow \qquad \hat{\boldsymbol{w}} = \boldsymbol{x} + (\boldsymbol{I} - \Psi_{\Sigma}^{\mathbf{x}})\hat{\boldsymbol{w}} = \boldsymbol{x} + (\boldsymbol{I} - \sum_{i=1}^{N} \Psi_{i}^{\mathbf{x}} \boldsymbol{G}_{i})\hat{\boldsymbol{w}}$$

$$\Leftrightarrow \qquad \hat{\boldsymbol{w}} = \boldsymbol{x} + (\sum_{i=1}^{N} \boldsymbol{G}_{i} - \sum_{i=1}^{N} \Psi_{i}^{\mathbf{x}} \boldsymbol{G}_{i})\hat{\boldsymbol{w}} = \boldsymbol{x} + (\sum_{i=1}^{N} (\boldsymbol{G}_{i} - \Psi_{i}^{\mathbf{x}} \boldsymbol{G}_{i}))\hat{\boldsymbol{w}}$$

$$\Leftrightarrow \qquad \hat{\boldsymbol{w}} = \boldsymbol{x} + (\sum_{i=1}^{N} (\boldsymbol{I} - \Psi_{i}^{\mathbf{x}})\boldsymbol{G}_{i})\hat{\boldsymbol{w}} = \boldsymbol{x} + \sum_{i=1}^{N} (\boldsymbol{I} - \Psi_{i}^{\mathbf{x}})\boldsymbol{G}_{i}\hat{\boldsymbol{w}} \qquad (2.53)$$

$$\Leftrightarrow \qquad \hat{\boldsymbol{w}} = \boldsymbol{x} + \sum_{i=1}^{N} (\boldsymbol{I} - \Psi_{i}^{\mathbf{x}})\tilde{\boldsymbol{w}}^{i}$$

and decompose the control action $\boldsymbol{u} = \boldsymbol{\Psi}^{\mathrm{u}}_{\Sigma}(\boldsymbol{\hat{w}})$ as

$$\boldsymbol{u} = \boldsymbol{\Psi}_{\Sigma}^{\mathrm{u}}(\boldsymbol{\hat{w}}) = (\sum_{i=1}^{N} \boldsymbol{\Psi}_{i}^{\mathrm{u}} \boldsymbol{G}_{i}) \boldsymbol{\hat{w}} = \sum_{i=1}^{N} (\boldsymbol{\Psi}_{i}^{\mathrm{u}} \boldsymbol{G}_{i} \boldsymbol{\hat{w}}) = \sum_{i=1}^{N} \boldsymbol{\Psi}_{i}^{\mathrm{u}} \boldsymbol{\tilde{w}}^{i}, \qquad (2.54)$$

where we define $\tilde{w}^i = G_i(\hat{w})$ as a partial disturbance term corresponding to the *i*th cCLM Ψ_i . This leads to the following implementation of the realizing controller K:

Definition 2.13. Let $\mathbf{K} = \Psi_{\Sigma}^{u}(\Psi_{\Sigma}^{x})^{-1}$ for a cCLM of the form $\Psi_{\Sigma} = \sum_{i=1}^{N} \Psi_{\Sigma} \mathbf{G}_{i}$, where $\{\Psi_{i}\}_{i=1}^{N}$ are cCLMs, and $\{\mathbf{G}_{i}\}_{i=1}^{N}$ are a collection of causal operators such that $\sum_{i=1}^{N} \mathbf{G}_{i} = \mathbf{I}$. Then, the following realization of \mathbf{K} is defined as its blended SL implementation:

$$\hat{w}_t = x_t - \sum_{i=1}^N \Psi_{i,t}^x(0, \tilde{w}_{t-1:0}^i)$$
(2.55a)

$$\tilde{w}_t^i = G_t^i(\hat{w}_{t:0}) \tag{2.55b}$$

$$u_t = \sum_{i=1}^{N} \Psi_{i,t}^u(\tilde{w}_{t:0}^i).$$
(2.55c)

In the later sections, we will consider the special case where Ψ_{Σ} is a blend of linear cCLMs $\{\Psi_i\}_{i=1}^N$ and static nonlinearities $G_{i,t}(z_{t:0}) := g_i(z_t)$. The corresponding blended SL implementation takes the form:

$$\hat{w}_t = x_t - \sum_{i=1}^N \sum_{k=2}^{t+1} R^i_{t,k} \tilde{w}^i_{t+1-k}$$
(2.56a)

$$\tilde{w}_t^i = g_i(\hat{w}_t) \tag{2.56b}$$

$$u_t = \sum_{i=1}^{N} \sum_{k=1}^{t+1} M_{t,k}^i \tilde{w}_{t+1-k}^i$$
(2.56c)

where the matrices $R_{t,k}^i \in \mathbb{R}^{n \times n}$ and $M_{t,k}^i \in \mathbb{R}^{n \times m}$ are associated with the linear component functions of the cCLMs Ψ_i as follows:

$$\Psi_{i,t}^{\mathbf{x}}(z_{1:t+1}) = \sum_{k=1}^{t+1} R_{t,k}^{i} z_{k}, \qquad \Psi_{i,t}^{\mathbf{u}}(z_{1:t+1}) = \sum_{k=1}^{t+1} M_{t,k}^{i} z_{k}.$$
(2.57)

Internal Stability of the Closed Loop

To investigate the stability of the overall closed loop, as described in the previous section, we need to add the internal states of the implementation \hat{w} and \tilde{w}^i to our closed loop model and analyze the stability of the corresponding closed loop maps.

To derive the dynamics of the lumped disturbance, we substitute $\boldsymbol{x} = \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w}$ into $\hat{\boldsymbol{w}} = \boldsymbol{x} + (\boldsymbol{I} - \boldsymbol{\Psi}_{\Sigma}^x)\hat{\boldsymbol{w}}$. Furthermore, since $(\boldsymbol{x}, \boldsymbol{u}) = \boldsymbol{\Psi}_{\Sigma}(\hat{\boldsymbol{w}})$ we have $\boldsymbol{x} = \boldsymbol{F}^+ \boldsymbol{\Psi}_{\Sigma} \hat{\boldsymbol{w}} + \boldsymbol{w}$ which yields:

$$egin{aligned} \hat{m{w}} &= (m{F}^+ m{\Psi}_\Sigma) \hat{m{w}} + m{w} + (m{I} - m{\Psi}^{\mathrm{x}}_\Sigma) \hat{m{w}} = (m{F}^+ m{\Psi}_\Sigma + m{I} - m{\Psi}^{\mathrm{x}}_\Sigma) \hat{m{w}} + m{w} \ &= m{w} + m{\Delta} [m{F}, m{\Psi}_\Sigma] \hat{m{w}} = \sum_{i=1}^N m{\Delta} [m{F}, m{\Psi}_i] m{G}_i \hat{m{w}} + m{w}. \end{aligned}$$

Finally, since we established that Ψ_{Σ} is a candidate CLM, $\Delta[F, \Psi_{\Sigma}] = F^+(\Psi_{\Sigma}) + (I - \Psi^x)$ is strictly causal and assures that $I - \Delta[F, \Psi_{\Sigma}]$ is causally invertible. Hence, the mapping $w \mapsto \hat{w}$ is defined by the following equation:

$$\hat{\boldsymbol{w}} = (\boldsymbol{I} - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}_{\Sigma}])^{-1} \boldsymbol{w} = (\boldsymbol{I} - \sum_{i=1}^{N} \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}_{i}]\boldsymbol{G}_{i})^{-1} \boldsymbol{w}$$
(2.58)

$$= \left(\sum_{i=1}^{N} (\boldsymbol{I} - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}_i]) \boldsymbol{G}_i\right)^{-1} \boldsymbol{w}.$$
(2.59)

Since F is assumed linear, we can invoke Corollary 10 of Theorem 8 to obtain conditions for closed loop internal stability. Recall the perturbed closed loop model δ CL with SL[$\Psi_{\Sigma}^{x}, \Psi_{\Sigma}^{u}$] as the controller. We have an internal ℓ_{p} -stable closed loop under the following conditions:

Theorem 15. Let Ψ_{Σ} be a \mathcal{G} -blended cCLM corresponding to a family of ℓ_p stable cCLMs $\mathcal{F} = {\{\Psi_i\}_{i=1}^N}$ and a family of ℓ_p -stable weights $\mathcal{G} = {\{G_i\}_{i=1}^N}$. Consider the closed loop system δ CL described by the equations (2.40) for a fixed ℓ_p -stable $\mathbf{F} \in \mathcal{LC}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and controller $\mathbf{K}_{\Sigma} = \Psi_{\Sigma}^{\mathrm{u}}(\Psi_{\Sigma}^{\mathrm{x}})^{-1}$ with implementation $\mathrm{SL}[\Psi_{\Sigma}^{\mathrm{x}}, \Psi_{\Sigma}^{\mathrm{u}}]$.

Then, the map $\Phi_{\delta CL} : (\boldsymbol{w}, \boldsymbol{v}, \boldsymbol{d}) \mapsto (\hat{\boldsymbol{w}}, \boldsymbol{x}, \boldsymbol{u})$ of the closed loop δCL is ℓ_p -stable if the operator $(\sum_{i=1}^{N} (\boldsymbol{I} - \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}_i])\boldsymbol{G}_i)^{-1}$ is ℓ_p -stable.

In the next section, we discuss how blending inspires a nonlinear control synthesis procedure for the constrained LQR problem, which strictly outperforms any linear controller.

2.10 Conclusion

This chapter highlights a general and fundamental relationship between closed loop maps and corresponding realizing controllers in general nonlinear discrete-time systems. The key findings are as follows: 1. All closed loop maps are solutions to an operator equation, and all solutions of the equation are achievable closed loop maps. 2. Given a solution of the operator equation, we can obtain a realizing controller by parameterizing a system-level controller with the solution. This controller then imposes the given solution as the closed loop map of the system. 3. This same procedure produces robust closed loop stability even when the system-level controllers are parameterized with approximate solutions of the operator equation.

We discuss an important consequence for the special case of LTV system dynamics and nonlinear dynamics in the controller: The space of (nonlinear) CLMs is closed under "blending"-type combinations. This observation informs a procedure for constructing complex (or more expressive) CLMs from a collection of simpler CLMs. This idea can be leveraged for nonlinear SL-controller synthesis and provides, for example, systematic instructions to synthesize nonlinear controllers by "blending" multiple linear SL controllers into one.

In the next chapter, we will see that this technique has important implications for the problem setting of linear systems with actuator saturation and provides new ways to perform simple stability and performance analysis of the closed loop.

2.A Proofs

Let $(\mathcal{X}, |\cdot|_{\mathcal{X}})$ and $(\mathcal{Y}, |\cdot|_{\mathcal{Y}})$ be some finite-dimensional banach spaces, $\mathcal{X}^{\mathbb{N}}, \mathcal{Y}^{\mathbb{N}}$ be the corresponding vector space of sequences over \mathbb{N} . Let $\mathcal{X}_p^{\mathbb{N}} \subset \mathcal{X}^{\mathbb{N}}$ and $\mathcal{Y}_p^{\mathbb{N}} \subset \mathcal{Y}^{\mathbb{N}}$ denote the normed vectorspaces w.r.t. to the usual ℓ_p -norms defined for sequences in \mathbb{N} . Let $\mathcal{C}(\mathcal{X}^{\mathbb{N}}, \mathcal{Y}^{\mathbb{N}})$ be the space of all causal operators $\mathcal{X}^{\mathbb{N}} \to \mathcal{Y}^{\mathbb{N}}$ and define $||| \cdot |||_p^* : \mathcal{C}(\mathcal{X}^{\mathbb{N}}, \mathcal{Y}^{\mathbb{N}}) \to \mathbb{R}_0^+ \cup \infty$ and $||| \cdot |||_p^* : \mathcal{C}(\mathcal{X}^{\mathbb{N}}, \mathcal{Y}^{\mathbb{N}}) \to \mathbb{R}_0^+ \cup \infty$, with $p \in \{1, 2, \dots, \infty\}$ for some operator $\mathbf{A} \in \mathcal{C}(\mathcal{X}^{\mathbb{N}}, \mathcal{Y}^{\mathbb{N}})$:

$$\|\|m{A}\|\|_p^\star := \sup_{m{x}\in\mathcal{X}_p^\mathbb{N}:\,m{x}
eq 0} \; rac{\|m{A}(m{x})-m{A}(m{0})\|_p}{\|m{x}\|_p} \qquad \|\|m{A}\|\|_p := \|\|m{A}\|\|_p^\star + \|m{A}(m{0})\|_p.$$

Lemma 11. $\|\| \cdot \|\|_p$ is a norm on $\mathcal{C}(\mathcal{X}_p^{\mathbb{N}}, \mathcal{Y}_p^{\mathbb{N}})$ and $\|\| \cdot \|\|_p^*$ is a norm on $\{A \in \mathcal{C}(\mathcal{X}_p^{\mathbb{N}}, \mathcal{Y}_p^{\mathbb{N}}) \mid A(0) = 0\}$ or the quotientspace $\mathcal{C}(\mathcal{X}_p^{\mathbb{N}}, \mathcal{Y}_p^{\mathbb{N}}) / \sim$ with the equivalence relation $A \sim B \Leftrightarrow A - B \equiv const$. Furthermore the following holds:

- (i) $\|\|\mathbf{A} \mathbf{B}\|\|_{p}^{\star} = 0$ if and only if \mathbf{A} and \mathbf{B} are the same up to a constant offset.
- (ii) Submultiplicativety holds for products TA and AB if T is linear and B(0) = 0:

$$\|\|TA\|\|_p^\star \leqslant \|T\|\|_p^\star \|A\|\|_p^\star \qquad \|\|AB\|\|_p^\star \leqslant \|A\|\|_p^\star \|B\|\|_p^\star.$$

Proof. If $|||\mathbf{A}|||_p^* = 0$, then for all $\mathbf{x} \in \mathcal{X}_p^{\mathbb{N}}, \mathbf{x} \neq 0$ holds $\mathbf{y} = \mathbf{A}(\mathbf{x}) = \mathbf{y}_0 := \mathbf{A}(\mathbf{0})$ because otherwise $(||\mathbf{y} - \mathbf{y}_0||_p)/||\mathbf{x}||_p \neq 0$ leads to a contradiction. Let $r \neq 0$ then $r\mathbf{A} = (r\mathbf{I}) \circ \mathbf{A}$ means that $||r\mathbf{A}(\mathbf{x}) - r\mathbf{A}(\mathbf{0})||_p = ||r(\mathbf{A}(\mathbf{x}) - \mathbf{A}(\mathbf{0}))||_p =$ $|r|||(\mathbf{A}(\mathbf{x}) - \mathbf{A}(\mathbf{0}))||_p$ and therefore $|||r\mathbf{A}|||_p^* = |r||||\mathbf{A}|||_p^*$. It remains to show a notion of sub-additivity. Let $\mathbf{A}, \mathbf{B} \in \mathcal{C}(\mathcal{X}^{\mathbb{N}}, \mathcal{Y}^{\mathbb{N}})$, then for all $\mathbf{x} \in \mathcal{X}_p^{\mathbb{N}}$ holds $(\mathbf{A} + \mathbf{B})(\mathbf{x}) - (\mathbf{A} + \mathbf{B})(\mathbf{0}) = \mathbf{A}(\mathbf{x}) - \mathbf{A}(\mathbf{0}) + \mathbf{B}(\mathbf{x}) - \mathbf{B}(\mathbf{0})$ and by triangle inequality of the ℓ_p -norm follows

$$\|(A+B)(x) - (A+B)(0)\|_p \le \|A(x) - A(0)\|_p + \|B(x) - B(0)\|_p.$$

Let $\mathcal{X} = \mathcal{Y}$ and consider $AB \in \mathcal{C}(\mathcal{X}_p^{\mathbb{N}}, \mathcal{X}_p^{\mathbb{N}})$ with B(0) = 0. Then

$$\begin{split} \|\|\boldsymbol{A}\|\|_{p}^{\star} &= \sup_{\boldsymbol{x}\in\mathcal{X}_{p}^{\mathbb{N}}:\,\boldsymbol{x}\neq\boldsymbol{0}} \quad \frac{\|\boldsymbol{A}\boldsymbol{B}(\boldsymbol{x})-\boldsymbol{A}\boldsymbol{B}(\boldsymbol{0})\|_{p}}{\|\boldsymbol{x}\|_{p}} \\ &= \sup_{\boldsymbol{x}\in\mathcal{X}_{p}^{\mathbb{N}}:\,\boldsymbol{B}(\boldsymbol{x})\neq\boldsymbol{0}} \quad \frac{\|\boldsymbol{A}\boldsymbol{B}(\boldsymbol{x})-\boldsymbol{A}(\boldsymbol{0})\|_{p}}{\|\boldsymbol{B}(\boldsymbol{x})\|_{p}} \frac{\|\boldsymbol{B}(\boldsymbol{x})\|_{p}}{\|\boldsymbol{x}\|_{p}} \\ &\leq \sup_{\boldsymbol{y}\in\mathcal{X}_{p}^{\mathbb{N}}:\,\boldsymbol{y}\neq\boldsymbol{0}} \quad \frac{\|\boldsymbol{A}(\boldsymbol{y})-\boldsymbol{A}(\boldsymbol{0})\|_{p}}{\|\boldsymbol{y}\|_{p}} \sup_{\boldsymbol{x}\in\mathcal{X}_{p}^{\mathbb{N}}:\,\boldsymbol{x}\neq\boldsymbol{0}} \frac{\|\boldsymbol{B}(\boldsymbol{x})-\boldsymbol{B}(\boldsymbol{0})\|_{p}}{\|\boldsymbol{x}\|_{p}} = \|\|\boldsymbol{A}\|_{p}^{\star}\|\|\boldsymbol{B}\|_{p}^{\star}. \end{split}$$

Now, let $T \in \mathcal{LC}(\mathcal{X}_p^{\mathbb{N}}, \mathcal{X}_p^{\mathbb{N}})$ be a linear operator and notice that TA(x) - TA(0) = T(A(x) - A(0)). We repeat the same process as above and obtain $|||TA|||_p^* \leq ||T||_p^* |||A|||_p^*$.

Chapter 3

NONLINEAR BLENDING APPROACH TO CONTROL OF LARGE-SCALE SYSTEMS

The theory developed in the previous chapter provides a framework to systematically "blend" multiple (possibly) nonlinear controllers into one stabilizing nonlinear controller. Blending happens at the controller implementation level and can be used to optimize the performance of the nonlinear closed-loop map by combining the desired properties of multiple closed-loop maps. In this chapter, we explore three applications to demonstrate the utility of this approach in control settings challenged by complex dynamics and constraints in the system or controller.

3.1 Introduction

As a warm-up, we discuss the time-varying design of the SL controller and the local stability analysis for closed loops involving nonlinear continuous-time systems, which is the first rigorous discussion of this topic. As an example, we use the problem of trajectory tracking for nonlinear continuous-time systems through discrete-time zero-order hold feedback control. Our empirical case study evaluates the SL controller on the cart-pole system and demonstrates that despite using only a rough model for synthesis, the resulting controller demonstrates very robust closed-loop performance.

In the later part, we consider nonlinear control settings, where complexity in the system dynamics is caused by sheer scale and distributedness of the system, rather than high-order nonlinearity in the dynamics. In particular, we show that the framework is particularly well-suited for large-scale systems subjected to input saturation and state constraints, but with otherwise linear dynamics. We investigate some first application scenarios where our approach naturally provides significant benefits over existing methods: distributed constrained LQR and distributed anti-windup control.

In the case of constrained LQR, we derive a synthesis procedure for blended SLS controllers that outperforms any optimal linear controller for the constrained LQR problem [36, 86, 91, 140]. As a second application, we discuss how the blended SLS technique provides a natural remedy for controller-windup in a way that is easily scalable for use in large-scale control systems. We discuss the efficacy of the methods with simulations and show that synthesis and implementation enjoy

the same benefits as previous SLS synthesis methods: both are distributed, handle delays, sparse actuation, and allow for localized disturbance rejection. The presented work is based on [64] and [137].

Setup and Recap of Key Concepts

For this chapter, we consider linear dynamical systems described by the operator equation

$$\boldsymbol{x} = \mathcal{S}^+ \boldsymbol{F}(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w} =: \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w}, \tag{3.1}$$

where $F^+ = S^+(F)$ is a strictly causal operator formed by a right shift of causal linear operator F which we call dynamics. As linearity allows, we split F into two causal linear operators $A : \ell^n \mapsto \ell^n$, $B : \ell^m \mapsto \ell^n$ and write F as F(x, u) :=(A(x) + B(u)). We represent A and B by a sequence of matrices $\{A_{t,k} \in \mathbb{R}^{n \times n} | t \ge$ $0, k \ge 1\}, \{B_{t,k} \in \mathbb{R}^{n \times n} | t \ge 0, k \ge 1\}$ parametrizing the component functions of A_t and B_t as:

$$A_t(x_{t:0}) = \sum_{k=0}^t A_{t,k+1} x_{t-k}, \qquad B_t(u_{t:0}) = \sum_{k=0}^t B_{t,k+1} u_{t-k}.$$

Correspondingly, F has the component functions $F_t(x_{t:0}, u_{t:0}) = A_t(x_{t:0}) + B_t(x_{t:0})$, the components of the dynamics F^+ are

$$F_t^+(x_{t:0}, u_{t:0}) = \sum_{k=1}^t A_{t-1,k} x_{t-k} + \sum_{k=1}^t B_{t-1,k} u_{t-k}$$
(3.2)

and the difference equation of the system dynamics takes the form:

$$x_t = \sum_{k=1}^t A_{t-1,k} x_{t-k} + \sum_{k=1}^t B_{t-1,k} u_{t-k} + w_t.$$
(3.3)

For the sake of defining the closed-loop maps, we view w as an input and the pair (x, u) as an output. Then, as in the previous chapter, for a causal controller $K : \ell^n \mapsto \ell^m$ and fixed plant F, the corresponding $w \mapsto \{x, u\}$ map of the closed-loop CL is the operator $\Psi = \begin{bmatrix} \Psi^x \\ \Psi^u \end{bmatrix}$ defined by the partial maps Ψ^x and Ψ^u :

$$\Psi^{\mathrm{x}} := \left(\boldsymbol{I} - \mathcal{S}^{+} \boldsymbol{F}(\boldsymbol{I}, \boldsymbol{K}) \right)^{-1}, \qquad \Psi^{\mathrm{u}} := \boldsymbol{K} \left(\boldsymbol{I} - \mathcal{S}^{+} \boldsymbol{F}(\boldsymbol{I}, \boldsymbol{K}) \right)^{-1}.$$

An operator $\Psi = (\Psi^x, \Psi^u)$ with partial maps Ψ^x and Ψ^u is a candidate closed-loop map (cCLM) of F, if it is causal and of conforming domain and co-domains, i.e., $\Psi^x : \ell^n \mapsto \ell^n$ and $\Psi^u : \ell^n \mapsto \ell^m$ and $\Psi^x - I$ is strictly causal. The residual Δ is a $\begin{array}{l} \text{map } \boldsymbol{\Delta} : \mathcal{C}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}}) \times \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}}) \rightarrow \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}}) \text{ where for } \boldsymbol{F} \in \mathcal{C}(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}}) \text{ and} \\ \boldsymbol{\Psi} = \begin{bmatrix} \boldsymbol{\Psi}^{\mathrm{x}} \\ \boldsymbol{\Psi}^{\mathrm{u}} \end{bmatrix} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}}), \, \boldsymbol{\Delta}[\boldsymbol{F}, \boldsymbol{\Psi}] \text{ is defined as:} \end{array}$

$$\Delta[F, \Psi] = F^+(\Psi) + I - \Psi^x =: S^+F(\Psi) + I - \Psi^x$$

The set of all closed-loop maps of F is denoted as $\Phi_{CL}^{w \mapsto xu}[F]$ and by Theorem 8 can be equivalently defined as:

$$\Phi_{\mathrm{CL}}^{\mathrm{w}\mapsto\mathrm{xu}}[F] = \{\Psi \mid \Delta[F,\Psi] = 0\}$$

3.2 Discrete-Time Trajectory-Tracking Control for Nonlinear Continuous-Time Systems

We derived that, in theory, we only need operators (Ψ^x, Ψ^u) to approximately satisfy the CLM condition (2.33) to obtain robustly stabilizing controllers SL[Ψ^x, Ψ^u]. The generality of the robustness result argues that system level controllers could be a promising design tool in practical control applications. However, more research is needed to quantify the trade-off between the approximation grade of the condition (2.33) and the corresponding achievable control performance. As a first step towards that, we present some first empirical results that show that system-level controllers can achieve good robust control performance in challenging-to-control nonlinear systems while using only crude models of the system for synthesis. Figure 3.1 shows simulation results of using a system level controller SL[Ψ^x, Ψ^u] at 30 Hz sampling time to swing up a cart pole system under small and large closed-loop perturbations. Rather than satisfying the CLM condition (2.33) of the zero-order hold actuated cart pole system, the maps Ψ are synthesized using the following approximations:

- Ψ are taken to be affine operators, where the affine term is a sampled continuoustime desired trajectory $(x^d(t), u^d(t))$ for the system and linear part is chosen to be finite memory (2 s window in continuous-time).
- The continuous-time trajectories $(x^d(t), u^d(t))$ are low-grade approximations of swing-up motions of the cartpole.
- Ψ are chosen as CLMs of an *approximation* of the linearized system around the desired trajectory.

The above simplifications make clear that Ψ serves only as a very coarse approximation of the exact CLM condition (2.33). On the other hand, the above approximations allow to synthesize the linear part of Ψ analytically and in parallel, allowing for



Figure 3.1: Swing-up control for cart-pole system with system level controller at 30Hz sampling rate. $(x, \dot{x}, \theta, \dot{\theta}, f)$ stand for cart position / velocity, pole angle / velocity, and cart force with units (m, m/s, deg, deg/s, N). $\theta = 180^{\circ}$ stands for the upward pole position. The weight of the cart and the pole is chosen as 1 kg and 0.1 kg, the length of the pole is chosen as 0.5 m. Consult [122] for detailed system description and equations. *Left*: Desired trajectory (red) vs. closed-loop performance under initial condition error $\psi(0) = 45^{\circ}$ and two scenarios of small (orange) and large (blue) system perturbations w, v, d. *Middle*: Evolution of internal state \hat{w} for both scenarios and normalized disturbance due to trajectory error \bar{e}_i . *Right*: i.i.d Gaussian perturbations w, v, d.

efficient computation.

Leaning on the discussion in Section 2.7, the closed-loop S_2 of the cart pole simulation can be written as

$$x_t = \phi_{t_s}(x_{t-1}, u_{t-1}) + w_t + e_t, \tag{3.4a}$$

$$\hat{w}_t = x_t + v_t - x^d(t\tau_s) - \sum_{k=2}^{t+1} R_{t,k} \hat{w}_{t+1-k}, \qquad (3.4b)$$

$$u_t = u^d(t\tau_s) + \sum_{k=1}^{t+1} M_{t,k} \hat{w}_{t+1-k} + d_t, \qquad (3.4c)$$

where w, d and v are state, input and internal controller state perturbations and e_t is due to errors in the trajectory synthesis. The matrices $R_{t,k} \in \mathbb{R}^{n \times n}$, $M_{t,k} \in \mathbb{R}^{m \times n}$ parameterize the linear part of Ψ . Due to the approximation steps taken in the synthesis of Ψ , there is a considerable gap between the real system and the model used for the synthesis. Nevertheless, Figure 3.1 shows that despite the large uncertainty of the model, the closed loop provides robust performance against a variety of perturbations: Large initial condition errors, large perturbation signals, and errors in trajectory.

Derivation and Nonlinear Stability Analysis

Using the cart pole system as an example, we demonstrate how to develop a systemlevel controller to track trajectories for nonlinear continuous-time systems. We use the description of the cart pole as presented in [122] and refer to the same reference for detailed derivations. The dynamic equations of the cart pole are

$$(m_c + m_p)\ddot{x}_c + m_p l\dot{\theta}_p \cos\theta_p - m_p l\dot{\theta}_p^2 \sin\theta_p = f$$
(3.5a)

$$m_p l\ddot{x}_c \cos\theta + m_p l^2 \ddot{\theta}_p + m_p g l \sin\theta_p = 0$$
(3.5b)

where x_c and θ_p stand for cart position and pole angle in counterclockwise direction and f represents the force exerted on the cart. Furthermore, $\theta = 0$ denotes the downward position. The parameters (m_c, m_p, l, g) are chosen as $(1 \text{ kg}, 0.1 \text{ kg}, 0.5 \text{ m}, 9.81 \text{m/s}^2)$ and represent the mass of the cart and the pole, the length of the pole and the gravity constant, respectively. Furthermore, (3.5) can be converted into the input affine standard form

$$\dot{x} = F(x) + g(x)u \tag{3.6}$$

where $x = [x_c, \theta_p, \dot{x}_c, \dot{\theta}_p]^T$, u = f, (see [122] for description of F(x) and g(x)). As in practice, controllers are usually implemented digitally, we assume zero-order hold on the input u with a sampling time of $\tau_s = 0.033 \text{sec} (1/\tau_s = 30 \text{Hz})$. Because of this discretization, we can equivalently represent the system (3.6) at sampling times through the discrete-time system

$$x_{t} = \phi_{\tau_{s}}(x_{t-1}, u_{t-1}), \quad \phi_{\tau_{s}}(x, u) := \alpha(\tau_{s}), \text{ s.t. } : \dot{\alpha} = F(\alpha, u), \alpha(0) = x, \quad (3.7)$$

where we denote $x_t := x(t\tau_s)$ and $u_t := u(t\tau_s)$ ($t \in \mathbb{N}$) to be samples of the continuous-time signals $x(\tau)$, $u(\tau)$ at time $t\tau_s$. To put (3.7) in operator form, define $\mathbf{F}^{\phi} \in \mathcal{C}_s(\ell^n \times \ell^m, \ell^n)$ with component functions $F_t^{\phi}(x_{t:0}, u_{t:0}) := \phi_{\tau_s}(x_{t-1}, u_{t-1})$ and equation (3.7) can be written in terms of the trajectories (\mathbf{x}, \mathbf{u}) as $\mathbf{x} = \mathbf{F}^{\phi}(\mathbf{x}, \mathbf{u})$.

Remark 16. We use the variable τ to indicate that a variable $a(\tau)$ is a continuous-time signal and use a_t to refer to the discrete-time samples $a_t := a(t\tau_s)$.

We use a continuous-time trajectory $x^d(\tau), u^d(\tau)$ as a reference, which is shown in red in Figure 3.1. The trajectories approximately satisfy the continuous-time dynamics, i.e., $\dot{x}^d(\tau) \approx F(x^d(\tau)) + g(x^d(\tau))u^d(\tau)$, and are designed to swing up the pole in 3 seconds and then keep the cart pole at $x_c = 0$, $\theta_p = \pi$.

Remark 17. Notice that even if $\dot{x}^d(\tau) = F(x^d(\tau)) + g(x^d(\tau))u^d(\tau)$, it still does not hold $x_t^d \neq \phi_{\tau_s}(x_{t-1}^d, u_{t-1}^d)$, since the continuous-time trajectories are computed without consideration of the zero-order hold actuation.

We take the following linear approximation of ϕ_{τ_s} around the reference trajectory:

$$\phi_{\tau_s}(x_{t-1}, u_{t-1}) \approx x_t^d + \underbrace{\exp(\nabla F|_{x_{t-1}^d} \tau_s)}_{=:\hat{A}_{t-1}} * (x_{t-1} - x_{t-1}^d) \dots$$
(3.8)
+
$$\underbrace{\int_0^{\tau_s} \exp(\nabla F|_{x_{t-1}^d} \tau) g(x_{t-1}^d) d\tau}_{=:\hat{B}_{t-1}} * (u_{t-1} - u_{t-1}^d).$$
(3.9)

Denote $[A_{t-1}, B_{t-1}] := \nabla_{x,u} \phi_{\tau_s}|_{(x_{t-1}^d, u_{t-1}^d)}$ the true linearization of ϕ_{τ_s} at (x_{t-1}^d, u_{t-1}^d) and notice that the above approximation (3.8) is only an approximation of the linearization, since $[\hat{A}_{t-1}, \hat{B}_{t-1}] \neq [A_{t-1}, B_{t-1}]$. Using Taylor's theorem and assuming ϕ_{τ_s} is differentiable, we can write ϕ_{τ_s} as

$$\phi_{\tau_s}(x_{t-1}, u_{t-1}) = \phi_{\tau_s}(x_{t-1}^d, u_{t-1}^d) + A_{t-1}(x_{t-1} - x_{t-1}^d) + B_{t-1}(u_{t-1} - u_{t-1}^d) + r_{t-1}(x_{t-1} - x_{t-1}^d, u_{t-1} - u_{t-1}^d)$$

where $\lim_{|z|\to 0} |r_{t-1}(z)|/|z| = 0$. We can factor out equation (3.7) into the following components

$$x_{t} = x_{t}^{d} + \hat{A}_{t-1}(x_{t-1} - x_{t-1}^{d}) + \hat{B}_{t-1}(u_{t-1} - u_{t-1}^{d}) + e_{t} + e_{t}'(x_{t-1}, u_{t-1})$$
(3.10)

where e_t and $e'_t(x_{t-1}, u_{t-1})$ are disturbance terms introduced due to errors in the reference trajectory and linearization

$$e_t = \phi_{\tau_s}(x_{t-1}^d, u_{t-1}^d) - x_t^d$$

$$e'_t(x_{t-1}, u_{t-1}) = (A_{t-1} - \hat{A}_{t-1})(x_{t-1} - x_{t-1}^d) + (B_{t-1} - \hat{B}_{t-1})(u_{t-1} - u_{t-1}^d)$$

$$\cdots + r_{t-1}(x_{t-1} - x_{t-1}^d, u_{t-1} - u_{t-1}^d)$$

and the remaining terms represent our linear approximation of the dynamics. we can express this more compactly in operator form: Define $\tilde{F}^{\phi} \in C_s(\ell^n, \ell^n)$ with the components

$$\tilde{F}_t^{\phi}(x_{t:0}, u_{t:0}) := x_t^d + \hat{A}_{t-1}(x_{t-1} - x_{t-1}^d) + \hat{B}_{t-1}(u_{t-1} - u_{t-1}^d)$$

and the residual $\Delta^{\phi} \in C(\ell^n, \ell^n)$ with the components $\Delta^{\phi}_t(x_{t:0}, u_{t:0}) = e_t + e'_t(x_{t-1}, u_{t-1})$, then we can factor out the original equations of (3.7) as

$$\boldsymbol{x} = \tilde{\boldsymbol{F}}^{\phi}(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{\Delta}^{\phi}(\boldsymbol{x}, \boldsymbol{u})$$
(3.12)

where per definition we have the decomposition $ilde{F}^{\phi} + \Delta^{\phi} = F^{\phi}.$

Our approach for synthesis is now to use \tilde{F}^{ϕ} as a model to design a system level controller and treat Δ^{ϕ} as disturbance terms we want to be robust against. We do this, first solving for the CLMs $\tilde{\Psi} = (\tilde{\Psi}^{x}, \tilde{\Psi}^{u})$ of \tilde{F}^{ϕ} and then choosing our feedback controller as $SL[\tilde{\Psi}^{x}, \tilde{\Psi}^{u}]$.

Due to Theorem 8, $\tilde{\Psi}$ is a CLM of \tilde{F}^{ϕ} if and only if it satisfies the CLM equation (2.33) for \tilde{F}^{ϕ} , i.e., $\tilde{\Psi}^{x} = \tilde{F}^{\phi}(\tilde{\Psi}) + I$. We restrict $\tilde{\Psi}$ to be of affine form $\tilde{\Psi} : \boldsymbol{w} = \begin{bmatrix} \boldsymbol{R} \\ \boldsymbol{M} \end{bmatrix}(\boldsymbol{w}) + \begin{bmatrix} \boldsymbol{r} \\ \boldsymbol{m} \end{bmatrix}$ with $\begin{bmatrix} \boldsymbol{r} \\ \boldsymbol{m} \end{bmatrix} \in \ell^{n} \times \ell^{m}$ and $\boldsymbol{R} \in \mathcal{LC}(\ell^{n}, \ell^{n}), \boldsymbol{M} \in \mathcal{LC}(\ell^{n}, \ell^{m})$. Thus the component functions of $\tilde{\Psi}$ take the from

$$\tilde{\Psi}_{t}^{x}(\alpha_{1:t+1}) = \sum_{k=1}^{t+1} R_{t,k}\alpha_{k} + r_{t}$$
(3.13)

$$\tilde{\Psi}_{t}^{u}(\alpha_{1:t+1}) = \sum_{k=1}^{t+1} M_{t,k}\alpha_{k} + m_{t}$$
(3.14)

where $R_{t,j} \in \mathbb{R}^{n \times n}$, $M_{t,j} \in \mathbb{R}^{n \times m}$, r_t , m_t are some fix sequences and $\alpha_{1:t+1}$ denote the t + 1 arguments of the component function. Structuring $\tilde{\Psi}$ in this form and using linearity of \tilde{F}^{ϕ} reduces the original operator equation $\tilde{\Psi}^x = \tilde{F}^{\phi}(\tilde{\Psi}) + I$ simply to the following set of linear equations for $R_{t,j}$, $M_{t,j}$ and r_t , m_t :

$$R_{t,k} = \hat{A}_{t-1}R_{t-1,k-1} + \hat{B}_{t-1}M_{t-1,k-1} \text{ for all } k \leq t, \quad R_{t,1} = I$$
(3.15)

$$r_t = x_t^d, \quad m_t = u_t^d.$$
 (3.16)

Equation (3.15) is an affine subspace constraint and opens up many possible ways to synthesize for solutions $R_{t,k}$, $M_{t,k}$. In fact, (3.15) matches the linear time-varying formulation of SLS as discussed in [66], [11] and for our case-study here, we are synthesizing for $R_{t,k}$, $M_{t,k}$ by solving the following \mathcal{H}_2 / LQR problem for the LTV system \tilde{F}^{ϕ} subject to an FIR constraint with horizon T = 60 time-steps:

$$\min_{\substack{R_{t,k}, M_{t,k} \\ s.t.}} \sum_{\substack{0 \le t \le H \ 1 \le k \le T \\ k = A_{t-1}R_{t-1,k-1} + \hat{B}_{t-1}M_{t-1,k-1} \\ R_{t,1} = I, \quad R_{t,T} = 0.} \sum_{\substack{0 \le t \le H \ 1 \le k \le T \\ k = A_{t-1}R_{t-1,k-1} + \hat{B}_{t-1}M_{t-1,k-1} \\ R_{t,1} = I, \quad R_{t,T} = 0.}$$
(3.17)

Remark 18. See [11] for details of the \mathcal{H}_2/LQR problem setup and derivation of the convex optimization problem. The FIR horizon can be understood as a time window [t, t + T] given to the controller to kill off the disturbance \hat{w}_t . Considering the sampling time of our example, T = 60 here translates to a 2-second window in continuous time.

Furthermore, H denotes the length of the trajectory in sampling time steps. The above problem can be solved in closed form since it is a QP without inequality constraints. Moreover, the change of variables $R_{j+h,j+1}$, $M_{j+h,j+1}$ with $0 \le j \le T-1$, $0 \le h \le H$ shows that (3.17) can be decomposed over h into H separate QP's that can be solved analytically and in parallel, hence showing that the computational complexity of our synthesis approach is **independent** of the trajectory length H.

The solutions of (3.17) are taken to parameterize the operators (3.13) which give us the system-level controller $SL[\tilde{\Psi}^x, \tilde{\Psi}^u]$. The resulting closed loop of the cart pole system (3.7) and controller $SL[\tilde{\Psi}^x, \tilde{\Psi}^u]$ can be put into the form of our robust stability analysis in Section 2.7:

$$x_t = \phi_{t_s}(x_{t-1}, u_{t-1}) + w_t \tag{3.18a}$$

$$\hat{w}_t = x_t + v_t - x^d(t\tau_s) - \sum_{k=2}^{t+1} R_{t,k} \hat{w}_{t+1-k}$$
(3.18b)

$$u_t = u^d(t\tau_s) + \sum_{k=1}^{t+1} M_{t,k} \hat{w}_{t+1-k} + d_t.$$
(3.18c)

Furthermore, referring to Theorem 11, it can be verified that the residual operator Δ^{ϕ} we defined earlier matches the residual operator of Theorem 11, i.e., $\Delta[\tilde{F}, \tilde{\Psi}] = \Delta^{\phi}$. Thus, Theorem 11 applies directly to our problem setting. More specifically, the local result Lem. 7 can be used to obtain robust stability guarantees. If the lumped residual terms are $(\gamma, \beta) \ell_p$ -stable with $\gamma < 1$, then the closed-loop system is f.g. ℓ_p -stable for small enough perturbations.

3.3 The Constrained LQR Problem

We use the idea of blended CLMs to derive a novel distributed synthesis procedure that outperforms any optimal linear controller for the constrained LQR problem [36, 86, 91, 140]. A significant advantage of the approach is that despite being a nonlinear synthesis method it naturally enjoys the same benefits as the linear system level approach introduced in [11], which allows for localized controller implementation, making it scalable to large networks. For the following discussion,

we assume that F is a standard linear time-invariant system, that is: F = Ax + Bu, where $A = I \otimes A$ and $B = I \otimes B$ for some fixed matrices $A \in \mathbb{R}^{n \times n}$ and $B \in \mathbb{R}^{n \times m}$.

Consider a control problem where we wish to minimize an average LQR cost, but also want that the closed loop meets certain safety guarantees against a set of rare yet possible worst-case disturbances. Ideally, we would like to synthesize a controller that can guarantee the necessary safety constraints without too much loss in performance compared to the unconstrained LQR controller. We will phrase this design goal as the following constrained LQR problem:

$$\min_{\boldsymbol{K}} \lim_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} \mathbb{E}_{w_t^i \sim p(w)} [\mathcal{J}(x_t, u_t)]$$
(3.19a)

s.t.
$$x_t = Ax_{t-1} + Bu_{t-1} + w_t$$
 (3.19b)

$$u_t = K_t(x_{t:0}) \tag{3.19c}$$

$$\forall \boldsymbol{w} : ||\boldsymbol{w}||_{\infty} \leqslant \eta_{max} : \tag{3.19d}$$

$$\sup_{k} |x_k| \leqslant x_{max}, \quad \sup_{k} |u_k| \leqslant u_{max}$$

where \mathcal{J} abbreviates the quadratic stage cost $\mathcal{J}(x, u) = x^T Q x + u P u$ with Q, P > 0. We will assume that the disturbance is stochastic but bounded such that $\|\boldsymbol{w}\|_{\infty} \leq \eta_{max}$ with known distribution which satisfies the following

Assumption 3.4. Disturbance w_t^i are i.i.d. drawn from the scalar centered distribution p(w) and uncorrelated in time t and coordinate i.

We can equivalently phrase the optimal control problem (3.19) in terms of closed loop maps as defined in chapter 2. Recalling Def. 2.6, the optimal control problem (3.19) can be described as an optimization over the set of feasible CLMs $\Psi \in \Phi_{CL}^{W \mapsto xu}[F]$, where $F: (x, u) \mapsto Ax + Bu$ and by using the characterization Theorem 8 we obtain:

$$\min_{\boldsymbol{\Psi}^{\mathbf{x}}, \boldsymbol{\Psi}^{\mathbf{u}}} \lim_{T \to \infty} \frac{1}{T} \sum_{t=1}^{T} \mathbb{E}[\mathcal{J}(\Psi_{t}^{x}(w_{t:0}), \Psi_{t}^{u}(w_{t:0}))]$$
(3.20a)

s.t.
$$\Psi_t^x(w_{t:0}) = \Psi_t^x(0, w_{t-1:0}) + w_t$$
 (3.20b)

$$\Psi_{t+1}^{x}(0, w_{t:0}) = A\Psi_{t}^{x}(w_{t:0}) + B\Psi_{t}^{u}(w_{t:0})$$

$$\forall t \mid w_{t} \mid \leq n \qquad (3.20c)$$

$$\forall t, |w_t| \leq \eta_{max} \cdot |\Psi_t(w_{t:0})| \leq x_{max} \quad (5.200)$$

$$\forall t, |w_t| \leq \eta_{max} : \quad |\Psi_t^u(w_{t:0})| \leq u_{max}. \tag{3.20d}$$

As in the linear SLS case [11], we do not need to have the controller K be a decision variable, since we can always realize the optimal solution $(\Psi^{(x)*}, \Psi^{(u)*})$ to (3.20) with a system level controller $SL[\Psi^{(x)*}, \Psi^{(u)*}]$.

Conservativeness of Linear Solutions

We will first discuss properties of solutions to our original problem (3.19), if we restrict ourselves to only LTI controllers K. Consider the equivalent problem formulation (3.20) with the CLMs (Ψ^x, Ψ^u) restricted to being linear. This poses a convex problem and, as shown in [36], it can be approximately solved by searching for FIR CLMs (Ψ^x, Ψ^u) with enough large horizon T. However, the corresponding linear optimal CLMs ($\Psi^{x,lin*}, \Psi^{u,lin*}$) come with undesirable restrictions:

- (Ψ^{x,lin*}, Ψ^{u,lin*}) impose stricter safety restrictions than the required restrictions (3.20c) and (3.20d).
- $(\Psi^{x,\text{lin}*}, \Psi^{u,\text{lin}*})$ do not depend on the disturbance distribution p(w).

To see the first point, we have the following result as a consequence of linearity:

Lemma 12. For any linear $(\Psi^{x,lin}, \Psi^{u,lin})$, the constraint (3.20c),(3.20d) is equivalent to

$$\sup_{t} |\Psi_t^{x,lin}(w_{t:0})| \leq \sup_{t} \frac{x_{max}}{\eta_{max}} |w_t|$$
(3.21a)

$$\sup_{t} |\Psi_t^{u,lin}(w_{t:0})| \leq \sup_{t} \frac{u_{max}}{\eta_{max}} |w_t|.$$
(3.21b)

Proof. Clearly, (3.21) implies (3.20c),(3.20d). The reverse implication follows from the assumed linearity of $(\Psi^{x,lin}, \Psi^{u,lin})$ and the homogeneity of the norms.

Lem. 12 shows that the linearity restriction in CLMs imposes stricter safety conditions (3.21) than (3.20c),(3.20d). To elaborate on the second point, notice that for linear CLMs ($\Psi^{x,lin}, \Psi^{u,lin}$), the objective function (3.20a) can be expressed equivalently as

$$(3.20a) = \sigma^2 \left\| \begin{array}{c} Q^{1/2} \Psi^{x} \\ P^{1/2} \Psi^{u} \end{array} \right\|_{\mathcal{H}_{2}}^{2}, \quad \sigma^2 := \mathbb{E}_{w \sim p(w)}[w^2]$$
(3.22)

where σ^2 denotes the variance of the scalar distribution p(w) and $\|.\|_{\mathcal{H}_2}$ denotes the \mathcal{H}_2 norm for linear operators. Since the objective function only gets scaled by a constant factor σ^2 for different distributions p(w), this shows that for linear CLMs, the solutions $(\Psi^{x,\lim}, \Psi^{u,\lim})$ to (3.20) are independent of the distribution p(w).

Taking the Nonlinear System Level Approach

Now we extend the search space of the problem (3.20), by allowing candidate CLMs Ψ that are blends of linear CLMs of F.

We structure the search space of blended cCLMs as follows. Let $\{\Psi_i\}_{i=1}^N$ be a collection of linear-time invariant cCLMs of finite horizon T. Correspondingly, we parametrize their component functions of in terms of the matrices $R_k^{(i)} \in \mathbb{R}^{n \times n}$ and $M_{t,k}^{(i)} \in \mathbb{R}^{n \times m}$ as follows:

$$\Psi_{i,t}^{\mathbf{x}}(z_{1:t+1}) = \sum_{k=1}^{\min\{T,t+1\}} R_k^{(i)} z_k \qquad \Psi_{i,t}^{\mathbf{u}}(z_{1:t+1}) = \sum_{k=1}^{\min\{T,t+1\}} M_k^{(i)} z_k.$$
(3.23)

We assume the saturation weighting of the N zone mentioned in Def. 2.12. For fixed $\eta_{N-1} \ge \cdots \ge \eta_1$, $P_{\eta_i} : \mathbb{R}^n \to \mathbb{R}^n$ denotes functions that parameterize the weights $G_i := I \otimes g_i$ where $g_0 = P_{\eta_0}, g_N(x) := x - P_{\eta_{N-1}}(x)$ and g_i for $i \in \{1, \ldots, N-1\}$ are defined as $g_i = P_{\eta_i} - P_{\eta_{i-1}}$. For P_{η} and we consider two specific classes functions, both representing different types of projection maps:

Definition 3.1 (Saturation Projection). Let vector $w = [w^1, ..., w^n]^T \in \mathbb{R}^n$. The saturation projection is an element-wise projection:

$$P_{\eta}(w) := \begin{bmatrix} \operatorname{sat}(w^{1}, \eta) \\ \vdots \\ \operatorname{sat}(w^{n}, \eta) \end{bmatrix}$$
(3.24)

where $\operatorname{sat}(w, \eta) = \operatorname{sign}(w) \max\{|w|, \eta\}.$

Definition 3.2 (Radial Projection). The radial projection is defined as:

$$P_{\eta}(w) := \frac{\operatorname{sat}(|w|/\eta, 1)}{|w|/\eta} w.$$
(3.25)

Unless otherwise specified, the results derived in the rest of the chapter hold for both projections.

Remark 19. For n = 1, the radial projection and the saturation projection coincide with each other. The radial and saturation projection operator act as the identity whenever $|w| \leq \eta$. Otherwise, the radial projection rescales w so that $|P_{\eta}(w)| = \eta$ whereas the saturation projection performs the elemental radial projection. This completes our setup and we arrive at the structure of our candidate CLMs as:

$$\Psi_{t}^{x}(\cdot) = \sum_{i=1}^{N} \sum_{k=1}^{\min\{T,t+1\}} R_{k}^{(i)}(P_{\eta_{i}} - P_{\eta_{i-1}})(w_{t+1-k})$$

$$\Psi_{t}^{u}(\cdot) = \sum_{i=1}^{N} \sum_{k=1}^{\min\{T,t+1\}} M_{k}^{(i)}(P_{\eta_{i}} - P_{\eta_{i-1}})(w_{t+1-k}).$$
 (3.26)

The corresponding blended SL implementation of Ψ_{Σ} takes the form:

$$u_{t} = \sum_{i=1}^{N} \sum_{k=1}^{\min\{T,t+1\}} M_{k}^{(i)} \tilde{w}_{t+1-k}^{i}$$
$$\tilde{w}_{t}^{(i)} = g_{i}(\hat{w}_{t})$$
$$\hat{w}_{t+1} = x_{t+1} - \sum_{i=1}^{N} \sum_{k=2}^{\min\{T,t+2\}} R_{k}^{(i)} \tilde{w}_{t+2-k}^{i}.$$

With regards to our optimization problem, we enforce the (linear) constraint that $\Psi^{x,i}, \Psi^{u,i}, i \in [N]$ are CLMs of the linear system of interests:

$$x_t = Ax_{t-1} + Bu_{t-1} + w_t, (3.27)$$

with $x_t \in \mathbb{R}^n$, $w_t \in \mathbb{R}^n$, $u \in \mathbb{R}^m$.

The overall nonlinear controller $SL[\Psi^{x}, \Psi^{u}]$ can be thought of as a *nonlinear blend* of the *linear* FIR controllers $SL[\Psi^{x,i}, \Psi^{u,i}]$, $i \in [N]$. Although the nonlinear operator Ψ^{x}, Ψ^{u} differs from its linear components $\Psi^{x,i}, \Psi^{u,i}$ only by the static nonlinear function $P_{\eta_{i}}(w)$, the upcoming sections will demonstrate that this simple additional nonlinearity proves surprisingly useful. In particular, η_{i} 's separate any disturbance w_{t} into N zones such that for each *i*th linear controller $SL[\Psi^{x,i}, \Psi^{u,i}]$, only the portion of w_{t} that "falls" between η_{i} and η_{i-1} is acted upon. Intuitively, one could choose different behaviors for various portions of the disturbance signal, specifying either performance or safety properties.

For ease of exposition, we focus on the two-zone case of the proposed controller $SL[\Psi^x, \Psi^u]$ although all the analysis naturally extends to the *N*-zone case.

Consider the general problem (3.20), where we now search for CLMs (Ψ^x, Ψ^u) of the form presented in (3.26) with N = 3 and the choice of $\eta_2 = \eta_{max}$, with some $\eta_1 < \eta_2$. However, since we have the assumption $\|\boldsymbol{w}\| \leq \eta_2$, $R_j^{(3)}$ and $M_j^{(3)}$ drop out of the objective and safety constraint and can be chosen as arbitrary FIR CLMs for now. In the later section we will discuss that this extra degree of freedom can be used for anti-windup control.

Restricting ourselves to this form of CLM leads to the convex problem (3.20), which is a relaxation of the general problem (3.20).

$$\min_{R^{(i)},M^{(i)}} \left\| \begin{bmatrix} Q & 0 \\ 0 & P \end{bmatrix}^{1/2} \begin{bmatrix} R^{(1)} & R^{(2)} \\ M^{(1)} & M^{(2)} \end{bmatrix} \Sigma_w^{1/2} \right\|_F^2$$
(3.28a)

s.t.
$$\eta_1 |R^{(1)}| + (\eta_2 - \eta_1) |R^{(2)}| \le x_{max}$$
 (3.28b)

$$\eta_1 |M^{(1)}| + (\eta_2 - \eta_1) |M^{(2)}| \le u_{max}$$
(3.28c)

$$R_{k+1}^{(i)} = AR_k^{(i)} + BM_k^{(i)}$$
(3.28d)

$$R_1^{(i)} = I, \quad R_T^{(i)} = 0$$

where

$$\Sigma_w = \begin{bmatrix} \alpha_1 I & \alpha_2 I \\ \alpha_2 I & \alpha_3 I \end{bmatrix}$$

with $\alpha_1 = \mathbb{E}[P_{\eta_1}(w)^2]$, $\alpha_2 = \mathbb{E}[P_{\eta_1}(w)(P_{\eta_2}(w) - P_{\eta_1}(w))]$, and $\alpha_3 = \mathbb{E}[(P_{\eta_2}(w) - P_{\eta_1}(w))^2]$, where $w \sim p(w)$ and $||w||_{\infty} \leq \eta_{max}$. Moreover $R^{(i)}$ and $M^{(i)}$ are abbreviations for the row-wise concatenation of the matrices associated with the linear CLMs $\Psi^{x,i}$, $\Psi^{u,i}$, *i.e*, $R^{(i)} = [R_T^{(i)}, R_{T-1}^{(i)}, \dots, R_1^{(i)}]$, $M^{(i)} = [M_T^{(i)}, M_{T-1}^{(i)}, \dots, M_1^{(i)}]$. Therefore, only constraints (3.28b), (3.28c) are sufficient conditions of the constraint (3.20c), (3.20d) through the multiplicativity of the norm. All other equations in the above optimization are equivalent to the original problem (3.20). Finally, solving the convex problem (3.28) gives the suboptimal nonlinear CLMs (Ψ^{*x}, Ψ^{*u}) for the system dynamics (3.19b), realized by an internally stabilizing controller SL[Ψ^{*x}, Ψ^{*u}]. The next theorem states a main result of this chapter:

Theorem 20. For all $\eta_1 \in [0, \eta_2]$, the nonlinear system level controller $SL[\Psi^{*x}, \Psi^{*u}]$ synthesized from (3.28) achieves lower optimal LQR cost for (3.19) than any linear solutions.

Proof. First, recall that restricting K to be linear in the problem (3.19) is equivalent to restricting Ψ^x and Ψ^u to be linear in the equivalent formulation (3.20). Furthermore, notice that under the restriction of linear (Ψ^x , Ψ^u), the problem (3.20) is equivalent to (3.28) with the added constraint $R^{(1)} = R^{(2)}$, $M^{(1)} = M^{(2)}$, which shows that any solution (Ψ^{*x} , Ψ^{*u}) of the problem (3.28) achieves a lower cost than a linear solution ($\Psi^{x,\text{lin*}}$, $\Psi^{u,\text{lin*}}$) of (3.20).

Remark 21. This argument extends directly to the N-blend case.

Localized Controller for Constrained LQR

Thanks to the particular form of (3.26), when the projection is chosen to be the saturation projection Def. 3.1, structural constraints of the controller, such as the sparsity and delay constraints, can be added in a convex way to the synthesis procedure described in Section 3.3. This is because imposing structural constraints on the nonlinear controller (3.26) is equivalent to imposing them on the linear CLM components of (3.26). Detailed in [11], localization of disturbance, communication, and actuation delay, as well as sparsity pattern, are all convex constraints in terms of linear CLMs in the linear System Level Synthesis framework. Specifically, all the constraints mentioned could be cast as a convex subspace S_x and S_u for linear CLMs $\Psi^{x,i}, \Psi^{u,i}, i \in [N]$. The corresponding system-level controller SL[$T^{(u)}, T^{(x)}$] can then be implemented in a localized fashion conforming to the subspace constraints on $\Psi^{x,i}, \Psi^{u,i}$. Therefore, the nonlinear controller synthesis in Section 3.3 naturally inherits all capabilities of the linear system level controllers in terms of distributed controller synthesis and implementation.

Simulation

To corroborate the results presented in the previous sections, we demonstrate the performance of a four-zone nonlinear blending controller with radial projection compared against the optimal linear controller for the constrained LQR problem of an open-loop unstable system:

$$x_{t} = \begin{bmatrix} 1 & 1 & 0 \\ 1 & 2 & 1 \\ 0 & 1 & 1 \end{bmatrix} x_{t-1} + \begin{bmatrix} 0 \\ 0 \\ 1 \end{bmatrix} u_{t-1} + w_{t}$$
(3.29)

with $u_{max} = 40$, $x_{max} = 15$, $\eta_{max} = 1$, $Q = I_3$, P = 10. The disturbances w_k are chosen to be a truncated i.i.d. Gaussian random variable with variance σ^2 . Figure 3.2 shows the optimal cost improvement of the presented nonlinear approach over the optimal linear controller for different choices of variance σ^2 . Figure 3.2 showcases that the proposed controller can exploit the knowledge of the disturbance distribution to achieve performance improvement over the linear optimal linear controller: For small σ the proposed controller gains more than 30%.

3.5 Distributed Anti-Windup Controller for Saturated Systems

In the constrained LQR problem, we considered a linear system $x_t = Ax_{t-1} + Bu_{t-1} + w_t$ and designed a controller that ensures that the closed-loop system state x_t and input u_t remained within the specified bounded sets $\mathcal{X} = \{x \mid |x| \leq x_{max}\}$ and



Figure 3.2: Performance improvement of nonlinear controller $SL[\Psi^{*x}, \Psi^{*u}]$ over optimal linear controller $SL[\Psi^{x,lin*}, \Psi^{u,lin*}]$ for different variances σ^2 of the non-truncated disturbance. The nonlinear blending controller synthesizes over 4 linear controllers w.r.t. to the projection parameters $\eta_1 = 0.05, \eta_2 = 0.1, \eta_3 = 0.2, \eta_4 = \eta_{max} = 1$

 $\mathcal{U} = \{u \mid |u| \leq u_{max}\}$, for any bounded sequence $w \in \ell^{\infty}$, $w_t \in \mathcal{W}$, $\mathcal{W} = \{w \mid |w| \leq \eta_{max}\}$. This partial problem is one of robust set invariance [96]. In this section, we discuss this problem in a broader context, where input constraints are not part of the problem specification, but rather enforced by a saturation nonlinearity sat_{\mathcal{U}} in our input. Hence, we consider our system to be nonlinear and of the form:

$$H': \quad x_t = Ax_{t-1} + B\operatorname{sat}_{\mathcal{U}}(u_{t-1}) + w_t.$$
(3.30)

If we can be sure that the disturbances of the system and the initial condition remain within the specified bounds W, then solving the robust invariance problem, for example, with one of the techniques from the previous chapter and the general SLS framework [36], is sufficient. However, in practice, it is more realistic to consider that our assumptions are only *mostly* true and it is possible that our assumptions are temporarily violated on rare occasions. In such a scenario, we have to accept some degradation in our guarantees, such as state constraints and performance bounds. However, we would like this degradation to happen gracefully and at least ensure that some basic properties of the closed-loop such as stability are still preserved.

In this section, we discuss how control design through blended CLMs provides a new perspective on this topic. In particular, blending with *N*-zone saturations naturally ensures graceful degradation in the presence of saturation and offers an elegant solution to the issue of controller wind-up [70, 78, 81], an important problem that commonly arises when input saturation is ignored during linear control design.

The main results of this section show that with the appropriate choice of blending and weight operators, closed-loop stability and convergence to the target set \mathcal{X} are guaranteed even in the saturated regime, provided that disturbance w violates our assumption "occasionally"; mathematically, we phrase this as $w = w_1 + w_2$, where $||w_1||_{\infty} \leq \eta_{max}$ and $w_2 \in \ell_p$.

Setup of Candidate CLMs for Unsaturated Regime

First, we consider the requirements of the unsaturated regime and assume that it is a feasible control problem. That is, for the given tuple of sets $(\mathcal{X}, \mathcal{U}, \mathcal{W})$, we can design a linear controller such that for any disturbance sequence w in \mathcal{W} , (that is, $w_t \in \mathcal{W}$ for all t), the state x_t is always guaranteed to stay within the set \mathcal{X} if $x_0 \in \mathcal{X}$. As shown in the previous section, we can pose this as a feasibility problem of convex constraints. Another general approach to address this problem in the context of SLS has been presented in [36], which allows polytopic sets \mathcal{X}, \mathcal{U} , and \mathcal{W} . We assume that the control design for the unsaturated regime is feasible and we formulate this in terms of cCLMs Ψ next.

We define the operators $F \in \mathcal{LC}_s(\ell^n \times \ell^m, \ell^n)$ and $F' \in \mathcal{C}_s(\ell^n \times \ell^m, \ell^n)$ to distinguish the dynamics of the linear system with and without actuator saturation. Let F(x, u) =A(x) + B(u), where $A = I \otimes A$ and $B = I \otimes B$. Let $F'(x, u) = A(x) + B_{\mathcal{U}}(u)$, where $B_{\mathcal{U}} := B \circ (I \otimes \operatorname{sat}_{\mathcal{U}})$. In general, a saturation nonlinearity sat is characterized by the properties described in (3.3), that is, as long as the input is in the set \mathcal{U} , the nonlinearity has no effect, and the system behaves linearly. However, we will focus our discussion on projection-based saturations $\Pi_{\mathcal{U}}$ as shown in the example below.

Definition 3.3. Given some closed bounded convex set \mathcal{U} with $0 \in \mathcal{U}$, a saturation function is a Lipshitz continuous map $\operatorname{sat}_{\mathcal{U}} : \mathbb{R}^m \to \mathcal{U}$ onto \mathcal{U} , which satisfies

 $\forall u \in \mathbb{R}^m : \operatorname{sat}_{\mathcal{U}}(\operatorname{sat}_{\mathcal{U}}(u)) = \operatorname{sat}_{\mathcal{U}}(u) \text{ and } \forall u \in \mathcal{U} : \operatorname{sat}_{\mathcal{U}}(u) = u.$

Example. Let \mathcal{U} be some convex body in a normed vector space $(||, \mathbb{R}^m)$ containing the origin in its interior. Then, the map $\Pi_{\mathcal{U}} : u \mapsto \underset{u' \in \mathcal{U}}{\arg \min} |u - u'|$ is a saturation.

For our blended cCLM $\Psi_{\Sigma} = \sum_{i=1}^{N} \Psi_{\Sigma} G_i$ we consider a collection of linear cCLM $\{\Psi_i\}_{i=1}^{N}$ and weight operators $\{G_i\}_{i=1}^{N}$ constructed from saturation functions as described in Def. 2.12.

Let $\mathcal{W}_1 \subset \mathcal{W}_2 \cdots \subset \mathcal{W}_{N-1}$ be a collection of nested convex sets and pick $\mathcal{W}_{N-1} := \mathcal{W}$. Denote $\Pi_{\mathcal{W}_k} : x \mapsto \underset{u \in \mathcal{W}_k}{\operatorname{arg min}} |x - u|, \forall k \in \{1, \ldots, N-1\}$ as the projection maps onto the sets \mathcal{W}_k . Define the weights as $G_k = I \otimes g_k$, where $g_k = \Pi_{\mathcal{W}_k} - \Pi_{\mathcal{W}_{k-1}}$ for all $k \in \{2, \ldots, N-1\}$ and $g_0 = \Pi_{\mathcal{W}_1}, g_N : x \mapsto x - \Pi_{\mathcal{W}_{N-1}}(x)$.

We purposely picked the last weight operator such that $G_N = I \otimes (I - \Pi_W)$, hence in the unsaturated regime, the last cCLM Ψ_N does not contribute to the closed-loop state and input behavior. This is due to the fact that for any sequence $w \subset W$, it holds that

$$oldsymbol{w} \subset \mathcal{W}: \quad \Longrightarrow \ oldsymbol{w} = \sum_{i=1}^N oldsymbol{G}_i(oldsymbol{w}) = \sum_{i=1}^{N-1} oldsymbol{G}_i(oldsymbol{w})$$

and therefore

$$oldsymbol{w} \subset \mathcal{W}: \quad \Longrightarrow \ oldsymbol{\Psi}_\Sigma oldsymbol{w} = \sum_{i=1}^{N-1} oldsymbol{\Psi}_i oldsymbol{G}_i(oldsymbol{w}).$$

We assume that $\{\Psi_i\}_{i=1}^{N-1}$ have been designed to satisfy the desired notion of robust set invariance in the unsaturated regime:

Assumption 3.6. $\{\Psi_i\}_{i=1}^{N-1}$ are CLMs of the linear system F, and for all $w \in W$, the following holds:

$$\sum_{i=1}^{N-1} \boldsymbol{\Psi}_{i}^{\mathbf{x}} \boldsymbol{G}_{i}(\boldsymbol{w}) \in \mathcal{X} \qquad \qquad \sum_{i=1}^{N-1} \boldsymbol{\Psi}_{i}^{\mathbf{u}} \boldsymbol{G}_{i}(\boldsymbol{w}) \in \mathcal{U}. \qquad (3.31)$$

For notational convenience, we split the blended CLM into two parts. Let $\begin{bmatrix} R \\ M \end{bmatrix}$ stand for

$$oldsymbol{R} := \sum_{i=1}^{N-1} oldsymbol{\Psi}_i^{\mathrm{x}} oldsymbol{G}_i \qquad \qquad oldsymbol{M} := \sum_{i=1}^{N-1} oldsymbol{\Psi}_i^{\mathrm{u}} oldsymbol{G}_i$$

so we can write:

$$oldsymbol{\Psi}_{\Sigma}^x = oldsymbol{R} + oldsymbol{\Psi}_N^{\mathrm{x}} oldsymbol{G}_N \qquad \qquad oldsymbol{\Psi}_{\Sigma}^u = oldsymbol{M} + oldsymbol{\Psi}_N^{\mathrm{u}} oldsymbol{G}_N.$$

It is clear from this setup that for sequences $w \in \mathcal{W}$, the map $\begin{bmatrix} R \\ M \end{bmatrix}$ is a CLM for the linear system F, and *if* we restricted the domain of the CLMs to sequences in \mathcal{W} , it would also be a CLM of the nonlinear system F'.

Design Approach for Stability in the Saturated Regime

We now consider the case where the disturbance leaves the set W occasionally, leading to actuator saturation. It is commonly known that graceful performance degradation cannot be taken for granted, as instability phenomena like the "wind-up" effect can occur if the controller synthesis improperly deals with actuator saturation.

For stable system matrices A, a modification based on the IMC principle of the controller is shown in [36], which guarantees closed-loop stability when $w \notin W$.

In the case of our blended SL controller $\Psi_{\Sigma}^{u}(\Psi_{\Sigma}^{x})^{-1}$, it turns out that stability in the saturated regime is ensured if we make two simple modifications. First, we require $\Psi_{N}^{u} = 0$, and Ψ_{N}^{x} to be an LTI FIR cCLM with the linear components

$$\Psi_{N,t}^{\mathbf{x}}(w_{t:0}) = \sum_{k=0}^{\min\{T-1,t\}} A^k w_{t-k} \qquad \Psi_{N,t}^{\mathbf{u}} = 0, \qquad (3.32)$$

for some finite horizon $T \ge 1$. Second, we prepend the projection to $\Pi_{\mathcal{W}}$ to \mathbf{R} and \mathbf{M} .

Next, we show that with the above rule, ℓ_p stability of the nonlinear closed loop is guaranteed. Moreover, global stability results (for stable A) and local stability results (for unstable A), along with the corresponding transient bounds, can be derived for the closed loop. The convergence to \mathcal{X} in finite time is shown for ℓ_p perturbations with $p < \infty$.

Dynamics of Lumped Disturbances in the Saturated Regime

The basis for our stability analysis is the provided by the following lemma:

Lemma 13. Given assumption (3.6) and assuming Ψ_N satisfies (3.32), then for $\Psi'_{\Sigma} = \begin{bmatrix} \mathbf{R} \\ M \end{bmatrix} \Pi_{W} + \Psi_N \mathbf{G}_N$ holds:

$$oldsymbol{\Delta}[oldsymbol{F}', oldsymbol{\Psi}_{\Sigma}'] = oldsymbol{\Delta}[oldsymbol{F}, oldsymbol{\Psi}_N]oldsymbol{G}_N.$$

Proof.

$$egin{aligned} oldsymbol{\Delta} egin{aligned} oldsymbol{\Delta} egin{aligned} oldsymbol{\Delta} egin{aligned} oldsymbol{F}', oldsymbol{\Psi}_{\Sigma} egin{aligned} &= oldsymbol{A}^+ oldsymbol{R} \Pi_{\mathcal{W}} + oldsymbol{A}^+ oldsymbol{\Psi}_N^{ ext{x}} oldsymbol{G}_N + oldsymbol{B}_{\mathcal{U}}^+ oldsymbol{(M} \Pi_{\mathcal{W}} + oldsymbol{\Psi}_N^{ ext{u}} oldsymbol{G}_N) \ &\cdots + oldsymbol{I} - oldsymbol{R} \Pi_{\mathcal{W}} - oldsymbol{\Psi}_N^{ ext{x}} oldsymbol{G}_N. \end{aligned}$$

Now since, $Mw \in \mathcal{U}$ for all $w \in \mathcal{W}$ and $\Pi_{\mathcal{W}}w \in \mathcal{W}, \forall w$, we can conclude $M\Pi_{\mathcal{W}}w \in \mathcal{U}$ for all w. Furthermore, since $\Psi_N^u = 0$, we have $\Psi_N^u G_N = 0$. It is therefore trivial to rewrite the term $B_{\mathcal{U}}^+(M\Pi_{\mathcal{W}} + \Psi_N^u G_N)$ as:

$$\boldsymbol{B}^+_{\boldsymbol{\mathcal{U}}}(\boldsymbol{M}\Pi_{\boldsymbol{\mathcal{W}}}+\boldsymbol{\Psi}^{\mathrm{u}}_N\boldsymbol{G}_N)=\boldsymbol{B}^+_{\boldsymbol{\mathcal{U}}}(\boldsymbol{M}\Pi_{\boldsymbol{\mathcal{W}}})=\boldsymbol{B}^+(\boldsymbol{M}\Pi_{\boldsymbol{\mathcal{W}}})+\boldsymbol{B}^+\boldsymbol{\Psi}^{\mathrm{u}}_N\boldsymbol{G}_N.$$

Moreover, using the fact that $I = G_N + \prod_W$ we substitute in the above equation and yield the decomposition:

$$oldsymbol{\Delta}[oldsymbol{F}', oldsymbol{\Psi}_{\Sigma}'] = (oldsymbol{A}^+oldsymbol{R} + oldsymbol{B}^+oldsymbol{M} - oldsymbol{R} + oldsymbol{I}) \Pi_{\mathcal{W}} + oldsymbol{\Delta}[oldsymbol{F}, oldsymbol{\Psi}_N]oldsymbol{G}_N.$$

Since, the first term equals $\Delta[F, \begin{bmatrix} R \\ M \end{bmatrix}] \Pi_{\mathcal{W}}$ and the $\begin{bmatrix} R \\ M \end{bmatrix}$ are CLMs of the linear system F we have $\Delta[F, \begin{bmatrix} R \\ M \end{bmatrix}] \Pi_{\mathcal{W}} = 0$. This leaves us with the desired result: $\Delta[F', \Psi'_{\Sigma}] = \Delta[F, \Psi_N] G_N$.

Applying Theorem 11 from the previous chapter, we obtain the dynamics of the lumped disturbance as:

$$\hat{oldsymbol{w}} = oldsymbol{\Delta} [oldsymbol{F}, oldsymbol{\Psi}_N] oldsymbol{G}_N \hat{oldsymbol{w}} + oldsymbol{w}$$

We analyze the stability of these dynamics by rewriting them as the difference equation:

$$\hat{w}_t = \begin{cases} A^T (\hat{w}_{t-T} - \Pi_{\mathcal{W}}(\hat{w}_{t-T})) + w_t & \text{for } t \ge T \\ w_t & \text{else} . \end{cases}$$
(3.33)

Here is a quick derivation for clarity: $\Psi_N^u = 0$, means $\Delta[F, \Psi_N] = A^+ \Psi_N^x + I - \Psi_N^x$. Define the auxiliary variables $\tilde{w}^N = G_N \hat{w}$, $\alpha = \Psi_N^x \tilde{w}^N \beta = A^+ \Psi_N^x \tilde{w}^N$, $\gamma = (I - \Psi_N^x) \tilde{w}^N$ and following calculations to compute $\beta + \gamma$

$$\begin{split} \hat{w}_{t} &= \beta_{t} + \gamma_{t} + w_{t} \\ \alpha_{t} &= \sum_{k=0}^{\min\{T-1,t\}} A^{k} \tilde{w}_{t-k}^{N} \\ \beta_{t} &= A \alpha_{t-1} \\ \gamma_{t} &= \sum_{k=1}^{\min\{T-1,t\}} -A^{k} \tilde{w}_{t-k}^{N} \end{split} \Leftrightarrow \qquad \begin{aligned} \alpha_{t} &= \sum_{k=1}^{\min\{T,t+1\}} A^{k-1} \tilde{w}_{t+1-k}^{N} \\ \beta_{t} &= \sum_{k=1}^{\min\{T,t\}} A^{k} \tilde{w}_{t-k}^{N} \\ \beta_{t} &= \sum_{k=1}^{\min\{T,t\}} A^{k} \tilde{w}_{t-k}^{N} \\ \beta_{t} + \gamma_{t} &= A^{T} \tilde{w}_{t-T}^{N}, \text{ if } t \ge T, \text{ else } 0 \\ \hat{w}_{t} &= A^{T} \tilde{w}_{t-T}^{N} + w_{t}, \text{ if } t \ge T, \text{ else } w_{t} \end{aligned}$$

Finally, substituting $\tilde{w}_t^N = \hat{w}_t - \Pi_W(\hat{w}_t)$ into the last expression on the right yields the difference equation (3.33).

Stability and Convergence in Saturated Regime

Next, we analyze the stability of the dynamic system (3.33) and its implications for the overall closed-loop. We show that we can always choose T to achieve closed-loop stability. The choice of T depends on the matrix A and the norm used in projection Π_{W} .

From equation (3.33), it is easy to see that we can decompose the sequence \hat{w} and w into T subsequences $\hat{\omega}_k^{[i]} := \hat{w}_{i+kT}, i \in \{0, \ldots, T-1\}, \omega_k^{[i]} := w_{i+kT}$, whose dynamics can be analyzed entirely separately. In fact, for each i and k holds

$$\hat{\omega}_{k}^{[i]} = A^{T}(\hat{\omega}_{k-1}^{[i]} - \Pi_{\mathcal{W}}(\hat{\omega}_{k-1}^{[i]})) + \omega_{k}^{[i]}, \text{ for all } t \ge 1 \text{ and } \hat{\omega}_{0}^{[i]} = \omega_{0}^{[i]}.$$
(3.34)

Notice that f.g. ℓ_p/ℓ_{∞} -stability of the above system is sufficient for fg. ℓ_p/ℓ_{∞} -stability of the original system (3.33), since:

$$\|\hat{\boldsymbol{\omega}}^{[i]}\|_{p} \leq \gamma \|\boldsymbol{\omega}^{[i]}\|_{p} \quad \text{for all } i \in \{0, \dots, T-1\}$$
$$\implies \|\hat{\boldsymbol{w}}\|_{p} = \sqrt[p]{\sum_{i=0}^{T-1} \|\hat{\boldsymbol{\omega}}^{[i]}\|_{p}^{p}} \leq \sqrt[p]{\sum_{i=0}^{T-1} \gamma^{p} \|\hat{\boldsymbol{\omega}}^{[i]}\|_{p}^{p}} = \gamma \sqrt[p]{\sum_{i=0}^{T-1} \|\hat{\boldsymbol{\omega}}^{[i]}\|_{p}^{p}} = \gamma \|\boldsymbol{w}\|_{p}.$$

Hence, we drop the index i and analyze the auxiliary system

$$\hat{\omega}_{k+1} = A^T (\hat{\omega}_k - \Pi_{\mathcal{W}}(\hat{\omega}_k)) + \omega_k.$$

For the next lemma, define $\mathcal{B}_{\eta} := \{w | |w| < \eta\}$ as the ball of radius η corresponding to the norm $|\cdot|$ used in the definition of the projection $\Pi_{\mathcal{W}}$. The next result proves the conditions for closed-loop stability dependent on the design parameter T.

Lemma 14. Define $\bar{\eta} := \sup\{\eta | \mathcal{B}_{\eta} \subset \mathcal{W}\}$ and define $|A| := \sup_{|x|=1} |Ax|$. If $\bar{\eta} > 0$ and assuming ℓ_p , ℓ_{∞} are formulated also w.r.t. the norm $|\cdot|$, then:

1. If $|A^T| < 1$, then the following bound holds for all w:

$$\|\hat{\boldsymbol{w}}\|_p \leqslant rac{1}{1-|A^T|} \|\boldsymbol{w}\|_p$$

2. For any $0 \leq \gamma < \min\{1, |A^T|\}$ holds:

$$\|\boldsymbol{w}\|_{p} \leq (1-\gamma) \frac{|A^{T}|\bar{\eta}}{|A^{T}|-\gamma} \implies \|\hat{\boldsymbol{w}}\|_{p} \leq \frac{1}{1-\gamma} \|\boldsymbol{w}\|_{p}.$$

Proof. First observe the following property of the projection map $\Pi_{\mathcal{W}}$:

$$\begin{aligned} |\Pi_{\mathcal{W}}(w) - w| &= \min |w'| &\leq \min |w'| \\ \text{s.t. } w + w' \in \mathcal{W} &\text{s.t. } |w + w'| < \bar{\eta} \\ &\stackrel{a)}{\leq} \min_{t \ge 0} t|w| &= \max\{0, |w| - \bar{\eta}\} \\ &\text{s.t. } (1 - t)|w| < \bar{\eta} \end{aligned}$$

where inequality a) follows by restricting the search to the line w' = tw, $t \ge 0$. This shows that $|\Pi_{\mathcal{W}}(w) - w| < |w|$ for all $w \in \mathbb{R}^n$. Hence, it follows that for any ω holds:

$$|A^T(\hat{\omega} - \Pi_{\mathcal{W}}(\hat{\omega}))| \leq |A^T| \max\{0, |\omega| - \bar{\eta}\} \leq \max\{0, |A^T||\omega| - |A^T|\bar{\eta}\}.$$

This proves the first statement. On the other hand, for any $\gamma < \min\{1, |A^T|\}$, the following local small-gain property holds:

$$\|\boldsymbol{a}\|_{p} < \frac{|A^{T}|\bar{\eta}}{|A^{T}| - \gamma} \implies \|(\boldsymbol{I} \otimes A^{T}(\mathrm{id} - \Pi_{\mathcal{W}}))\boldsymbol{a}\|_{p} \leqslant \gamma \|\boldsymbol{a}\|_{p}$$
(3.35)

where $id : x \mapsto x$. Lastly, the desired local and global results follow by direct application of the small gain result lemma 7.

Lemma 14 formulates stability conditions in terms of T and a requirement of $|A^T| < 1$ to ensure global stability. Due to Gelfand's theorem, a standard result from Linear Algebra, we are guaranteed that as long as A is Schur, it is always possible to pick a T large enough such that $|A^T| < 1$, however T might depend on the particular norm $|\cdot|$ we chose in the projection Π_W .

Lemma 15 (Gelfand's Theorem). Denote by $\rho(A)$ the spectral radius (max absolute value of eigenvalue) of $A \in \mathbb{R}^{n \times n}$, then for any matrix norm |.| holds $\lim_{k\to\infty} (|A^k|)^{1/k} = \rho(A)$.

The lemma 14 tells us that if A is Schur, we can always choose T such that $|A^T| < 1$, then we are guaranteed f.g. ℓ_p stability for the lumped disturbances \hat{w} and by Theorem 11, this shows overall closed-loop stability for our system since F is trivially Lipschitz and therefore i.f.g. ℓ_p stable everywhere. Furthermore, even in the case where $|A^T| > 1$, the second result offers at least local ℓ_p -stability for the closed-loop. Hence, we have stability for small disturbances in the saturated regime, even if the system is open-loop unstable.

As a corollary of the above result, we can prove that as long as our disturbance only exits the set \mathcal{W} a finite number of times, then we are also guaranteed to violate the state constraint \mathcal{X} a finite number of times. Our strategy is to show that there exists a time t' for which x_t is guaranteed to stay in \mathcal{X} for all time t > t' if w is composed of $w_1 + w_2$ where $||w_1||_{\infty} < \overline{\eta}$ and $w_2 \in \ell_p$.

Corollary 16. Assume $\bar{\eta} > 0$ and $|A^T| \leq 1$ as used in Lemma 14. If $\boldsymbol{w} = \boldsymbol{w}_1 + \boldsymbol{w}_2$ where $\|\boldsymbol{w}_1\|_{\infty} \leq \bar{\eta} - \varepsilon$ for some $\varepsilon > 0$ and $\boldsymbol{w}_2 \in \ell_p$, then there exists a time t' such that for all t > t': $x_t \in \mathcal{X}$. *Proof.* Recall the relationship $\hat{w} = (\Psi^x)^{-1} x \Leftrightarrow \Psi^x \hat{w} = x$ and $\Psi_{\Sigma}^x = R(I - G_N) + \Psi_N^x(G_N)$ to decompose x into the terms s and s':

$$\boldsymbol{x} = \underbrace{\boldsymbol{R}}_{\mathcal{W}}(\hat{\boldsymbol{w}}) + \underbrace{\boldsymbol{\Psi}_{N}^{\mathrm{x}}(\hat{\boldsymbol{w}} - \boldsymbol{\Pi}_{\mathcal{W}}(\hat{\boldsymbol{w}}))}_{=:\boldsymbol{s}'}.$$
(3.36)

Since $\mathbf{R}\mathbf{w} \in \mathcal{X}$ for all $\mathbf{w} \in \mathcal{W}$ and $\Pi_{\mathcal{W}}$ maps onto \mathcal{W} , it is clear that $\mathbf{s} = \mathbf{R}\Pi_{\mathcal{W}}(\hat{w}) \in \mathcal{X}$ for all \hat{w} . Denote $(s)_+ := \max\{0, s\}$ and retrace the proof of Lem. 14, to arrive at the inequality:

$$|\hat{w}_t| \le (|\hat{w}_{t-1}| - \bar{\eta})_+ + |w_t|$$

From the above, we obtain the following inequalities

$$\begin{split} |\hat{w}_{t}| &\leqslant (|\hat{w}_{t-1}| - \bar{\eta})_{+} + |w_{t}| \\ \Leftrightarrow & |\hat{w}_{t}| - \bar{\eta} \leqslant (|\hat{w}_{t-1}| - \bar{\eta})_{+} + |w_{t}| - \bar{\eta} \\ \Leftrightarrow & \leqslant (|\hat{w}_{t-1}| - \bar{\eta})_{+} + |w_{2,t}| - \varepsilon \\ \Rightarrow & \leqslant (|\hat{w}_{t-1}| - \bar{\eta})_{+} + (|w_{t}| - \bar{\eta})_{+} \\ \Rightarrow & (|\hat{w}_{t}| - \bar{\eta})_{+} \leqslant (|\hat{w}_{t-1}| - \bar{\eta})_{+} + (|w_{t}| - \bar{\eta})_{+} \\ \Rightarrow & \forall \tau_{f} : \sum_{t=0}^{\tau_{f}} (|\hat{w}_{t}| - \bar{\eta})_{+} - (|\hat{w}_{t-1}| - \bar{\eta})_{+} \leqslant \sum_{k=0}^{\tau_{f}} (|w_{t}| - \bar{\eta})_{+} \\ \Leftrightarrow & \forall \tau_{f} : (\hat{w}_{\tau_{f}} - \bar{\eta})_{+} \leqslant \sum_{k=0}^{\tau_{f}} (|w_{t}| - \bar{\eta})_{+} \leqslant \sum_{k=0}^{\tau_{f}} (|w_{t}| - \bar{\eta})_{+} \\ \end{split}$$

Recall $w = w_2 + w_1$, where $w_2 \in \ell_p$ and $|w_1|_{\infty} < \bar{\eta}$. The last line shows $\|\hat{w}\|_{\infty} < \infty$, since there exists some timestep T_f such that $\forall t > T_f$, $|w_t| - \bar{\eta} \leq -\frac{\varepsilon}{2}$. Furthermore, for all $k \geq T_f$ holds:

$$|\hat{w}_k| - \bar{\eta} \leq (|\hat{w}_{k-1}| - \bar{\eta})_+ - \frac{\varepsilon}{2}.$$

The shows that after some finite time T_2 , $|\hat{w}_t| < \bar{\eta}$ and therefore $\hat{w}_t - \Pi_{\mathcal{W}}(\hat{w}_t) = 0$ for all $t \ge T_2$. Finally, since cCLM Ψ_N^x is FIR, it also implies s'(t) = 0 after at most T time steps afterward. This proves that eventually, that is, after some t' > 0, $x_t \in \mathcal{X}$ for all $t \ge t'$.

3.7 Example

As an example, we revisit the structure of the optimal blended SL controller of the constrained LQR problem and augment it with the anti-windup technique discussed

in the previous section. In the control design procedure, we discussed that the last CLM Ψ_N (in the small example, it was Ψ_3) was a degree of freedom that does not affect the LQR cost. However, in light of our previous discussion, we can use that extra degree of freedom to ensure graceful degradation in the event that the disturbance violates our assumptions.

As discussed in our previous results, we pick $\Psi_N^u = 0$ and pick Ψ_N^x according to the rule (3.32). The blended cCLM Ψ_{Σ} changes only the component Ψ_{Σ}^x :

$$\Psi_{t}^{\mathbf{x},\mathbf{a}}(w_{t:0}) = \sum_{i=1}^{N} \left(\sum_{k=1}^{\min\{T,t+1\}} R_{k}^{(i)} (P_{\eta_{i}} - P_{\eta_{i-1}})(w_{t+1-k}) \right) + \sum_{k=1}^{\tau+1} A^{k-1} \left(w_{t+1-k} - P_{\eta_{N}}(w_{t-k+1}) \right).$$
(3.37)

In the above, τ is a design parameter, and recall that by design, we have chosen $\eta_N = \eta_{max}$, the expected norm bound on disturbances. This extra term added to the CLM accounts for residual disturbances that are not attenuated by the original controller $SL[\Psi^x, \Psi^u]$ because the disturbances are larger than expected by projection mapping, that is, $|w_t| > \eta_{max}$. Therefore, $SL[\Psi^{x,a}, \Psi^u]$ considers the τ -step propagation of the unaccounted for disturbances from $SL[\Psi^x, \Psi^u]$.

The resulting lumped dynamics under the augmented controller $SL[\Psi^{x,a}, \Psi^{u}]$ take the form:

$$\hat{w}_t = A^{\tau+1} (\hat{w}_{t-\tau} - P_{\eta_{max}}(\hat{w}_{t-\tau})) + w_t$$
(3.38)

where $P_{\eta_{max}}()$ is the entry-wise saturation used in the constrained LQR problem. $P_{\eta_{max}}()$ can be equivalently viewed as an instance of a projection $\Pi_{\mathcal{W}}$ with respect to the ∞ -norm $|\cdot|_{\infty}$, and where \mathcal{W} denotes the scaled ∞ -norm ball $\eta_{max}\{x \mid ||x||_{\infty} \leq 1\}$. Leveraging the stability theorem Lem. 14 and using Gelfand's Lemma, closed-loop stability for the saturated regime is ensured under the following condition stated in the lemma below:

Lemma 17. Assume $\rho(A) < 1$ and pick τ such that $|A^{\tau+1}|_{\infty} < 1$. Then the internal dynamics (3.38) are globally finite-gain ℓ_{∞} -stable, where for all $w \in \ell_{\infty}^{n}$,

$$\|\hat{\boldsymbol{w}}\|_{\infty} \leq \frac{1}{1 - |A^{\tau+1}|_{\infty}} \|\boldsymbol{w}\|_{\infty}$$

Proof. Pick $|\cdot| := |\cdot|_{\infty}$ and choose τ such that $|A^{\tau+1}|_{\infty} < 1$. Then apply the lemma (14).

Therefore, if the system is open-loop stable, $SL[\Psi^{x,a}, \Psi^{u}]$ guarantees graceful degradation when the closed-loop is saturated. Similarly, we obtain local stability results in the case where *A* is open-loop unstable. Similarly to the large-scale constrained LQR case in Section 3.3, the augmentation for anti-windup allows for distributed and localized implementation.

Simulation

We end this chapter with some simulation experiments demonstrating the effectiveness of anti-windup augmentation. We consider a bi-directional chain system with the *i*th node's dynamics being

$$x_{t+1}^{i} = (1 - 0.4|\mathcal{N}(i)|)x_{t}^{i} + 0.4\sum_{j\in\mathcal{N}_{i}} x_{t}^{j} + \operatorname{sat}(u_{t}^{i}, u_{max}) + w_{t}^{i}$$

where $\mathcal{N}(i)$ denotes the set of vertices that has an edge connected to *i*th vertex and w_t^i is the *i*th coordinate of disturbance vector at time *t*. In particular, $\|\boldsymbol{w}\|_{\infty} \leq 1$ and $x_0 = 0$. One can check that the overall chain system is open-loop marginally (un)stable. In this chain example, we allow 1 time step communication delay between nodes and actuation delay with 50%.

We illustrate the anti-windup property of the nonlinear controller (3.26) in the decentralized setting with additional sparsity, locality, and delay constraints in Figure 3.3c. First, a nominal integral controller for this system is designed and called the *Integral Controller*. Due to its integral structure, the *Integral Controller* for the unconstrained closed loop guarantees convergence of the state to the origin under persistent disturbance, i.e., step rejection. In comparison, a second linear controller synthesized from the standard constrained LQR problem is generated that guarantees stability for all admissible w under saturation. We refer to this linear controller as the *non-integral controller* since the states only stay bounded under persistent admissible disturbance.

The nonlinear controller with the saturation projection here is chosen to be a two-zone blending controller. The simulation shows the anti-windup property as well as the preservation of step rejection in both large- and small-disturbance schemes of the proposed method. Figure 3.3c shows that the blended SL-controller stabilizes the system while the integral controller becomes unstable under worst-case bounded disturbance. On the other hand, in Figure 3.4c, the proposed blending controller preserves the performance of step rejection while the linear Non-integral Controllers


Figure 3.3: *Worst-case Response*: The heatmaps show how a worst-case disturbance is propagated through space-time for the saturated chain system. The integral controller becomes unstable due to saturation and the naive blending controller possesses has the anti-windup property of the non-integral controller.



Figure 3.4: Step disturbance rejection: Response to small step disturbances at node 8,10,12 entering at time 2,6,10, respectively. As in the scalar case, the proposed blending controller not only stabilizes under saturation but also recovers the performance objective of rejecting small step disturbances. This contrasts against the non-integral controller, which sacrifices small-signal performance for stability.

forfeit the performance objective in order to preserve stability in the saturated closed loop. For more details on the simulation, see [137].

3.8 Conclusion

We showcase the nonlinear system-level approach developed in [64] and illustrate the use cases for a class of nonlinear system-level controllers. We propose a tractable nonlinear control synthesis method that outperforms any optimal linear controller for constrained LQR problems. It is further shown that such a controller naturally possesses an anti-windup property for linear systems with input saturation. A key highlight is that the presented approach is compatible with locality/delay constraints and distributed implementation, similar to the linear system-level approach [11]. Overall, this chapter is a first step in exploring the full potential of the new nonlinear control synthesis framework developed in [64] and highlights that even just the presented special case of the framework, called "nonlinear blending" of linear controllers, offers many benefits.

Chapter 4

ROBUSTNESS AND SENSITIVITY OF CERTAINTY-EQUIVALENT ADAPTATION

This chapter investigates closed-loop in learning-to-control problem settings, where the decision-making of the algorithm is structured in a certainty equivalent way. Certainty Equivalence (CE) is a common principle underlying many learning and control algorithms: At each time step, we hypothesize a model for the system dynamics and then act according to the *nominal* control law designed for that model. In other words, we pretend to have found a model that will remain accurate going forward. We show that there is a natural way to perform certainty-equivalent adaptation for SL controllers, which simplifies the analysis and is instructive for the design of the overall learning-to-control algorithm. The particular structure of the nonlinear closed-loop maps reveals three general design principles sufficient for designing stable learning-to-control algorithms:

- 1. **Smooth parametrizations of nominal CLMs and Dynamics**: The nominal CLMs and dynamics chosen by CE-adaptation should change smoothly with small changes in the nominal model.
- 2. Consistency of Selected Models: The sequence of models that we hypothesize should be consistent with the online data (up to some ℓ_p -bounded error).
- 3. Efficient Model Selection: Our model hypothesis should eventually converge; however, it is <u>not</u> required to converge to the true model.

In the second part of the chapter, we explore applications of this result for problem settings related to online learning of optimal controllers. To this end, we focus on the setting of linear time-invariant systems and linear-quadratic costs, a problem setting that has received immense recent attention in the learning and control literature. Guided by the theoretical findings in the first part, we follow the principle of certainty equivalence to design a learning-to-control scheme with nominal LQ-optimal system-level controllers. We analyze the closed-loop stability of the learning-to-control scheme and provide conditions for model selections that are sufficient for closed-loop stability, which are closely related to consistent model chasing, a core topic in Part 2 of the thesis. The main technical result underlying this analysis is on perturbation analysis of LQ-optimal CLMs. The result in itself is new and characterizes the

sensitivity (i.e., analytic bounds on the Lipschitz constant) of LQ-optimal closed-loop maps for LTI systems in terms of system-theoretic properties such as controllability and observability.

4.1 Closed-Loop Dynamics in the Presence of CE-Adaptation

We assume a nonlinear system in the standard form we introduced in Chapter 2 and that the disturbance w belongs to $\mathcal{W}^{\mathbb{N}} := \{w \in \ell^{\mathcal{X}} \mid \forall k \in \mathbb{N} : w_k \in \mathcal{W}\}$ for some closed-bounded set $\mathcal{W} \subset \mathbb{R}^n$.

$$oldsymbol{x} = oldsymbol{F}^+(oldsymbol{x},oldsymbol{u}) + oldsymbol{w}, \quad oldsymbol{w} \in \mathcal{W}^{\mathbb{N}}.$$

We restrict ourselves to diagonal operators F, i.e., F can be decomposed as:

$$\boldsymbol{F} = \sum_{t=0}^{\infty} \boldsymbol{P}^{[t]} \boldsymbol{F} \boldsymbol{P}^{[t]}$$
(4.1)

and let $\{f_k : \mathbb{R}^{n+m} \mapsto \mathbb{R}^n\}_{k=0}^{\infty}$ be the unique functions formed by dropping the obsolete arguments of the component functions; alternatively, each f_{τ} is well-defined by the mapping $f_{\tau} : (x_{\tau}, u_{\tau}) \mapsto F_{\tau}(x_{\tau:0}, u_{\tau:0})$. Moreover, we assume that the dynamics F are unknown, however, belong to a known set of dynamics $\{\hat{F}(\omega) \mid \omega \in \Omega\} =: \hat{F}[\Omega]$ parametrized over some compact metric space (Ω, d_{Ω}) with some fixed map $\hat{F} : \Omega \to C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$. Correspondingly with our assumption of F, we assume that $\mathbb{D}[\Omega]$ consists of diagonal operators and there exists some $\omega^* \in \Omega$ such that $\mathbb{D}(\omega^*) = F$. We summarize these assumptions and definitions below: Moreover, we assume that the dynamics $\{\hat{F}(\omega) \mid \omega \in \Omega\} =: \hat{F}[\Omega]$ parameterized over some compact metric space (Ω, d_{Ω}) with some fixed map $\hat{F} : \Omega \to C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$. Similarly to our assumption of F, we assume that $\mathbb{D}[\Omega]$ consists of diagonal operators and there exists and that there exists some $\omega^* \in \Omega$ such that $\mathbb{D}(\omega, d_{\Omega})$ with some fixed map $\hat{F} : \Omega \to C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$. Similarly to our assumption of F, we assume that $\mathbb{D}[\Omega]$ consists of diagonal operators and that there exists some $\omega^* \in \Omega$ such that $\mathbb{D}(\omega^*) = F$. We summarize these assumptions and definitions and definitions below:

Assumption 4.2. We are given a compact metric space (Ω, d_{Ω}) and a map \hat{F} : $\Omega \rightarrow C(\ell^{X \times U}, \ell^{X})$, where each $\omega \in \Omega$ represents a dynamic system with diagonal dynamics operator $\hat{F}[\omega]$.

As we discussed, taking the CE-approach is the most common (often also efficient [92]) way to tackle learning of feedback control in the closed-loop. The defining feature of CE-adaptation is the simple structure of the decision-making process: At each time-step k, we hypothesize a model parameter θ_k and evaluate the *nominal*

controller $u_k = K_k[\theta_k](x_{k:0})$ designed for the assumed dynamics $\hat{F}[\theta_k] := \mathbb{D}(\theta_k)$; one can view the nominal controllers as a parametrized collection $\{\mathbb{K}(\omega) \mid \omega \in \Omega\}$, $\mathbb{K} : \Omega \to C(\ell^{\mathcal{X}}, \ell^{\mathcal{U}})$ of specified (or synthesized online) control laws, where each control law $K[\omega] := \mathbb{K}(\omega)$ achieves some desired nominal closed-loop behavior, assuming the system dynamics are $\hat{F}[\omega]$. Selecting θ_k is commonly performed via online system identification [16], or gradient-based adaptation rules [71], and we shall denote this process as the causal operator $S \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \Omega^{\mathbb{N}}), \Omega^{\mathbb{N}} := \{\omega \mid \omega_k \in \Omega, \forall k \in N\}$ which we refer to as the *model selection/selector* S.

To summarize, under CE-adaptive control, the closed-loop dynamics are governed by the following equations at each time-step k:

$$x_k = f_{k-1}(x_{k-1}, u_{k-1}) + w_k$$
(4.2a)

$$\theta_k = S_k(x_{k:0}, u_{k-1:0}) \tag{4.2b}$$

$$u_k = K_k[\theta_k](x_{k:0}).$$
 (4.2c)

It is hard to gauge the behavior of the closed-loop dynamics from the above equations; however, that changes with a slight tweak to how we perform adaptation. Next, we formulate CE-adaptation in terms of nominal closed-loop maps and system-level controllers.

CE-Adaptation in System Level Implementations

Instead of encoding the desired nominal behavior as a parameterization $\mathbb{K} : \omega \mapsto \mathbf{K}[\omega]$ of control laws $\{\mathbf{K}[\omega] | \omega \in \mathbf{\Omega}\}$, we can equivalently consider a representation in the form of nominal CLMs $\{\Psi[\omega] | \omega \in \mathbf{\Omega}\}, \Psi[\omega] \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$. Hence, we assume each nominal CLM $\Psi[\omega]$ is a CLM of the dynamics $\hat{F}[\omega]$:

$$\forall \omega \in \mathbf{\Omega} : \qquad \Psi[\omega] \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}}) \text{ s.t.: } \Psi[\omega] \in \Phi_{\mathrm{CL}}^{\mathrm{w} \mapsto \mathrm{xu}}[\hat{F}[\omega]]$$
(4.3)

and corresponds to the nominal control law $K[\omega] = \Psi^u[\omega](\Psi^x[\omega])^{-1}$. However, the System Level Implementation $SL[\Psi^u[\omega], \Psi^x[\omega]]$ of the nominal controller offers alternative (to (4.2c)) ways to perform adaptation, simply by swapping out the nominal CLMs in the SL control structure. The resulting closed-loop dynamic equations over time k take the form:

$$x_k = f_{k-1}(x_{k-1}, u_{k-1}) + w_k$$
(4.4a)

$$\hat{w}_k = x_k - \Psi_k^x [\theta_k](0, \hat{w}_{k-1:0})$$
(4.4b)

$$u_k = \Psi_k^u [\theta_k](\hat{w}_{k:0}) \tag{4.4c}$$

$$\theta_k = S_k(x_{k:0}, u_{k-1:0}). \tag{4.4d}$$

Despite the correspondence $K[\omega] = \Psi^u[\omega](\Psi^x[\omega])^{-1}$, the closed-loops (4.4) and (4.2) are *not* the same because CE-adaptation is performed differently.

Remark. In fact, the adaptive control law (4.2c) parameterized by CLMs takes the form $\boldsymbol{u} = \sum_{t=0}^{\infty} \boldsymbol{P}^{[t]} \boldsymbol{\Psi}^{u}[\theta_{t}] \boldsymbol{P}^{t} \boldsymbol{\Psi}^{-x}[\theta_{t}] \boldsymbol{P}^{t} \boldsymbol{x}.$

However, the above closed-loop dynamics can be manipulated into a form that turns out to be insightful, particularly for the design of S. In the next section, we discuss sufficient conditions for closed-loop stability provided by appropriate design of the model selection.

Model Selection Conditions for Closed-Loop Stability

For this section, we consider S a design variable, and therefore we treat the parameter sequence θ as an input of the closed-loop model (the set of all $(x, u, \theta, \hat{w}, w)$ conforming with equations (4.4)). Regardless of θ , the usual identity still remains for all k:

$$x_k = \Psi_k^x[\theta_k](\hat{w}_{k:0}) \qquad \qquad u_k = \Psi_k^u[\theta_k](\hat{w}_{k:0}). \tag{4.5}$$

Moreover, treating the sequence θ as an index set, we denote $\Psi_{|\theta}$ as the operator with component functions $(\Psi_{|\theta})_k : (\hat{w}_{k:0}) \mapsto \Psi_t[\theta_t](\hat{w}_{k:0})$. We can rewrite (4.5) in sequence space as:

$$\boldsymbol{x} = \boldsymbol{\Psi}_{\boldsymbol{\mid}\boldsymbol{\theta}}^{\boldsymbol{x}} \hat{\boldsymbol{w}} \qquad \qquad \boldsymbol{u} = \boldsymbol{\Psi}_{\boldsymbol{\mid}\boldsymbol{\theta}}^{\boldsymbol{u}} \hat{\boldsymbol{w}}. \tag{4.6}$$

An intuitive design criterion for model selection and choice of θ is the *1-step* prediction error e. At time step k, the prediction error is defined as $e_k = x_k - \hat{f}_{k-1}[\theta_k](x_{k-1}, u_{k-1})$, i.e., it measures how well the current selected model θ_k matches with the most recent system transition $(x_{k-1}, u_{k-1}) \mapsto x_k$. However, the selection of θ_k is allowed to depend on e_k , since we get to observe x_k before choosing θ_k . In particular, it is easy to see that we can always control the size of this error to be at most the size of w_k . Since there exists a true $\omega^* \in \Omega$ such that $F = \hat{F}[\omega^*]$, given x_k and x_{k-1}, u_{k-1} at time k, ω^* accurately models the latest state transition $x_{k-1}, u_{k-1} \mapsto x_k$. That is, $e_k = x_k - \hat{f}_k[\theta_k](x_{k-1}, u_{k-1}) = w_k$, therefore it is always possible to find a parameter θ_k such that $|e_k| \leq |w_k|$. The same argument applies to the entire history of observed transitions $(x, u)_{k-1:0} \mapsto x_{k:1}$, and therefore it is always possible to select a parameter θ_k such that the prediction error in hindsight

$$e_{[k:0]|k} := x_{k:1} - F_{k-1:0}[\theta_k](x_{k-1:0}, u_{k-1:0})$$

is always smaller in norm than $w_{k-1:0}$. Hence, at each time k, the history of past observations $x_{k:0}$, $u_{k-1:0}$ provides sufficient knowledge to find some θ_k^{*p} , for example, by explicit minimization, such that $||e_{[k:0]|k}||_p \leq ||w_{k-1:0}||_p$, regardless of what value w takes. We will call a sequence θ for which the hindsight prediction errors are uniformly bounded over time, i.e., $\sup_k \{||e_{[k:0]|k}||_p\} < \infty$, to be an ℓ_p -consistent parameter selection, and $\theta^{*p} = (\theta_0^{*p}, \theta_1^{*p}, \dots)$ to be ℓ_p -optimal. Similarly, we call a sequence θ to be a finite horizon ℓ_p -consistent selection for some T, if the norm of the truncated hindsight prediction errors

$$e_{\mathcal{I}_k|k} := x_{\mathcal{I}_k} - \hat{F}_{\mathcal{I}_k-1}[\theta_k](x_{k-1:0}, u_{k-1:0}), \quad \mathcal{I}_k = [k - T, k]$$

forms a scalar ℓ_p -bounded sequence $(|e_{\mathcal{I}_0|0}|, |e_{\mathcal{I}_1|1}|, ...)$. We summarize these definitions below and define causal operators S to be ℓ_p -consistent model selectors if the output parameter sequence $\boldsymbol{\theta} = S(\boldsymbol{x}, \boldsymbol{u})$ is always a ℓ_p -consistent selection, whenever $\boldsymbol{x} = \boldsymbol{F}^+(\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w}, \boldsymbol{w} \in \ell_p^{\mathcal{X}}$.

Definition 4.1. Let (Ω, d) be a parameter space, $\hat{F} : \Omega \ni \omega \mapsto \hat{F}[\omega] \in C((\mathcal{X} \times \mathcal{U})^{\mathbb{N}}, \mathcal{X}^{\mathbb{N}})$ be parameterization of dynamic functions, and $S \in C((\mathcal{X} \times \mathcal{U})^{\mathbb{N}}, \Omega^{\mathbb{N}})$ be a parameter selector. Denote $E_T : (\Omega \times \mathcal{X} \times \mathcal{U})^{\mathbb{N}} \to \mathbb{R}^+_0$ as the causal operator defined by its component functions as

$$\boldsymbol{P}^{[k]}\boldsymbol{E}_T: \ (\boldsymbol{\theta}, \boldsymbol{x}, \boldsymbol{u}) \mapsto \|\boldsymbol{P}^{\mathcal{I}_k}(\boldsymbol{x} - \hat{\boldsymbol{F}}^+[\theta_k](\boldsymbol{x}, \boldsymbol{u}))\|_p, \ \mathcal{I}_k := [k - T, k].$$

- (i) S is called an ℓ_p -consistent parameter selector if for all $x \in \ell^{\mathcal{X}}$, $u \in \ell^{\mathcal{U}}$, $w \in \ell_p^{\mathcal{X}}$ such that $x = \hat{F}^+[\omega](x, u) + w$ for some $\omega \in \Omega$, $E_{\infty}(S(x, u), x, u) \in \ell_{\infty}$.
- (ii) S is called an ℓ_p -consistent parameter selector with a finite T-horizon if for all $x \in \ell^{\mathcal{X}}$, $u \in \ell^{\mathcal{U}}$, $w \in \ell_p^{\mathcal{X}}$ such that $x = \hat{F}^+[\omega](x, u) + w$ for some $\omega \in \Omega$, $E_T(S(x, u), x, u) \in \ell_p$.

Remark. We discuss consistent model selections in depth in Chapter 6.

As alluded to previously, the following decision rule always defines a corresponding ℓ_p -consistent selector:

$$S_t: \boldsymbol{P}^t(\boldsymbol{x}, \boldsymbol{u}) \mapsto \theta_t \in \operatorname*{arg\,min}_{\omega} \| \boldsymbol{P}^{\mathcal{I}_k}(\boldsymbol{P}^t \boldsymbol{x} - \hat{\boldsymbol{F}}^+[\omega] \boldsymbol{P}^t(\boldsymbol{x}, \boldsymbol{u})) \|_p$$

 $\boldsymbol{\theta}^{*p} = (\theta_0^{*p}, \theta_1^{*p}, \dots)$, and therefore the model selectors defined above always exist. Next, we consider \boldsymbol{S} , which are just ℓ_{∞} -consistent selectors of horizon T = 1, and consider \boldsymbol{F} , which are diagonal operators, i.e., for any t, $\boldsymbol{P}^{[t]}\boldsymbol{F} = \boldsymbol{P}^{[t]}\boldsymbol{F}\boldsymbol{P}^{[t]}$. In other words, the selector S is assumed to guarantee the uniform boundedness of the 1-step prediction error $e_k = f_{k-1}(x_{k-1}, u_{k-1}) + w_k - \hat{f}_{k-1}[\theta_k](x_{k-1}, u_{k-1})$. Provided that assumption, e can replace the role of w as input, and as summarized in Theorem 22, we can analyze the closed-loop stability in terms of the stability of the map $(e, \theta) \mapsto (\hat{w}, x, u)$.

To derive these equations, first, notice that the following statement is vacuous:

$$x_k = \hat{f}_{k-1}[\theta_k](x_{k-1}, u_{k-1}) + e_k.$$

By substituting the above into (4.4b) and using the identity (4.5), we arrive at the following set of equations:

$$\hat{w}_k = \hat{f}_{k-1}[\theta_k](\Psi_{k-1}[\theta_{k-1}](\hat{w}_{k-1:0})) - \Psi_k^x[\theta_k](0, \hat{w}_{k-1:0}) + e_k.$$
(4.7)

For a fixed k, the right-hand side of the equation corresponds with the k-th component of the operator $\hat{F}^+[\theta_k]\Psi[\theta_{k-1}] + \overline{\Psi^x}[\theta_k]$, where $\overline{\Psi^x} := I - \Psi^x$ hence we can write the above set of equations in sequence space as:

$$egin{aligned} \hat{m{w}} &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(\hat{m{w}}) + m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{w}}) + m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{W}}) + m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{W}}) + m{e} \ &= \sum_{k=0}^\infty m{P}^{[k]}(\hat{m{F}}^+[heta_k] \Psi[heta_{k-1}] + \overline{\Psi^x}[heta_k])(m{P}^{k-1}\hat{m{W}}) + m{P}^{[k]}(m{P}^{k-1}\hat{m{P}}) +$$

where the second equation follows from the strict causality of \hat{F}^+ and $\overline{\Psi^x}$. The operator on the right-hand side is what we refer to as $\Delta_{|\theta}$, and it determines the stability of the disturbance dynamics. The terms of the sum are parameterized by the function $\Lambda^+[\omega, \nu] := \hat{F}^+[\omega]\Psi[\nu] + \overline{\Psi^x}[\omega]$.

Theorem 22 below summarizes our findings so far:

Theorem 22. Consider a fixed sequence $\theta \in \Omega^{\mathbb{N}}$ and let (\hat{w}, x, u) be governed by the closed-loop equations (4.4a)-(4.4c) where F and $\hat{F}[\omega], \omega \in \Omega$ are all diagonal operators. Then (\hat{w}, x, u) satisfy:

$$x = \Psi^x_{|\theta} \hat{w}$$
 $u = \Psi^u_{|\theta} \hat{w}$ $\hat{w} = (I - \Delta^+_{|\theta})^{-1} e$ (4.8)

where $\Psi_{|\theta} \in \mathcal{C}(\mathcal{X}^{\mathbb{N}}, (\mathcal{X} \times \mathcal{U})^{\mathbb{N}})$ and $\Delta_{|\theta}^+ \in \mathcal{C}_s((\mathcal{X} \times \mathcal{U})^{\mathbb{N}}, \mathcal{X}^{\mathbb{N}})$ denote the operators:

$$\Psi|_{\boldsymbol{\theta}} = \sum_{t=0}^{\infty} \boldsymbol{P}^{[t]} \Psi[\theta_t] \boldsymbol{P}^t \qquad \boldsymbol{\Delta}_{|\boldsymbol{\theta}}^+ = \sum_{k=1}^{\infty} \boldsymbol{P}^{[t]} \boldsymbol{\Lambda}^+ [\theta_t, \theta_{t-1}] \boldsymbol{P}^{t-1}$$
(4.9)

and $\Lambda^+ : \Omega^2 \to C_s((\mathcal{X} \times \mathcal{U})^{\mathbb{N}}, \mathcal{X}^{\mathbb{N}})$, $\Omega^2 := \Omega \times \Omega$, for fixed $\omega, \nu \in \Omega$, is defined as

$$oldsymbol{\Lambda}^+[\omega,
u]:= \hat{oldsymbol{F}}^+[\omega]\Psi[
u]+\overline{\Psi^x}[\omega]$$

The stability of the closed-loop systems is determined by the stability of the maps $\Psi|_{\theta}$ and $(I - \Delta_{|\theta}^+)^{-1}$ in equation (4.8). If $\Psi[\omega], \omega \in \Omega$ are all CLMs of the corresponding dynamics $\hat{F}[\omega], \omega \in \Omega$ and \hat{F} are linear we can split $\Delta_{|\theta}^+$ into two components, as stated below:

Lemma 18. If $\Psi[\omega] \in \Phi_{CL}^{w \mapsto xu}(\hat{F}[\omega]), \forall \omega \in \Omega$, the operator $\Delta|_{\theta}$ can be written as the sum $\Delta_{|\theta}^{+\varphi} + \Delta_{|\theta}^{+f}$, where where $\Delta_{|\theta}^{+f}, \Delta_{|\theta}^{+\varphi}$ are defined by their component functions for $k \in \mathbb{N}$ as:

$$\begin{aligned} \boldsymbol{U}^{*[k]} \boldsymbol{\Delta}_{|\boldsymbol{\theta}}^{+\boldsymbol{f}} \boldsymbol{U}^{k} : \ (\hat{w}_{k:0}) &\mapsto (\hat{f}_{k-1}[\theta_{k}] - \hat{f}_{k-1}[\theta_{k-1}]) \circ (\Psi_{k-1}[\theta_{k-1}])(\hat{w}_{k-1:0}) \\ \boldsymbol{U}^{*[k]} \boldsymbol{\Delta}_{|\boldsymbol{\theta}}^{+\boldsymbol{\varphi}} \boldsymbol{U}^{k} : \ (\hat{w}_{k:0}) &\mapsto (\overline{\Psi}_{k}[\theta_{k}] - \overline{\Psi}_{k}[\theta_{k-1}])(0, \hat{w}_{k-1:0}). \end{aligned}$$

Moreover, if $\{\hat{f}_k[\omega] \mid \omega \in \Omega, k \in \mathbb{N}\}$ are linear functions, then the components of $\Delta^+_{|\theta}$ become

$$\boldsymbol{U}^{*[k]}\boldsymbol{\Delta}_{|\boldsymbol{\theta}}^{+}\boldsymbol{U}^{k} = \hat{f}_{k-1}[\boldsymbol{\theta}_{k}] \circ (\Psi_{k-1}[\boldsymbol{\theta}_{k-1}] - \Psi_{k-1}[\boldsymbol{\theta}_{k}]).$$
(4.10)

Proof. Per definition, $\Psi[\theta_k] \in \Phi_{CL}^{w \mapsto xu}[F[\theta_k]]$ and $\Psi[\theta_{k-1}] \in \Phi_{CL}^{w \mapsto xu}[F[\theta_{k-1}]]$ holds, and therefore the operators have to satisfy the respective CLM equation:

$$\boldsymbol{F}^{+}[\theta_{k-1}](\boldsymbol{\Psi}[\theta_{k-1}]) = \boldsymbol{\Psi}^{x}[\theta_{k-1}] - \boldsymbol{I} \qquad \boldsymbol{F}^{+}[\theta_{k}](\boldsymbol{\Psi}[\theta_{k}]) = \boldsymbol{\Psi}^{x}[\theta_{k}] - \boldsymbol{I}.$$

With the above, we can rewrite (4.7) as

$$\hat{w}_k = \Delta_{k-1}^f(\hat{w}_{k-1:0}) + \Delta_{k-1}^{\varphi}(\hat{w}_{k-1:0}) + e_k$$

with the operators defined below:

$$\Delta_{k-1}^{f} := (\hat{f}_{k-1}[\theta_{k}] - \hat{f}_{k-1}[\theta_{k-1}]) \circ (\Psi_{k-1}[\theta_{k-1}])$$
$$\Delta_{k-1}^{\varphi}(\hat{w}_{k-1:0}) := (\Psi_{k}[\theta_{k-1}] - \Psi_{k}[\theta_{k}])(0, \hat{w}_{k-1:0})$$
$$e_{k} := x_{k} - \hat{f}_{k-1}[\theta_{k}](x_{k-1}, u_{k-1}).$$

Lastly, if $\{\hat{f}_k[\omega] | \omega \in \Omega, k \in \mathbb{N}\}$ are linear functions, the components of $\Delta_{|\theta}^f + \Delta_{|\theta}^{\varphi+}$ take on a simpler a form:

$$(\Delta_{|\boldsymbol{\theta}}^{f} + \Delta_{|\boldsymbol{\theta}}^{\varphi+})_{k-1} = \hat{f}_{k-1}[\theta_{k}] \circ (\Psi_{k-1}[\theta_{k-1}] - \Psi_{k-1}[\theta_{k}]).$$

Hence, CE-adaptation using the SL implementation of the nominal controllers allows us to represent the closed-loop dynamics in the form:

$$x=\Psi^x_{| heta}\hat{w} \qquad u=\Psi^u_{| heta}\hat{w} \qquad \hat{w}=\left(I-\mathcal{S}^+\Delta^f_{| heta}-\mathcal{S}^+\Delta^{arphi+}_{| heta}
ight)^{-1}e^{-1}$$

From this representation, we can derive conditions for closed-loop stability in terms of θ . As expected, a necessary condition for closed-loop stability is sensible design of the nominal CLMs that assures at least some type of stability of the maps $\Psi[\omega]$. Moreover, boundedness of the closed-loop trajectory (θ, w, x, u) is implied, if θ is chosen such that the derived operator $(I - S^+ \Delta^f_{|\theta} - S^+ \Delta^{\varphi+}_{|\theta})^{-1}$ is $\ell^{\mathcal{X}}_{\infty}$ -stable and the prediction error e is bounded. This observation provides an objective for the design of the model selection: Finding a causal selection rule S which guarantees the former conditions for any selected sequence $\theta = S(x, u)$. A natural design approach is to view S as a means to prove stability via the small-gain theorem. For example, if S guarantees sufficiently small gains of $\Delta^{arphi^+}_{| heta}$ and $\Delta^{f^+}_{| heta}$, then we can prove stability of $(I - S^+ \Delta^f_{|\theta} - S^+ \Delta^{\varphi^+}_{|\theta})^{-1}$ via small-gain theorems such as Theorem 2. From this representation, we can derive conditions for closed-loop stability in terms of θ . As expected, a necessary condition for closed-loop stability is a sensible design of the nominal CLMs that guarantees at least some type of stability of the maps $\Psi[\omega]$. Moreover, the boundedness of the closed-loop trajectory (θ, w, x, u) is implied if θ is chosen such that the derived operator $(I - S^+ \Delta^f_{|\theta} - S^+ \Delta^{\varphi+}_{|\theta})^{-1}$ is ℓ^{χ}_{∞} stable and the prediction error e is bounded. This observation provides an objective for the design of the model selection: Finding a causal selection rule S that guarantees the former conditions for any selected sequence $\theta = S(x, u)$. A natural design approach is to view S as means to prove stability via the small-gain theorem. For example, if S guarantees sufficiently small gains of $\Delta^{\varphi_+}_{|\theta}$ and $\Delta^{f_+}_{|\theta}$, then we prove the stability of $(I - S^+ \Delta^f_{|\theta} - S^+ \Delta^{\varphi_+}_{|\theta})^{-1}$ using small-gain theorems such as Theorem 2. However, we obtain less restrictive design criteria if we conduct analysis with the adapted small-gain conditions of Lem. 8:

Theorem 23. Let w, x, u, θ be a trajectory of the closed-loop dynamical system described by the equations (4.4) for some disturbance $w \in \ell_{\infty}^{\chi}$ and sequence of selected parameters θ . The state and input sequence x, u are bounded if the following conditions are met:

- *i)* Consistent model selection: S is an ℓ_{∞} -consistent model selector with a horizon of at least T = 1.
- *ii)* Efficient selection: the selection θ converges in Ω .

- iii) *CE-nominal design:* Ψ *is a parametrization of CLMs: for each* $\omega \in \Omega$ *holds* $\Delta[\Psi[\omega], \hat{F}[\omega]] = 0.$
- iv) Smooth/robust parametrization of CLMs: The "oracle" map $\Psi : \Omega \rightarrow C(\ell^{\chi}, \ell^{\chi \times U})$ is continuous over the metric space (Ω, d) w.r.t. the operator norm $\| \cdot \|_{\infty}$.
- *v)* Smooth/robust parametrization of Dynamics : The map **D** defined by the correspondence

$$\boldsymbol{D}: (\omega_1, \omega_2) \mapsto \boldsymbol{F}[\omega_1] \circ \boldsymbol{\Psi}[\omega_2]$$

between the domain $(\Omega^2, d_{\Omega^2}), d_{\Omega^2}(x, y) := d(x_1, y_1) \lor d(x_2, y_2)$ and codomain $(\mathcal{C}(\ell^n, \ell^n), |||(\cdot) - (\cdot)|||_{\infty})$ is continuous at every point p of the subset $\{(\omega, \omega) \mid \omega \in \Omega\} \subset \Omega^2.$

Proof. We assumed (Ω, d) is a compact metric space and defined for the family of dynamics $\{\hat{F}[\omega] | \omega \in \Omega\}$ and CLMs $\{\Psi[\omega] | \omega \in \Omega\}$ the map $\Lambda^+ : \Omega^2 \to C_s(\ell^{\chi}, \ell^{\chi})$, where

$$oldsymbol{\Lambda}^+[\omega, \mathbf{v}] \coloneqq oldsymbol{F}^+[\omega] \Psi[\mathbf{v}] + \overline{\Psi^x}[\omega] = oldsymbol{D}^+[\omega, \mathbf{v}] + oldsymbol{I}_n - \overline{\Psi^x}[\omega]$$

for each $(\omega, \nu) \in \Omega^2$ in the metric space (Ω^2, d_{Ω^2}) . Recall the definitions of the operator norm $\| \cdot \|_p$ and pseudo-norm $\| \cdot \|_p^*$ from Lem. 11:

$$\|\|m{A}\|\|_p^\star := \sup_{m{x}\in\mathcal{X}_p^{\mathbb{N}}:\,m{x}
eq 0} \; rac{\|m{A}(m{x})-m{A}(m{0})\|_p}{\|m{x}\|_p} \qquad \|\|m{A}\|\|_p := \|\|m{A}\|\|_p^\star + \|m{A}(m{0})\|_p$$

and let $\mathbb{E}_0 : \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}}) \to \ell^{\mathcal{X}}$ denote the linear evaluation map such that for each $\forall \mathbf{A} \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}}) : \mathbb{E}_0(\mathbf{A}) = \mathbf{A}(0)$. Since per assumption, $\mathbf{D} : (\mathbf{\Omega}^2, d_{\mathbf{\Omega}^2}) \to (\mathcal{C}(\ell^n, \ell^n), \| \cdot \| _{\infty})$ is continuous for all $p = (p_1, p_2) \in \mathbf{\Omega}^2$, s.t.: $p_1 = p_2$ and $\Psi[\cdot]$ is a continuous map $(\mathbf{\Omega}, d) \to \| \cdot \| _{\infty}$, it follows that $\mathbf{\Lambda}^+ : (\mathbf{\Omega}^2, d_{\mathbf{\Omega}^2}) \to (\mathcal{C}(\ell^n, \ell^n), \| \cdot \| _{\infty})$ has to be continuous for all $p = (p_1, p_2) \in \mathbf{\Omega}^2$, s.t.: $p_1 = p_2$ as well.

Per assumption, we are assured that our parameter sequence converges to some parameter $\theta_{\infty} = \lim_{t \to \infty} \theta_t$, $\theta_{\infty} \in \Omega$. Pick $\varepsilon = \mu < 1$ and invoke continuity of Λ^+ at θ_{∞} . Thus, there exists some $\delta > 0$ such that

$$\forall \boldsymbol{\omega}, \boldsymbol{\nu} \in \mathsf{B}_{\delta}[\theta_{\infty}] : \quad |||\boldsymbol{\Lambda}^{+}[\boldsymbol{\omega},\boldsymbol{\nu}] - \boldsymbol{\Lambda}^{+}[\theta_{\infty},\theta_{\infty}]|||_{\infty} \leq \varepsilon.$$
(4.11)

Since $\lim_{t\to\infty} \theta_t = \theta_{\infty}$, there exists some $N \in \mathbb{N}$ such that $\forall k > N$ holds $d(\theta_{k-1}, \theta_{\infty}) < \delta$. Now decompose $\Delta|_{\theta}$ into the sum

$$\boldsymbol{\Delta}|_{\boldsymbol{\theta}} = \sum_{k=1}^{\infty} \boldsymbol{P}^{k} \boldsymbol{\Lambda}^{+} [\theta_{k}, \theta_{k-1}] = \sum_{k=1}^{\infty} \boldsymbol{P}^{[k]} \boldsymbol{\Lambda}^{+} [\theta_{k}, \theta_{k-1}] \boldsymbol{P}^{k-1}$$

and assume some arbitrary $\boldsymbol{z} \in \ell_{\infty}^{\mathcal{X}}$. Next, rewrite $\|\overline{\boldsymbol{P}^{N}}\boldsymbol{\Delta}|_{\boldsymbol{\theta}}\boldsymbol{z}\|_{\infty}$ as

$$\|\overline{\boldsymbol{P}^{N}}\boldsymbol{\Delta}|_{\boldsymbol{\theta}}\boldsymbol{z}\|_{\infty} = \sup_{h>N} \|\boldsymbol{P}^{[h]}\boldsymbol{\Delta}|_{\boldsymbol{\theta}}\boldsymbol{z}\|_{\infty} = \sup_{h>N} \|\boldsymbol{P}^{[h]}\sum_{k=1}^{\infty} \boldsymbol{P}^{[k]}\boldsymbol{\Lambda}^{+}[\theta_{k},\theta_{k-1}]\boldsymbol{P}^{k-1}\boldsymbol{z}\|_{\infty}$$
(4.12)

$$= \sup_{h>N} \|\boldsymbol{P}^{[h]}\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\boldsymbol{P}^{h-1}\boldsymbol{z}\|_{\infty}.$$
(4.13)

Next we bound each individual term $\|\boldsymbol{P}^{[h]}\boldsymbol{\Lambda}^+[\theta_h,\theta_{h-1}]\boldsymbol{P}^{h-1}\boldsymbol{z}\|_{\infty}$:

$$\|\boldsymbol{P}^{[h]}\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\boldsymbol{P}^{h-1}\boldsymbol{z}\|_{\infty}$$

$$\leq \|\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\boldsymbol{P}^{h-1}\boldsymbol{z}-\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}](\boldsymbol{0})\|_{\infty}+\|\mathbb{E}_{0}\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\|_{\infty}. \quad (4.14)$$

By construction h > N, $\theta_h, \theta_{h-1} \in \mathsf{B}_{\delta}[\theta_{\infty}]$ and by continuity of Λ^+ we have

$$\|\mathbb{E}_{0}\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\|_{\infty} \leq \|\mathbb{E}_{0}\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}] - \mathbb{E}_{0}\boldsymbol{\Lambda}^{+}[\theta_{\infty},\theta_{\infty}]\|_{\infty} \leq \varepsilon$$

where $\Lambda^+[\theta_{\infty}, \theta_{\infty}] = \Delta[\hat{F}, \Psi][\theta_{\infty}] = 0$, since $\Psi[\theta_{\infty}]$ is designed as a CLM of $\hat{F}[\theta_{\infty}]$. Similarly continuity of Λ^+ implies that

$$\begin{split} \|\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\boldsymbol{P}^{h-1}\boldsymbol{z} - \boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}](\boldsymbol{0})\|_{\infty} &\leq \mu \|\boldsymbol{P}^{h-1}\boldsymbol{z}\|_{\infty} \\ &\leq \mu \|\boldsymbol{P}^{N}\boldsymbol{z}\|_{\infty} + \mu \|\boldsymbol{P}^{(N,h-1)}\boldsymbol{z}\|_{\infty} \end{split}$$

and by substituting this bound into (4.14) we obtain a bound for each term on the right-hand side of (4.13):

$$\|\boldsymbol{P}^{[h]}\boldsymbol{\Lambda}^{+}[\theta_{h},\theta_{h-1}]\boldsymbol{P}^{h-1}\boldsymbol{z}\|_{\infty} \leq \mu \|\boldsymbol{P}^{(N,h-1)}\boldsymbol{z}\|_{\infty} + \mu \|\boldsymbol{P}^{N}\boldsymbol{z}\|_{\infty} + \varepsilon.$$
(4.15)

Finally, we apply the above to (4.13) and bound $\|\overline{P^N}\Delta|_{\theta} z\|_{\infty}$ from above as:

$$\|\overline{\boldsymbol{P}^{N}}\boldsymbol{\Delta}|_{\boldsymbol{\theta}}\boldsymbol{z}\|_{\infty}$$

$$\leq \sup_{h>N} \mu \|\boldsymbol{P}^{(N,h-1]}\boldsymbol{z}\|_{\infty} + \mu \|\boldsymbol{P}^{N}\boldsymbol{z}\|_{\infty} + \varepsilon$$

$$\leq \mu \sup_{h>N} \sup_{j\in(N,h-1]} |z_{j}| + \mu \|\boldsymbol{P}^{N}\boldsymbol{z}\|_{\infty} + \varepsilon$$

$$\leq \mu \sup_{h>N} |z_{h}| + \mu \|\boldsymbol{P}^{N}\boldsymbol{z}\|_{\infty} + \varepsilon = \mu \|\overline{\boldsymbol{P}^{N}}\boldsymbol{z}\|_{\infty} + \mu \|\boldsymbol{P}^{N}\boldsymbol{z}\|_{\infty} + \varepsilon.$$
(4.16)

We are ready to apply our findings for stability analysis of the lumped disturbance dynamics. Recall from our derivations, that \hat{w} is governed by the dynamic equations $\hat{w} = \Delta^+|_{\theta}\hat{w} + e$ and therefore:

$$\overline{oldsymbol{P}^N}oldsymbol{\hat{w}}=\overline{oldsymbol{P}^N}oldsymbol{\Delta}^+|_{oldsymbol{ heta}}oldsymbol{\hat{w}}+\overline{oldsymbol{P}^N}oldsymbol{e}.$$

Denote the sequence of the intervals $\mathcal{I}_k = [N+1, k], k \ge N+1$ and apply the truncation $P^{\mathcal{I}_t}$ to the above equation for some t > N and use causality of $\Delta|_{\theta}$ to arrive at the equality:

$$oldsymbol{P}^{\mathcal{I}_t} \hat{oldsymbol{w}} = oldsymbol{P}^{\mathcal{I}_t} \overline{oldsymbol{P}^N} \Delta^+ |_{oldsymbol{ heta}} oldsymbol{P}^t \hat{oldsymbol{w}} + oldsymbol{P}^{\mathcal{I}_t} oldsymbol{e}.$$

Now, we apply (4.16) by setting $\boldsymbol{z} = \boldsymbol{P}^t \hat{\boldsymbol{w}}$ and obtain the inequality

$$\begin{aligned} \| \boldsymbol{P}^{\mathcal{I}_{t}} \hat{\boldsymbol{w}} \|_{\infty} &\leq \mu \| \overline{\boldsymbol{P}^{N}} \boldsymbol{P}^{t} \hat{\boldsymbol{w}} \|_{\infty} + \mu \| \boldsymbol{P}^{N} \boldsymbol{P}^{t} \hat{\boldsymbol{w}} \|_{\infty} + \| \boldsymbol{P}^{\mathcal{I}_{t}} \boldsymbol{e} \|_{\infty} + \varepsilon \\ &\leq \mu \| \boldsymbol{P}^{\mathcal{I}_{t}} \hat{\boldsymbol{w}} \|_{\infty} + \mu \| \boldsymbol{P}^{N} \hat{\boldsymbol{w}} \|_{\infty} + \| \boldsymbol{P}^{\mathcal{I}_{t}} \boldsymbol{e} \|_{\infty} + \varepsilon \end{aligned}$$
$$\Leftrightarrow \quad (1-\mu) \| \boldsymbol{P}^{\mathcal{I}_{t}} \hat{\boldsymbol{w}} \|_{\infty} &\leq \mu \| \boldsymbol{P}^{N} \hat{\boldsymbol{w}} \|_{\infty} + \| \boldsymbol{P}^{\mathcal{I}_{t}} \boldsymbol{e} \|_{\infty} + \varepsilon. \end{aligned}$$

The constant $\varepsilon = \mu$ was chosen such that $\mu < 1$. Hence from the above we get the inequality

$$\| \boldsymbol{P}^{\mathcal{I}_t} \hat{\boldsymbol{w}} \|_{\infty} \leq rac{\mu \| \boldsymbol{P}^N \hat{\boldsymbol{w}} \|_{\infty} + \| \boldsymbol{e} \|_{\infty} + \mu}{1 - \mu}.$$

The right-hand side of the above inequality is bounded because: 1. we assumed a bounded prediction error $e \in \ell_{\infty}^{\mathcal{X}}$ and 2. N is finite and thus $\|P^N P^t \hat{w}\|_{\infty}$ ¹. Since the scalar sequence $(s_N, s_{N+1}, ...)$, where $s_k = \|P^{\mathcal{I}_t} \hat{w}\|_{\infty}$ is non-decreasing and bounded above, it converges and proves that $\overline{P^N} \hat{w} \in \ell_{\infty}^{\mathcal{X}}$ and therefore $\hat{w} \in \ell^{\mathcal{X}}$. Moreover, $\|\hat{w}\|_{\infty}$ can be bounded above in terms of the norm of its finite T-truncation as:

$$\|\hat{\boldsymbol{w}}\|_{\infty} \leq \left\{ \|\boldsymbol{P}^{N}\hat{\boldsymbol{w}}\|_{\infty}, \frac{\mu \|\boldsymbol{P}^{N}\hat{\boldsymbol{w}}\|_{\infty} + \|\boldsymbol{e}\|_{\infty} + \varepsilon}{1-\mu}
ight\}$$

Finally, we utilize the continuity of the CLM parametrization map $\Psi : (\Omega, d) \rightarrow (\ell^{\mathcal{X} \times \mathcal{U}}, \|\cdot\|_{\infty})$. Since Ψ is continuous, it also follows that $\mathbb{E}_0 \circ \Psi$ is continuous. Due to the compactness of Ω , there exists some C > 0 and $c_0 > 0$ such that $\||\Psi[\omega]|\|_{\infty} \leq C$ and $\|\Psi[\omega](0)\|_{\infty} \leq c_0 \ \forall \omega \in \Omega$. This also implies that the component functions have to obey the inequality:

$$|\Psi_t[\omega](z_{t:0})| \leq C \|z_{t:0}\|_{\infty} + c_0$$

This proves the boundedness of \boldsymbol{x} and \boldsymbol{u} , since $(x_t, u_t) = \Psi_t[\theta_t](\hat{w}_{t:0}) \leq C \max_{k \leq t} |\hat{w}_k| + c_0$, and therefore $\Psi \in \mathcal{C}(\ell_{\infty}^{\mathcal{X}}, \ell_{\infty}^{\mathcal{X} \times \mathcal{U}})$.

The stability conditions above admit a natural interpretation and provide us with a concrete guideline for nominal control design and the design of model selectors. To

¹Since we are in the discrete-time setting, we do not have to worry about finite-time escapes.

assure closed-loop stability, the design and parametrization of the nominal control behavior, represented by the maps \hat{F} and Ψ , and the online adaptation strategy, denoted by the model selector S, have to fulfill the following requirements:

- Smooth parameterization maps for dynamics and CLMs: The set of system dynamics *F* = {*F*[ω] | ω ∈ Ω}, which describe the model uncertainty, is usually dictated by the specific problem setting and is therefore fixed. However, we have design freedom in how we choose to cover the space *F*, that is, the choice of (Ω, d) and the parametrization mapping *F* : Ω → *C*((*X* × *U*)^N, *X*^N) such that ⋃_{ω∈Ω} *F*[ω] = *F* is not unique. Moreover, for a given Ω and *F*, there is even more freedom in how we assign nominal-CLMs Ψ[ω] for each *F*[ω]. Conditions iii), iv), and v) of Theorem 23 state that the parameter space Ω and the maps *F*, Ψ should be designed so that the parametrization of nominal CLMs and dynamics is as smooth as possible over the space Ω. An intuitive notion of smoothness is presented in the form of continuity of Ψ and the map *D*, which parametrizes the composition of *F*.
- ℓ_∞-consistent selector S with convergent selection θ: Conditions i) and ii) of Theorem 23 impose intuitive design requirements for the model selector S. The operator S, i.e., the algorithm in charge of the adaptation, should always select parameters θ_t that are consistent with our observations up until time t (allowing for some small ℓ_∞-bounded error), and our selection strategy should be efficient in that it eventually settles on a fixed parameter θ_∞, i.e., S should guarantee the convergence of the sequence θ. As discussed in Part 2 of the thesis, designing such selectors, and even ones with stronger properties, is possible and requires us to consider the general problem of *Consistent Model Chasing*, which we discuss in Chapter 6.

CE-based adaptive control schemes are commonly used in problem settings concerned with learning optimal controllers [1, 5, 92, 94]. In such scenarios, it is common that each nominal CLM $\Psi[\omega]$ is chosen to be an optimal closed-loop map with respect to the specific optimal control problem (OCP). A general OCP can be succinctly defined by a dynamics operator $F \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\mathcal{X}})$ and a functional $\mathcal{J} : C(\ell^{\mathcal{W}}, \ell^{\mathcal{X} \times \mathcal{U}}) \mapsto \mathbb{R}^+, ^2$

²In our formulation, we always set $W = \mathcal{X}$.

as the following infinite-dimensional optimization problem

$$OCP[\mathcal{J}, \boldsymbol{F}]: \min_{\boldsymbol{\Phi}} \mathcal{J}(\boldsymbol{\Phi})$$
(4.17)
s.t.:
$$\boldsymbol{\Phi}^{x} = \boldsymbol{F}^{+}(\boldsymbol{\Phi}) + \boldsymbol{I}$$
$$\boldsymbol{\Phi} \in \mathcal{C}(\ell_{p}^{\mathcal{W}}, \ell_{q}^{\mathcal{X} \times \mathcal{U}})$$

over all stable closed-loop maps $\Phi \in C(\ell_p^{\mathcal{W}}, \ell_q^{\mathcal{X} \times \mathcal{U}})$ of F^+ , where p and q represent some specified notion of closed-loop stability. Correspondingly, the nominal CLM parametrization Ψ is a map $\Omega \ni \omega \mapsto \Phi^* \in \arg \min \operatorname{OCP}[\mathcal{J}, \hat{F}[\omega]]$ which corresponds dynamics $\hat{F}[\omega]$ with the respective optimal CLM solutions $\Phi^* \in \arg \min \operatorname{OCP}[\mathcal{J}, \hat{F}[\omega]]$. Moreover, if such solutions are unique, i.e., the set $\arg \min \operatorname{OCP}[\mathcal{J}, \hat{F}[\omega]]$ is a singleton, then the nominal CLM parametrization map Ψ is entirely determined by the OCP functional ${\mathcal J}$ and the parametrization of dynamics \hat{F} . In light of Theorem 23, establishing the smoothness conditions iv) and v) falls entirely on sensitivity analysis of the solutions to the optimal control problem $OCP[\mathcal{J}, \mathbf{F}[\omega'_{+\delta}]]$ to perturbations $\omega'_{+\delta} \in \mathsf{B}_{\delta}[\omega]$ around fixed parameters $\omega \in \Omega$. Thus, having a well-conditioned OCP formulation over the parametrization **F** is a prerequisite for using the optimal controllers for online adaptation. This is, again, states an intuitive requirement: Assume, for example, that for some $\omega_{\mathfrak{A}} \in \Omega$ the dynamics $\hat{F}[\omega_{\mathbf{g}}]$ lose u-controllability (in an essential way important to the problem), then we can expect $OCP[\mathcal{J}, \hat{F}[\omega_{\mathfrak{B}}]]$ to exhibit discontinuities in any neighborhood of $\omega_{\mathbf{g}}$. We explore this connection between the conditioning of optimal control problems and well-posedness for learning-to-control in the wellknown and important class of OCPs defined by LQ-functionals \mathcal{J} and LTI dynamics $\hat{F}: \Omega \to \mathcal{LC}(\ell^{\mathcal{W}}, \ell^{\mathcal{X} \times \mathcal{U}})$, where we assume Ω to be a compact subset of some finite-dimensional Euclidean space. The main result assumes \hat{F} to be affine and states that as long as the set of linear dynamics $\{\hat{F}\omega\}_{\omega\in\Omega}$ are all of sufficient and of equal degree of controllability, then the natural parametrization of LQ-optimal linear CLMs $\Psi : \Omega \ni \omega \mapsto \Phi^*[\omega] \in \mathcal{LC}(\ell^{\mathcal{W}}, \ell^{\mathcal{X} \times \mathcal{U}})$ fulfills the smoothness conditions iv) and v) of Theorem 23. Therefore, CE-adaptation with any model selector Sthat meets conditions i) and ii) of Theorem 23 yields stable learning-to-control in closed-loop (1).

4.3 Certainty Equivalent Adaptation with LQ-Optimal SL Controllers

LQR is a canonical and well-studied problem of optimal control [77, 85, 132]. In recent years, it has received revived attention in the context of learning and control problems [2, 37, 38, 92]. The general problem setup assumes an LTI system of

the form $x_k = Ax_k + Bu_{k-1} + w_k$, with state $x \in \mathbb{R}^n$, input $u \in \mathbb{R}^m$, i.i.d standard Gaussian disturbance $w_t \sim \mathcal{N}(0, I_n)$, and system matrices $A \in \mathbb{R}^{n \times n}$ and $B \in \mathbb{R}^{n \times m}$. Our objective is to find a linear causal controller $\mathbf{K} \in \mathcal{LC}(\ell^{\mathcal{U}}, \ell^{\mathcal{X}})$ that solves the optimization problem

$$\min_{\boldsymbol{K} \in \mathcal{LC}(\ell^{\mathcal{U}}, \ell^{\mathcal{X}})} \quad \limsup_{T \to \infty} \frac{1}{T} \sum_{k=0}^{T} \quad \mathbb{E}_{w_k^i \sim \mathcal{N}(0, I)} [x_t^\top C^\top C x_t + u_t^\top D^\top D u_t] \quad (4.18a)$$

s.t.:

$$x_k = Ax_{k-1} + Bu_{k-1} + w_k$$
, for $k \ge 1$ (4.18b)

$$u_k = K_k(x_{k:0}), \text{ for } k \in \mathbb{N}, \qquad x_0 = w_0, \qquad (4.18c)$$

where C and D are fixed matrices of full rank, i.e., $C^{\top}C > 0$, $D^{\top}D > 0$. As stated, it is well-known that the solution to the above problem is the static linear feedback control law $u_{lqr} : x \mapsto -K^*x$, where the gain matrix K^* is obtained by solving the Discrete Algebraic Ricatti-Equation (DARE). However, in applications with systems of large scale (such as, for example, the power grid), it is not feasible to implement the former in practice due to technical limitations in sensing, actuation, and communication. To account for that, one has to translate the former into constraints on the controller structure, which we then incorporate into the problem formulation. A common constraint is communication delay between sensors and actuators of different subsystems, which can be translated into spatial-temporal subspace constraints on the operator K, as shown in the example below:

Example. Assume we have p separate actuators and q different sensors that partition the input- and state-space into orthogonal subspaces as $\mathcal{V}_1 \oplus \cdots \oplus \mathcal{V}_p = \mathbb{R}^m$ and $\mathcal{U}_1 \oplus \cdots \oplus \mathcal{U}_q = \mathbb{R}^n$, respectively. Furthermore, let $V_i V_i^{\top}$ and $U_j U_j^{\top}$ denote orthogonal projection maps onto the subspaces \mathcal{V}_i and \mathcal{U}_j , and denote $\Pi_i^V = \mathbf{I}_m \otimes V_i V_i^{\top}$ and $\Pi_j^U = \mathbf{I}_n \otimes U_j U_j^{\top}$ their diagonal extensions to $\mathcal{LC}(\ell^{\mathcal{U}}, \ell^{\mathcal{U}})$ and $\mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X}})$, respectively. Now, if each actuator i receives the measurement from sensor j with d(i, j) time-steps delay, we can formulate these $i \cdot j$ number of constraints as:

$$\forall i, j, \tau : \qquad \mathbf{P}^{\tau} \mathbf{\Pi}_{i}^{V} \mathbf{K} \mathbf{\Pi}_{j}^{U} = \mathbf{P}^{\tau} \mathbf{\Pi}_{i}^{V} \mathbf{K} \mathbf{\Pi}_{j}^{U} \mathbf{P}^{\tau - d(i, j)}. \tag{4.19}$$

Adding constraints, however, such as (4.19) to the LQR problem (4.18), makes the problem hard to solve with traditional approaches. With the introduction of SLS [11, 128], this changed: By changing the search space from K to the CLMs Φ , we can formulate tractable subspace constraints on Φ , which are sufficient for $K = \Phi^u (\Phi^x)^{-1}$ to fulfill (4.18). A common approximation [11] is to restrict the CLM search to FIR maps of a fixed finite horizon t. Then, we obtain the following finite-dimensional QP problem for Φ : The resulting optimization problem takes the form:

$$LQ(A, B, t): \min \left\| \begin{bmatrix} C & 0 \\ 0 & D \end{bmatrix} \begin{bmatrix} R_1 & R_2 & \dots & R_t \\ M_1 & M_2 & \dots & M_t \end{bmatrix} \right\|_F^2$$
(4.20)
s.t.: $R_1 = I$
 $R_{k+1} = AR_k + BM_k, \quad \forall \ k: 1 \le k \le t$
 $R_{t+1} = 0$

where $R_k \in \mathbb{R}^{n \times n}$ and $M_k \in \mathbb{R}^{m \times n}$ represent matrices associated with the component functions of cCLMs

$$\Phi_k^x(w_{k:0}) := \sum_{j=1}^{\min\{t,k+1\}} R_j w_{k+1-j} \qquad \Phi_k^u(w_{k:0}) = \sum_{j=1}^{\min\{t,k+1\}} M_j w_{k+1-j}.$$
(4.21)

For fixed A, B, and horizon t, the optimal CLMs $\Phi^*(A, B, t) \in \mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$, exist if [A, B] is t-controllable (i.e., it is possible to drive the system state from any initial condition to the origin within t time-steps) and are unique if we assume that C, D, and B have a trivial null-space.

LQ-Optimal CLMs for Nominal SL Control

Motivated by the scalability of SLS in complex system settings, such as large-scale systems, we investigate the LQ-optimal SL controllers of the problem (4.20) for CE-based learning and control.

Setup. We assume that we are given an uncertainty set of LTI dynamics, described by a compact metric space (Ω, d) of LTI systems, and an affine parameterization map \hat{F} : $\Omega \to \mathcal{LC}(\ell^{\chi \times U}, \ell^{\chi})$, where each operator $\hat{F}[\omega]$ is linear, diagonal and has component functions $\{\hat{F}_k[\omega] : (x_{k:0}, u_{k:0}) \mapsto f[\omega](x_k, u_k)\}_{k \in \mathbb{N}}, f[\omega] : (x, u) \mapsto A[\omega]x + B[\omega]u$ for some fixed continuous matrix-valued functions $A : (\Omega, d) \to (\mathbb{R}^{n \times n}, |\cdot|)$ and $B : (\Omega, d) \to (\mathbb{R}^{m \times n}, |\cdot|)$. We consider a *fixed horizon* t and let the nominal CLM parameterization $\Psi : \Omega \to \mathcal{LC}(\ell^{\chi}, \ell^{\chi \times U})$ assign to each parameter $\omega \in \Omega$ the LQ-optimal CLM $\Psi[\omega] := \Phi^*(A[\omega], B[\omega], t)$ w.r.t. to the optimization problem LQ $(A[\omega], B[\omega], t)$ described by (4.20) and (4.21). We leave the model selector $S \in C(\ell^{\chi \times U}, \ell^{\Omega})$ unspecified and assume that it is ℓ_{∞} consistent and guarantees a converging parameter selection $\theta \in \ell^{\Omega}$. As we discuss in detail in Chapter 6 in Part 2 of the thesis, this assumption is justified, as we can tackle the design of S in *isolation* from the question of nominal control design, as a problem instance of *Consistent Model Chasing*.

Algorithm 1 Learning-to-Control with LQ-optimal SL Controllers

Setup: model selector $S \in C(\ell^{\mathcal{X} \times \mathcal{U}}, \ell^{\Omega})$, compact parameter space (Ω, d) , continuous maps $A: \Omega \to \mathbb{R}^{n \times n}$ and $B: \Omega \to \mathbb{R}^{m \times n}$, specified LQ(A', B', t) with fixed invertible $C, D, t \in \mathbb{N}$ and online solver $\Phi^*(A', B', t)$, CLM parameterization $\Psi : \omega \mapsto$ $\Phi^*(A[\omega], B[\omega], t)$ **Initialization:** choose some $\theta_0 \in \Omega$ 1: for k = 0, 1, ... to ∞ do 2: observe x_k $\theta_k \leftarrow S_k(x_{k:0}, u_{k-1:0})$ \succ update model to $\theta_k \in \mathbf{\Omega}$ 3: solve $LQ(A[\theta_k], B[\theta_k], t)$ from (4.20) ▷ Solve LQ-OCP 4: Set $\Psi[\theta_k] \leftarrow \Phi^*(A[\theta_k], B[\theta_k], t)$ from (4.21) ▷ Synthesize optimal CLM 5: $\hat{w}_k \leftarrow x_k - \Psi_k^x[\theta_k](0, \hat{w}_{k-1:0})$ ⊳ adapt SL-controller 6: $u_k \leftarrow \Psi_k^u[\theta_k](\hat{w}_{k:0})$ 7: apply action u_k 8: 9: end for

Remark. $\hat{F}[\omega]$ can be expressed as $\sum_{k=0}^{\infty} U_n^{*[k]} A[\omega] U_n^{[k]} + U_n^{*[k]} B[\omega] U_m^{[k]}$.

The CE-Adaptive Control Algorithm. The learning-to-control algorithm is described in Algorithm 1 and naturally inherits many of the benefits and features of basic System Level Synthesis. [66] first recognized the importance of SL controllers for online learning and control and showed that by extending the system-level robustness analysis of [127], [11] to linear time-varying systems, one can obtain scalable and easy-to-analyze adaptive control methods via system-level controllers. A key difference from [66] is that in [66], the parameterization Ψ consisted of candidate CLMs synthesized to be robust to a limited amount of parametric uncertainty and, therefore, closed-loop stability was only guaranteed if Ω was not too large. As illustrated in Algorithm 1, in each time step k we select the system matrices $A_k = A[\theta_k], B_k = B[\theta_k]$ and synthesize an LQ-optimal LQ controller by solving the problem $LQ(A_k, B_k, t)$ presented in (4.20). Note that whether or not the parameterization Ψ is available offline or computed on demand online does not make any difference for closed-loop stability analysis. As described in detail in [11], we can add – at no cost – linear constraints to ensure that the controller implementation is aligned with real-world restrictions such as delay and sparsity of sensing, actuation and communication. Almost all attractive features of the SLS approach [11] carry over into our online learning and control setting.

Closed-Loop Dynamics. Appealing to our previous discussion and derivation of Theorem 22 and Lem. 22, the closed-loop dynamics are described by the following

set of equations:

$$\forall k \in \mathbb{N} : \quad x_k = \Psi_k^x [\theta_k](\hat{w}_{k:0}) \tag{4.22a}$$

$$\theta_k = S_k(x_{k:0}, u_{k-1:0}) \tag{4.22b}$$

$$e_k = x_k - A[\theta_k] x_{k-1} - B[\theta_k] u_{k-1} + w_k$$
(4.22c)

$$\hat{w}_{k} = \underbrace{f[\theta_{k-1}](\Psi_{k-1}[\theta_{k-1}] - \Psi_{k-1}[\theta_{k}])}_{U^{*[k]}\Delta^{+}_{|\theta}U^{[k-1]}}(\hat{w}_{k-1:0}) + e_{k}$$
(4.22d)

$$u_k = \Psi_k^u[\theta_k](\hat{w}_{k:0}) \tag{4.22e}$$

where $\Psi : \Omega \ni \omega \mapsto \Phi^*(A[\omega], B[\omega], t)$ and f represents the linear, time-invariant transition function $f[\omega] : (x, u) \mapsto A[\omega]x + B[\omega]u$. Let $\mathsf{M}^{*\infty}_{\Omega} \subset \ell^{\mathcal{X} \times \mathcal{U} \times \mathcal{W} \times \mathcal{X} \times \mathcal{W}}$ denote the model set of the above closed-loop and $\mathsf{M}^{\infty}_{\Omega} \subset \ell^{\mathcal{X} \times \mathcal{U} \times \mathcal{W}}$ the model of the uncertain open-loop dynamics for bounded disturbance w:

$$\mathsf{M}_{\Omega}^{*\infty} := \{ \boldsymbol{\tau}^* = (\boldsymbol{x}, \boldsymbol{u}, \hat{\boldsymbol{w}}, \boldsymbol{e}, \boldsymbol{w}) \text{ s.t.: } (4.22) \text{ and } \boldsymbol{w} \in \ell_{\infty}^{\mathcal{W}} \}$$
(4.23)
$$\mathsf{M}_{\Omega}^{\infty} := \{ \boldsymbol{\tau} = (\boldsymbol{x}, \boldsymbol{u}, \boldsymbol{w}) \text{ s.t.: } \boldsymbol{x} = \hat{\boldsymbol{F}}[\omega](\boldsymbol{x}, \boldsymbol{u}) + \boldsymbol{w} \text{ for some } \omega \in \Omega, \boldsymbol{w} \in \ell_{\infty}^{\mathcal{W}} \}.$$
(4.24)

Investigating closed-loop stability in the ℓ_{∞} -BIBO sense now means proving that all trajectories $\tau^* \in \mathsf{M}^{*\infty}_{\Omega}$ are bounded, i.e., that $\mathsf{M}^{*\infty}_{\Omega}$ is a subset of $\ell_{\infty}^{\mathcal{X} \times \mathcal{U} \times \mathcal{W} \times \mathcal{X} \times \mathcal{W}}$.

 ℓ_{∞} -BIBO-Stability of Closed-Loop Dynamics. We appeal to the conditions of Theorem 23 for stability analysis of the closed-loop dynamical system. Since we assumed that S guarantees $e \in \ell^{\infty}$, $\exists \theta_{\infty}$ s.t.: $\lim_{t\to\infty} \theta_t = \theta_{\infty}$ and we have $\Delta[\hat{F}[\omega], \Psi[\omega]] = 0, \forall \omega \in \Omega$ due to $\Psi[\omega] := \Phi^*(A[\omega], B[\omega], t)$, verifying stability reduces down to checking iv) and v). Furthermore, it can be shown that in our setup iv) actually implies v), and therefore, checking the continuity of the CLM parameterization map Ψ over Ω is sufficient for verifying both conditions. This is easily seen by first rewriting the difference $\hat{F}[p_1]\Psi[p_2] - \hat{F}[\omega]\Psi[\omega]$ as

$$\hat{\boldsymbol{F}}[p_1]\boldsymbol{\Psi}[p_2] - \hat{\boldsymbol{F}}[\omega]\boldsymbol{\Psi}[\omega]$$

$$= \hat{\boldsymbol{F}}[p_1]\boldsymbol{\Psi}[p_2] - \hat{\boldsymbol{F}}[p_1]\boldsymbol{\Psi}[p_1] + \hat{\boldsymbol{F}}[p_1]\boldsymbol{\Psi}[p_1] - \hat{\boldsymbol{F}}[\omega]\boldsymbol{\Psi}[\omega]$$

$$= \hat{\boldsymbol{F}}[p_1](\boldsymbol{\Psi}[p_2] - \boldsymbol{\Psi}[p_1]) + \boldsymbol{\Psi}^{\mathrm{x}}[p_1] - \boldsymbol{\Psi}^{\mathrm{x}}[\omega], \qquad (4.25)$$

where the last equation follows from $\Delta[\hat{F}[\omega], \Psi[\omega]] = 0$ and the linearity of \hat{F} , and second by noticing that the $\|\cdot\|_{\infty}$ -norm of $\hat{F}[p_1](\Psi[p_2] - \Psi[p_1])$ is bounded above as

$$\| \hat{F}[p_1](\Psi[p_2] - \Psi[p_1]) \|_{\infty} \leq c_{\Omega} \cdot \| \Psi[p_2] - \Psi[p_1] \|_{\infty}$$
(4.26)

where $c_{\Omega} = \max_{p \in \Omega} \{|A[p]| + |B[p]|\}$; the maximum c_{Ω} exists because (Ω, d) is a compact metric space and the parameterizations A and B are continuous maps on (Ω, d) . Hence, from (4.25) and (4.26) and since A and B were already assumed to be continuous, the stability conditions iv) and v) would be satisfied if $\Phi^*(A, B, t)$ is continuous in A and B. Therefore, we need to investigate the sensitivity of LQ optimal CLM solutions $\Phi^*(A + \partial A, B + \partial B, t)$ with respect to small matrix perturbations $\partial A \in \mathbb{R}^{n \times n}$, $\partial B \in \mathbb{R}^{n \times m}$. This perturbation analysis constitutes the second part of this chapter and culminates in the theorem stated below:

Theorem 24 (Lipschitzness of LQ-optimal CLM oracle). Let $t \in \mathbb{N}$ be fixed, $C \in \mathbb{R}^{n \times n}$, $D \in \mathbb{R}^{m \times m}$ be fixed invertible matrices, and $\Phi^*(A, B, t)$ represent the unique optimum of the optimal control problem LQ(A, B, t) described by (4.20), (4.21). Assume that S_{ab} is a compact subset of $\mathbb{R}^{n \times (n+m)}$ such that each pair $[A, B] \in S_{ab}$ is t-controllable, i.e., the matrix $P_t(A, B) = [A^{t-1}B, A^{t-2}B, \cdots, B] \in \mathbb{R}^{n \times mt}$ is of rank n. Then there exists a fixed positive constant $L \in \mathbb{R}^+$ such that for all $[A_1, B_1], [A_2, B_2] \in S_{ab}$ holds

$$\|\mathbf{\Phi}^*(A_1, B_1, t) - \|\mathbf{\Phi}^*(A_2, B_2, t)\|_{\infty} \le L(|\partial_{12}A| + |\partial_{12}B|)$$

where $\partial_{12}A := A_1 - A_2$ and $\partial_{12}B := B_1 - B_2$.

The above result, which will be stated in much greater detail later, states that the map $\Phi^*(A', B', t)$ is Lipshitz-continuous over compact sets of *t*-controllable pairs of matrices [A', B'], and therefore our CLM parametrization map Ψ is indeed continuous over (Ω, d) . Hence, in summary, as long as all system matrix pairs $\{[A[\omega], B[\omega]]\}_{\omega \in \Omega}$ are *t*-controllable, all stability conditions imposed on the nominal control design, i.e., iii), iv), v) of Theorem 23, are met, and the closed-loop system (4.22) is ℓ_{∞} -BIBO stable (with a fitting choice of S). We summarize our discussion in the theorem stated below:

Theorem 25 (ℓ_{∞} -stability of Closed-Loop Alg.1). *Recall the Closed-Loop setup* described in Algorithm 1 and the corresponding closed-loop equations (4.22) and sets $M_{\Omega}^{*\infty}$ and M_{Ω}^{∞} of closed-loop (4.23) and open-loop (4.24) trajectories, respectively. All closed-loop trajectories $\tau \in M_{\Omega}^{*\infty}$ are bounded if the following conditions are met:

1. If $\boldsymbol{\theta} = \boldsymbol{S}(\boldsymbol{x}, \boldsymbol{u})$, where $(\boldsymbol{x}, \boldsymbol{u}, \boldsymbol{w}) \in \mathsf{M}_{\Omega}^{\infty}$, then $\boldsymbol{\theta}$ converges in (Ω, d) and $\sup_{k \in \mathbb{N}} |x_k - A[\theta_k] x_{k-1} - B[\theta_k] u_{k-1}| < \infty.$

2. Each pair of matrices $[A[\omega], B[\omega]], \omega \in \mathbf{\Omega}$ is t-controllable.

Proof. If the model selector S satisfies the consistency Assumption (1), then the conditions (i)) and (ii)) of Theorem 23 hold for any sequence of selections θ and consistency errors e of a closed-loop trajectory $(x, u, \hat{w}, e, w) \in \mathsf{M}^{*\infty}_{\Omega}$. Condition Theorem 23(iii)) is automatically satisfied, since $\Psi[\omega] = \Phi^*(A[\omega], B[\omega], t)$ are always CLMs of the corresponding dynamics $\hat{F}[\omega]$. Since (Ω, d) is compact and A and B are continuous maps, it follows that the image $S_{\Omega} = \{[A[\omega], B[\omega]] \mid \omega \in \Omega\}$ is a compact subset of $\mathbb{R}^{n \times (n+m)}$. By Assumption (2), we also conclude that all pairs $[A', B'] \in S_{\Omega}$ are t-controllable, and therefore, by Lipshitzness of the LQ-optimal CLM map Φ^* (according to Theorem 24) and continuity of A and B, we conclude that Ψ is continuous over (Ω, d) , and therefore condition Theorem 23(iv)) is met. Moreover, as derived earlier via (4.25) and (4.26), the former implies also that condition Theorem 23(v)) is true. Finally, we invoke Theorem 23 and conclude that all closed-loop trajectories $(x, u, \hat{w}, e, w) \in \mathsf{M}^{*\infty}_{\Omega}$ have to be bounded.

The conditions stated above show that closed-loop stability is largely characterized by the properties of the model selection S. Moreover, it is important to notice that it is nowhere required that the selections θ converge to the true system parameter θ^* , merely that it converges to some $\theta_{\infty} \in \Omega$. This is a major distinction from most modern learning and control algorithms [3, 35, 50] which explicitly require accurate system identification in order to guarantee stability. In Part 2 of the thesis, we will discuss how to define such procedures S through Consistent Model Chasing. Moreover, aside from convergence, stronger properties (competitiveness) can be obtained for S which allow for sharper analysis of the transient and cost performance of the closed-loop. A key component towards establishing these types of guarantees is proving that the CE-LQ optimal CLMs Φ change indeed gradually with perturbations in θ . Sensitivity analysis of the (*oracle*) map $\Phi : \Omega \to \mathcal{LC}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ is an important part of what we later refer to as robust oracle design. In the remaining part of this chapter, we will prove the previously claimed Lipschitz property of the optimal CLM parametrization map $\Phi^*(A', B', t)$ and give analytic bounds of the Lipschitz constant L in terms of system-theoretic properties such as controllability and observability. We start the second part of this chapter with a discussion on basic control theory results regarding controllability, observability, and grammians, which we refine for the use-case of controlling large-scale systems.

4.4 Digression on Linear Algebra and Control Theory Basics

Here we discuss some basic linear algebra results that are frequently used in our arguments but are not always mentioned explicitly in our derivations. For the following definitions, let M denote an arbitrary matrix $M \in \mathbb{R}^{n \times m}$, where w.l.o.g. $n \ge m$.

Reduced SVD Decomposition

The factorization $M = U_r \Sigma_r V_r^{\top}$, $U_r \in \mathbb{R}^{n \times k}$, $\Sigma_r \in \mathbb{R}^{k \times k}$, $V^{\top} \in \mathbb{R}^{k \times m}$ is called a *reduced* SVD decomposition of M, if $U_r^{\top} U_r = I_k$, $\Sigma_r > 0$ is positive-definite and diagonal, and $V_r^{\top} V_r = I_k$. The diagonal entries of Σ_r are the non-zero singular values of M. We denote $\underline{\sigma}(M) \in \mathbb{R}^n$ as the vector $[\sigma_1, \sigma_2, \ldots, \sigma_k, \sigma_{k+1}, \ldots, \sigma_n]^{\top}$ of singular values of M, ordered in descending order; the diagonal entries (up to some permutation) of Σ_r are $\sigma_1, \ldots, \sigma_k$, while $\sigma_{k+1} = \cdots = \sigma_n = 0$. M is invertible if and only if the reduced SVD is such that k = n. Correspondingly, $\sigma_1 = \sigma_{\max}(M) = ||M||_2$, $\sigma_n = \sigma_{\min}(M)$ and we define $\sigma_{-1} := \sigma_k(M)$ as the smallest *non-zero* singular value of M. If M is invertible, then $\sigma_{-1}(M) = \sigma_{\min}(M)$ and vice versa.

Moore-Penrose Inverse

The Moore-Penrose inverse of M can be uniquely defined as $M^{\dagger} = V_r \Sigma_r^{-1} U_r^{\top}$ for any reduced SVD decomposition of M. It is also the unique matrix M^{\dagger} satisfying all of the following four conditions:

$$MM^{\dagger}M = M \qquad \qquad M^{\dagger}MM^{\dagger} = M^{\dagger}$$
$$(MM^{\dagger})^{\top} = MM^{\dagger} \qquad \qquad (M^{\dagger}M)^{\top} = M^{\dagger}M.$$

Here some of the key-properties we use of M^{\dagger} :

Lemma 19. For any matrix M and its pseudo-inverse M^{\dagger} holds:

- *i)* MM^{\dagger} and $M^{\dagger}M$ are orthogonal projections with rank k.
- *ii)* M^{\dagger} *is the (left/right)-inverse of* M *if and only if* M *is (left/right)-invertible.*
- *iii)* $M^{\dagger} = \lim_{t \searrow 0} (M^{\top}M + tI)^{-1}M^{\top}$ and $M^{\dagger}x = x^{\dagger} := \lim_{t \searrow 0} x_t^{\dagger}$ for any x where $x_t^{\dagger} := (M^{\top}M + tI)^{-1}M^{\top}x$ denotes the unique solution to the optimization problem

$$\min_{v} \|Mv - x\|_{2}^{2} + t\|v\|_{2}^{2}.$$

iv) The optimal value of the least-squares problem $\min_{v} ||Mv - x||_{2}^{2}$ is attained at $v^{*} = M^{\dagger}x$ and if there is more than one minimizer, then v^{*} is the unique one

of smallest $\| \|_2$ -norm.

v) $||M^{\dagger}||_{2}^{-1} = \sigma_{-1}(M) \text{ and } ||M||_{2}^{-1} = \sigma_{-1}(M^{\dagger}).$ vi) $M^{\dagger} = (M^{\top}M)^{\dagger}M^{\top} \text{ and } M^{\dagger} = M^{\top}(MM^{\top})^{\dagger}.$

4.5 Controllability and Observability at a Fixed Time

Large-scale systems often consist of many sparsely interconnected small systems, i.e., the dimension of the large system is $n = Nn_S$, where n_S denotes the typical dimensions of the small systems, and N denotes the number of subsystems. Denote $x_t^{(i)} \in \mathbb{R}^{n_S}$ as the state of the *i*th subsystem and $u^{(i_1)\top}, u^{(i_2)\top}, \ldots, u^{(i_M)\top}$ denote the inputs of the M (< N) systems i_1, i_2, \ldots, i_M which have actuation available. Let $x = [x^{(1)\top}, x^{(2)\top}, \ldots, x^{(N)\top}]$ denote the state of the overall interconnected system and $u = [u^{(i_1)\top}, u^{(i_2)\top}, \ldots, u^{(i_M)\top}] \in \mathbb{R}^m$ the total input of dimension $m = Mm_S$, where m_S is the dimension of the inputs of the subsystems. Typically, the large dimension of the overall system is due to N and M being large numbers, i.e., $n \gg n_S$, $m \gg m_S$.

If the overall system is controllable, we know that any initial condition x_0 can be controlled to the origin within at most n time-steps. However, in the above case $n = Nn_S$, this is often a conservative statement: If communication between subsystems is fast enough, interconnection is sparse, and actuation is available in sufficiently many subsystems (M is on the order of N), it is possible to control arbitrary initial conditions to the origin within much fewer steps $t \ll n$.

Motivated by applications in large-scale control settings, we refine the notion of controllability for this purpose. We shall call [A, B] to be *t*-controllable if within *t* time-steps we can control any initial condition x_0 to the origin - or equivalently we can reach any target state x_f at time *t* - regardless of the initial condition. We adopt basic controllability results to the finite time setting below:

Lemma 20. The following statements are equivalent:

- [A, B] is t-controllable.
- For any pair $\zeta_0, \zeta_f \in \mathbb{R}^n$, there exists $u_0, u_1, \ldots, u_{t-1}$ such that $x_0 = \zeta_0$, $x_t = \zeta_f$, where

$$x_{k+1} = Ax_k + Bu_k, \forall \quad k \le t.$$

• The matrix $P_t = [A^{t-1}B, A^{t-2}B, \dots, B] \in \mathbb{R}^{n \times tm}$ is full row rank.

The fourth condition gives us a necessary condition for *t*-controllability:

$$t > \frac{n}{m} = \frac{N}{M} \frac{n_S}{m_S}.$$

Similarly, we define that a pair (C, A) is t-observable if the sequence of observations $y_0, \ldots, y_{t-1}, y_k = Cx_k$ of an impulse response $x_k = A^k \xi_0$ are sufficient to compute the initial condition ξ_0 . As expected, t-observability has multiple equivalent definitions, which are dual to those of controllability.

Lemma 21. The following statements are equivalent:

- (C, A) is t-observable.
- Let $y_{[0:t-1]}^1$, $y_{[0:t-1]}^2$ be observations $y_k^i = CA^k\zeta_i$, of the state-trajectories $x_k^1 = CA^k\zeta_1$ and $x_k^2 = CA^k\zeta_2$, respectively. Then it holds:

$$\boldsymbol{y}_{[0:t-1]}^1 = \boldsymbol{y}_{[0:t-1]}^2 \Leftrightarrow \zeta_1 = \zeta_2.$$

• The matrix $Q_t = [C^{\top}, (CA)^{\top}, \dots, (CA^{t-1})^{\top}]^{\top}$ is full column rank.

Controllability and Observability Grammians

The relationship between controllability and observability and their corresponding grammians, defined below, will play an important role in our derivations. In contrast to standard literature, we discuss this interplay also in the setting of partial observability and partial controllability. To this end, we make use of the Moore-Penrose inverse.

Definition 4.2 (controllability/observability grammians). Let $C \in \mathbb{R}^{h \times n}$, $A \in \mathbb{R}^{n \times n}$, $B \in \mathbb{R}^{n \times m}$ be fixed system matrices. The sequence of positive-semidefinite matrices $\{W_t^c\}$ and $\{W_t^o\}$ are called controllability and observability grammians of the system $x_{k+1} = Ax_k + Bu_k$, $y_k = Cx_k$, respectively, if they satisfy the following set of dynamic equations:

$$W_0^c = BB^{\top}, \quad W_t^c = AW_{t-1}^c A^{\top} + BB^{\top},$$
 (4.27)

$$W_0^o = C^{\top}C, \quad W_t^o = A^{\top}W_{t-1}^oA + C^{\top}C.$$
 (4.28)

By substitution, we obtain the explicit form of W_t^c and W_t^o :

$$W_t^c \text{ s.t. (4.27)} \iff W_t^c = P_t P_t^\top = \sum_{k=0}^t A^k B B^\top A^{k\top}$$
$$W_t^o \text{ s.t. (4.28)} \iff W_t^o = Q_t^\top Q_t = \sum_{k=0}^t A^{k\top} C^\top C A^k.$$

It is easy to see, using standard arguments, that the controllability grammians obey the following properties: **Lemma 22.** Let $\{W_t^c\}$ be the controllability grammians of [A, B]. Then, for any $t \ge 1$,

- 1. [A, B] is t-controllable if and only if $W_t^c > 0$.
- 2. If [A, B] is t-controllable, then, given some target state ζ_f and initial condition $x_0 = 0$, we can compute the optimal (unique) input sequence that drives the system to $x_t = \zeta_f$ with minimal cost $\|\boldsymbol{u}\|_2^2 = \sum_{k=0}^{t-1} \|u_k\|_2^2$ as

$$\boldsymbol{u}^{*} = P_{t}^{\top} W_{t}^{-1} \zeta_{f}, \text{ where } \boldsymbol{u}^{*} = [u_{0}^{*\top}, u_{1}^{*\top}, \dots, u_{t-1}^{*\top}]^{\top}.$$
(4.29)

The optimal cost is $\|\boldsymbol{u}^*\|_2^2 = \zeta_f^\top W_t^c \zeta_f$.

More generally, for a given ζ_f ∈ ℝⁿ, x^{*}_t = P_tP[†]_tζ_f is the state closest in 2-norm among all states reachable within t time-steps, starting from the origin. Among all input sequences that reach x^{*}_t at time t, u^{*} = P^T_t(W^c_t)[†]ζ_f is the one with the smallest ℓ₂-norm. Furthermore, it holds ||u^{*}||²₂ = ζ_fW^c_tζ_f.

Proof. The first statement is obvious after recalling (4.27). For the second, notice that the desired optimal control problem reduces to the least squares problem:

$$\min_{\boldsymbol{u}} \quad \|\boldsymbol{u}\|_2^2$$

s.t.: $\zeta_f = P_t \boldsymbol{u}$.

Since [A, B] is *t*-controllable, it holds that P_t is full-column rank and $W_t = P_t P_t^\top > 0$, and the above problem is feasible for any $\zeta_f \in \mathbb{R}^n$. The unique solution is $u^* = P_t^\top (P_t P_t^\top)^{-1} \zeta_f = P_t^\top W_t^{-1} \zeta_f$. The last statement follows by recalling the properties of the pseudo-inverse.

Lemma 23. Let $\{W_t^o\}$ be the observability grammians of (C, A). Then, for any $t \ge 1$,

- 1. (C, A) is t-observable if and only if $W_t^o > 0$.
- 2. If (C, A) is t-observable, then for any sequence of observations $y_{[0:t-1]}$, $y_t = Cx_t$ of an impulse response $x_t = A^t \xi_0$, it holds that $\xi_0 = (W_t^o)^{-1} Q_t^\top y_{[0:t-1]}$. Moreover, for an arbitrary $\hat{y}_{[0:t-1]}$, $\hat{\xi}_0 = (W_t^o)^{\dagger} Q_t^\top \hat{y}_{[0:t-1]}$ is the initial condition ³ which produces the closest feasible impulse response $y_{[0:t-1]} = Q_t \hat{\xi}_0$, i.e., $\|y - \hat{y}\|_2^2 = \min_{\xi} \sum_{k=0}^{t-1} \|\hat{y}_t - CA^k \xi\|_2^2 = \hat{\xi}_0^\top W_T^o \xi_0$.
- 3. $\mathcal{R}(W_t^o)$ is the subspace of observable states in \mathbb{R}^n at time t.

³And is of smallest 2-norm if there are multiple such initial conditions.

4.6 Perturbation Analysis of Open-Loop Maps over Finite Time-Horizon

In this section, we discuss how linear system behavior changes with different system matrices A, B under *feedforward/open-loop* control. We consider the following LTI system with state $x \in \mathbb{R}^n$, control input $u \in \mathbb{R}^m$, disturbance $w \in \mathbb{R}^n$, and the two outputs $y \in \mathbb{R}^n$ and $z \in \mathbb{R}^{n+m}$:

$$x_k = A_i x_{k-1} + B_i u_{k-1} + w_{k-1}, \qquad z_k = \begin{bmatrix} C x_k \\ D u_k \end{bmatrix} \qquad y_k = C x_k.$$
 (4.30)

We consider the output matrices C and D to be fixed and invertible, and B_i to always be of full column rank. For a given pair A_i , B_i and a fixed horizon t, we define the open-loop map G_t as the causal map $G_t(A_i, B_i)(w, u)_{[0:t]} \mapsto (x, y)_{[0:t]}$ between the input signals w, u and the state and output signals x, y. Our goal is to analyze the sensitivity of the map $G_t(A_i, B_i)$ with respect to changes in (A_i, B_i) .

Setup and Notation

We represent the system (4.30) in batch form: Let \underline{x}_t denote the stacked vector $[x_0^{\top}, x_1^{\top}, \dots, x_t^{\top}]^{\top}$, and define $\underline{u}_t, \underline{w}_t, \underline{y}_t, \underline{z}_t$ accordingly. Let $\boldsymbol{E}_k^{\top} \in \mathbb{R}^{n \times (t+1)n}$, for $k \in \{0, 1, \dots, t\}$, denote the map $\boldsymbol{E}_k^{\top} : \underline{x}_t \mapsto x_k$, i.e., \boldsymbol{E}_k is a block-column matrix containing all zero matrices $\boldsymbol{0}_{n \times n}$, except for the k-th block row, which contains the identity matrix I_n :

We represent the open-loop map $G_t : (\underline{w}_t, \underline{u}_t) \mapsto (\underline{x}_t, \underline{y}_t)$ as a Toeplitz matrix, and, as shown below, we decompose it into block matrices, each representing a partial open-loop map:

Now we can express the relationship between the system trajectories over the time-horizon [0, t] as:

$$\underline{x}_{t} = \mathbf{Z}_{t}^{+} \mathbf{G}_{t}^{xu}(A, B) \underline{u}_{t} + \mathbf{G}_{t}^{xw}(A) \underline{w}_{t}
\underline{y}_{t} = \mathbf{Z}_{t}^{+} \mathbf{G}_{t}^{yu}(C, A, B) \underline{u}_{t} + \mathbf{G}_{t}^{yw}(C, A) \underline{w}_{t} \qquad \underline{z}_{t} = \begin{bmatrix} C \mathbf{G}_{t}^{yw}(\underline{w}_{t}) + C \mathbf{G}_{t}^{yu}(\underline{u}_{t}) \\ D \underline{u}_{t} \end{bmatrix}
D \underline{u}_{t} \qquad (4.32)$$

where Z_t^+ denotes the delay operator, and:

$$\boldsymbol{G}_{t}^{xw}(A) = \begin{bmatrix} I_{n} & 0 & 0 & \dots & 0 \\ A & I_{n} & 0 & \dots & 0 \\ A^{2} & A & I_{n} & \dots & 0 \\ \dots & \dots & \dots & \dots \\ A^{t} & A^{t-1} & A^{t-2} & \dots & I_{n} \end{bmatrix} \quad \boldsymbol{G}_{t}^{xu}(A,B) = \boldsymbol{G}_{t}^{xw}(A)(I_{t+1} \otimes B)$$

$$\boldsymbol{G}_{t}^{yw}(C,A) = (I_{t} \otimes C)\boldsymbol{G}^{xw}(A) \quad \boldsymbol{G}_{t}^{yu}(C,A,B) = (I_{t} \otimes C)\boldsymbol{G}^{xu}(A,B)$$

$$(4.33)$$

 $\boldsymbol{G}_{t}^{yw}(C,A) = (I_{t+1} \otimes C)\boldsymbol{G}_{t}^{xw}(A) \qquad \boldsymbol{G}_{t}^{yu}(C,A,B) = (I_{t+1} \otimes C)\boldsymbol{G}_{t}^{xu}(A,B).$

We denote the k-th controllability matrix and grammian with respect to the pair (A, B) as $P_k(A, B)$ and $W_k^c(A, B)$, respectively. Similarly, we define the k-th observability matrix and grammian of a pair (C, A) as $Q_k(C, A)$ and $W_t^o(C, A)$:

We refer to the rank of $P_k(A, B)$ and $Q_k(C, A)$ as the degree of k-controllability and k-observability of the system (C, A, B). We quantify the level of controllability and observability in terms of the singular values ⁴ of the matrices $P_k(A, B)$ and $Q_k(C, A)$: $\bar{\sigma}_t^c(A, B)$ and $\bar{\sigma}_t^o(C, A)$ denote the largest singular eigenvalues of $P_k(A, B)$ and $Q_k(C, A)$, while $\underline{\sigma}_t^c(A, B)$ and $\underline{\sigma}_t^o(C, A)$ denote the smallest *non-zero* ones. We summarize these definitions below:

Definition 4.3. For a fixed set of parameters (A, B, C) and time-horizon k, we define $\bar{\sigma}_k^c(A, B)$, $\underline{\sigma}_k^c(A, B)$, $\bar{\sigma}_k^o(C, A)$, and $\underline{\sigma}_k^o(C, A)$ as:

$$\bar{\sigma}_k^c(A,B) := \|P_k(A,B)\|_2 \quad \underline{\sigma}_k^c(A,B) := \sigma_{-1}(P_k(A,B)) = \|P_k^{\dagger}(A,B)\|_2^{-1}$$
$$\bar{\sigma}_k^o(C,A) := \|Q_k(C,A)\|_2 \quad \underline{\sigma}_k^o(C,A) := \sigma_{-1}(Q_k(C,A)) = \|Q_k^{\dagger}(C,A)\|_2^{-1}$$

⁴Which coincide with the eigenvalues of the grammians.

Moreover, we define $\bar{\rho}_k^{c/o}(A)$, $\underline{\rho}_k^{c/o}(A)$ as:

$$\bar{\rho}_{k}^{c}(A) := \|P_{k}(A, I)\|_{2} \qquad \qquad \underline{\rho}_{k}^{c}(A) := \sigma_{-1}(P_{k}(A, I))$$
$$\bar{\rho}_{k}^{o}(A) := \|Q_{k}(I, A)\|_{2} \qquad \qquad \rho_{k}^{o}(A) := \sigma_{-1}(Q_{k}(I, A)).$$

The quantities $\bar{\rho}_k^{c/o}(A)$ and $\underline{\rho}_k^{c/o}(A)$ measure the controllability and observability of a fictitious system, where *B* and *C* are replaced with the identity matrix *I*, while *A* is being kept the same:

$$x_{k+1} = Ax_k^* + u_k^* \qquad \qquad y_k^* = x_k^*$$

 $\bar{\rho}_k^{c/o}(A)$ and $\underline{\rho}_k^{c/o}(A)$ represent the largest and smallest non-zero singular values of the controllability and observability matrix $P_k(A, I)$ and $Q_k(I, A)$ and can be interpreted as describing the attainable level of controllability and observability for the system matrix A, if we had the ability to change the input and output matrix.

Remark 26. If we drop mentioning the dependence on t and/or the parameters (A, B, C) in some statement, it is assumed to hold for all t and/or parameters (A, B, C). For example, if we state, " \mathbf{G}^{xw} satisfies ...", we implicitly mean "For all A, B and t, $\mathbf{G}_t^{xw}(A, B)$ satisfies ...".

Since $W_t^c = P_t P_t^{\top}$ and $W_t^o = Q_t^{\top} Q_t$, the following relation between controllability/observability matrices P_t and Q_t and corresponding grammians W_t^c and W_t^o is always true.

Lemma 24.

(i)
$$||P_t||_2 = (\lambda_{\max}(W_t^c))^{\frac{1}{2}}$$
 and $||P_t^{\dagger}||_2 = (\lambda_{-1}^{\downarrow}(W_t^c))^{-\frac{1}{2}}$.
(ii) $||Q_t||_2 = (\lambda_{\max}(W_t^o))^{\frac{1}{2}}$ and $||Q_t^{\dagger}||_2 = (\lambda_{-1}^{\downarrow}(W_t^o))^{-\frac{1}{2}}$.

The next two sections are concerned with deriving key lemmas needed for the perturbation analysis.

Open-Loop Map Norm-Bounds

Here we derive approximations for the worst-case ℓ_2 -gain of the open-loop map G over a fixed time-horizon t. We will express the bounds in terms of the relation between the controllability/observability matrices P_t and Q_t and the corresponding Grammians.

With the help of the previous lemma and the following easily verified fact

Lemma 25. $(G_t^{xw}(A))^{-1} = I_t - Z_t^+ \otimes A.$

we can derive upper bounds and lower bounds of the toeplitz operators $m{G}^{(\,\cdot\,)}$:

Lemma 26. Any singular value σ_i of $\sigma(G_t^{xw})$ and G_t^{xu} lies in the range described by the inequalities below:

$$\frac{1}{1+\|A\|_2} \leqslant \sigma_i(\boldsymbol{G}_t^{xw}) \leqslant \sqrt{t+1}\bar{\rho}_t^c(A)$$
(4.35a)

$$\frac{\sigma_{\min}(B)}{1+\|A\|_2} \leqslant \sigma_i(\boldsymbol{G}_t^{xu}) \leqslant \sqrt{t+1}\bar{\sigma}_t^c(A,B)$$
(4.35b)

$$\frac{\sigma_{\min}(C)}{1+\|A\|_2} \leqslant \sigma_i(\boldsymbol{G}_t^{yw}) \leqslant \sqrt{t+1}\bar{\sigma}_t^o(C,A).$$
(4.35c)

Proof. $\|\boldsymbol{G}_t^{xu}\|_2^2 := \max_{\|\boldsymbol{u}\|_2=1} \|\boldsymbol{G}_t^{xu}\boldsymbol{u}\|_2^2$. By decomposing $\boldsymbol{u} = [\boldsymbol{u}_0^\top, \dots, \boldsymbol{u}_t^\top]^\top$ we can rewrite this as

$$\|\boldsymbol{G}_{t}^{xu}\|_{2}^{2} = \max_{\|\boldsymbol{u}\|_{2}=1} \left\| \begin{bmatrix} Bu_{0} \\ ABu_{0} + Bu_{1} \\ \dots \\ A^{t}Bu_{0} + \dots + Bu_{t} \end{bmatrix} \right\|_{2}^{2} = \max_{\|\boldsymbol{u}\|_{2}=1} \sum_{k=0}^{t} \|P_{k}\boldsymbol{u}\|_{2}^{2}$$
$$\leq \sum_{k=0}^{t} \max_{\|\boldsymbol{u}\|_{2}=1} \|P_{k}\boldsymbol{u}\|_{2}^{2} = \sum_{k=0}^{t} \|P_{k}\|_{2}^{2} \leq \sum_{k=0}^{t} (\bar{\sigma}_{k}^{c}(A, B))^{2}$$
$$\leq (t+1) \|P_{t}\|_{2}^{2} = (t+1)(\bar{\sigma}_{t}^{c}(A, B))^{2}$$

where we used the fact that $||P_k||_2^2$ increases in k since it is equal to the largest eigenvalue of the corresponding controllability Gramian $W_k^c = \sum_{i=0}^{k-1} A^i B B^\top A^{i\top}$. Thus, we obtain the bound

$$\|\boldsymbol{G}_t^{xu}(A,B)\|_2 \leqslant \sqrt{(t+1)}\bar{\sigma}_t^c(A,B),$$

and the bound on $\|\boldsymbol{G}^{xw}(A)\|_2$ follows by setting B = I. We apply a similar idea to bound $\|\boldsymbol{G}^{yw}_t\|_2$. Splitting the identity operator $\boldsymbol{I}_t = \sum_{k=0}^t \boldsymbol{E}_k \boldsymbol{E}_k^{\top}$ into a sum of the t+1 orthogonal projections $\boldsymbol{E}_k \boldsymbol{E}_k^{\top}$, we use the triangle inequality to obtain:

$$\|\boldsymbol{G}_{t}^{yw}\|_{2} = \|\sum_{k=0}^{t} \boldsymbol{G}_{t}^{yw} \boldsymbol{E}_{k} \boldsymbol{E}_{k}^{\top}\|_{2} \leqslant \sum_{k=0}^{t} \|\boldsymbol{G}_{t}^{yw} \boldsymbol{E}_{k} \boldsymbol{E}_{k}^{\top}\|_{2}$$
$$\leqslant \sum_{k=0}^{t} \|\boldsymbol{Q}_{k}\|_{2} \leqslant \sqrt{t+1} \|\boldsymbol{Q}_{t}\|_{2} = \sqrt{t+1} \bar{\sigma}_{t}^{o}.$$

For the lower bound, notice that the following chain of inequalities

$$(\sigma_{-1}(\boldsymbol{G}^{xw}(A)))^{-1} = \sigma_1((\boldsymbol{G}_w(A))^{\dagger}) = \|\boldsymbol{I}_t - \boldsymbol{Z}^+ \otimes A\|_2 \le 1 + \|A\|_2$$

implies $\sigma_{-1}(\mathbf{G}^{xw}(A)) \ge (1 + ||A||_2)$. Now for the operator $\mathbf{G}^{xu}(A, B)$, notice that for any vector \underline{u}_t we have:

$$\|\boldsymbol{G}_{t}^{xu}(A,B)\underline{u}_{t}\|_{2} = \|\boldsymbol{G}_{t}^{xw}(A)(I_{t}\otimes B)\underline{u}_{t}\|_{2} \ge \sigma_{\min}(\boldsymbol{G}_{t}^{xw}(A))\|(I_{t}\otimes B)\underline{u}_{t}\|_{2}$$
$$\ge \sigma_{\min}(\boldsymbol{G}_{t}^{xw}(A))\sigma_{\min}(B)\|\underline{u}_{t}\|_{2}.$$

Since $G_t^{xw}(A)$ is invertible and we assume $\sigma_{\min}(B)$ to be full column rank, we conclude that for all \underline{u}_t it holds

$$\|\boldsymbol{G}_t^{xu}(A,B)\underline{u}_t\|_2 \ge \frac{\sigma_{\min}(B)}{1+\|A\|_2} \|\underline{u}_t\|_2,$$

which yields the stated result.

Recall that $G_t^{yw}(C, A) = (I_{t+1} \otimes C)G_t^{xw}(A)$ and that G_t^{yu} can be decomposed in multiple ways:

$$\boldsymbol{G}_{t}^{yu} = (I_{t+1} \otimes C)\boldsymbol{G}_{t}^{xu} = \boldsymbol{G}_{t}^{yw}(I_{t+1} \otimes B) = \boldsymbol{G}_{t}^{yw}(\boldsymbol{I}_{t} - \boldsymbol{Z}_{t}^{+} \otimes A)\boldsymbol{G}_{t}^{xu}.$$

As a corollary of the previous lemma, we obtain bounds on the open loop maps for the outputs:

Lemma 27. For any singular value $\sigma(G_t^{yw})$, $\sigma(G_t^{yu})$ of G_t^{yw} and G_t^{yu} holds:

$$\frac{\sigma_{\min}(C)}{1+\|A\|_2} \leqslant \sigma(\boldsymbol{G}_t^{yw}) \leqslant \sqrt{(t+1)}\bar{\sigma}_t^o(C,A)$$
$$\frac{\sigma_{\min}(C)\sigma_{\min}(B)}{1+\|A\|_2} \leqslant \sigma(\boldsymbol{G}_t^{yu}) \leqslant \|C\|_2\sqrt{t+1}\bar{\sigma}_t^c(A,B)$$
$$\sigma(\boldsymbol{G}_t^{yu}) \leqslant \sqrt{t+1}\bar{\sigma}_t^o(C,A)\|B\|_2.$$

Furthermore, by definition, it is easy to see that $\sigma_{\max}^2(\boldsymbol{G}_t^{zu}) \leq \sigma_{\max}^2(\boldsymbol{G}_t^{yu}) + \sigma_{\max}^2(D)$ and $\sigma_{\min}^2(\boldsymbol{G}_t^{yu}) + \sigma_{\min}^2(D) \leq \sigma_{\min}^2(\boldsymbol{G}_t^{zu})$. We can apply Lem. 27 and conclude the following range for the singular values of \boldsymbol{G}_t^{zu} :

$$\sigma^{2}(\boldsymbol{G}_{t}^{zu}) \ge \left(\frac{\sigma_{\min}(C)\sigma_{\min}(B)}{1 + \|A\|_{2}}\right)^{2} + \sigma_{\min}^{2}(D)$$
(4.37a)

$$\sigma^{2}(\boldsymbol{G}_{t}^{zu}) \leq (t+1) \|C\|_{2} \|B\|_{2} \bar{\sigma}_{t}^{o}(C,A) \bar{\sigma}_{t}^{c}(A,B) + \|D\|_{2}^{2}.$$
(4.37b)

Perturbation Inequalities in Operator Norm

Here we discuss how the feedforward operators $G^{(\cdot)}$ vary with changes in the system matrices (A, B), while assuming C and D are fixed. For the next lemmas, fix two sets of parameters A_1, B_1 and A_2, B_2 and denote $G^{xw,i}, G^{xu,i}, G^{yu,i}, i \in \{1, 2\}$ etc. the corresponding operators $G^{xw}(A_i)$, $G^{xu}(A_i, B_i)$, $G^{yu}(A_i, B_i, C)$, etc.. Furthermore, denote $\partial_{12}A := A_1 - A_2, \partial_{12}B := B_1 - B_2$, and correspondingly write $\partial_{12}G = G(A_1, B_1) - G(A_2, B_2)$ to denote the corresponding changes in the openloop maps. Correspondingly, $\partial_{12}G_t^{xu} = G_t^{xu,1} - G_t^{xu,2}, \partial_{12}G_t^{xw} = G_t^{xw,1} - G_t^{xw,2}$, etc. First, we derive a useful decomposition of the terms $\partial_{12}G$. To that end, notice that for two invertible matrices A and B we can always write

$$A^{-1} - B^{-1} = A^{-1}(B - A)B^{-1} = B^{-1}(B - A)A^{-1}$$

Now using the fact that $G^{xw,i}$ is always invertible and $G_t^{xw,i} = (I_t - Z_t^+ \otimes A_i)^{-1}$, we can rewrite the differences $\partial_{12}G_t^{xw} = G_t^{xw,1} - G_t^{xw,2}$ and $G_t^{xu,1} - G_t^{xu,2}$ as:

$$\partial_{12} \boldsymbol{G}_t^{xw} = \boldsymbol{G}_t^{xw,2} (\boldsymbol{Z}_t^+ \otimes \partial_{21} A) \boldsymbol{G}_t^{xw,1}$$
(4.38a)

$$\partial_{12} \boldsymbol{G}_{t}^{xu} = (\boldsymbol{G}_{t}^{xw,1} - \boldsymbol{G}_{t}^{xw,2})(I_{t} \otimes B_{1}) + \boldsymbol{G}_{t}^{xw,2}(I_{t} \otimes \partial_{12}B)$$
(4.38b)
$$= \boldsymbol{G}_{t}^{xw,2}(\boldsymbol{Z}^{+} \otimes \partial_{21}A)\boldsymbol{G}_{t}^{xu,1} + \boldsymbol{G}_{t}^{xw,2}(I_{t} \otimes \partial_{12}B)$$

$$\partial_{12} \boldsymbol{G}_{t}^{yw} = \boldsymbol{G}_{t}^{yw,1} (\boldsymbol{Z}_{t}^{+} \otimes \partial_{21} A) \boldsymbol{G}_{t}^{xw,2}$$

$$(4.38c)$$

$$\partial_{12} \boldsymbol{G}_{t}^{yu} = \boldsymbol{G}_{t}^{yw,2} (\boldsymbol{Z}_{t}^{+} \otimes \partial_{21} A) \boldsymbol{G}_{t}^{xu,1} + \boldsymbol{G}_{t}^{yw,2} (I_{t} \otimes \partial_{12} B).$$
(4.38d)

Applying the triangle inequality to these equations, we can directly obtain the following perturbation inequalities in the induced 2-norm:

$$\begin{aligned} \|\partial_{12} \boldsymbol{G}_{t}^{xw}\|_{2} &\leq \|\boldsymbol{G}^{xw,1}\|_{2} \|\boldsymbol{G}^{xw,2}\|_{2} \|\partial_{12}A\|_{2} \\ \|\partial_{12} \boldsymbol{G}_{t}^{xu}\|_{2} &\leq \|\boldsymbol{G}^{xw,2}\|_{2} \|\boldsymbol{G}^{xu,1}\|_{2} \|\partial_{12}A\|_{2} + \|\boldsymbol{G}_{t}^{xw,2}\|_{2} \|\partial_{12}B\|_{2} \\ \|\partial_{12} \boldsymbol{G}_{t}^{yw}\|_{2} &\leq \|\boldsymbol{G}^{yw,1}\|_{2} \|\boldsymbol{G}^{xw,2}\|_{2} \|\partial_{12}A\|_{2} \\ \|\partial_{12} \boldsymbol{G}_{t}^{yu}\|_{2} &\leq \|\boldsymbol{G}^{yw,2}\|_{2} \|\boldsymbol{G}^{xu,1}\|_{2} \|\partial_{12}A\|_{2} + \|\boldsymbol{G}_{t}^{yw,2}\|_{2} \|\partial_{12}B\|_{2}. \end{aligned}$$

Applying the results of the previous section Lem. 26, we can further bound the above inequalities in terms of the singular values of the controllability grammians. For the following results, we will denote $\bar{\rho}_t^{c,i} = \bar{\rho}_t^c(A_i)$, $\bar{\sigma}_t^{c,i} = \bar{\sigma}_t^c(A_i, B_i)$, $\bar{\rho}_t^{o,i} = \bar{\rho}_t^o(C, A_i)$ corresponding to two sets of system parameters (A_1, B_1) and (A_2, B_2) ; for each variable, we will use $\bar{x}_t^{c/o*} := \max_i \{x_t^{c/oi}\}$ ($\underline{x}_t^{c/o*} := \min_i \{\underline{x}_t^{c/oi}\}$) to denote the maximum (minimum) of both cases, i.e., $\bar{\rho}_t^{c*} = \max\{\bar{\rho}_t^{c,1}, \bar{\rho}_t^{c,2}\}$, ($\underline{\rho}_t^{c*} = \min\{\underline{\rho}_t^{c,1}, \underline{\rho}_t^{c,2}\}$) etc.

Lemma 28. Let $G_{t,1}$ and $G_{t,2}$ be the open loop maps of two sets of system parameters (A_1, B_1) and (A_2, B_2) . Then:

$$\begin{aligned} \|\partial_{12} \boldsymbol{G}_{t}^{xw}\|_{2} &\leq (t+1)\bar{\rho}_{t}^{c*}\|\partial_{12}A\|_{2} \tag{4.39} \\ \|\partial_{12} \boldsymbol{G}_{t}^{xu}\|_{2} &\leq (t+1)\bar{\sigma}_{t}^{c*}\bar{\rho}_{t}^{c*}\|\partial_{12}A\|_{2} + \sqrt{t+1}\bar{\rho}_{t}^{c*}\|\partial_{12}B\|_{2} \\ \|\partial_{12} \boldsymbol{G}_{t}^{yw}\|_{2} &\leq (t+1)\bar{\sigma}_{t}^{o*}\bar{\rho}_{t}^{c*}\|\partial_{12}A\|_{2} \\ \|\partial_{12} \boldsymbol{G}_{t}^{yu}\|_{2} &\leq (t+1)\bar{\sigma}_{t}^{c*}\bar{\sigma}_{t}^{o*}\|\partial_{12}A\|_{2} + \sqrt{t+1}\bar{\sigma}_{t}^{o*}\|\partial_{12}B\|_{2}. \end{aligned}$$

We can utilize the above decomposition also for the controllability and observability matrices $P_t(A, B)$ and $Q_t(C, A)$. Notice that $P_k \in \mathbb{R}^{n \times (k+1)m} = \mathbf{E}_t^\top \mathbf{G}_t^{xu}$ is the last row of submatrices in \mathbf{G}_k^{xu} and $Q_k(C, A) \in \mathbb{R}^{\ell(k+1) \times n} = \mathbf{G}_t^{yw} \mathbf{E}_0$ is the first column of submatrices in \mathbf{G}_k^{yw} , hence from (4.38c) we obtain the decomposition:

$$Q_t(C, A_1) - Q_t(C, A_2) = \boldsymbol{G}_t^{yw,1}(C, A_1)(\boldsymbol{Z}_t^+ \otimes \partial_{21}A)\boldsymbol{G}_t^{xw,2}(A_2)\boldsymbol{E}_1$$
$$\boldsymbol{G}_t^{yw,1}(C, A_1)(\boldsymbol{Z}_t^+ \otimes \partial_{21}A)Q_t(I, A_2)$$

and from (4.38a) and (4.38b) we get:

$$P_t(A_1, B_1) - P_t(A_2, B_2) = P_t(A_2, I) \left((\mathbf{Z}_t^+ \otimes \partial_{21} A) \mathbf{G}_t^{xu, 1} + I_t \otimes \partial_{12} B \right).$$

Applying the triangle-inequality to these decompositions and substituting the operator norm bounds derived in Lem. 27 and Lem. 26, we obtain the following perturbation inequality for the controllability and observability matrices:

Lemma. Denote $\sigma^{u,i} = \lambda_{\max}(W_t^c(A_i, B_i))$ and $\sigma^{y,i} = \lambda_{\max}(W_t^o(C, A_i))$ corresponding to the two sets of system parameters (A_1, B_1) and (A_2, B_2) . Then $\partial_{12}P_t$ and $\partial_{12}Q_t$ are bounded as:

$$\|\partial_{12}P_t\|_2 \leqslant \bar{\rho}_t^{c,2} \left(\sqrt{(t+1)}\bar{\sigma}_t^{c,1}\|\partial_{12}A\|_2 + \|\partial_{12}B\|_2\right)$$
(4.40)

$$\|\partial_{12}Q_t\|_2 \leqslant \bar{\rho}_t^{o,2}\sqrt{(t+1)}\bar{\sigma}_t^{o,1}\|\partial_{12}A\|_2.$$
(4.41)

4.7 From \mathcal{H}_2 -Optimal Control to Least Squares

Having the basic definitions and background, our next step is to reduce the \mathcal{H}_2 problem described by the equations (4.20) to a Least-Squares problem. Denote $\phi_k^{j,x} \in \mathbb{R}^n$, $\phi_k^{j,u} \in \mathbb{R}^m$ as the *j*th column of $R_k \in \mathbb{R}^{n \times n}$, $M_k \in \mathbb{R}^{m \times n}$ and e_j the unit vector in the *j*-th coordinate axis. From now on, we fix the finite horizon *t*. As described in [11], we can separate the problem by columns and can equivalently restate (4.20) in terms of each column $\phi_k^{j,x}$ and $\phi_k^{j,u}$:

$$S_{j} := \min \left\| \begin{bmatrix} C & 0 \\ 0 & D \end{bmatrix} \begin{bmatrix} \phi_{1}^{j,x} & \phi_{2}^{j,x} & \dots & \phi_{t}^{j,x} \\ \phi_{1}^{j,u} & \phi_{2}^{j,u} & \dots & \phi_{t}^{j,u} \end{bmatrix} \right\|_{F}^{2}$$
(4.42)
s.t.: $\phi_{1}^{j,x} = e_{j}$
 $\phi_{k+1}^{j,x} = A\phi_{k}^{j,x} + B\phi_{k}^{j,u}, \quad \forall \ 1 \le k \le t$
 $\phi_{t+1}^{j,x} = 0.$

We rewrite (4.42) further and introduce new variables to avoid tedious notation. Define $u_k = \phi_k^{j,u}, \forall k : 0 \le k \le t - 1, u = [u_1^\top, \dots, u_t^\top]^\top$ and let C, D denote the lifted weight matrices $C = I_t \otimes C, D = I_t \otimes D$. Now we rewrite the subproblem S_j as

$$S_{j} = \min_{\boldsymbol{u}} \quad \left\| \begin{bmatrix} \boldsymbol{C}\boldsymbol{G}_{t}^{xu}(A,B) \\ \boldsymbol{D} \end{bmatrix} \boldsymbol{u} - \boldsymbol{\eta}_{t}(A) \right\|_{2}^{2} + (C^{\top}C)_{jj} \quad (4.43a)$$

s.t.:
$$0 = A^{t+1}e_j + P_t(A, B)\boldsymbol{u}$$
 (4.43b)

where $\boldsymbol{\eta}_t^{\top} = -[Q_t(C, A)Ae_j)^{\top}, \mathbf{0}]$ and $\boldsymbol{E}_1^{\top} = [I_n, \mathbf{0}_{n \times n}, \cdots, \mathbf{0}_{n \times n}]$. To simplify notation, we introduce the virtual outputs $z = [Cx, Du]^{\top} \in \mathbb{R}^{(n+m) \times m}$, y = Cx and the operators $\boldsymbol{G}^{yu}, \boldsymbol{G}^{zu}$ as

$$\boldsymbol{G}^{yu}(A,B) = \boldsymbol{C}\boldsymbol{G}^{xu}(A,B) \qquad \boldsymbol{G}^{zu}(A,B) = \begin{bmatrix} \boldsymbol{C}\boldsymbol{G}^{xu}(A,B) \\ \boldsymbol{D} \end{bmatrix}$$
 (4.44)

to rephrase the problem into (4.45), where we dropped the constant term $(C^{\top}C)_{jj}$ as it is not needed for analysis.

$$S_j = \min_{\boldsymbol{u}} \quad \|\boldsymbol{G}_t^{zu}(A, B)\boldsymbol{u} - \boldsymbol{\eta}_t(A)\|_2^2$$
(4.45a)

s.t.:
$$0 = A^{t+1}e_j + P_t(A, B)\boldsymbol{u}.$$
 (4.45b)

4.8 **Representation as a Least-Squares Problem**

We now rewrite (4.43) as a least square problem. Define $u_c^* := P_t^{\top} (P_t P_t^{\top})^{-1} A^t e_j$, which is the solution to the optimization problem

$$\min_{\boldsymbol{u}} \quad \|\boldsymbol{u}\|_2^2$$

s.t. $-A^{t+1}e_j = P_t\boldsymbol{u}.$

We can interpret u_c^* as the smallest control action, measured in ℓ_2 , that drives the system from the origin to $-A^{t+1}e_j$ in t time-steps. This relates to controllability

grammians as described in [49]. In terms of the Moore-Penrose Inverse, we can also write $\boldsymbol{u}_c^* := P_t^{\dagger} A^{t+1} e_j = P_t^{\top} (W_t^c)^{\dagger} A^{t+1} e_j$, where W_t^c is the *t*-th controllability grammian.

We drop the index j from ϕ^{*j} and reparameterize $\boldsymbol{u} = -\boldsymbol{u}_c^* + \boldsymbol{u}'$ where $\boldsymbol{u}' \in \operatorname{null}(P_t)$ and describe (4.43) as the optimization problem:

$$S_j := \min_{\boldsymbol{u}' \in \operatorname{null}(P_t(A,B))} \left\| \boldsymbol{G}_t^{zu}(A,B)(\boldsymbol{u}' - \boldsymbol{u}_c^*(A,B)) - \boldsymbol{\eta}_t(A) \right\|_2^2.$$
(4.46)

Let $u^*(A, B)$ be a minimizer of the above problem for fixed A, B, we are interested in the SLS solutions

$$\phi^*(A,B) := \begin{bmatrix} \boldsymbol{C}^{-1} & 0\\ 0 & \boldsymbol{D}^{-1} \end{bmatrix} (\boldsymbol{G}_t^{zu}(A,B)(\boldsymbol{u}^* - \boldsymbol{u}_c^*(A,B)) - \boldsymbol{\eta}_t(A))$$

and how these solutions are perturbed with changes in A, B. We (over-)parameterize \boldsymbol{u} as $\boldsymbol{u} = (I - P_t^{\dagger} P_t)\boldsymbol{\eta}$, to cast problem S_j into an unconstrained one:

$$\min_{\boldsymbol{\eta}} \left\| \underbrace{\boldsymbol{G}_{t}^{zu}(A,B)\boldsymbol{u}_{c}^{*}(A,B) + \boldsymbol{\eta}_{t}(A)}_{\boldsymbol{g}} - \underbrace{\boldsymbol{G}_{t}^{zu}(A,B)(I-P_{t}^{\dagger}P_{t})}_{\boldsymbol{H}} \boldsymbol{\eta} \right\|_{2}^{2}.$$
(4.47)

The optimal value of problem S_j is $\|\nu^*\|_2^2$, where $\nu^* := (\boldsymbol{H}\boldsymbol{H}^{\dagger} - I)\boldsymbol{g}$ and is achieved at $\eta^* = \boldsymbol{H}^{\dagger}\boldsymbol{g}$. The corresponding optimal ϕ^* takes the form min-norm solution η^* to the above problem is $\eta^* = \boldsymbol{H}^{\dagger}\boldsymbol{g}$ and $S_j = \|\nu^*\|_2$ and therefore the optimal solution ϕ^* takes the form:

$$\phi^* = \begin{bmatrix} \boldsymbol{C}^{-1} & 0\\ 0 & \boldsymbol{D}^{-1} \end{bmatrix} \nu^*, \text{ where } \nu^* = (\boldsymbol{H}\boldsymbol{H}^{\dagger} - I)\boldsymbol{g}.$$
(4.48)

Hence, up to the constant $\kappa_{CD} = \max\{\|C^{-1}\|_2, \|D^{-1}\|_2\}$, the sensitivity of ϕ^* scales linearly with the sensitivity of the corresponding optimal ν^* . Hence, if ϕ_1^*, ν_1^* and ϕ_2^*, ν_2^* are optimal solutions for two different sets of parameters (A_1, B_1) and (A_2, B_2) , then $\|\phi_1^* - \phi_2^*\|_2$ is bounded as:

$$\|\phi_1^* - \phi_2^*\|_2 \leqslant \kappa_{CD} \|\nu_1^* - \nu_2^*\|_2.$$

The goal of the next section is to bound $\|\nu_1^* - \nu_2^*\|_2$ in terms of $\|A_1 - A_2\|_2$ and $\|B_1 - B_2\|_2$.

4.9 Perturbation of Least Squares

Consider two optimal solutions $\nu_1^* = (\boldsymbol{H}_1 \boldsymbol{H}_1^{\dagger} - I)\boldsymbol{g}_1$ and $\nu_2^* = (\boldsymbol{H}_2 \boldsymbol{H}_2^{\dagger} - I)\boldsymbol{g}_2$, of parameters (A_1, B_1) and (A_2, B_2) . The difference $\nu_1^* - \nu_2^*$ can be written as

$$\nu_1^* - \nu_2^* = (\boldsymbol{H}_1 \boldsymbol{H}_1^{\dagger} - I)(\boldsymbol{g}_1 - \boldsymbol{g}_2) + (\boldsymbol{H}_1 \boldsymbol{H}_1^{\dagger} - \boldsymbol{H}_2 \boldsymbol{H}_2^{\dagger})\boldsymbol{g}_2$$

and the term $\boldsymbol{H}_1\boldsymbol{H}_1^\dagger - \boldsymbol{H}_2\boldsymbol{H}_2^\dagger$ can be further broken down into:

$$(I - \boldsymbol{H}_1 \boldsymbol{H}_1^{\dagger})(\boldsymbol{H}_1 - \boldsymbol{H}_2)\boldsymbol{H}_2^{\dagger} + \left[(I - \boldsymbol{H}_2 \boldsymbol{H}_2^{\dagger})(\boldsymbol{H}_1 - \boldsymbol{H}_2)\boldsymbol{H}_1^{\dagger}\right]^{\top}.$$

Now, since the operators $(MM^{\dagger} - I)$ and MM^{\dagger} are projections for any matrix M, we know that $\|\boldsymbol{H}_{i}\boldsymbol{H}_{i}^{\dagger} - I\|_{2} = 1$ ⁵. Therefore, we can bound the term $\boldsymbol{H}_{1}\boldsymbol{H}_{1}^{\dagger} - \boldsymbol{H}_{2}\boldsymbol{H}_{2}^{\dagger}$ in terms of the $\|\cdot\|_{2}$ norm as:

$$\|\boldsymbol{H}_{1}\boldsymbol{H}_{1}^{\dagger} - \boldsymbol{H}_{2}\boldsymbol{H}_{2}^{\dagger}\|_{2} \leq \|\boldsymbol{H}_{1} - \boldsymbol{H}_{2}\|_{2}(\|\boldsymbol{H}_{1}^{\dagger}\|_{2} + \|\boldsymbol{H}_{2}^{\dagger}\|_{2}).$$
(4.49)

Applying the triangle-inequality on (4.49) and substituting the above bound yields:

$$\|\nu_1^* - \nu_2^*\|_2 \leq \|\boldsymbol{g}_1 - \boldsymbol{g}_2\|_2 + \|\boldsymbol{H}_1 - \boldsymbol{H}_2\|_2 (\|\boldsymbol{H}_1^{\dagger}\|_2 + \|\boldsymbol{H}_2^{\dagger}\|_2)\|\boldsymbol{g}_2\|_2.$$
(4.50)

In the following, we proceed to analyze each of the terms in the above inequality separately. To this end, we will make frequent use of the Lemmas. The facts stated in (29) can be easily verified. The proof of Lem. 30 is more involved and can be found in [129].⁶

Lemma 29. For arbitrary matrices $X, Y \in \mathbb{R}^{n \times m}$ and $A, B \in \mathbb{R}^{n \times n}$, it holds that

(i) $A_1^k - A_2^k = \sum_{j=0}^{k-1} A_1^{k-1-j} (A_1 - A_2) A_2^j$. (ii) $XX^{\dagger} - YY^{\dagger} = (I - XX^{\dagger})(X - Y)Y^{\dagger} + [(I - YY^{\dagger})(X - Y)X^{\dagger}]^{\top}$. (iii) $X^{\dagger}X - Y^{\dagger}Y = Y^{\dagger}(X - Y)(I - X^{\dagger}X) + [X^{\dagger}(X - Y)(I - Y^{\dagger}Y)]^{\top}$. (iv) If A and B are invertible, then $A^{-1} - B^{-1} = A^{-1}(B - A)B^{-1}$.

The following is a corollary from Theorem 4.1 in [129]:

Lemma 30. Let X and Y be matrices with equal rank, let $\|\cdot\|_2$ denote the induced 2-norm and $\|\cdot\|_F$ denote the Frobenius norm. The following inequalities hold:

$$||X^{\dagger} - Y^{\dagger}||_{2} \leq \varphi ||X^{\dagger}||_{2} ||Y^{\dagger}||_{2} ||X - Y||_{2}$$
$$||X^{\dagger} - Y^{\dagger}||_{F} \leq \sqrt{2} ||X^{\dagger}||_{2} ||Y^{\dagger}||_{2} ||X - Y||_{F}$$

where $\varphi = \frac{1+\sqrt{5}}{2}$ denotes the golden ratio constant.

⁵Unless $H_i = 0$.

⁶See Corollary of Thm.4.1 in [129].

Bounding $\|\boldsymbol{H}_1 - \boldsymbol{H}_2\|_2$

First, notice that we can express \boldsymbol{H}^{\top} as $(I - P_t^{\dagger} P_t)(\boldsymbol{G}_t^{zu})^{\top}$ and that

$$\boldsymbol{H}_{1} - \boldsymbol{H}_{2} = \boldsymbol{G}_{t}^{zu,1}(P_{t,2}^{\dagger}P_{t,2} - P_{t,1}^{\dagger}P_{t,1}) + (\boldsymbol{G}_{t}^{zu,1} - \boldsymbol{G}_{t}^{zu,2})(I - P_{t,2}^{\dagger}P_{t,2}).$$

Now using the previous lemma and noticing that $\|\boldsymbol{G}_t^{zu,1} - \boldsymbol{G}_t^{zu,2}\|_2 = \|\boldsymbol{G}_t^{yu,1} - \boldsymbol{G}_t^{yu,2}\|_2$ we obtain the following inequality

$$\|\boldsymbol{H}_{1} - \boldsymbol{H}_{2}\|_{2} \leq \|\boldsymbol{G}_{t}^{yu,1} - \boldsymbol{G}_{t}^{yu,2}\|_{2} + \|\boldsymbol{G}_{t}^{zu,1}\|_{2} \left(\|\boldsymbol{P}_{t,1}^{\dagger}\|_{2} + \|\boldsymbol{P}_{t,2}^{\dagger}\|_{2}\right)\|\boldsymbol{P}_{t,2} - \boldsymbol{P}_{t,1}\|_{2}$$

and arrive at:

$$\|\boldsymbol{H}_{1}-\boldsymbol{H}_{2}\|_{2}(\|\boldsymbol{H}_{1}^{\dagger}\|_{2}+\|\boldsymbol{H}_{2}^{\dagger}\|_{2})\|\boldsymbol{g}_{2}\|_{2} \leqslant \mu_{1}\|\partial_{12}\boldsymbol{G}_{t}^{yu}\|_{2}+\mu_{2}\|\partial_{12}P_{t}\|_{2}$$

with the constants

=

$$\mu_1 = (\|\boldsymbol{H}_1^{\dagger}\|_2 + \|\boldsymbol{H}_2^{\dagger}\|_2)\|\boldsymbol{g}_2\|_2$$
(4.51a)

$$\mu_{2} = \|\boldsymbol{G}_{t}^{zu,1}\|_{2}(\|\boldsymbol{P}_{t,1}^{\dagger}\|_{2} + \|\boldsymbol{P}_{t,2}^{\dagger}\|_{2})(\|\boldsymbol{H}_{1}^{\dagger}\|_{2} + \|\boldsymbol{H}_{2}^{\dagger}\|_{2})\|\boldsymbol{g}_{2}\|_{2}$$
(4.51b)

Lemma 31. Recall that D is invertible. Therefore,

$$\|\boldsymbol{H}^{\dagger}(A,B)\|_{2} \leq \|\boldsymbol{G}_{t}^{zu\dagger}\|_{2} \leq (\sigma_{\min}(\boldsymbol{G}_{t}^{zu}))^{-1}.$$

Proof. In order to bound $||M^{\dagger}||_2$ from above, we have to bound $\sigma_{-1}(M)$ from below. Notice that $G_t^{zu}(A, B)$ is full-column rank, since D is invertible and thus rank $(G_t^{zu}(A, B)) = (t+1) \cdot n_u$. The projection $\Pi_{\mathcal{N}(P_t)} := (I - P_t^{\dagger}(A, B)P_t(A, B)) \in \mathbb{R}^{(t+1)n_u \times (t+1)n_u}$ has rank $r_H < (t+1)n_u = \operatorname{rank}(G_t^{zu}(A, B))$. Hence, $G_t^{zu}(A, B)$ and P_t have the same null space and therefore $H = G_t^{zu}(A, B)\Pi_{\mathcal{N}(P_t)}$ is of rank r_H . From these observations, we can equivalently say that $\sigma_{-1}(H)$ is the r_H -th largest singular eigenvalue of H. Using the Minimax principle, we can therefore represent $\sigma_{-1}(H)$ as the solution to the following max-min problem:

$$\sigma_{-1}(\boldsymbol{H}) = \max_{\text{proj.}\Pi, \text{ s.t.: rank}(\Pi) = r_H} \min_{x \text{ s.t.: } \|\Pi x\| = 1} x^\top \Pi \boldsymbol{H}^\top \boldsymbol{H} \Pi x$$
(4.52)

$$= \max_{\text{proj.}\Pi, \text{ s.t.: rank}(\Pi)=r_H} \min_{x \text{ s.t.: } \|\Pi x\|=1} x^\top \Pi \Pi_{\mathcal{N}(P_t)} \boldsymbol{G}_t^{zu\top} \boldsymbol{G}_t^{zu} \Pi_{\mathcal{N}(P_t)} \Pi x.$$
(4.53)

Now recall that $\Pi_{\mathcal{N}(P_t)}$ is of rank r_H , hence it is a feasible choice for the variable Π of the outer optimization problem. Furthermore, this leads us to the bound

$$\sigma_{-1}(\boldsymbol{H}) \geq \min_{x \text{ s.t.: } \|\Pi_{\mathcal{N}(P_t)} x\|=1} x^{\top} \Pi_{\mathcal{N}(P_t)} \boldsymbol{G}_t^{zu\top} \boldsymbol{G}_t^{zu} \Pi_{\mathcal{N}(P_t)} x$$
(4.54)

$$= \min_{\substack{x \text{ s.t.: } z \in \mathcal{N}(P_t), \|z\|_2 = 1}} z^\top \boldsymbol{G}_t^{zu\top} \boldsymbol{G}_t^{zu} z \tag{4.55}$$

$$\geq \min_{z \text{ s.t. } \|z\|=1} z^{\top} \boldsymbol{G}_{t}^{zu\top} \boldsymbol{G}_{t}^{zu} z = \sqrt{\lambda_{\min}(\boldsymbol{G}_{t}^{zu\top} \boldsymbol{G}_{t}^{zu})}.$$
(4.56)
Finally, the desired statement follows, since $\sqrt{\lambda_{\min}(G_t^{zu\top}G_t^{zu})} = \sigma_{\min}(G_t^{zu})$ and by inverting the inequality above.

To bound the worst-case of $\|g\|_2$ recall its definition:

$$\boldsymbol{g} = \boldsymbol{G}_t^{zu}(A, B)\boldsymbol{u}_c^*(A, B) + \boldsymbol{\eta}_t(A)$$
$$= \boldsymbol{G}_t^{zu}(A, B)P_t^{\dagger}(A, B)A^{t+1}\boldsymbol{e}_j - Q_t(C, A)A\boldsymbol{e}_j.$$

Definition 4.4. For a pair (A_i, B_i) define the constants $\alpha_{t,i}$, and $\bar{\sigma}_t^{zu,i}, \underline{\sigma}_t^{zu,i}$ as:

$$\alpha_t := \max_{0 \le k \le t+1} \|A^k\|_2 \quad \bar{\sigma}_t^{zu,i} := \sigma_{\max}(G_t^{zu}(A_i, B_i)) \quad \underline{\sigma}_t^{zu,i} := \sigma_{\min}(G_t^{zu}(A_i, B_i)).$$

We can obtain an upper bound for $\|g\|_2$, as a corollary of Lem. 27 and Lem. 24

$$\|\boldsymbol{g}\|_{2} \leq \left(\|C\|_{2}\sqrt{t}\bar{\sigma}_{t}^{c} + \|D\|_{2}\right) (\underline{\sigma}_{t}^{c})^{-1}\alpha_{t} + \bar{\sigma}_{t}^{o}\|A\|_{2},$$
(4.57)

and using the constants defined in Definition (4.4) we can bound the constant μ_1 as:

$$\boldsymbol{\mu}_1 \leqslant 2 \frac{\bar{\sigma}_t^{zu*}}{\underline{\sigma}_t^{zu*} \underline{\sigma}_t^{c*}} \alpha_t + \bar{\sigma}_t^{o*} \|A_*\|_2$$

$$(4.58)$$

$$\boldsymbol{\mu}_{2} \leqslant \frac{4}{\underline{\sigma}_{t}^{zu*}} \left(\frac{\bar{\sigma}_{t}^{zu*}}{\underline{\sigma}_{t}^{c*}}\right)^{2} \alpha_{t} + 2 \frac{\bar{\sigma}_{t}^{zu*}}{\underline{\sigma}_{t}^{c*}} \bar{\sigma}_{t}^{o*} \|A_{*}\|_{2}$$
(4.59)

where the expression with superscript $\underline{\sigma}_t^{c*} = \min\{\underline{\sigma}_t^{c,1}, \underline{\sigma}_t^{c,2}\}, \bar{\sigma}_t^{o*} = \max\{\bar{\sigma}_t^{o,1}, \bar{\sigma}_t^{o,2}\}$ represents the worst-case pick of i = 1, 2 for the corresponding expression. We can obtain explicit bounds by substituting the bounds of Lem. 27

Bounding $\|{\bm{g}}_1 - {\bm{g}}_2\|_2$

Recall that $g_1 - g_2$ takes the form:

$$\boldsymbol{g}_{1} - \boldsymbol{g}_{2} = \boldsymbol{G}_{t}^{zu,1} P_{t,1}^{\dagger} A_{1}^{t+1} e_{j} - \boldsymbol{G}_{t}^{zu,2} P_{t,2}^{\dagger} A_{2}^{t+1} e_{j} + \boldsymbol{\eta}_{t}(A_{1}) - \boldsymbol{\eta}_{t}(A_{2}).$$
(4.60)

We split $g_1 - g_2 = \Delta_1 + \Delta_2 + \Delta_3 + \Delta_4$ into the four telescoping terms $\Delta_1, \ldots, \Delta_4$ below and proceed to bound each of them individually.

$$\Delta_{1} = \left(\boldsymbol{G}_{t}^{zu,1} - \boldsymbol{G}_{t}^{zu,2}\right) P_{t,1}^{\dagger} A_{1}^{t+1} e_{j} \qquad \Delta_{2} = \boldsymbol{G}_{t}^{zu,2} \left(P_{t,1}^{\dagger} - P_{t,2}^{\dagger}\right) A_{1}^{t+1} e_{j}$$

$$\Delta_{3} = \boldsymbol{G}_{t}^{zu,2} P_{t,2}^{\dagger} \left(A_{1}^{t+1} e_{j} - A_{2}^{t+1} e_{j}\right) \qquad \Delta_{4} = \boldsymbol{\eta}_{t}(A_{1}) - \boldsymbol{\eta}_{t}(A_{2})$$
1. $\underline{\Delta}_{1} : \|\Delta_{1}\|_{2} \leq \|\boldsymbol{G}_{t}^{yu,1} - \boldsymbol{G}_{t}^{yu,2}\|_{2} \|P_{t,1}^{\dagger}\|_{2} \|A_{1}^{t+1} e_{j}\|_{2}.$
2. $\underline{\Delta}_{2} : \text{Applying the } \| \cdot \|_{2}\text{-bound from Lem. 30 yields:}$

$$\|\Delta_{2}\|_{2} \leq \varphi \|P_{t,1}^{\dagger}\|_{2} \|P_{t,2}^{\dagger}\|_{2} \|\boldsymbol{G}_{t}^{zu,2}\|_{2} \|A^{t+1} e_{j}\|_{2} \|P_{t,1} - P_{t,2}\|_{2}.$$

3. $\underline{\Delta_3}$: We apply Lem. 29 to obtain $A_1^{t+1} - A_2^{t+1} = \sum_{j=0}^t A_1^{t-j} (A_1 - A_2) A_2^j$. Notice we can rewrite that as

$$A_1^{t+1} - A_2^{t+1} = P_t(A_1, I)(A_1 - A_2)Q_t(I, A_2)$$

and apply the bounds of Lem. 24 to conclude:

$$\begin{aligned} \|\Delta_3\|_2 &\leqslant \sqrt{\lambda_{\max}(W_t^c(A_1, I))\lambda_{\max}(W_t^o(I, A_2))} \|\boldsymbol{G}_t^{zu,2}\|_2 \|\boldsymbol{P}_{t,2}^{\dagger}\|_2 \|A_1 - A_2\|_2 \\ &\leqslant \bar{\rho}_t^{o,2} \bar{\rho}_t^{c,1} \|\boldsymbol{G}_t^{zu,2}\|_2 \|\boldsymbol{P}_{t,2}^{\dagger}\|_2 \|A_1 - A_2\|_2. \end{aligned}$$

4. Δ_4 : We can bound $\|\Delta_4\|_2$ as

$$|\Delta_4||_2 \leq \|\partial_{12}Q_t\|_2 \|A_1e_j\|_2 + \|Q_{t,2}\|_2 \|(A_1 - A_2)e_j\|_2$$

Combining all the previous inequalities gives us:

$$\|\boldsymbol{g}_{1} - \boldsymbol{g}_{2}\|_{2} \leq \gamma_{1} \|\partial_{12}\boldsymbol{G}_{t}^{yu}\|_{2} + \gamma_{2} \|\partial_{12}P_{t}\|_{2} + \gamma_{3} \|\partial_{12}Q_{t}\|_{2} + \gamma_{4} \|\partial_{12}A\|_{2}$$

with the γ_i defined as:

$$\gamma_1 = \|P_{t,1}^{\dagger}\|_2 \|A_1^{t+1} e_j\|_2 \tag{4.61a}$$

$$\gamma_2 = \varphi \|P_{t,1}^{\dagger}\|_2 \|P_{t,2}^{\dagger}\|_2 \|\boldsymbol{G}_t^{zu,2}\|_2 \|A^{t+1}e_j\|_2$$
(4.61b)

$$\gamma_3 = \|A_1 e_j\|_2 \tag{4.61c}$$

$$\gamma_4 = \|Q_{t,2}\|_2 + \bar{\rho}_t^{o,2} \bar{\rho}_t^{c,1} \|\boldsymbol{G}_t^{zu,2}\|_2 \|P_{t,2}^{\dagger}\|_2$$
(4.61d)

with abbreviations $\bar{\rho}_t^{c,i} = \lambda_{\max}(W_t^c(A_i, I))$ and $\bar{\rho}_t^{o,i} = \lambda_{\max}(W_t^o(I, A_i))$. Combining the inequalities we proved in the last two subsections, we obtain a first bound on the optimal solutions $\|\phi_1^* - \phi_2^*\|_2 \leq \kappa_{CD} \|\nu_1^* - \nu_2^*\|_2$:

$$\kappa_{CD}^{-1} \|\phi_1^* - \phi_2^*\|_2 \leq \gamma_1 \boldsymbol{\mu}_1 \|\partial_{12} \boldsymbol{G}_t^{yu}\|_2 + \gamma_2 \boldsymbol{\mu}_2 \|\partial_{12} P_t\|_2 + \gamma_3 \|\partial_{12} Q_t\|_2 + \gamma_4 \|\partial_{12} A\|_2.$$

This inequality informs a natural interpretation: The sensitivity to parameter changes of the optimal solutions ϕ^* can be reduced to analyzing the change of four systemtheoretic quantities: the open-loop map from control input to output $\|\partial_{12} G_t^{yu}\|_2$, the controllability matrix $\partial_{12} P_t$, observability $\partial_{12} Q_t$, and the system matrix $\partial_{12} A$.

Lemma 32. Let C, D be fixed invertible matrices and let $[A_1, B_1]$ and $[A_2, B_2]$ be two t-controllable pairs of system and input matrices. Let ϕ_1^* and ϕ_2^* be the optimal solutions of the \mathcal{H}_2 -problem (4.45), corresponding to both sets of parameters (A_1, B_1) and (A_2, B_2) . Then, for ϕ_1^* and ϕ_2^* holds the inequality:

$$\kappa_{CD}^{-1} \|\phi_1^* - \phi_2^*\|_2 \leq \delta_1 \|\partial_{12} \boldsymbol{G}_t^{yu}\|_2 + \delta_2 \|\partial_{12} P_t\|_2 + \delta_3 \|\partial_{12} Q_t\|_2 + \delta_4 \|\partial_{12} A\|_2$$

where $\delta_1, \ldots, \delta_4$ and μ_1 are the constants defined below:

$$\mu_1 \leqslant \frac{2}{\underline{\sigma}_t^{zu*}} \frac{\bar{\sigma}_t^{zu*}}{\underline{\sigma}_t^{c*}} \alpha_t + \bar{\sigma}_t^{o*} \|A_*\|_2$$

$$(4.62a)$$

$$\mu_2 \leqslant \frac{4}{\underline{\sigma}_t^{zu*}} \left(\frac{\bar{\sigma}_t^{zu*}}{\underline{\sigma}_t^{c*}}\right)^2 \alpha_t + 2 \frac{\bar{\sigma}_t^{zu*}}{\underline{\sigma}_t^{c*}} \bar{\sigma}_t^{o*} \|A_*\|_2$$
(4.62b)

$$\delta_1 = \|P_{t,1}^{\dagger}\|_2 \|A_1^{t+1} e_j\|_2 + \mu_1 \tag{4.62c}$$

$$\delta_2 = \varphi \|P_{t,1}^{\dagger}\|_2 \|P_{t,2}^{\dagger}\|_2 \bar{\sigma}_t^{zu*} \|A^{t+1}e_j\|_2 + \mu_2$$
(4.62d)

$$\delta_3 = \|A_1 e_j\|_2 \tag{4.62e}$$

$$\delta_4 = \|Q_{t,2}\|_2 + \bar{\rho}_t^{o,2} \bar{\rho}_t^{c,1} \bar{\sigma}_t^{zu*} \|P_{t,2}^{\dagger}\|_2.$$
(4.62f)

Final Perturbation Bound

Our final step is to apply the results of section Section 4.6 to bound the ∂ -terms on the right-hand side of the perturbation inequality Lem. 32 in terms of $\partial_{12}A$ and $\partial_{12}B$:

$$\begin{aligned} \|\partial_{12} \boldsymbol{G}_{t}^{yu}\|_{2} &\leq (t+1)\bar{\sigma}_{t}^{c*}\bar{\sigma}_{t}^{o*}\|\partial_{12}A\|_{2} + \sqrt{t+1}\bar{\sigma}_{t}^{o*}\|\partial_{12}B\|_{2} \\ \|\partial_{12}P_{t}\|_{2} &\leq \bar{\rho}_{t}^{c,2}\left(\sqrt{(t+1)}\bar{\sigma}_{t}^{c,1}\|\partial_{12}A\|_{2} + \|\partial_{12}B\|_{2}\right) \\ \|\partial_{12}Q_{t}\|_{2} &\leq \bar{\rho}_{t}^{o,2}\sqrt{(t+1)}\bar{\sigma}_{t}^{o,1}\|\partial_{12}A\|_{2}. \end{aligned}$$

We recall also the lower and upper bounds of the singular values of $m{G}_t^{zu}$

$$\sigma^{2}(\boldsymbol{G}_{t}^{zu}) \geq \left(\frac{\sigma_{\min}(C)\sigma_{\min}(B)}{1+\|A\|_{2}}\right)^{2} + \sigma_{\min}^{2}(D) =: (\underline{\sigma}_{t}^{zu})^{2}$$
$$\sigma^{2}(\boldsymbol{G}_{t}^{zu}) \leq (t+1)\|C\|_{2}\|B\|_{2}\bar{\sigma}_{t}^{o}(C,A)\bar{\sigma}_{t}^{c}(A,B) + \|D\|_{2}^{2} =: (\bar{\sigma}_{t}^{zu})^{2}$$

and the relationship between controllability/observability matrices and grammians:

$$||P_t||_2 = \bar{\sigma}_t^c \qquad ||P_t^{\dagger}||_2 = (\underline{\sigma}_t^c)^{-1} \qquad ||Q_t||_2 = \bar{\sigma}_t^o \qquad ||Q_t^{\dagger}||_2 = (\underline{\sigma}_t^o)^{-1}.$$

We formulate the constants from (4.62) in terms of the controllability and observability singular values of the system:

$$\begin{split} \delta_{1} &\leq (\underline{\sigma}_{t}^{c*})^{-1} \alpha_{t*} + 2(\underline{\sigma}_{t}^{zu*})^{-1} \bar{\sigma}_{t}^{zu*} (\underline{\sigma}_{t}^{c*})^{-1} \alpha_{t} + \bar{\sigma}_{t}^{o*} \|A_{*}\|_{2} \\ \delta_{2} &\leq \varphi(\underline{\sigma}_{t}^{c*})^{-2} \bar{\sigma}_{t}^{zu*} \alpha_{t*} + 4(\underline{\sigma}_{t}^{zu*})^{-1} (\bar{\sigma}_{t}^{zu*})^{2} (\underline{\sigma}_{t}^{c*})^{-2} \alpha_{t} + 2 \bar{\sigma}_{t}^{zu*} (\underline{\sigma}_{t}^{c*})^{-1} \bar{\sigma}_{t}^{o*} \|A_{*}\|_{2} \\ \delta_{3} &\leq \|A_{*}\|_{2} \\ \delta_{4} &\leq \bar{\sigma}_{t}^{o*} + \bar{\rho}_{t}^{o*} \bar{\rho}_{t}^{c*} \bar{\sigma}_{t}^{zu*} (\underline{\sigma}_{t}^{c*})^{-1}, \qquad \alpha_{t*} := \max_{0 \leq k \leq t+1} \|A_{1}^{k}\|_{2} \vee \|A_{2}^{k}\|_{2}. \end{split}$$

After a bit of bookkeeping, we obtain a first bound quantifying the sensitivity of the solutions with respect to perturbations in the parameters A and B:

Lemma 33. Recall the setup of Lem. 32. Then the following inequality holds:

$$\kappa_{CD}^{-1} \|\phi_1^* - \phi_2^*\|_2 \leqslant \Gamma_a \|\partial_{12}A\|_2 + \Gamma_b \|\partial_{12}B\|_2$$

where Γ_a, Γ_b are defined as:

$$\Gamma_{a} = (t+1)\bar{\sigma}_{t}^{c*}\bar{\sigma}_{t}^{o*}\delta_{1} + \delta_{2}(\bar{\rho}_{t}^{c*}\sqrt{(t+1)}\bar{\sigma}_{t}^{c*}) + \delta_{3}\bar{\rho}_{t}^{o*}\sqrt{(t+1)}\bar{\sigma}_{t}^{o*} + \delta_{4} \\
\Gamma_{b} = \delta_{1}\sqrt{t+1}\bar{\sigma}_{t}^{o*} + \delta_{2}\bar{\rho}_{t}^{c*}.$$

4.10 Lipshitzness of \mathcal{H}_2 -Optimal CLM Parameterization

With Lem. 33, we have derived analytic bounds that quantify the difference between the two optimal CLMs $\Phi^*(A_1, B_1, t)$ and $\Phi^*(A_2, B_2, t)$ in terms of the difference in system parameters $\partial_{12}A = A_1 - A_2$, $\partial_{12}B = B_1 - B_2$, and control-theoretic properties of the linear dynamics corresponding to each set of parameters. The latter is measured in terms of singular eigenvalues measuring controllability and observability (w.r.t. C) of the individual pairs of system matrices $[A_1, B_1]$ and $[A_2, B_2]$. Our original motivation for this analysis was to investigate the continuity of the mapping Φ^* as a parametrization of optimal CLMs over a compact set S of system matrices $\{[A[\omega], B[\omega]] \mid \omega \in \Omega\}$ corresponding to the continuous parametrization functions $A: \Omega \to \mathbb{R}^{n \times n}$, $B: \Omega \to \mathbb{R}^{m \times n}$, and parameter space (Ω, d) used in Algorithm 1. With the help of Lem. 33, we will now derive that the mapping $\Phi^*(\cdot|t)$: $\mathcal{S} \ni [A', B'] \mapsto \Phi^*(A', B', t) \in \mathcal{C}(\ell^{\mathcal{X}}, \ell^{\mathcal{X} \times \mathcal{U}})$ is indeed Lipshitz over a compact set $S \subset \mathbb{R}^{n \times (n+m)}$ of matrices, as long as S consists all of *t*-controllable pairs [A, B]. To this end, in the next lemma, we verify that all constants used in Lem. 33, such as $\delta_i^{c/o}$, μ , $\rho_i^{c/o}$, etc., which depend on (A_i, B_i) can be uniformly bounded above for the described sets S.

Lemma 34. Let S be a compact⁷ subset of t-controllable matrix pairs $(A, B) \subset \mathbb{R}^{n \times n} \times \mathbb{R}^{n \times m}$ with full-column rank B, and let $C \in \mathbb{R}^{n \times n}$ and $D \in \mathbb{R}^{m \times m}$ be invertible matrices. Then, there exist positive constants $0 < \underline{\sigma}_{S}^{c} \leq \overline{\sigma}_{S}^{c}, 0 < \underline{\sigma}_{S}^{o} \leq \overline{\sigma}_{S}^{o}, 0 < \underline{\rho}_{S}^{o} \leq \overline{\rho}_{S}^{o}, 0 < \underline{b}_{S} \leq \overline{b}_{S}, and some 0 \leq \overline{\alpha}_{S}^{(1)} \leq \dots \overline{\alpha}_{S}^{(k)} \dots$ such that for all $(A, B) \in S$ and $i \in \{1, \dots, n\}$:

$\underline{\sigma}_{\mathcal{S}}^c \leqslant \sigma_i(P_t(A,B)) \leqslant \overline{\sigma}_{\mathcal{S}}^c$	$\underline{\sigma}^{o}_{\mathcal{S}} \leqslant \sigma_{i}(Q_{t}(C,A)) \leqslant \overline{\sigma}^{o}_{\mathcal{S}}$
$\underline{\rho}_{\mathcal{S}}^{c} \leqslant \sigma_{i}(P_{t}(A, I)) \leqslant \overline{\rho}_{\mathcal{S}}^{c}$	$\underline{\rho}^o_{\mathcal{S}} \leqslant \sigma_i(Q_t(I,A)) \leqslant \overline{\rho}^o_{\mathcal{S}}$
$\max_{1 \le j \le k} \ A^j\ _2 \le \overline{\alpha}_{\mathcal{S}}^{(k)}, \forall k \ge 1$	$\underline{b}_{\mathcal{S}} \leqslant \sigma_i(B) \leqslant \overline{b}_{\mathcal{S}}.$

⁷Take the usual norm for the product-space is defined as |(A, B)| := |A| + |B|.

Proof. All bounds follow from the simple fact that continuous scalar functions over compact sets achieve a maximum and a minimum, and by letting the corresponding constant be defined by that maximum/minimum value. Thus we need to verify the continuity of the corresponding functions. It is obvious that $||A^j||_2$ and P_t and Q_t are continuous functions of A and B. It is well-known [25], that the k-th largest singular value $\sigma_k : \mathbb{R}^{p \times q} \to \mathbb{R}^+$ is a continuous function over the space of matrices; a quick way to convince ourselves is by writing σ_k as a difference of the k-th and the k - 1-th Ky-Fan-Norm [25] and using continuity of norms. Lastly, we know that the minimum singular values $\underline{\sigma}_{S}^c$, $\underline{\sigma}_{S}^o$, $\underline{\rho}_{S}^c$, $\underline{\rho}_{S}^o$, \underline{b}_{S} are all positive, because they are achieved at respective pairs $\{(A_i, B_i)\}$ all of which are t-controllable, i.e., $\sigma_{\min}(P_t(A_i, B_i)) > 0$, and B_i is full column rank, i.e., $\sigma_{\min}(B_i) > 0$.

The concluding theorem of our perturbation analysis follows as a corollary of the previous two lemmas, and establishes the Lipshitzness of the LQ-optimal CLM parametrization map Φ^* over compact sets of *t*-controllable LTI systems:

Theorem (Lipshitzness of LQ-optimal CLM oracle). Let $t \in \mathbb{N}$ be fixed, $C \in \mathbb{R}^{n \times n}$, $D \in \mathbb{R}^{m \times m}$ be fixed invertible matrices, and $\Phi^*(A, B, t)$ represent the unique optimum of the optimal control problem LQ(A, B, t) described by (4.20), (4.21). Assume that S_{ab} is a compact subset of $\mathbb{R}^{n \times n} \times \mathbb{R}^{n \times m}$ such that for all $(A, B) \in S_{ab}$, [A, B] is t-controllable and B is full column rank. Then, there exist fixed constants $L_a, L_b \in \mathbb{R}^+$ such that for all $(A_1, B_1), (A_2, B_2) \in S_{ab}$, the following inequality holds:

$$\| \Phi^*(A_1, B_1, t) - \Phi^*(A_2, B_2, t) \|_{\infty} \leq L_a |\partial_{12}A| + L_b |\partial_{12}B|$$

where $\partial_{12}A := A_1 - A_2$ and $\partial_{12}B := B_1 - B_2$.

Proof. Using Lem. 33 and the constants defined in Lem. 34 for $S = S_{ab}$, we can see that there are constants $\overline{\Gamma}_a$ and $\overline{\Gamma}_b$ which are functions of the constants $t, \overline{\sigma}_S^c, \underline{\sigma}_S^c, \overline{\sigma}_S^o, \underline{\sigma}_S^c, \underline{\rho}_S^c, \underline{\rho}_S^c, \underline{\rho}_S^o, \underline{\rho}_S^o, \underline{h}_S, \overline{h}_S, \overline{\alpha}_S^{(k)}$ such that the difference between any two vectorized solutions $\phi_1^* := \phi^*(A_1, B_1, t), \phi_2^* := \phi^*(A_2, B_2, t)$, where (A_1, B_1) and (A_2, B_2) belong to S_{ab} , is bounded in $|\cdot|_2$ as $|\phi_1^* - \phi_2^*|_2 \leq \overline{\Gamma}_a |\partial_{12}A|_2 + \overline{\Gamma}_b |\partial_{12}B|_2$. Thus, ϕ^* is a Lipshitz-continuous map over S_{ab} . This means that the matrices $M_k^{\Phi^*}$ associated with component functions of $\Phi_k^* = U^{*[k]} \Phi^* P^k$ are $|\cdot|_F$ -Lipshitz continuous and, by equivalence of norms, also $|\cdot|_\infty$ -Lipshitz continuous. Finally, because $\Phi^*(A_i, B_i, t)$ is always t-FIR and time-invariant, there are fixed L_a and

 L_b such that for any component Φ_k^* , $k \in \mathbb{N}$, and perturbations $\partial_{12}A$ and $\partial_{12}B$, the following inequality holds:

$$\sup_{w_{k:0}} \frac{|\partial_{12} \Phi_k^*(w_{k:0})|}{\max_{j \le k} |w_j|} \le L_a |\partial_{12} A| + L_b |\partial_{12} B|.$$

The desired conclusion follows, since $\|| \Phi^*(A_1, B_1, t) - \Phi^*(A_2, B_2, t) \||_{\infty}$ is the supremum over k of the left-hand side of the above inequality.

Analytic expressions of L_a and L_b in terms of the constants can be easily derived, however, doing so is very tedious. The asymptotic behavior in terms t and controllability/observability singular values is given below:

$$L_{a} = \mathcal{O}\left(\left(\frac{t\overline{\sigma}_{\mathcal{S}}^{c}\overline{\sigma}_{\mathcal{S}}^{o}}{\underline{\sigma}_{\mathcal{S}}^{c}}\right)^{\frac{3}{2}} \left(\frac{\overline{\sigma}_{\mathcal{S}}^{o}}{\underline{\sigma}_{\mathcal{S}}^{c}}\right)^{\frac{1}{2}} \overline{\alpha}_{\mathcal{S}}^{(t)}\right) \quad L_{b} = \mathcal{O}\left(\left(\frac{t\overline{\sigma}_{\mathcal{S}}^{c}\overline{\sigma}_{\mathcal{S}}^{o}}{\underline{\sigma}_{\mathcal{S}}^{c}}\right) \left(\frac{\overline{\sigma}_{\mathcal{S}}^{o}}{\underline{\sigma}_{\mathcal{S}}^{c}}\right) \overline{\alpha}_{\mathcal{S}}^{(t)}\right).$$

Part II

Learning to Control Unknown Systems

OVERVIEW

In regards to robustness to model uncertainty, the previous part primarily focuses on problem settings where the system dynamics are uncertain but can be narrowed down to a set of possible models. In this setting, all the models can be simultaneously stabilized by a fixed controller using methods of robust control theory. On the other hand, this part of the thesis focuses on the more general "large uncertainty" scenario, where the set of possible models is bounded but can be arbitrarily large. In practical terms, Part 2 of the thesis assumes that the dynamics of the system are almost entirely unknown. In this second part, we introduce new theory and algorithms for a general framework of learning-to-control with worst-case safety and performance guarantees, even in settings where dynamic uncertainty is very large. We approach this problem from two complementary perspectives: Chapter 5 pursues a model-free, data-driven approach, while Chapter 6 follows a model-based learning and control approach.

The results presented in Chapter 5 are based on the work published in [5] and develop new methods for stability analysis and control design without the need for a model. In particular, we demonstrate the first instance of an all model-free formulation of controller, closed-loop dynamics, and robust stability analysis. We present a simple model-free control algorithm that can robustly learn and stabilize an unknown discrete-time linear system with full control and state feedback, subject to arbitrary bounded disturbance and noise sequences. The controller does not require any prior knowledge of the system dynamics, disturbances, or noise, yet it can guarantee robust stability and provides asymptotic and worst-case bounds on the state and input trajectories. To the best of our knowledge, this is the first model-free algorithm to come with such robust stability guarantees without the need to make any prior assumptions about the system. Simulation results also show that despite its generality and simplicity, the controller demonstrates good closed-loop performance, including fast convergence, small learning transients, and nearly optimal asymptotic gain.

In Chapter 6, we approach the problem of learning-to-control unknown systems from a model-based perspective. In this case, we are given a compact parameterization of all possible system dynamics, which can be arbitrarily large. The results presented are based on the work published in [5] and introduce a new modular framework for model-based learning-to-control. This framework provides robust safety and cost-performance guarantees for closed loop under worst-case scenarios of realized disturbances, noise, or other environmental conditions. Our approach involves decomposing the problem into two subproblems: online learning, referred to as "consistent model chasing," and the underlying control problem in the absence of model uncertainty, known as "oracle design." Each subproblem can be addressed separately, and its solutions (a control oracle and model chaser) are used to instantiate a certainty-equivalent learning-to-control scheme. This scheme inherits both control-and learning-theoretic guarantees, certifying robustness of the closed-loop, even for large model uncertainty in the system dynamics. We discuss how the control oracle is implicitly given by standard nominal control design, provided that this procedure satisfies certain regularity properties over the space of models. Designing the corresponding model chaser represents a problem which we term "consistent model chasing" and discuss how to solve it with existing techniques from online learning.

Chapter 5

ROBUST MODEL FREE LEARNING AND CONTROL OF LINEAR SYSTEMS

We present a simple model-free control algorithm that is able to robustly learn and stabilize an unknown discrete-time linear system with full control and state feedback subject to arbitrary bounded disturbance and noise sequences. The controller does not require any prior knowledge of the system dynamics, disturbances, or noise, yet it can guarantee robust stability and provides asymptotic and worst-case bounds on the state and input trajectories. To the best of our knowledge, this is the first model-free algorithm that comes with such robust stability guarantees without the need to make any prior assumptions about the system. We would like to highlight the new convex geometry-based approach taken towards robust stability analysis, which served as a key enabler in our results. We will conclude with simulation results that show that despite generality and simplicity, the controller demonstrates good closed-loop performance.

5.1 Introduction

Motivation and Problem Statement

Learning to stabilize unknown dynamical systems from online data has been an active research area in the control community since the 1950s [74] and has recently attracted the attention of the machine learning community, in particular in the context of reinforcement learning. Although extensive research has been conducted on this topic, very few of the algorithms developed have reached the level of adoption in real-world applications, as one would expect. In particular in areas where frequent interaction with the physical world is necessary, system failure is costly and the deployment of control algorithms is only possible if the algorithm can guarantee that minimal safety and performance specifications will be met during operation. Although there have been previous research [132],[18] and recent research efforts [57],[43], [2], [24], [38],[44] to address this problem, very few algorithms have come with the necessary performance and safety guarantees to be deployed in real-world applications so far. Motivated by this, we revisit the basic problem of learning to stabilize a linear system and aim to find learning and control strategies with the least restrictive assumptions that can still give robust stability bounds for the closed loop.

In this chapter, we focus on the problem of adaptively stabilizing a linear discrete-time system

$$z_{k+1} = A_0 z_k + u_k + d_k \tag{5.1a}$$

$$x_k = z_k + n_k \tag{5.1b}$$

with state z_k , bounded disturbance d_k , bounded noise n_k , and control action u_k that is only allowed to depend on noisy state measurements until time k, i.e., x_0, \ldots, x_k . We are interested in finding controllers that can stabilize (in the sense of BIBOor input-to-state stability guarantees) without requiring any additional assumptions about the unknown system matrix A_0 and the disturbance/noise sequences (d_k) , (n_k) . Although the system (5.1) admits a very restrictive class of linear systems (full state feedback and control), nearly all available learning and control approaches need to make some prior assumptions about this system in order to state stability and performance guarantees. Most commonly, these assumptions come in the form of a priori bounds on d_k , n_k , and/or A_0 .

Related Work

We will review the relevant literature in the context of our problem setting. Classical control approaches are found in the literature on adaptive control with [72], [71], [112] focusing on the deterministic setting and [132] on the stochastic setting. The self-tuning regulator [18] and its variations come with asymptotic optimality [60], yet robust stability guarantees without restrictive assumptions are few and can only be made in the probabilistic sense. On the deterministic side, [72], [71] point out that instabilities can occur with traditional adaptive schemes and provide improved versions of adaptive controllers that come with robust stability and performance guarantees. However, the desired guarantees depend on knowing some limits of the system parameters and disturbance signals. Other challenges associated with classical adaptive control approaches are discussed in [8], [10]. Methods in safe reinforcement learning [57], [24], [5], [53] have made great progress toward methods that guarantee robust safety properties for classes of non-linear systems, yet the synthesis procedures involved are computationally expensive, and require knowledge of an initially robust stabilizing controller, even in the case of a simple linear system (5.1). Recent work [2], [42], [43], [92], [38], [44] has made significant progress in providing algorithms with robust finite-time performance guarantees for the adaptive linear quadratic Gaussian regulator problem. However, in the context of our simple linear problem setting, all methods require that the uncertainty in the system dynamics

(i.e., A_0) is small enough at the outset to provide stability guarantees of the closed loop.

Main Contribution and Overview of the Chapter

In this work, we present a simple controller that can adaptively stabilize (5.1) without any additional assumptions about disturbance, noise, or the system matrix A_0 . The presented algorithm performs tractable computations (solving a linear program each time step) and provides both uniform asymptotic and worst-case guarantees on the state and input trajectories. An additional surprising feature of the presented algorithm is that it is not based on the certainty-equivalence principle and has a completely model-free formulation. To the best of our knowledge, this is the first model-free adaptive controller that can give our robust stability guarantees without requiring prior knowledge about the unknown system (5.1).

Our core theoretical contribution is a novel approach towards stability analysis. We first show that in any closed-loop trajectory (x_t) , there are only a finite number of time instances t_i at which x_{t_i+1} is significantly larger than x_{t_i} . We term those timeinstances as "unstable transitions" and our first main theorem shows an upper-bound on the occurrence of these unstable transitions in the closed loop state trajectory. Then, our stability and performance guarantees follow as corollaries of this result. We develop a new technique based on convex geometry to bound the occurrence of unstable transitions in the closed loop. Vaguely speaking, our main idea is to show that if an unstable transition occurs at some time t', our proposed adaptive controller learns enough from this observation to prevent similar unstable transitions from occurring in the future. Mathematically, we formulate this idea in two steps: 1. We define a distance function d between unstable transitions and show that, w.r.t. to d, we can identify the set of unstable transitions with a bounded separated set P of equal cardinality. 2. We bound the cardinality of P by a metric-entropy type of quantity, which leads to an upper bound on the maximum number of times that unstable transitions can occur. We discuss the convex geometry-based techniques in detail, to emphasize their potential use for robust design and analysis of learning and control algorithms, particularly in the model-free setting. The chapter is organized as follows. We formulate our problem in Section 5.2 and give a brief overview of our main results in Section 5.3. In Section 5.5, we derive the model-free closed-loop equation and explain the intuition behind the proposed control law. In Section 5.6, we present and discuss our main results in detail. Section 5.7 and Section 5.8 highlight the main techniques and ideas used to prove our results. Section 5.9 highlights a

parallel between the role of metric entropy in the context of our results and in the context of statistical learning theory, which could serve as an interesting intersection for future research. We conclude with some experimental results in Section 5.10.

5.2 Problem Setup

For our discussion, we transform the system (5.1) w.r.t to the measurements x_t and obtain the equivalent¹ system

$$x_{t+1} = A_0 x_t + u_t + w_t$$

$$w_t := d_t + n_{t+1} - A_0 n_t,$$
(5.2)

where w_t represents the lumped bounded disturbance at time t which summarizes the influence on the system of the original noise and disturbance. A causal controller can be represented as a collection $\mathbf{K} = (K_0, K_1, ...)$ of control laws $K_t : (x_t, ..., x_0) \mapsto u_t$. The closed loop of \mathbf{K} and (5.2) is then described by the equation

$$x_{t+1} = A_0 x_t + K_t(x_t, \dots, x_0) + w_t.$$
(5.3)

Our goal is now to design K such that the closed-loop (5.3) is bounded-input boundedoutput stable for any A_0 and any bounded sequence² (w_t). More specifically, we want to design K such that any closed-loop trajectory (x_t) satisfies bounds of the form

$$\sup_{t} \|x_t\| \leq f_1(A_0, \sup_{t} \|w_t\|)$$
$$\limsup_{t \to \infty} \|x_t\| \leq f_2(A_0, \sup_{t} \|w_t\|)$$

for some fixed functions f_1 and f_2 .

5.3 Preview of Main Result

We will start by describing the implementation of our proposed controller and a summary of its performance guarantees in a closed-loop with system (5.2).

Proposed control strategy

For adaptive stabilization of (5.2) we propose a dynamic controller $K^{cc} \in C(\ell^{\chi}, \ell^{U})$, which at every time step t computes the input as

$$u_t = K_t^{\rm cc}(x_t, \dots, x_0, u_{t-1}, \dots, u_0)$$

¹Since $x_t = z_t + n_t$, controlling the system state z_t is equivalent to controlling the noisy measurement x_t .

²We use bracket notation to distinguish a sequence (w_t) from its element w_t at time t.

based on all previous measurements x_t, \ldots, x_0 and previously taken actions u_{t-1}, \ldots, u_0 . The controller \mathbf{K}^{cc} computes the input u_t at time t as

$$K_t^{\rm cc}(x_t,\ldots,x_0,u_{t-1},\ldots,u_0) := \left(U_{t-1} - X_{t-1}^+\right)\lambda_{t-1}(x_t) \tag{5.4}$$

where $\lambda_{t-1}(x_t)$ is defined as the solution of the convex optimization problem

$$\min_{\lambda} \|\lambda\|_{1}$$
s.t. $X_{t-1}\lambda = x_{t}$

$$(5.5)$$

and where the matrices U_t , X_t and X_t^+ are composed of state x_t and input u_t measurements up until time t as

$$X_t := [x_t, x_{t-1}, \dots, x_0, X_{-1}]$$
(5.6a)

$$U_t := [u_t, u_{t-1}, \dots, u_0, U_{-1}]$$
(5.6b)

$$X_{t-1}^{+} := [x_t, x_{t-1}, \dots, x_1, X_{-1}^{+}]$$
(5.6c)

with fixed chosen matrices X_{-1} , U_{-1} , $X_{-1}^+ \in \mathbb{R}^{n \times n_0}$ such that $n_0 > n$ and $\operatorname{rank}(X_{-1}) = n$. The matrices X_{-1} , U_{-1} , X_{-1}^+ with columns \hat{x}_i , \hat{u}_i and \hat{x}_i^+ defined as

$$X_{-1} := [\hat{x}_1, \dots, \hat{x}_{n_0}], \quad U_{-1} := [\hat{u}_1, \dots, \hat{u}_{n_0}], \\ X_{-1}^+ := [\hat{x}_1^+, \dots, \hat{x}_{n_0}^+]$$
(5.7)

serve to initialize the controller K^{cc} . Depending on the application scenario, the matrices can be chosen as follows:

- (I1) No prior knowledge: choose $\hat{u}_i = 0$, $\hat{x}_i^+ = 0$, $\hat{x}_i = \varepsilon e_i$ for $1 \le i \le n$ where e_i is the *i*th Cartesian unit vector and $\varepsilon > 0$ is some positive scalar. The parameter ε can be viewed as an initial guess on $\sup_t ||w_t||_1$, the supremum of the disturbance sequence in 1-norm.
- (I2) **Prior data available:** Assume we had noisy data available x_{-j} , x_{-j}^+ , u_{-j} , $1 \le j \le k$ collected from the system (5.2) before t = 0. i.e., the data satisfies

$$x_{-j}^{+} = A_0 x_{-j} + u_{-j} + w_{-j}$$
(5.8)

with w_{-j} denoting the corresponding lumped disturbances. Then, in addition to the initialization (I1) $\hat{u}_i = 0$, $\hat{x}_i^+ = 0$, $\hat{x}_i = \varepsilon e_i$ for $1 \le i \le n$, we can incorporate the data x_{-j} , x_{-j}^+ , u_{-j} by appending additional columns as

$$\hat{u}_{n+i} := u_{-i}, \ \hat{x}_{n+i} := x_{-i}, \ \hat{x}_{n+i}^+ := x_{-i}^+, \ 1 \le i \le k.$$

Closed-Loop Guarantees

In this chapter, we will show that the controller K^{cc} stabilizes system (5.1) without requiring any knowledge of A_0 or (w_t) . We will term the controller K^{cc} defined by (5.4), (5.5) the *causal cancellation controller*, as it can be interpreted at time t to cancel out the part of the dynamics that can be inferred from all previously collected observations x_t, \ldots, x_0 and actions u_{t-1}, \ldots, u_0 .

As presented in detail in Section 5.6, for **any** initialization X_{-1} , U_{-1} , X_{-1}^+ (only assuming rank $(X_{-1}) = n$), the controller (5.5) **always** ensures a closed loop for which:

- (i) the state (x_t) and input (u_t) are uniformly bounded.
- (ii) an analytic upper-bound can be derived for the worst-case state-deviation.
- (iii) after some finite time, (x_t) and (u_t) converge exponentially to a bounded limit set.

The above guarantees will be phrased w.r.t. a norm $\|\cdot\|_W$ which measures (x_t) and (u_t) relative to the size of the disturbance (w_t) that produced them. Moreover, as described in (I2), we can incorporate prior data into the initialization of the controller \mathbf{K}^{cc} . In the case where the provided data is "more informative" than the default initialization (I1), the closed loop guarantees and bounds tighten. Hence, the proposed control scheme \mathbf{K}^{cc} does not need prior knowledge to give closed-loop stability guarantees, but if prior knowledge is available, it can be leveraged through the initialization (I2) to improve closed-loop guarantees.

5.4 Preliminaries of Convex Geometry

Convex Bodies and Norms

It is a well-known fact in convex geometry that there is a one-to-one relationship between symmetric convex bodies (see Def. 5.1) and norms in \mathbb{R}^n . Here we will discuss this equivalence and how it relates to the properties of the $\|\cdot\|_{S}$ -norms which we used in this chapter. The following discussion is adapted from chapter 1.7 of the standard text [113], to which we refer for more detail.

Definition 5.1 (Symmetric Convex Body). *A set* $B \subset \mathbb{R}^n$ *is a symmetric convex body if* B *is a closed, bounded, convex set with non-empty interior and* $z \in B \Leftrightarrow -z \in B$.

Symmetric convex bodies and norms are equivalent in \mathbb{R}^n , in the sense that any norm on \mathbb{R}^n is uniquely defined by its corresponding unit norm ball, and the set of all possible unit norm balls in \mathbb{R}^n is precisely the set of symmetric convex bodies in \mathbb{R}^n . We summarize this in the following lemma:

Lemma 35. For any norm $\|\cdot\|$ in \mathbb{R}^n , the corresponding norm ball $\{x | \|x\| \le 1\}$ is a symmetric convex set in \mathbb{R}^n . On the contrary, for any symmetric convex body B in \mathbb{R}^n , the function $g(\mathsf{B}, \cdot) : \mathbb{R}^n \mapsto \mathbb{R}^+$ defined by

$$g(\mathsf{B}, x) := \min\left\{r \ge 0 \left| x \in r\mathsf{B} \right\}, \quad \forall x \in \mathbb{R}^n$$
(5.9)

is a norm on \mathbb{R}^n .

In convex geometry, the function $g(B, \cdot)$ is often called the *gauge function* or the *Minkowski functional* of B and describes a concrete way to evaluate a norm based on knowing its unit ball. For our purposes, it will be convenient to extend the above definition to derive norms from general bounded sets $S \subset \mathbb{R}^n$ in the following way: Given an arbitrarily bounded set S, we will refer to $\|\cdot\|_S$ as the norm $g(c(S), \cdot)$, obtained by the Minkowski functional of the *absolute convex hull* c(S) of the set S. We define this formally in Def. 5.4 and Def. 5.5:

Definition. Let S be a set in \mathbb{R}^n , then the set of all finite linear combinations $\sum_{i=1}^{N} \lambda_i x_i$ of elements x_i in S with $\sum_{i=1}^{N} |\lambda_i| \leq 1$ is called the absolute convex hull of S, and we will refer to its closure as c(S):

$$c(\mathsf{S}) := cl \left\{ \sum_{i=1}^{N} \lambda_i x_i \, \middle| \quad \{x_i\}_{i=1}^{N} \subset \mathsf{S}, \quad \sum_{i=1}^{N} |\lambda_i| \leq 1 \right\}.$$
(5.10)

Remark. Equivalently, c(S) is the closure of the convex hull of the set $(-S) \cup S$.

Definition. For a fixed bounded set $S \subset \mathbb{R}^n$, let $\|\cdot\|_S : \mathbb{R}^n \mapsto \mathbb{R}_{\geq 0}$ be the norm *defined for all* $x \in \mathbb{R}^n$ *as*

$$\|x\|_{\mathsf{S}} := \begin{cases} \min \{r \ge 0 | x \in rc(\mathsf{S})\}, & \text{for } x \in \operatorname{span}(\mathsf{S}) \\ \infty, & \text{else} \end{cases}$$

and for sets $S' \subset \mathbb{R}^n$, define $\|S'\|_S$ as the quantity

$$\|\mathsf{S}'\|_{\mathsf{S}} := \max_{z \in c(\mathsf{S}')} \|z\|_{\mathsf{S}} \,. \tag{5.11}$$

The definition is overloading the common notation for the norm $\|\cdot\|_{S}$ as used in [113] where S is required to be a symmetric convex body. Applying Def. 5.5 to the disturbance set W, (5.26) defines the previously introduced norm $\|\cdot\|_{W}$ that we use to formulate the stability analysis. Figure 5.1 illustrates an example of how the set c(W) and the norm $\|x\|_{W}$ are related in two dimensions. The following lemma summarizes some key properties following from the above definitions. Note property (iii) and (iv), which show that we can verify the set-membership and set inclusions of symmetric convex bodies in terms of the norm. Furthermore, property (ii) describes a practical evaluation of $\|\cdot\|_{S}$ for finite sets S and shows

$$\|\lambda_{t-1}(x)\|_{1} = \min_{\lambda} \|\lambda\|_{1} = \|x\|_{\mathsf{X}_{t-1}}$$

s.t. $X_{t-1}\lambda = x.$ (5.12)

Lemma 36. Norms according to Definition 5.5 satisfy:

- (*i*) For any set S in Def. 5.5, c(S) is a symmetric convex body in \mathbb{R}^n . Moreover, c(S) is the unit norm ball of $\|\cdot\|_S$, so we can equiv. Write $\|\cdot\|_S = \|\cdot\|_{c(S)}$.
- (ii) If S is a finite set $S = \{p_1, \ldots, p_N\}$, then for any $x \in \mathbb{R}^n$, $||x||_S$ can be computed as:

$$\|x\|_{\mathsf{S}} = \min\left\{\sum_{i=1}^{N} |\lambda_i| \ \left|\lambda_1, \ldots, \lambda_N \text{ s.t. } \sum_{i=1}^{N} \lambda_i p_i = x\right\}\right\}.$$

- (iii) for all $x \in \mathbb{R}^n$ holds $x \in c(\mathsf{S}) \Leftrightarrow ||x||_{\mathsf{S}} \leq 1$.
- (iv) $c(S_1) \subset c(S_2)$ holds if and only if for all $x \in \mathbb{R}^n$ holds $||x||_{S_1} \ge ||x||_{S_2}$.
- (v) for all $\gamma > 0$, holds $\|\cdot\|_{\frac{1}{\gamma}\mathsf{S}} = \gamma \|\cdot\|_{\mathsf{S}}$.

Proof. The statements of Lem. 36 are easy to verify: (i), (iii), (iv) follow directly from Lem. 35 and (ii) follows by using the description of the set shown in (5.10) to rewrite the definition of $\|\cdot\|_{S}$.

We use the definition (5.11) of $\|S_1\|_{S_2}$ to measure the size of a set S_1 w.r.t. the norm $\|\cdot\|_{S_2}$ of another set S_2 . The following properties can be easily verified:

Lemma 37. Let S_1 , S_2 be some bounded sets in \mathbb{R}^n and recall the definitions Def. 5.5. Then we have the following.

(*i*)
$$\|S_1\|_{S_2} := \min\{ t \mid S_1 \subset tc(S_2), t \ge 0 \}.$$

- (ii) equivalence of norms in \mathbb{R}^n : $\frac{1}{\|\mathsf{S}_1\|_{\mathsf{S}_2}} \| \cdot \|_{\mathsf{S}_2} \leq \| \cdot \|_{\mathsf{S}_1} \leq \|\mathsf{S}_2\|_{\mathsf{S}_1} \| \cdot \|_{\mathsf{S}_2}$.
- (*iii*) $\|\mathsf{S}_1\|_{\mathsf{S}_2} \leq 1 \Leftrightarrow \mathsf{S}_1 \subset \mathrm{c}(\mathsf{S}_2).$

(*iv*)
$$S_1 \subset ||S_1||_{S_2} c(S_2)$$
.

Property (i) states that we can equivalently define $||S_1||_{S_2}$ as the smallest factor *t* such that S_1 is contained in $tc(S_2)$. This definition is visualized in the middle plot of Figure 5.1 for some exemplary sets W and X₁ in \mathbb{R}^2 . The other properties can be derived as immediate consequences of property (i): (i) \Rightarrow (iv) \Rightarrow (iii), (ii).

Distance Between Norms

For bounded sets S_1 , S_2 we define $d(S_1, S_2)$ as a multiplicative distance $d(S_1, S_2)$ between the two norms $\|\cdot\|_{S_1}$ and $\|\cdot\|_{S_2}$:

Definition 5.2. Let $S_1, S_2 \subset \mathbb{R}^n$ be sets with norms $\|\cdot\|_{S_1}$, $\|\cdot\|_{S_2}$ defined as in Def. 5.5. Then, define $d(S_1, S_2)$ as

$$d(\mathsf{S}_1,\mathsf{S}_2) := \max\{\|\mathsf{S}_1\|_{\mathsf{S}_2}, \|\mathsf{S}_2\|_{\mathsf{S}_1}\}.$$
(5.13)

Lemma 38. *The definitions 5.2 imply the following.*

- (*i*) d(S, S) = 1.
- (*ii*) $d(S_1, S_2) = d(S_2, S_1)$.
- (*iii*) $d(S_1, S_2) \leq d(S_1, S')d(S', S_2)$.

Proof. Statement (i) and (ii) are trivial. Part (iii) follows using (iv) and (i) of Lem. 37: Note that

$$\mathsf{S}_1 \subset d(\mathsf{S}_1,\mathsf{S}')c(\mathsf{S}') \subset d(\mathsf{S}_1,\mathsf{S}')d(\mathsf{S}',\mathsf{S}_2)c(\mathsf{S}_2) \tag{5.14}$$

$$\mathsf{S}_2 \subset d(\mathsf{S}_2,\mathsf{S}')c(\mathsf{S}') \subset d(\mathsf{S}_2,\mathsf{S}')d(\mathsf{S}',\mathsf{S}_1)c(\mathsf{S}_1) \tag{5.15}$$

leads to $\max\{\|\mathsf{S}_1\|_{\mathsf{S}_2}, \|\mathsf{S}_2\|_{\mathsf{S}_1}\} \le d(\mathsf{S}_2, \mathsf{S}')d(\mathsf{S}', \mathsf{S}_1).$

Lem. 38 shows that the map $d(\cdot, \cdot)$ can be viewed as a multiplicative distance between sets: These properties imply that $\log d(\cdot, \cdot)$ is a pseudometric over the space of bounded sets in \mathbb{R}^n .

5.5 The Model-Free Adaptive Controller and Closed-Loop Equations

We will start by first discussing the key idea and intuition behind the causal cancelation controller K^{cc} . Section 5.5 is deriving that all (X_t^+, X_t, U_t) satisfies the open loop equation of the unknown system for some appropriately defined disturbance matrix W_t . This is used to show in Section 5.5 that at time t and state x_t , the causal cancelation control law K_t^{cc} approximates the ideal deadbeat control action $u_t^* = -A_0 x_t$ directly from online data $(X_{t-1}^+, X_{t-1}, U_{t-1})$ without requiring an explicit estimate A_0 . This relation leads to a model-free form of the closed loop equation, shown in Section 5.5, which is used for the later stability analysis.

Open Loop Equation for Data Matrices

Recall from (5.6), that (X_t^+, X_t, U_t) are constructed from some fixed initialization $(X_{-1}^+, X_{-1}, U_{-1})$ and some state (x_t) and input (u_t) sequences of the system (5.2) with respect to some fixed lumped disturbance (w_t) . Define the disturbance matrix $W_t \in \mathbb{R}^{n \times (t+1+n_0)}$ as the matrix

$$W_t := [w_t, w_{t-1}, \dots, w_0, W_{-1}]$$
(5.16a)

$$W_{-1} := [\hat{w}_1, \dots, \hat{w}_{n_0}] := X_{-1}^+ - A_0 X_{-1} - U_{-1}$$
(5.16b)

of lumped disturbances w_t, \ldots, w_0 and the matrix W_{-1} which is composed of the columns $\hat{w}_1, \ldots, \hat{w}_{n_0}$. With the above auxiliary definition, we can easily see that the matrices U_t, X_t, X_t^+ , and W_t satisfy the linear equation

$$X_{t-1}^{+} = A_0 X_{t-1} + U_{t-1} + W_{t-1}, (5.17)$$

which resembles the open-loop dynamics of the unknown system. We will term \hat{w}_i as "virtual" disturbances, which are defined to account for errors introduced through the initial guesses X_{-1} , U_{-1} , X_{-1}^+ . If we take $\hat{x}_i, \hat{u}_i, \hat{x}_i^+$ to be the *i*th columns of the initialization matrices X_{-1} , U_{-1} , X_{-1}^+ and W_{-1} , we can rewrite the definition (5.16b) columnwise in the form

$$\hat{x}_{i}^{+} = A_{0}\hat{x}_{i} + \hat{u}_{i} + \hat{w}_{i}, \quad 1 \le i \le n_{0}$$
(5.18)

to see that each pair of (\hat{x}_i, \hat{u}_i) and \hat{x}_i^+ can be posed as a transition of the true unknown system (5.2), w.r.t. the virtual disturbance \hat{w}_i .

Example 2. If we initialize according to procedure (11), then $\hat{w}_i = -\varepsilon A_0 e_i$ and $W_{-1} = -\varepsilon A_0$.

In a compact form, the components $\{K_t^{cc}\}$ of the causal cancellation controller are mappings $(x_t, X_{t-1}^+, X_{t-1}, U_{t-1}) \mapsto u_t$ such that:

$$u_{t} = \left(U_{t-1} - X_{t-1}^{+}\right)\lambda_{t-1}(x_{t})$$
where $\lambda_{t-1}(x_{t}) := \underset{\lambda \text{ s.t. } X_{t-1}\lambda = x_{t}}{\arg\min} \|\lambda\|_{1}.$
(5.19)

Remark. The technical issue that a minimizer of (5.5) might not be unique is not relevant for the analysis and for simplicity will be ignored.

The function $\lambda_{t-1}(x_t)$ is defined to always satisfy

$$X_{t-1}\lambda_{t-1}(x_t) = x_t$$
 (5.20)

and represents a decomposition of the state x_t as a linear combination of the columns of X_{t-1} .

Rewriting equation (5.17) as

$$U_{t-1} - X_{t-1}^{+} = -A_0 X_{t-1} - W_{t-1}$$
(5.21)

and substituting the right hand side of equation (5.21) into (5.19) and using (5.20) allows us to rewrite the controller equivalently as

$$u_{t} = \left(U_{t-1} - X_{t-1}^{+}\right)\lambda_{t-1}(x_{t}) \stackrel{(5.20)}{=} -A_{0}x_{t} - W_{t-1}\lambda_{t-1}(x_{t}).$$
(5.22)

The above shows that the control law (5.19) is a direct way to approximate the ideal deadbeat control action $-A_0x_t$ from the online data matrices U_{t-1} , X_{t-1}^+ , X_{t-1} . The additional term $-W_{t-1}\lambda_{t-1}(x_t)$ is the corresponding approximation error at time t. As will become clear later, the optimization step in (5.19) is minimizing an upper bound of $-W_{t-1}\lambda_{t-1}(x_t)$ relative to the norm $\|\cdot\|_{W_{t-1}}$ (see definition in Section 5.6).

The Model-Free Closed-Loop equations

Setting $K = K^{cc}$ and using (5.22), the closed loop equations (5.3) take the form of

$$x_{t+1} = -W_{t-1}\lambda_{t-1}(x_t) + w_t$$
(5.23a)

$$u_t = \left(U_{t-1} - X_{t-1}^+\right)\lambda_{t-1}(x_t)$$
(5.23b)

which will serve as the basis for our stability analysis. The above equation says that the closed-loop dynamics is entirely determined by W_{t-1} (containing virtual and past lumped disturbances), as well as how we choose to decompose x_t as a linear combination $X_{t-1}\lambda_{t-1}(x_t)$ of past data. Suitably, we could call equation (5.23a) to be a model-free description of the closed loop, since the dynamics are formulated independent of the underlying true unknown system A_0 .

5.6 Main Results

Theorem 27 and Theorem 29 are the main theoretical results. Theorem 27 states that any state trajectory (x_t) of the closed loop has finitely many "unstable transitions" (defined in Def. 5.6). Theorem 29 is a consequence of Theorem 27 and presents our main stability bounds for state and input trajectories of the closed loop. To formulate our results, we first introduce the necessary notation and definitions.

Notation and Definitions

Definition 5.3. If $M \in \mathbb{R}^{n \times N}$ is a matrix with N columns m_i , then define the corresponding variable M in sans serif font to denote the set $M := \{m_1, \ldots, m_N\}$.

Definition 5.4. Let S be a set in \mathbb{R}^n , then the set of all finite linear combinations $\sum_{i=1}^{N} \lambda_i x_i$ of elements x_i in S with $\sum_{i=1}^{N} |\lambda_i| \leq 1$ is called the absolute convex hull of S and we will refer to its closure as c(S):

$$c(\mathsf{S}) := cl\left\{\sum_{i=1}^{N} \lambda_i x_i \middle| \quad \{x_i\}_{i=1}^{N} \subset \mathsf{S}, \quad \sum_{i=1}^{N} |\lambda_i| \leq 1\right\}.$$
(5.24)

Definition 5.5. For a fixed bounded set $S \subset \mathbb{R}^n$, let $\|\cdot\|_S : \mathbb{R}^n \mapsto \mathbb{R}_{\geq 0}$ be the norm *defined for all* $x \in \mathbb{R}^n$ *as*

$$\|x\|_{\mathsf{S}} := \begin{cases} \min\{r \ge 0 | x \in rc(\mathsf{S})\}, & \text{for } x \in \operatorname{span}(\mathsf{S}) \\ \infty, & \text{else} \end{cases}$$
(5.25)

and for sets $S' \subset \mathbb{R}^n$, define $\|S'\|_S$ as the quantity

$$\|S'\|_{S} := \max_{z \in c(S')} \|z\|_{S}.$$

Key properties of the above norm and relevant concepts from convex geometry are discussed in the appendix Section 5.4. For a fixed disturbance (w_t) and virtual disturbance \hat{w}^i , define W as the corresponding fixed set.

$$\mathsf{W} := \{ w_t | t \in \mathbb{N} \} \cup \{ \hat{w}_i | 1 \leqslant i \leqslant n_0 \}.$$
(5.26)

Let $\|\cdot\|_{W}$ and $\|\cdot\|_{X_t}$ denote the norms constructed from the fixed disturbances and



Figure 5.1: Examples in \mathbb{R}^2 . Left: c(W) and $||x||_W$ for $W = \{\hat{w}_1, \hat{w}_2\} \cup \{w_t | r \in \mathbb{N}\}$ and $(w_t) = (w_0, w_1, 0, 0, ...)$. Right: $c(X_1)$ and $|| \cdot ||_{X_1}$ for $X_1 = [x_1, x_0, \hat{x}_2, \hat{x}_1]$. Middle: $\kappa_2 = ||W||_{X_1}$ is the smallest factor r such that $c(W) \subset rc(X_1)$.

data matrix X_t according to Def. 5.5 and Def. 5.3. For a fixed trajectory (x_t) , W and fixed initial time τ , the constant κ_{τ} refers to the quantity

$$\kappa_{\tau} := \|\mathbf{W}\|_{\mathbf{X}_{\tau-1}}.$$
(5.27)

Figure 5.1 shows an example in \mathbb{R}^2 that illustrates the geometric relationship between the sets c(W), $c(X_t)$ and the evaluation of their respective norms at some point x. The arrows indicate that one set is a scaled copy of the other set. The middle picture in Figure 5.1 shows a geometric interpretation of the corresponding constant κ_{τ} for $\tau = 2$: κ_{τ} is the smallest scaling factor r such that the set $r \times c(X_{\tau-1})$ contains the set c(W).

Finite Occurrence of Unstable Transitions

Our approach is to analyze the behavior of the closed loop by quantifying how many "unstable transitions" can occur in the future time window $[\tau, \infty)$ of a closed-loop trajectory (x_t) , given $(X_{\tau}, X_{\tau-1}^+, U_{\tau-1})$, which represents the data collected up to time τ . For a fixed $0 < \mu < 1$ and a trajectory (x_t) , we define the occurrence of a μ -unstable transition as follows:

Definition 5.6 (μ -unstable transition). *The trajectory* (x_t) *has a* μ *-unstable transition at time t if the pair of consecutive states* (x_{t+1}, x_t) *satisfies*

$$\|x_{t+1}\|_{\mathsf{W}} > \max\left\{\frac{1}{1-\mu}, \mu \,\|x_t\|_{\mathsf{W}} + 1\right\}.$$
(5.28)

In other words, $(x_{t+1}, x_t) \in \mathcal{U}_{\mu}$, where \mathcal{U}_{μ} denotes the set of all pairs $(x, x^+) \in \mathbb{R}^n \times \mathbb{R}^n$ that satisfy the inequality (5.28):

$$\mathcal{U}_{\mu} := \left\{ (x^{+}, x) | \|x^{+}\|_{\mathsf{W}} > \max\left\{ \frac{1}{1-\mu}, \mu \|x\|_{\mathsf{W}} + 1 \right\} \right\}.$$
 (5.29)

The condition (5.28) represents a growth condition on a transition (x_t, x_{t+1}) on the trajectory (x_t) . For each trajectory (x_t) , we define a corresponding set \mathcal{X}_{μ} that collects all states x_t at which (x_{t+1}, x_t) belongs to \mathcal{U}_{μ} :

Definition 5.7. *Given a trajectory* (x_t) *, an initial time* τ *and some* $0 < \mu < 1$ *, define* $\mathcal{X}_{\mu}((x_t); \tau) \subset \mathbb{R}^n$ *as*

$$\mathcal{X}_{\mu}((x_{t});\tau) := \{ x_{t} \mid (x_{t}, x_{t+1}) \in \mathcal{U}_{\mu}, t \ge \tau \}.$$
(5.30)

Remark. Note that if $\mu < \mu'$, then $\mathcal{X}_{\mu}((x_t), \tau) \supset \mathcal{X}_{\mu'}((x_t), \tau)$.

The core technical contribution of our chapter is Theorem 27, which places an upper bound on the number of μ -unstable transitions that can occur in the closed loop trajectory (x_t) :

Theorem 27. For any trajectory (x_t) of the closed loop (5.23a) and any $\tau \ge 0$, the set $\mathcal{X}_{\mu}((x_t); \tau)$ is a finite set for any $\mu \in \mathcal{I}_{\kappa_{\tau}}$, where $\mathcal{I}_{\kappa_{\tau}}$ is the open interval.

$$\mathcal{I}_{\kappa_{\tau}} := \left(\left(\sqrt{\frac{1}{4} + \frac{1}{\kappa_{\tau}}} + \frac{1}{2} \right)^{-1}, 1 \right).$$
(5.31)

Moreover, the cardinality is bounded above as $|\mathcal{X}_{\mu}((x_t);\tau)| \leq N(\mu;\kappa_{\tau})$, where $N: \mathbb{R} \times \mathbb{R} \mapsto \mathbb{R}$ stands for the function

$$N(\mu;\kappa_{\tau}) := \frac{1}{2} \left(\frac{\mu}{\mu - \sqrt{\kappa_{\tau}(1-\mu)}} \right)^n \max\{1, \frac{\mu}{1-\mu}\}^n$$
(5.32)

and κ_{τ} is a constant computed from $X_{\tau-1}$ as:

$$\kappa_{\tau} = \|\mathbf{W}\|_{\mathbf{X}_{\tau-1}}.$$
(5.33)

Remark 28. Recall, that we initialize X_{-1} such that $\operatorname{rank}(X_{-1}) = n$; This guarantees $\operatorname{rank}(X_{\tau-1}) = n$, assures $\kappa_{\tau} < \infty$ and that the interval $\mathcal{I}_{\kappa_{\tau}}$ is always non-empty. In addition, it can be verified that $N(\mu; \kappa_{\tau}) < \infty$ for any feasible μ .

Theorem 27 states that for the suitably chosen μ , the set $\mathcal{X}_{\mu}((x_t); \tau)$ is finite for any closed loop trajectory (x_t) . The constant κ_{τ} controls the interval of feasible μ as well as the total number of unstable transitions \mathcal{U}_{μ} that can occur in the time interval $[\tau, \infty)$. As κ_{τ} decreases, the bound $N(\mu; \kappa_{\tau})$ tightens $(N(\mu; \kappa_{\tau}) \leq N(\mu; \kappa_{\tau}')$ for $\kappa_{\tau} \leq \kappa_{\tau}')$ and the interval (5.31) widens. Geometrically, κ_{τ} describes the size of the disturbance set W relative to the set $c(X_{\tau-1})$ (see Figure 5.1 for an example in \mathbb{R}^2) and the result states that we have fewer unstable transitions if the collected observations are larger in size than the disturbance. Therefore, we can view κ_{τ} as a constant that quantifies how informative the data $X_{\tau-1}$ observed before τ are to control the system for time $t \geq \tau$.

The proof of Theorem 27 is postponed to Section 5.8.

Closed Loop Stability Bounds

As a consequence of Theorem 27, we obtain our main closed loop stability bounds presented in Theorem 29. The result gives bounds on the trajectories (x_t) and (u_t) in terms of the fixed disturbance (w_t) and virtual disturbance \hat{w}_i .

Remark. Recall from (5.16b) that instead of analyzing the closed loop dynamics for fixed A_0 , X_{-1}^+ , X_{-1} , U_{-1} , we can equivalently analyze the closed loop dynamics for fixed \hat{w}_i .

Theorem 29. Let (x_t) , (u_t) be the trajectories of the closed loop (5.23a) for some fixed (w_t) and \hat{w}_i with the corresponding set W defined as (5.26). Let τ be some fixed time and let $\kappa_{\tau} := \|W\|_{X_{\tau-1}}$. Then, for any $\mu \in \mathcal{I}_{\kappa_{\tau}}$, where $\mathcal{I}_{\kappa_{\tau}}$ is the interval

$$\mathcal{I}_{\kappa_{\tau}} := \left(\left(\sqrt{\frac{1}{4} + \frac{1}{\kappa_{\tau}}} + \frac{1}{2} \right)^{-1}, 1 \right), \tag{5.34}$$

the trajectories (x_t) and (u_t) satisfy the bounds (i), (ii) and (iii):

- (i)
 $$\begin{split} \lim \sup_{t \to \infty} \|x_t\|_{\mathsf{W}} &\leq \frac{1}{1-\mu}, \\ \limsup_{t \to \infty} \|u_t\|_{\mathsf{W}} &\leq (\|A_0\|_{\mathsf{W}} + \kappa_{\tau}) \frac{1}{1-\mu}. \end{split}$$
- (ii) there exists an T' > 0 such that for all k > 0 holds

$$V_1(x_{T'+k}) \le \mu^k V_1(x_{T'}) \tag{5.35}$$

where $V_1(x) := \max\{0, \|x\|_{\mathsf{W}} - \frac{1}{1-\mu}\}.$

(iii) the worst-case norm of (x_t) and (u_t) is bounded above as ³

$$\sup_{t \ge \tau} \|x_t\|_{\mathsf{W}} \le f(\kappa_{\tau}, \mu, \|x_{\tau}\|_{\mathsf{W}}) + g(\mu, \kappa_{\tau})$$

$$\sup_{t \ge \tau} \|u_t\|_{\mathsf{W}} \le (\|A_0\|_{\mathsf{W}} + \kappa_{\tau}) \sup_{t \ge \tau} \|x_t\|_{\mathsf{W}}$$
(5.36)

where $N(\mu; \kappa_{\tau})$ is defined as the function

$$N(\mu;\kappa_{\tau}) := \frac{1}{2} \left(\frac{\mu}{\mu - \sqrt{\kappa_{\tau}(1-\mu)}} \right)^n \max\{1, \frac{\mu}{1-\mu}\}^n,$$
(5.37)

 $||A_0||_{\mathsf{W}} := \max_{x \in \mathsf{W}} ||A_0x||_{\mathsf{W}}$ is a constant and f and g abbreviate the functions

$$f(\kappa_{\tau}, \mu, \|x_{\tau}\|_{\mathsf{W}}) = \max\{1, \kappa_{\tau}^{N(\mu;\kappa_{\tau})}\} \max\{\frac{1}{1-\mu}, \|x_{\tau}\|_{\mathsf{W}}\}$$
$$g(\kappa_{\tau}, \mu) = \frac{1 - \kappa_{\tau}^{N(\mu;\kappa_{\tau})}}{1 - \kappa_{\tau}}.$$
(5.38)

³For $\tau = 0$ and $x_0 \notin \operatorname{span}(\mathsf{W})$, replace $||x_0||_{\mathsf{W}}$ with $||A_0x_0||_{\mathsf{W}}$ in (5.36).

The bounds in Theorem 29 are phrased w.r.t. to the norm $\|\cdot\|_W$ that is constructed from the set W (see Figure 5.1 as an example of $\|\cdot\|_W$ in \mathbb{R}^2). The set W captures disturbances due to (w_t) and due to \hat{w}_i , where \hat{w}_i describes the mismatch between the initial guess matrices X_{-1}^+ , X_{-1} , U_{-1} and the true system matrix A_0 . $\|x_t\|_W$ measures x_t relative to the underlying set of (lumped and virtual) disturbances W that realized it.

The result also quantifies how the bound guarantees improvement with online data: Given some initial time τ , the above result gives stability bounds on the future trajectories of x_t , u_t , $t \ge \tau$ which depend on the total states observed X_{τ} before time τ , the constant κ_{τ} and $\mu \in \mathcal{I}_{\kappa_{\tau}}$, which acts as a free variable. The constant κ_{τ} can be interpreted as a signal-to-noise ratio between state observations X_{τ} and the disturbance set W (see Figure 5.1 for an example in \mathbb{R}^2). A smaller κ_{τ} indicates that the data $X_{\tau-1}^+$, $X_{\tau-1}$, $U_{\tau-1}$ collected before time τ are more informative about how to stabilize the system for future time-steps $t \ge \tau$. κ_{τ} is always nonincreasing in τ and the bounds (iii), (i) of Theorem 29 tighten as τ increases. The bounds in Theorem 29 depend on a free variable μ which can be chosen in the interval $\mathcal{I}_{\kappa_{\tau}}$. We can tighten the bounds (i) and (iii) by minimizing the right-hand side over $\mu \in \mathcal{I}_{\kappa_{\tau}}$. For bound (i), the choice

$$\mu^* = \left(\sqrt{\frac{1}{4} + \frac{1}{\kappa_\tau}} + \frac{1}{2}\right)^{-1} \tag{5.39}$$

minimizes $\frac{1}{1-\mu}$ over $\mu \in \mathcal{I}_{\kappa_{\tau}}$ and achieves a minimal value which is almost linear in κ_{τ} :

$$\frac{1}{1-\mu^*} = \kappa_\tau \left(\frac{1}{2} + \sqrt{\frac{1}{4} + \frac{1}{\kappa_\tau}} \right) + 1 \quad (\leqslant \kappa_\tau + 2).$$
 (5.40)

For $\tau = 0$ we get the following improved asymptotic upper bound for the state trajectory:

Corollary 39. If (x_t) satisfies (5.23a) then

$$\limsup_{t \to \infty} \|x_t\|_{\mathsf{W}} \leq \kappa_0 \left(\frac{1}{2} + \sqrt{\frac{1}{4} + \frac{1}{\kappa_0}}\right) + 1 =: m(\kappa_0).$$

Example

Assume n = 1 and the scalar system $x_{t+1} = a_0 x_t + u_t + w_t$. Pick $X_{-1} = \varepsilon$ with some $\varepsilon > 0$ and X_{-1}^+ , $U_{-1} = 0$. Let (w_t) be some fixed bounded scalar disturbance with $||(w_t)||_{\infty} = 1$. Then $W = c(-a_0 \varepsilon \cup \{w_t | t \in \mathbb{N}\})$ and $||x||_W = \frac{|x|}{\max\{|a_0|\varepsilon,1\}}$. The constant κ_0 takes the value

$$\kappa_0 = \|\mathbf{W}\|_{\mathbf{X}_{-1}} = \max\{\varepsilon^{-1}, |a_0|\}$$

If we substitute this into Cor. 39, and rewrite it in terms of $|x_t|$ we obtain the bound

$$\limsup_{t \to \infty} |x_t| \leq \varepsilon \kappa_0^2 \left(\frac{1}{2} + \sqrt{\frac{1}{4} + \frac{1}{\kappa_0}} \right) + \varepsilon \kappa_0.$$
(5.41)

5.7 Proving Closed Loop Stability

In this section, we derive the closed loop stability bounds presented in Theorem 29 from the results of Theorem 27. The derivation of Theorem 27 is postponed to Section 5.8. First, we will derive some useful inequalities that are used frequently in the derivations.

Bounding One-Time Step Closed Loop Transitions

Recall the closed loop equation (5.23a) and the definition of the norm Def. 5.5 and the sets W and X_t. In the Appendix, Lem. 36 summarizes some important properties of the norms $\|\cdot\|_{S}$. We use these to obtain the following bounds on the one time-step growth of the state:

Lemma 40. Consider a state trajectory (x_t) of the closed loop for a fixed W, then at each time step $t > \tau$ holds:

$$\|x_{t+1}\|_{\mathsf{W}} \leq \left\|W_{t-1}\frac{\lambda_{t-1}}{\|\lambda_{t-1}(x_t)\|_1}\right\|_{\mathsf{W}} \|\lambda_{t-1}(x_t)\|_1 + 1$$
(5.42a)

$$\leq \left\|\lambda_{t-1}(x_t)\right\|_1 + 1 \tag{5.42b}$$

$$\leq \|x_t\|_{\mathsf{X}_{t-1}} + 1 \tag{5.42c}$$

$$\leq \|\mathbf{W}\|_{\mathbf{X}_{t-1}} \|x_t\|_{\mathbf{W}} + 1$$
 (5.42d)

$$\leq \kappa_{\tau} \|x_t\|_{\mathsf{W}} + 1. \tag{5.42e}$$

Recall that the vector $\lambda_{t-1}(x_t)$ poses as a linear decomposition of x_t in terms of the previous observations X_{t-1} , which is obtained through the minimization in (5.19). The right-hand side of the inequality (5.42c) and (5.42b) are equivalent. This follows from the equivalence relation

$$\|\lambda_{t-1}(\cdot)\|_1 = \|\cdot\|_{\mathsf{X}_{t-1}},$$
(5.43)

which follows from property (ii) of Lem. 36 and is discussed in the appendix. The inequality (5.42c) offers valuable insight into the closed loop behavior: The smaller x_t is relative to the absolute convex hull of all previous observations X_{t-1} , the tighter the bound is on $||x_{t+1}||_W$. Hence, $||x_t||_{X_{t-1}}$ captures how well we can control a certain state x_t given the observations made up until time t. If we rewrite $||x_t||_{X_{t-1}}$ as $||x_t||_W ||x_t/||x_t||_W ||_{X_{t-1}}$ and use the fact that the normalized vector $x_t/||x_t||_W$ lies in

the set W, we obtain the looser upper-bound (5.42d). Finally, (5.42e) is obtained by recalling that, per definition $\|W\|_{X_t}$ is non-increasing in t and therefore for all $t > \tau$ it holds $\|W\|_{X_{t-1}} \leq \|W\|_{X_{\tau-1}} = \kappa_{\tau}$.

Obtaining Bounds on Closed Loop Trajectories

Recall the definition of a μ -unstable transition in Def. 5.6 and consider Lem. 41: If a μ -unstable transition does not occur, (5.44) and (5.45) show that the quantities $V_1(x_t; \mu)$ and $V_2(x_t; \mu)$ do not increase for that time-step; On the other hand, (5.46) provides a bound on the increase of $V_2(x_t; \mu)$ if a μ -unstable transition does occur.

Lemma 41. Let (x_t) be a trajectory of (5.23a) with $t \ge 0$ and define the scalar functions $V_1(x;\mu) := \max\{0, \|x\|_W - \frac{1}{1-\mu}\}$ and $V_2(x;\mu) := \max\{\|x\|_W, \frac{1}{1-\mu}\}$. Then

(i) if $(x_{t+1}, x_t) \notin \mathcal{U}_{\mu}$, then

$$V_1(x_{t+1};\mu) \le \mu V_1(x_t;\mu)$$
 (5.44)

$$V_2(x_{t+1};\mu) \le V_2(x_t;\mu).$$
(5.45)

(*ii*) if $(x_{t+1}, x_t) \in \mathcal{U}_{\mu}$, then

$$V_2(x_{t+1};\mu) \le \kappa_t V_2(x_t;\mu) + 1$$
 (5.46)

$$V_1(x_{t+1};\mu) > \mu V_1(x_t;\mu).$$
(5.47)

Proof. See Appendix.

The bounds of Theorem 29 follow by combining the results of Theorem 27 with the above lemma. To highlight the main proof techniques, we focus only on the derivation of (i) and (ii) of Theorem 29 and refer to the Appendix for a detailed proof of the remaining statements.

Consider some arbitrary closed-loop trajectory (x_t) , fix $\tau = 0$, and choose some $\mu \in \mathcal{I}_{\kappa_0}$, where κ_{τ} depends on the set W and the initial guess matrix $X_{-1} = [\hat{x}_1, \ldots, \hat{x}_{n_0}]$. Recall that κ_{τ} measures the relative size between the disturbance set W and the set $c(X_{-1})$. According to Theorem 27, the trajectory (x_t) is guaranteed to have at most $N(\mu; \kappa_0)$ -many μ -unstable transitions. Hence, there is some finite time, call it $T'((x_t))$, such that for all time $t > T'((x_t))$ it holds $(x_{t+1}, x_t) \notin \mathcal{U}_{\mu}$, and therefore the reverse inequality of (5.28) holds, that is:

$$\|x_{t+1}\|_{\mathsf{W}} \le \max\left\{\frac{1}{1-\mu}, \mu \|x_t\|_{\mathsf{W}} + 1\right\}, \quad \forall t > T'((x_t)).$$
 (5.48)

Now, apply the statement (i) of Lem. 41, to conclude that for all $t > T'((x_t))$ holds $V_1(x_{t+1}; \mu) \leq \mu V_1(x_t; \mu)$. Therefore, we get the convergence bound.

$$V_1(x_{T'((x_t))+k};\mu) \le \mu^k V_1(x_{T'((x_t))};\mu), \quad k \ge 0$$
(5.49)

which proves that the trajectory (x_t) has to be bounded. We also conclude that $\lim_{t\to\infty} V_1(x_t;\mu) = 0$, which leads to the asymptotic bound.

$$\limsup_{t \to \infty} \|x_t\|_{\mathsf{W}} \le \limsup_{t \to \infty} (V_1(x_t; \mu) + \frac{1}{1-\mu}) = \frac{1}{1-\mu}.$$
 (5.50)

Similar type of arguments are used to derive the other statements of Theorem 29 and are presented in the Appendix.

5.8 Proving Finite Occurrence of Unstable Transitions

Here, we will discuss the key steps in proving Theorem 27. The general idea will be to first argue that if an unstable transition occurred at time t' and state $x_{t'}$, (i.e., $(x_{t'+1}, x_{t'}) \in \mathcal{U}_{\mu}$) then any future unstable transitions $(x_{t+1}, x_t) \in \mathcal{U}_{\mu}, t > t'$ must originate from some state x_t which is significantly different from $x_{t'}$; in a second step, we then prove that there is a finite upper bound on how many significantly "different" unstable transitions can occur in the same trajectory, which leads to the result presented in Theorem 27. In the following derivations we will make use of various simple facts from convex geometry, which are summarized in the appendix, Section 5.4. Matching the presentation of the theorem, in the derivations we will use the constant $\kappa_{\tau} := \|W\|_{X_{\tau-1}}$ corresponding to some fixed set W, the trajectory (x_t) of the closed loop (5.23a) and the initial time τ . Throughout the discussion, μ will represent some fixed value in the open interval.

$$\mathcal{I}_{\kappa_{\tau}} := \left(\left(\sqrt{\frac{1}{4} + \frac{1}{\kappa_{\tau}}} + \frac{1}{2} \right)^{-1}, 1 \right)$$
(5.51)

and δ will refer to the corresponding transformed variable $\delta := \frac{\mu^2}{1-\mu} \frac{1}{\kappa_{\tau}}$, which always satisfies $\delta > 1$. The following one-to-one relationship between both constants μ and δ will be frequently used and can be easily verified:

$$\delta = \frac{\mu^2}{1-\mu} \frac{1}{\kappa_{\tau}}, \quad \text{for } \mu \in \left(\left(\sqrt{\frac{1}{4} + \frac{1}{\kappa_{\tau}}} + \frac{1}{2} \right)^{-1}, 1 \right)$$
(5.52a)

$$\Rightarrow \qquad \mu = \left(\sqrt{\frac{1}{4} + \frac{1}{\delta\kappa_{\tau}}} + \frac{1}{2}\right)^{-1}, \quad \text{for } \delta \in (1, \infty). \tag{5.52b}$$

Our argument can be structured into the following three statements, which we prove separately in the next sections:

\$

- (a) We can radially project the set \mathcal{X}_{μ} onto the ball $\frac{\delta}{\mu}W$ and show that the resulting set, called \mathcal{P}_{μ} , has the same cardinality as \mathcal{X}_{μ} .
- (b) The set \mathcal{P}_{μ} forms a δ -separated subset of $\frac{\delta}{\mu}W$ with respect to a particularly chosen distance function $d(\cdot, \cdot; X_{\tau-1})$.
- (c) There are some constants c and C, such that for any δ separated subset P of $\frac{\delta}{\mu}W$ we can construct a superset $P \subset \mathcal{N}(P)$ in \mathbb{R}^n whose volume can be bounded above and below as $|\mathsf{P}|c_{\text{in}} \leq \operatorname{Vol}(\mathcal{N}(\mathsf{P})) \leq C_{\text{out}}$; hence, the cardinality of any δ separated set, included \mathcal{P}_{μ} , is bounded above by $\frac{C_{\text{out}}}{c_{\text{in}}}$.

Projection Onto the Ball $\frac{\delta}{u}W$

Define the projection $\Pi_{\mu} : \mathbb{R}^n \mapsto \frac{\delta}{\mu} \mathbb{W}$ as $\Pi_{\mu}(p) := \frac{\delta}{\mu \|p\|_{W}} p$ and define $\mathcal{P}_{\mu}((x_t); \tau)$ as the set resulting from applying Π_{μ} to every point in $\mathcal{X}_{\mu}((x_t); \tau)$:

$$\mathcal{P}_{\mu}((x_t);\tau) := \{ \Pi_{\mu}(x_t) \mid x_t \in \mathcal{X}_{\mu}((x_t);\tau) \}.$$
(5.53)

Remark. To limit the notational burden, we will state the explicit dependency on the trajectory (x_t) and τ only in lemmas and theorems. For the derivations, we will simply write \mathcal{X}_{μ} , \mathcal{P}_{μ} instead of $\mathcal{P}_{\mu}((x_t); \tau)$, $\mathcal{X}_{\mu}((x_t); \tau)$.

Per construction, for every point $p \in \mathcal{P}_{\mu}$ holds $||p||_{W} = \frac{\delta}{\mu}$ and therefore each $p \in \mathcal{P}_{\mu}$ lies on the surface of the ball $\frac{\delta}{\mu}W$. Recall that for a time instance t, where $x_t \in \mathcal{X}_{\mu}$ holds.

$$\|x_{t+1}\|_{\mathsf{W}} > \max\left\{\frac{1}{1-\mu}, \mu \|x_t\|_{\mathsf{W}} + 1\right\}$$

$$\|x_{t+1}\|_{\mathsf{W}} \leq \|x_t\|_{\mathsf{Y}} + 1,$$
(5.54a)

$$\leq \|\mathbf{W}\|_{\mathbf{X}_{t-1}} \|x_t\|_{\mathbf{W}} + 1$$
(5.54b)

where (5.54a) is due to the definition of the set \mathcal{X}_{μ} and (5.54b) follows from Lem. 40. Combining the above inequalities, we can further establish that any $x_t \in \mathcal{X}_{\mu}$ also satisfies the inequalities (5.55a):

Lemma 42.

$$\mu \|x_t\|_{\mathsf{W}} > \frac{\mu^2}{(1-\mu)} \frac{1}{\|\mathsf{W}\|_{\mathsf{X}_{t-1}}}$$
(5.55a)

$$\left\|\frac{1}{\mu \|x_t\|_{\mathsf{W}}} x_t\right\|_{\mathsf{X}_{t-1}} > 1.$$
(5.55b)

Proof. Combining the lower-bound (5.54a) and the upper-bound (5.54b) yields

$$\|x_t\|_{\mathsf{X}_{t-1}} + 1 > \mu \|x_t\|_{\mathsf{W}} + 1$$
$$\|\mathsf{W}\|_{\mathsf{X}_{t-1}} \|x_t\|_{\mathsf{W}} > \frac{1}{1-\mu} - 1 = \mu \frac{1}{1-\mu}.$$

Using the above inequalities we can show in Lem. 43 that \mathcal{P}_{μ} has the same cardinality as \mathcal{X}_{μ} . Hence, instead of reasoning about the size of \mathcal{X}_{μ} directly, we can equivalently study the size of the set \mathcal{P}_{μ} . As will become apparent in the following sections, the main advantage of analyzing the projected set \mathcal{P}_{μ} rather than \mathcal{X}_{μ} is that we can leverage \mathcal{P}_{μ} as a subset of $\frac{\delta}{\mu}W$.

Lemma 43. $|X_{\mu}| = |P_{\mu}|.$

Proof. From the definition of \mathcal{P}_{μ} it is clear that \mathcal{P}_{μ} has at most as many elements as \mathcal{X}_{μ} , hence trivially we have $|\mathcal{P}_{\mu}| \leq |\mathcal{X}_{\mu}|$. To establish $|\mathcal{P}_{\mu}| \geq |\mathcal{X}_{\mu}|$, we have to show that there are no two time instances $t_1 \neq t_2$ for which $x_{t_1}, x_{t_2} \in \mathcal{X}_{\mu}$ gets mapped to the same point $p \in \mathcal{P}_{\mu}$. For the sake of proof by contradiction, assume for some $x_{t_1}, x_{t_2} \in \mathcal{X}_{\mu}$ where w.l.o.g. $t_1 < t_2$, holds $\delta(\mu ||x_{t_1}||_W)^{-1}x_{t_1} = \delta(\mu ||x_{t_2}||_W)^{-1}x_{t_2}$. Then, using Lem. 42 it follows:

$$\left\| \frac{1}{\mu \| x_{t_1} \|_{\mathsf{W}}} x_{t_1} \right\|_{\mathsf{X}_{t_2-1}} = \left\| \frac{1}{\mu \| x_{t_2} \|_{\mathsf{W}}} x_{t_2} \right\|_{\mathsf{X}_{t_2-1}} \overset{(5.55b)}{>} 1$$

$$\Rightarrow \| x_{t_1} \|_{\mathsf{X}_{t_2-1}} > \mu \| x_{t_1} \|_{\mathsf{W}} \overset{(5.55a)}{>} \frac{\mu^2}{(1-\mu)} \frac{1}{\|\mathsf{W}\|_{\mathsf{X}_{t_1-1}}}. \tag{5.56}$$

Now, since $t_2 > t_1$, it is clear that $x_{t_1} \in c(X_{t_2-1})$ and therefore $||x_{t_1}||_{X_{t_2-1}} \leq 1$. Furthermore, with (5.56) and since μ is in the interval $\mathcal{I}_{\kappa_{\tau}}$, we are forced to conclude the following

$$\|\mathbf{W}\|_{\mathbf{X}_{t_{1}-1}} > \frac{\mu^{2}}{(1-\mu)} \ge \|\mathbf{W}\|_{\mathbf{X}_{\tau-1}}$$
(5.57)

which is a contradiction, since $t_1 \ge \tau$ and we know that $\|\mathbf{W}\|_{\mathbf{X}_t}$ is non-increasing in t.

Separateness of the Set \mathcal{P}_{μ}

The previous section established, that the bounded set $\mathcal{P}_{\mu} \subset \frac{\delta}{\mu} W$ has equal number of elements as \mathcal{X}_{μ} . Here, we will show that the points in the set \mathcal{P}_{μ} are "evenly spread" across the surface of $\frac{\delta}{\mu} W$. Formally, we will term \mathcal{P}_{μ} to be a δ -separated subset of $\frac{\delta}{\mu}W$. This property will ultimately lead to the cardinality bound derived in the next Section 5.8. The next lemma shows that any two points $p, p' \in \mathcal{P}_{\mu}, p \neq p'$ respect the inequality (5.58).

Lemma 44. Let $\mathcal{P}_{\mu}((x_t); \tau)$ be the projected set (5.53) and recall the definitions of the variables δ , μ and κ_{τ} in (5.52). Then, for any two distinct points $p_1, p_2 \in \mathcal{P}_{\mu}((x_t); \tau)$, $p_1 \neq p_2$ holds:

$$\max\{\|p_2\|_{\mathsf{X}_{\tau-1}\cup\mathsf{p}_1}, \|p_1\|_{\mathsf{X}_{\tau-1}\cup\mathsf{p}_2}\} > \delta.$$
(5.58)

Proof. Fix two arbitrary and distinct points $p_1, p_2 \in \mathcal{P}_{\mu}, p_1 \neq p_2$, then according to the definition of \mathcal{P}_{μ} there are two corresponding elements $x_{t_1}, x_{t_2} \in \mathcal{X}_{\mu}((x_t); \tau)$ with $t_1 \neq t_2$ such that $p_1 = \delta(\mu \| x_{t_1} \|_W)^{-1} x_{t_1}$ and $p_2 = \delta(\mu \| x_{t_2} \|_W)^{-1} x_{t_2}$. We will prove the desired statement, by showing that depending on which unstable transition occurred first, i.e., $t_2 > t_1$ or $t_1 < t_2$, either $\| p_2 \|_{\mathsf{X}_{\tau-1} \cup \mathsf{p}_1} > \delta$ or $\| p_1 \|_{\mathsf{X}_{\tau-1} \cup \mathsf{p}_2} > \delta$ has to be satisfied. The inequality (5.58) then follows taking the maximum of both cases. Therefore, to complete our argument, we will assume the case $t_2 > t_1$ and proceed to prove $\| p_2 \|_{\mathsf{X}_{\tau-1} \cup \mathsf{p}_1} > \delta$; The case $t_1 < t_2$ then follows by interchanging t_1 and t_2 : First, notice that since $\| W \|_{\mathsf{X}_{t_1-1}} \leq \| W \|_{\mathsf{X}_{\tau-1}}$, we can conclude that for any $x_t \in$ $\mathcal{X}_{\mu}((x_t); \tau)$, the following inequality is satisfied:

$$\frac{\delta}{\mu \|x_t\|_{\mathsf{W}}} \leqslant \frac{\mu^2}{(1-\mu)} \frac{1}{\|\mathsf{W}\|_{\mathsf{X}_{t-1}}} \frac{1}{\mu \|x_t\|_{\mathsf{W}}} < 1.$$
(5.59)

Now, for x_{t_2} recall from (5.55b) that

$$\|p_2\|_{\mathsf{X}_{t_2-1}} = \left\|\frac{\delta}{\mu \|x_{t_2}\|_{\mathsf{W}}} x_{t_2}\right\|_{\mathsf{X}_{t_2-1}} > \delta.$$
(5.60)

We will now use repeatedly the property (iv) of Lem. 36, to bound the left-hand side of (5.60) from above. To this end, consider first the following chain of inclusions:

$$c(\mathsf{X}_{t_2-1}) \stackrel{a)}{\supset} c(\mathsf{X}_{t_1}) \stackrel{b)}{\supset} c(\mathsf{X}_{\tau-1} \cup \mathsf{x}_{t_1}) \dots$$
$$\dots \stackrel{c)}{\supset} c(\mathsf{X}_{\tau-1} \cup \mathsf{p}_1).$$
(5.61)

The inclusions a), b) follow directly from the definition of X_t . For inclusion c), observe that $p_1 = \delta(\mu ||x_t||_W)^{-1}x_t$ and recall from (5.59) that the scalar constant $\delta(\mu ||x_t||_W)^{-1}$ is less than one. Now, since $c(X_\tau \cup x_{t_1})$ is a symmetric convex body, we know that it contains 0. Hence, we can view p_1 as a convex combination of 0 and x_t , which proves that $p_1 \in c(X_{\tau-1} \cup x_t)$ and therefore the set inclusion c). Now,

$$\|p_2\|_{\mathsf{X}_{\tau-1}\cup\mathsf{p}_1} > \delta.$$

The term on the left hand side of inequality (5.58) can be seen as a binary operation $d(\cdot, \cdot; X_{\tau-1})$ on the points p_1 and p_2 which measures a particular notion of distance characterized by the symmetric convex body $c(X_{\tau-1})$. We will define this operation more generally for some set B below and can use it to restate inequality (5.58) as

$$d(p_1, p_2; \mathsf{X}_{\tau-1}) > \delta$$

Definition 5.8. Let B be some bounded set in \mathbb{R}^n and define the map $d(\cdot, \cdot; B)$: $\mathbb{R}^n \times \mathbb{R}^n \mapsto \mathbb{R}_{\geq 0}$ for each $x, y \in \mathbb{R}^n$ as

$$d(x, y; \mathsf{B}) := \min \left\{ \begin{array}{c} r & c(\mathsf{B} \cup \mathsf{x}) \subset rc(\mathsf{B} \cup \mathsf{y}) \\ c(\mathsf{B} \cup \mathsf{y}) \subset rc(\mathsf{B} \cup \mathsf{x}) \end{array} \right\}$$
(5.62)

or equivalently as

$$d(x, y; \mathsf{B}) := \max\{\|\mathsf{B} \cup \mathsf{x}\|_{\mathsf{B} \cup \mathsf{y}}, \|\mathsf{B} \cup \mathsf{y}\|_{\mathsf{B} \cup \mathsf{x}}\}.$$
 (5.63)

The definition of $d(\cdot, \cdot; B)$ in the form of equation (5.62) gives a geometric intuition as to why the value d(x, y; B) can be viewed as a notion of distance between xand y. As an example, consider in Figure 5.2 the two points $x, y \in \mathbb{R}^2$ that satisfy d(x, y; B) > r and where B is taken as the box $[-1, 1] \times [-1, 1]$ in \mathbb{R}^2 ; Equation (5.62) then implies that x lies outside the set $rc(B \cup y)$ and y lies outside of $rc(B \cup x)$. This scenario is presented in Figure 5.2 for r = 1.3 and illustrates how condition d(x, y; B) > r enforces a separation between x and y. We will call x and y to be (r; B)-separated. More generally, we will introduce the following terminology:

Definition 5.9. A set $P \subset \mathbb{R}^n$ is $(\varepsilon; B)$ -separated (with suitable set B), if $d(p, p'; B) > \varepsilon$ holds for any $p, p' \in P$, $p \neq p'$.

In terms of the above definition, Lem. 44 states that \mathcal{P}_{μ} is a $(\delta; X_{\tau-1})$ -separated subset of $\frac{\delta}{\mu}W$.



Figure 5.2: Geometry of the distance function $d(\cdot, \cdot; B)$: The points x and y satisfy the inequality d(x, y; B) > r in n = 2, where B is taken as the two dimensional cube $B := \{(x_1, x_2) | |x_i| \le 1\}$ and r = 1.3.

Bounding the Cardinality of \mathcal{P}_{μ} Through Volume Bounds

In this section we will complete the proof of Theorem 27, by showing that any (ε, B) -separated subset of some bounded set S has to be a finite set. This argument then leads to the results in Theorem 27, since \mathcal{P}_{μ} is a $(\delta; X_{\tau-1})$ -separated subset of $\frac{\delta}{\mu}W$.

To illustrate the general idea, assume that we would like to construct a (ε ; B) separated set P = { p_1, p_2, \ldots , }, contained within some larger bounded set S $\in \mathbb{R}^2$. In particular, assume that we start with some $p_1 \in$ S, pick $p_2 \in$ S such that $d(p_1, p_2; B) > \varepsilon$ and proceed to select each p_n st. $d(p_n, p_k; B) > \varepsilon$ holds for all previous k < n. As illustrated in Figure 5.2, it becomes intuitively clear that any constructed (ε ; B)separated subset P in S has to have finite cardinality, as it becomes increasingly harder to find "enough" room for a new point $p_n \in$ S that respects the separation condition w.r.t. previous points $d(p_n, p_k; B) > \varepsilon$, k < n. In the next section, we will show by means of a volumetric argument that this intuition extends to *n*-dimensions and leads to a cardinality bound on the set \mathcal{P}_{μ} . Denote P to represent some (δ ; X_{τ -1})-*separated* subset of $\frac{\delta}{\mu}W$, i.e., not necessarily \mathcal{P}_{μ} . We will bound |P| first by constructing a corresponding cover set $\mathcal{N}(P) \supset P$ and then showing that the volume of $\mathcal{N}(P)$ is bounded below and above as

$$|\mathsf{P}|c_{\rm in} \leqslant \operatorname{Vol}(\mathcal{N}(\mathsf{P})) \leqslant C_{\rm out},$$

with some constants $c_{\rm in}$, $C_{\rm out}$ independent of P. The desired cardinality bound then

takes the form $|\mathsf{P}| \leq \frac{C_{\text{out}}}{c_{\text{in}}}$. The following sections will discuss: 1) the set $\mathcal{N}(\mathsf{P})$, 2) establishing the lower bound $|\mathsf{P}|c_{\text{in}}$, 3) proving the upper bound C_{out} , and 4) formulating the statement of Theorem 27.

Covering Set $\mathcal{N}(\mathsf{P})$

For some point $p \in \mathbb{R}^n$ and $\varepsilon > 1$, define $N(p; \varepsilon, B)$ to stand for the set

$$\mathsf{N}(p;\varepsilon,\mathsf{B}) := \{ p' \in \mathbb{R}^n \, | \, d(p,p';\mathsf{B}) \leqslant \varepsilon \} \,. \tag{5.64}$$

See Figure 5.3 as an example of the geometry of the set $N(p; \varepsilon, B)$ in \mathbb{R}^2 with $B = [-1, 1] \times [-1, 1]$. For a set P, correspondingly define the set $\mathcal{N}(P)$ as the following union of sets.

$$\mathcal{N}(\mathsf{P}) := \bigcup_{p \in \mathsf{P}} \mathsf{N}(p; \delta^{\frac{1}{2}}, \mathsf{X}_{\tau-1}).$$
(5.65)

 $\mathcal{N}(\mathsf{P})$ is a cover of P, since it can be easily verified that $\mathcal{N}(\mathsf{P}) \subset \mathsf{P}$. It can be easily seen that the map $d(\cdot, \cdot; \mathsf{B})$ inherits the following properties from Lem. 38:

Lemma 45. For all $x, y, z \in \mathbb{R}^n$ holds:

(*i*)
$$d(x, x; B) = 1$$

(*ii*)
$$d(x, y; B) = d(y, x; B) = d(y, -x; B)$$

(iii)
$$d(x, y; \mathsf{B}) \leq d(x, z; \mathsf{B})d(z, y; \mathsf{B})$$

As shown in Corollary 46, the property (iii) of Lemma 45 can be used to show that for (δ, B) -separated sets P, the sets in the union (5.65) are pairwise disjoint and we can therefore evaluate the volume $Vol(\mathcal{N}(P))$ as the sum:

$$\operatorname{Vol}(\mathcal{N}(\mathsf{P})) = \sum_{p \in \mathsf{P}} \operatorname{Vol}(\mathsf{N}(p; \delta^{\frac{1}{2}}, \mathsf{X}_{\tau-1})).$$
(5.66)

Corollary 46 (of Lem. 45). If for some $x, y \in \mathbb{R}^n$ holds $d(x, y; B) > \varepsilon$, then $N(x; \varepsilon^{\frac{1}{2}}, B) \cap N(y; \varepsilon^{\frac{1}{2}}, B) = \emptyset$.

Proof. For the sake of proving the statement through contradiction, assume that there was some point $z \in N(x; \varepsilon^{\frac{1}{2}}, B) \cap N(y; \varepsilon^{\frac{1}{2}}, B)$. Then we know that z satisfies both $d(x, z; B) \leq \varepsilon^{\frac{1}{2}}$ and $d(y, z; B) \leq \varepsilon^{\frac{1}{2}}$. But from the property (iii) of Lem. 45, we also have to conclude

$$d(x, y; \mathsf{B}) \leqslant d(x, z; \mathsf{B}) d(z, y; \mathsf{B}) \leqslant \varepsilon$$

which leads to the intended contradiction.



Figure 5.3: $N(x; \varepsilon, B)$ is the intersection of $\{p \mid ||x||_{B \cup p} \leq \varepsilon\}$ and $\{p \mid ||p||_{B \cup x} \leq \varepsilon\}$ and contains two translates of the set $(\varepsilon - 1)B$. In the picture, B is taken as the two dimensional cube $B := \{(x_1, x_2) \mid |x_i| \leq 1\}$ and $\varepsilon = 1.6$.

Lower Bound on Volume of $\mathcal{N}(\mathsf{P})$

To lower-bound the quantity (5.66), we will make use of the following lemma:

Lemma 47. Let $x, y \in \mathbb{R}^n$, then if $y = x + (\varepsilon - 1)p$ for some $p \in B$ and $\varepsilon > 1$, then it holds $d(x, y; B) \leq \varepsilon$.

Proof. We need to prove $||x||_{B\cup y} \leq \varepsilon$ and $||y||_{B\cup x} \leq \varepsilon$. $||y||_{B\cup x} \leq \varepsilon$: From the triangle inequality, we obtain

$$\|y\|_{\mathsf{B}\cup\mathsf{x}} = \|x + (\varepsilon - 1)p\|_{\mathsf{B}\cup\mathsf{x}} \leqslant \|x\|_{\mathsf{B}\cup\mathsf{x}} + (\varepsilon - 1)\|p\|_{\mathsf{B}\cup\mathsf{x}}$$

and using the fact that $x, p \in c(B \cup x)$ by the norm definition (5.25) we get $\|y\|_{B \cup x} \leq 1 + (\varepsilon - 1) = \varepsilon$. $\|x\|_{B \cup y} \leq \varepsilon$: Rewrite x as $x = \varepsilon \left(\frac{1}{\varepsilon}(y) + \frac{\varepsilon - 1}{\varepsilon}(-p)\right)$ and notice that $-p, y \in c(B \cup y)$, which shows that $x \in \varepsilon c(B \cup y)$. Hence, via the norm definition (5.25) we conclude $\|x\|_{B \cup y} \leq \varepsilon$.

If we use the property (ii) of Lem. 45, then Lem. 47 tells us that each set $N(p; \varepsilon, B)$ contains the sets $x \oplus (\varepsilon - 1)B$ and $-x \oplus (\varepsilon - 1)B$, where the operator \oplus denotes the Minkowski sum of two sets. For \mathbb{R}^2 , Figure 5.3 illustrates the geometric relationship between the set $N(x; \varepsilon, B)$ and the set B which is taken again to be the ∞ -norm unit ball. We can see that the set $N(p; \varepsilon, B)$ is a union of two symmetrical polytopes

and contains two non-overlapping translations (by the vector x and -x) of the set $(\varepsilon - 1)B$. Now, using the fact that n-dimensional volume $Vol(\cdot)$ is a homogenous function of degree n, we can obtain the following lower bound on the volume of any set $N(p; \varepsilon, B)$:

$$\operatorname{Vol}(\mathsf{N}(p;\varepsilon,\mathsf{B})) \ge 2(\varepsilon-1)^n \operatorname{Vol}(\mathsf{B}).$$
(5.67)

Combining this observation with our previous finding (5.66), we obtain the following lower bound on the volume of $Vol(\mathcal{N}(\mathsf{P}))$:

Lemma 48. Let $\mathcal{N}(\mathsf{P})$ be the collection (5.65) corresponding to a $(\delta, \mathsf{X}_{\tau-1})$ -separated $(\delta > 1)$ set P , then the volume $\operatorname{Vol}(\mathcal{N}(\mathsf{P}))$ is bounded below by

$$\operatorname{Vol}(\mathcal{N}(\mathsf{P})) \ge 2(\delta^{\frac{1}{2}} - 1)^{n} \operatorname{Vol}(c(\mathsf{X}_{\tau-1}))|\mathsf{P}|,$$
(5.68)

where $|\mathsf{P}|$ denotes the cardinality of the set P .

Proof. Apply (5.67) to every term in the sum (5.66).

Upper Bound on Volume of $\mathcal{N}(\mathsf{P})$

Consider some arbitrary point $q \in N(p; \delta^{\frac{1}{2}}, X_{\tau-1})$ for some p in the δ -separated set P and recall that $p \in \frac{\delta}{\mu}W$. Then, from the construction of the sets N as (5.64) we can conclude that

$$\|q\|_{\mathsf{X}_{\tau-1}\cup p} \leqslant d(q, p; \mathsf{X}_{\tau-1}) \leqslant \delta^{\frac{1}{2}}.$$
(5.69)

Moreover, since we can upper bound W as $W \subset \kappa_{\tau} c(X_{\tau-1})$, we also obtain $X_{\tau-1} \cup p \subset \max\{1, \frac{\kappa_{\tau}\delta}{\mu}\} c(X_{\tau-1})$ and therefore the point q satisfies

$$\|q\|_{\max\{1,\frac{\kappa_{\tau}\delta}{\mu}\}\mathsf{X}_{\tau-1}} \leq \|q\|_{\mathsf{X}_{\tau-1}\cup p} \leq \delta^{\frac{1}{2}}$$

$$\Leftrightarrow \qquad q \in \delta^{\frac{1}{2}}\max\{1,\frac{\kappa_{\tau}\delta}{\mu}\}\mathsf{c}(\mathsf{X}_{\tau-1}). \tag{5.70}$$

Hence, (5.70) shows that the collection $\mathcal{N}(\mathsf{P})$ is a subset of $\delta^{\frac{1}{2}} \max\{1, \frac{\kappa_{\tau}\delta}{\mu}\}c(\mathsf{X}_{\tau-1})$ which proves the following upperbound on the volume $\operatorname{Vol}(\mathcal{N}(\mathsf{P}))$:

Lemma 49. Let $\mathcal{N}(\mathsf{P})$ be the collection (5.65) corresponding to a $(\delta, \mathsf{X}_{\tau-1})$ -separated $(\delta > 1)$ set P , then the volume $\operatorname{Vol}(\mathcal{N}(\mathsf{P}))$ is bounded above by

$$\operatorname{Vol}(\mathcal{N}(\mathsf{P})) \leqslant \delta^{\frac{n}{2}} \max\{1, \frac{\kappa_{\tau}\delta}{\mu}\}^{n} \operatorname{Vol}(\operatorname{c}(\mathsf{X}_{\tau-1})).$$
(5.71)


Figure 5.4: The geometry of a $(\delta, c(X_{\tau-1}))$ -separated set $P = \{p_1, \ldots, p_5\}$. All sets are geometrically accurate, assuming $c(X_{\tau-1})$ is the box $[-1, 1] \times [-1, 1]$ and $\delta = 1.6$.

Cardinality Bound for $(\delta, X_{\tau-1})$ -Separated Sets P

Finally, the lower bound (5.68) and upper bound (5.71) imply the following bound on the cardinality of any $(\delta, X_{\tau-1})$ -separated set $\mathsf{P} \subset \frac{\delta}{\mu}\mathsf{W}$ with $\delta > 1$:

$$|\mathsf{P}| \leq \frac{1}{2} \left(\frac{\sqrt{\delta}}{\sqrt{\delta}-1}\right)^n \max\{1, \frac{\delta\kappa_{\tau}}{\mu}\}^n.$$
(5.72)

Figure 5.4 shows a pictorial summary of our derivation of the above inequality in \mathbb{R}^2 . A $(\delta, c(X_{\tau-1}))$ -separated set $\mathsf{P} = \{p_1, \ldots, p_5\}$ is defined to satisfy $p_i \notin \mathsf{N}(p_j, \delta, \mathsf{X}_{\tau-1})$, for all $i \neq j$ and as a consequence, we showed in Cor. 46 that the sets in the cover $\mathcal{N}(\mathsf{P}) = \bigcup_j \mathsf{N}(p_j, \sqrt{\delta}, \mathsf{X}_{\tau-1})$, are all disjoint. Then, Lem. 47 helped us establish that $\mathcal{N}(\mathsf{P})$ contains $2|\mathsf{P}|$ many translations of the set $(\sqrt{\delta} - 1)c(\mathsf{X}_{\tau-1})$, which lead to the volume bound (5.67). We obtain the upperbound (5.71) by showing that $\mathcal{N}(\mathsf{P})$ has to be contained in the bigger box $\delta^{\frac{1}{2}} \max\{1, \frac{\kappa_{\tau}\delta}{\mu}\}c(\mathsf{X}_{\tau-1})$. So, in the context of the picture Figure 5.4, we obtained our final cardinality bound (5.72) by dividing the volume of the outer larger box by the volume of the smaller boxes.

Recalling the relationship between the variables δ , μ and κ_{τ} in (5.52), we can express μ in terms of some $\delta > 1$ and κ_{τ} as

$$\mu = \delta \kappa_{\tau} \left(\sqrt{\frac{1}{4} + \frac{1}{\delta \kappa_{\tau}}} - \frac{1}{2} \right) = \left(\sqrt{\frac{1}{4} + \frac{1}{\delta \kappa_{\tau}}} + \frac{1}{2} \right)^{-1}$$
(5.73)

and can equivalently rewrite (5.72) in terms of constant κ_{τ} and $\delta > 1$ as a free

166

variable :

$$|\mathsf{P}| \leq \frac{1}{2} \left(\frac{\sqrt{\delta}}{\sqrt{\delta}-1}\right)^n \max\{\frac{1}{\delta\kappa_{\tau}}, \sqrt{\frac{1}{4} + \frac{1}{\delta\kappa_{\tau}}} + \frac{1}{2}\}^n (\delta\kappa_{\tau})^n.$$
(5.74)

In summary, (5.74) establishes a bound on the cardinality of $|\mathsf{P}|$ which serves as an upper-bound on $|\mathcal{P}_{\mu}((x_t), \tau)| = |\mathcal{X}_{\mu}((x_t), \tau)|$, thus the total number of unstable transitions \mathcal{U}_{μ} that can occur in the interval $[\tau, \infty)$ of any closed loop trajectory (x_t) . We conclude by restating the results Theorem 27 again in terms of δ :

Theorem. For any trajectory (x_t) of the closed loop (5.23a) and any $\tau \ge 0$, the cardinality $|\mathcal{X}_{\mu}((x_t); \tau)|$ of the set $\mathcal{X}_{\mu}((x_t); \tau)$ is finite for any μ chosen as

$$\mu = \left(\sqrt{\frac{1}{4} + \frac{1}{\delta\kappa_{\tau}}} + \frac{1}{2}\right)^{-1}, \delta > 1$$
(5.75)

for some $\delta > 1$ and bounded above as $|\mathcal{X}_{\mu}((x_t); \tau)| \leq N(\delta; \kappa_{\tau})$, where N stands for the function

$$N(\delta;\kappa_{\tau}) := \frac{1}{2} \left(\frac{\sqrt{\delta}}{\sqrt{\delta}-1}\right)^n \max\{\frac{1}{\delta\kappa_{\tau}}, \sqrt{\frac{1}{4} + \frac{1}{\delta\kappa_{\tau}}} + \frac{1}{2}\}^n (\delta\kappa_{\tau})^n$$
(5.76)

and κ_{τ} is a constant computed from $X_{\tau-1}$ as:

$$\kappa_{\tau} = \|\mathbf{W}\|_{\mathbf{X}_{\tau-1}} := \max_{z \in \mathbf{W}} \|z\|_{\mathbf{X}_{\tau-1}}.$$
(5.77)

5.9 A Connection Between Metric Entropy Bounds and Model-Free Stability Analysis

The notion of metric entropy⁴ dates back to early work of A.N. Kolmogorov [80] in 1959 and more recently has been proven useful for studying stochastic processes in the field of high-dimensional statistics. As an example, Chap. 5 of [125] discusses how bounds on the metric entropy of a metric space can be leveraged to obtain probabilistic bounds on the supremum of sub-Gaussian processes over that same metric space; Particularly in machine learning applications, these mathematical results can then be used to derive learning theoretic guarantees of algorithms.

Reexamining the line of arguments that lead to our theoretical guarantees suggests that there might be a possibly fruitful connection between metric entropy bounds and worst-case performance bounds in the context of learning and control problems. In retrospect, the main technique for stability analysis can be described as representing the collection X_{μ} of unstable transitions as a packing set P_{μ} in the *totally bounded*

⁴This is to be distinguished from the Kolmogorov-Sinai entropy of a dynamical system introduced in [79].

metric space $(\frac{\delta}{\mu}W, d(\cdot, \cdot; X_{\tau-1}))$; The bound on the cardinality \mathcal{X}_{μ} presented in Theorem 27 and derived in Section 5.8 can be viewed as the corresponding metric entropy bound. In hindsight, this inspires a new potential approach to algorithm design for learning and control: Synthesizing a control law for which unstable transitions (potentially as broader defined than \mathcal{U}_{μ} considered here) form a packing in some totally bounded metric space. Similar to our presented result, we could also hope that smaller metric entropy translates to improved closed loop performance guarantees.

We will proceed by introducing the metric entropy and related concepts based on [125] and [47]. In the next section, we will draw the connection to our stability analysis presented in Section 5.8.

Metric Entropy of Pseudo-Metric Spaces

A pseudometric space (S, d) consists of a set S and a pseudometric $d : S \times S \mapsto \mathbb{R}_{\geq 0}$, which satisfies the following properties:

(i) d(x, x) = 0 for any $x \in S$.

(ii)
$$d(x, y) = d(y, x)$$
 for any $x, y \in S$

(iii)
$$d(x,y) \leq d(x,y) + d(y,z)$$
 for any $x, y, z \in S$.

If in addition d(x, y) = 0 holds only if x = y, then d is called a metric and correspondingly (S, d) is a metric space. The ε -packing of S w.r.t to d is a set $P \subset S$ such that for each two distinct points $p_1, p_2 \in P$, $p_1 \neq p_2$ holds $d(p_1, p_2) > \varepsilon$. Correspondingly, the ε -packing number of S is the cardinality of the largest ε -packing set P of S. Formally, this is defined in Def. 5.10

Definition 5.10. Let (S, d) be a metric (or pseudo-metric) space. Then the ε -packing number $D(S, \varepsilon)$ (or $D(S, \varepsilon, d)$) of S is defined as

$$D(\mathsf{S},\varepsilon) := \sup \left\{ m \left| \begin{array}{c} \text{for some } p_1, \dots, p_m \in \mathsf{S}, \\ d(p_i, p_j) > \varepsilon \text{ for } 1 \leqslant i < j \leqslant m \end{array} \right\}.$$
(5.78)

If $D(S, \varepsilon)$ is finite for any $\varepsilon > 0$, then (S, d) is often called *totally bounded*. In this case, we define the quantity $\log(D(S, \varepsilon))$ as the *metric entropy* of the set S w.r.t. the metric d.

Remark 30. An alternative definition of metric entropy as in [125], is $log(N(S, \varepsilon))$ where $N(S, \varepsilon)$ is the covering number of the set S. The distinction between both (and other) definitions is of conventional matter, as it is well-known that packing numbers and covering numbers behave in equivalent manners. For our purpose, we will use the more fitting definition (5.10), which is for example used in the works of *R.M.* Dudley [47].

Bounding Occurrence of Unstable Transitions Through Metric Entropy

The key result behind our analysis in Section 5.8 was to show that any $(\delta, X_{\tau-1})$ separated subset $P \subset \frac{\delta}{\mu}W$ respects the cardinality bound (5.74). From Lem. 45 we
can directly see that the operation $\log(d(\cdot, \cdot; B)$ satisfies the properties of a pseudometric Def. 5.9 and therefore $(\frac{\delta}{\mu}W, \log(d(\cdot, \cdot; X_{\tau-1})))$ is a pseudo-metric space.
Correspondingly, the set $P \subset \frac{\delta}{\mu}W$ is $\log(\delta)$ -packing in that same pseudo-metric
space and our cardinality bound can be seen as an upper bound on the packingnumber $D(\frac{\delta}{\mu}W, \log(\delta))$ of the set $\frac{\delta}{\mu}W$ w.r.t. to the pseudometric $\log(d(\cdot, \cdot; X_{\tau-1}))$.
Moreover, since we established the bound (5.74) for every $\delta > 1$ (or $\log(\delta) > 0$),
the space $(\frac{\delta}{\mu}W, \log(d(\cdot, \cdot; X_{\tau-1})))$ is a totally bounded pseudometric space with
inequality (5.74) implying a particular metric entropy bound. Hence in hindsight,
our approach to stability analysis relied on mapping the set of unstable transitions \mathcal{X}_{μ} onto a fitting pseudometric space in which the metric entropy imposes a direct
bound on the cardinality of the set \mathcal{X}_{μ} . An interesting topic of further research is
whether this general principle could be leveraged for model-free stability analysis
and controller synthesis in broader learning and control problem settings.

5.10 Simulation

We conducted N = 1000 simulations of the causal cancelation controller K^{cc} defined in (5.19). For the *k*th experiment, the trajectories (x_t^k) , (u_t^k) are produced by the closed loop equations

$$x_{t+1}^{k} = A_0^k x_t^k + K_t^{\text{cc}}(x_t^k, X_{t-1}^{k+1}, X_{t-1}^k, U_{t-1}^k) + w_t^k,$$
(5.79)

and the system matrix $A_0^k \in \mathbb{R}^{3\times 3}$, initial condition $x_0^k \in \mathbb{R}^3$ and disturbance w_t^k is picked at random. All entries of A_0^k and x_0^k are picked i.i.d. from the standard Gaussian distribution $\mathcal{N}(0, 1)$. In each experiment, the causal cancellation controller (5.19) is initialized as $X_{-1} = \varepsilon I$, $X_{-1}^+ = 0$, $U_{-1} = 0$ with fixed choice $\varepsilon = 0.1$.



Figure 5.5: Results of 1000 closed-loop simulations with random A_0^k , x_0^k and disturbances w_t^k drawn from $[-1, 1]^3$. The plots on the left show the largest 1%, 10%, 50% percentile values of $||x_t^k||_{W^k}$, $||u_t^k||_{W^k}$, κ_{τ}^k , $m(\kappa_{\tau})$. The right plot shows the same percentile values for the state x_t^k and input u_t^k measured in 2-norm.

Figure 5.7 shows the simulation results of a single experiment where A_0 is chosen as

$$A_{0} = \begin{bmatrix} 1.4 & 0.2 & 1 \\ 0.2 & 1.3 & 1 \\ 0.5 & 0.3 & 2 \end{bmatrix} \qquad \qquad \lambda(A_{0}) = \begin{bmatrix} 2.7 \\ 1.13 \\ 0.86 \end{bmatrix}. \tag{5.80}$$

and has a large unstable eigenvalue $\lambda_i(A_0)$. Figure 5.5 and Figure 5.6 summarize the N closed-loop experiments for two scenarios of disturbances. The graphs show, as a function of t, the highest 1%, 10%, 50% percentiles of the values $\|x_t^k\|_{W^k}, \|u_t^k\|_{W^k}, \kappa_t^k = \|W^k\|_{X_{t-1}}$ and $m(\kappa_\tau)$ among the N experiments; the quantity $m(\kappa_\tau) := \kappa_\tau(\frac{1}{2} + \sqrt{\frac{1}{4} + \frac{1}{\kappa_\tau}}) + 1$ represents the updated theoretical asymptotic bound given the data collected up until time τ . In experiment k, the set W^k is constructed according to equation (5.26) from the disturbance sequence⁵ $(w_t^k)_{t=0}^{T-1}$ and the virtual disturbances \hat{w}_i^k . For our initialization of X_{-1}, X_{-1}^+, U_{-1} , the vectors \hat{w}_i^k take the values $-\varepsilon A_0^k e_i, 1 \leq i \leq n$, where e_i denotes the *i*th axis of the standard basis in \mathbb{R}^n :

$$\mathsf{W}^{k} = \left\{ w_{t}^{k} \mid 0 \leq t < T \right\} \cup \left\{ -\varepsilon A_{0}^{k} e_{i} \mid 1 \leq i \leq n \right\}.$$
(5.81)

⁵We assume that after t > T the disturbance w_t^k stays in the set $c(W^k)$.



Figure 5.6: Results of 1000 closed-loop simulations with random A_0^k , x_0^k and $w_t^k = 0$. The plots on the left show the largest 1%, 10%, 50%* percentile values of $||x_t^k||_{W^k}$, $||u_t^k||_{W^k}$, κ_{τ}^k , $m(\kappa_{\tau})$. (* this percentile is too small to visualize for $||x_t^k||_{W^k}$ and $||u_t^k||_{W^k}$) The right plot shows the same percentile values for the state and input measured in 2-norm.

For Figure 5.5, the disturbance sequence (w_t^k) of each experiment is picked i.i.d. uniformly from the interval $[-1, 1]^3$. For Figure 5.6, the initial condition $x_{0,i}$ is chosen i.i.d. according to the Gaussian distribution $\mathcal{N}(0, \sigma^2)$, $\sigma = 10^{-3}$ and $w_t^k = 0$. Since we have no disturbance, for this case, W^k is simply the set of virtual disturbances $\{-\varepsilon A_0^k e_i | 1 \le i \le n\}$. We discussed that, as a corollary of our main result (see (5.40)), the causal cancelation controller \mathbf{K}^{cc} guarantees for each experiment the asymptotic bound

$$\limsup_{t \to \infty} \|x_t^k\|_{\mathsf{W}^k} \leqslant m(\kappa_t^k), \quad \forall t$$
(5.82)

where we take the function $m(\cdot)$ to abbreviate the expression

$$m(s) := s\left(\frac{1}{2} + \sqrt{\frac{1}{4} + \frac{1}{s}}\right) + 1.$$
(5.83)

In Figure 5.5 and Figure 5.6, we overlayed the percentiles of $||x_t^k||_{W^k}$ (blue) and $m(\kappa_t^k)$ (red) to show that qualitatively the experiments match the theoretical guarantee above. In each experiment, the controller eventually learns to stabilize the unknown system (consistently after 10 time-steps) and eventually (see t > 20 in Figure 5.5)

is bounded above by the asymptotic bound $m(\kappa_t^k)$. Note also that as more online data is observed, the asymptotic bound $m(\kappa_t^k)$ tightens. Figure 5.6 is showing the closed loop performance in the no disturbance regime. This is to investigate how the controller K^{cc} performs in the absence of excitation by the disturbance. We see in Figure 5.6 that the controller K^{cc} stabilizes the system in all experiments, but compared to (5.5), we have a longer transient of learning. Note that in Figure 5.6, the percentiles of the constant κ_t^k do not decrease over time as much as in the experiments of Figure 5.5. Recall that for time t, the constant κ_t^k can be seen to approximate the remaining uncertainty of the unknown system A_0^k . Therefore, Figure 5.6 shows that despite the remaining uncertainty in the system, the controller still manages to stabilize the system. This reflects that K^{cc} does not primarily care about identifying the unknown matrix A_0^k , but rather collects *only* enough data about the matrix A_0^k to be able to stabilize the closed loop.



Figure 5.7: $||x_t||_W$ and $||u_t||_W$ trajectories for closed loop with uniform disturbance $w_{i,t}$ in [-1, 1] and $x_0 = [0.2, 0, 0.1]^T$.

5.11 Conclusion

In this chapter we derive a simple model-free controller that can adaptively and robustly stabilize a linear system with full actuation without any additional knowledge on disturbance, noise or parameter bounds. The controller comes with uniform asymptotic and worst-case guarantees on the state-deviation. The control design and stability analysis is enabled by a novel approach inspired by convex geometry, and simulations show that the controller is able to simultaneously learn and control the system in an efficient manner, even when applied to an open loop system with large unstable eigenvalues. Future work will further explore how this new perspective on adaptive control can provide more learning and control algorithms with robustness guarantees and non-restrictive assumptions in a more general setting. In addition, we will investigate how the presented ideas can help in providing robustness and performance bounds for present methods in adaptive control and reinforcement learning.

5.A Proofs

Theorem 29

Proof. The proof follows by applying Lem. 41. The corresponding bounds for (u_t) are then obtained using the equation (5.22).

According to the setting of the theorem, consider some fixed trajectories (x_t) , (u_t) , reference time τ , $\mu \in \mathcal{I}_{\kappa_{\tau}}$ with $\kappa_{\tau} = \|W\|_{X_{\tau-1}}$. Then, as discussed before, a direct consequence of Theorem 27 is that there is some trajectory-dependent finite time $T' < \infty$, such that in the time interval [0, T'] there are at most $N(\mu; \kappa_{\tau})$ -many time instances $\mathcal{T} := \{t'_1, \ldots, t'_M\}$, where

$$\tau \leqslant t'_1 < t'_2 < \dots < t'_M \leqslant T', \quad M < N(\mu; \kappa_\tau)$$
(5.84)

at which μ -unstable transitions occur and for all other time-instances $t \neq t'_i$ holds the opposite inequality of (5.28). Thus, depending on whether t belongs to \mathcal{T} , the transitions (x_{t+1}, x_t) of the trajectory (x_t) satisfy

$$||x_{t+1}||_{\mathsf{W}} > \max\left\{\frac{1}{1-\mu}, \mu ||x_t||_{\mathsf{W}} + 1\right\}, \quad \forall t \in \mathcal{T}$$
 (5.85a)

$$||x_{t+1}||_{\mathsf{W}} \le \max\left\{\frac{1}{1-\mu}, \mu \,||x_t||_{\mathsf{W}} + 1\right\}, \quad \forall t \notin \mathcal{T}.$$
 (5.85b)

Moreover, combining Lem. 41 with the above, we find that, w.r.t. the function $V_1(x;\mu) := \max\{0, \|x\|_{\mathsf{W}} - \frac{1}{1-\mu}\}$, the transitions (x_{t+1}, x_t) respect the inequality

$$V_1(x_{t+1};\mu) > \mu V_1(x_t;\mu), \quad \forall t \in \mathcal{T}$$
(5.86a)

$$V_1(x_{t+1};\mu) \leqslant \mu V_1(x_t;\mu), \quad \forall t \notin \mathcal{T}$$
(5.86b)

and for function $V_2(x;\mu) := \max\{\|x\|_W, \frac{1}{1-\mu}\}\$, the transitions (x_{t+1}, x_t) satisfy

$$V_2(x_{t+1};\mu) \leqslant \|\mathbf{W}\|_{\mathbf{X}_{t-1}} V_2(x_t;\mu) + 1, \qquad \forall t \in \mathcal{T}$$
(5.87a)

$$V_2(x_{t+1};\mu) \leqslant V_2(x_t;\mu), \qquad \forall t \notin \mathcal{T}.$$
(5.87b)

The bounds on (x_t) in part (i) and (ii) were derived before from (5.86), (see (5.49) and (5.50)). For (iii), notice that (5.87) implies that for any t, $V_2(x_t; \mu)$ can be

bounded above by $V_2(x_{t'_M+1}; \mu)$, since apart from the time intervals $[t'_i, t'_{i+1}]$, the quantity $V_2(x_t; \mu)$ is guaranteed to be nonincreasing. Moreover (5.87) also shows that $V_2(x_{t'_M+1}; \mu)$ can be bounded above as

$$V_{2}(x_{t'_{M}+1};\mu) \leq \alpha V_{2}(x_{\tau};\mu) + \beta$$

$$\alpha := \prod_{k=1}^{M} \|\mathbf{W}\|_{\mathbf{X}_{t'_{k}-1}}, \quad \beta := \sum_{k=0}^{M-1} \prod_{j=1}^{k} \|\mathbf{W}\|_{\mathbf{X}_{t'_{M-j}-1}}.$$
(5.88)

Recall that $\|W\|_{X_t}$ is not increasing (hence $\|W\|_{X_{t'_k-1}} \leq \kappa_{\tau} := \|W\|_{X_{\tau-1}}$) and the bound $M \leq N(\mu; \kappa_{\tau})$, to see that the constants α and β are bounded above as

$$\alpha \leq \max\{1, \kappa_{\tau}^{N(\mu;\kappa_{\tau})}\} \qquad \beta \leq \frac{1 - \kappa_{\tau}^{N(\mu;\kappa_{\tau})}}{1 - \kappa_{\tau}}.$$
(5.89)

We then obtain the final inequality (5.36) by substituting the above bounds into (5.88) and observing that

$$\sup_{t \ge \tau} \|x_t\|_{\mathsf{W}} \le \sup_{t \ge \tau} V_2(x_t;\mu) \le V_2(x_{t'_M+1};\mu).$$
(5.90)

To obtain the corresponding bounds for the input (u_t) , recall that u_t can be rewritten as

$$u_{t} = (U_{t-1} - X_{t:1}) \lambda_{t-1}(x_{t})$$

= $(-A_{0}X_{t-1} - W_{t-1})\lambda_{t-1}(x_{t})$
= $-A_{0}x_{t} - W_{t-1}\lambda_{t-1}(x_{t})$

and that $\|\lambda_{t-1}\|_1 = \|x_t\|_{X_{t-1}}$. This allows us to upper-bound $\|u_t\|_W$ by

$$\|u_{t}\|_{\mathsf{W}} \leq \left\|A_{0}\frac{x_{t}}{\|x_{t}\|_{\mathsf{W}}}\right\|_{\mathsf{W}} \|x_{t}\|_{\mathsf{W}} + \left\|\frac{x_{t}}{\|x_{t}\|_{\mathsf{W}}}\right\|_{\mathsf{X}_{t-1}} \|x_{t}\|_{\mathsf{W}}$$
$$\leq \left(\max_{x \in \mathsf{W}} \|A_{0}x\|_{\mathsf{W}} + \|\mathsf{W}\|_{\mathsf{X}_{\tau-1}}\right) \|x_{t}\|_{\mathsf{W}}$$
$$\leq \left(\|A_{0}\|_{\mathsf{W}} + \kappa_{\tau}\right) \|x_{t}\|_{\mathsf{W}}$$

and obtain desired bounds for (u_t) by adding the bounds already derived for (x_t) . \Box

Lemma 41

Proof. Part (i) and (5.47): We can expand the inequality as

$$\|x_{t+1}\|_{\mathsf{W}} \le \max\left\{\frac{1}{1-\mu}, \mu \,\|x_t\|_{\mathsf{W}} + (1-\mu)\frac{1}{1-\mu}\right\}$$
(5.91)

and can subtract $\frac{1}{1-\mu}$ on both sides to obtain

$$\|x_{t+1}\|_{\mathsf{W}} - \frac{1}{1-\mu} \leq \max\left\{0, \mu(\|x_t\|_{\mathsf{W}} - \frac{1}{1-\mu})\right\}$$

$$\Leftrightarrow \max\{0, \|x_{t+1}\|_{\mathsf{W}} - \frac{1}{1-\mu}\} \leq \mu \max\left\{0, \|x_t\|_{\mathsf{W}} - \frac{1}{1-\mu}\right\}$$

Similarly, noticing that the second term on the right hand sight of (5.91) is a convex combination of $||x_t||_W$ and $\frac{1}{1-\mu}$, we can conclude

$$\max\{\|x_{t+1}\|_{\mathsf{W}}, \frac{1}{1-\mu}\} \leq \max\{\|x_t\|_{\mathsf{W}}, \frac{1}{1-\mu}\}.$$

Part (ii): We previously derived that the inequality (5.42d) holds for all time t:

$$||x_{t+1}||_{\mathsf{W}} \leq ||\mathsf{W}||_{\mathsf{X}_{t-1}} ||x_t||_{\mathsf{W}} + 1.$$

Now, if in addition inequality (5.28) holds, then we obtain

$$\max\left\{\frac{1}{1-\mu}, \mu \|x_t\|_{\mathsf{W}} + 1\right\} < \|\mathsf{W}\|_{\mathsf{X}_{t-1}} \|x_t\|_{\mathsf{W}} + 1.$$

Combining both the previous inequalities, we get the following result.

$$\max\left\{\frac{1}{1-\mu}, \|x_{t+1}\|_{\mathsf{W}}\right\} \leq \|\mathsf{W}\|_{\mathsf{X}_{t-1}} \|x_t\|_{\mathsf{W}} + 1$$
$$\leq \|\mathsf{W}\|_{\mathsf{X}_{t-1}} \max\left\{\frac{1}{1-\mu}, \|x_t\|_{\mathsf{W}}\right\} + 1.$$

Chapter 6

ROBUST MODEL-BASED LEARNING AND CONTROL OF UNKNOWN SYSTEMS

In this chapter, we introduce a new framework "PixSel" for adaptive control, and, more generally, for one-shot control design of nonlinear discrete-time systems. Most notably, the theory and design methods can provide worst-case closed-loop guarantees on safety- and cost-performance even in the presence of arbitrarily large model uncertainty, and allow for problem settings with nonlinear time-varying dynamics in both system and controller. Our framework reveals a promising connection between online learning and robust control theory, which enables systematic and modular design of robust learning and control algorithms with provided safety and performance guarantees in the large uncertainty setting. To the best of our knowledge, this is the first time that a fundamental connection between the fields of online learning and control theory has ever been discovered in this context.

Our approach is based on decomposing one-shot control design into two separate sub-problems: Designing a "robust oracle" π , which encapsulates application specific nominal control design and desired guarantees, and designing a "consistent model chaser" SEL, a pure online learning problem which embodies the issue of stable and efficient adaptation. If each individual problem can be solved, we can use the resulting subroutines π and SEL to instantiate a CE¹-based adaptive controller $A_{\pi \times SEL}$ that inherits worst-case guarantees from nominal control design, which surprisingly still hold for arbitrarily large model uncertainty.

Our discussion will begin with studying the one-shot control design problem "Online Control with Mistake Guarantees" (OC-MG) first introduced in our work [5], which will serve as a motivation and introductory case study of the general "PixSel"-design framework (which ultimately has a far broader scope than the problem setting of OC-MG). Some of the broader implications of the "PixSel"-framework are discussed in Section 6.7 and Section 6.8, others are presented in recent [9] and topic of ongoing work.

¹Certainty-Equivalence principle.



Figure 6.1: PixSel Framework for adaptive control.

6.1 Introduction

We study the problem of online control for nonlinear systems with large model uncertainty, under the requirement to provide upfront control-theoretic guarantees for the worst-case online performance; by large uncertainty, we mean to say that we are given an arbitrarily large set of potential models, of which an unknown few are exact descriptions of the true system dynamics. Algorithms with such capabilities can enable us to (at least partially) sidestep undertaking laborious system identification tasks prior to robust controller design. Motivated by real-world control applications, we formulate a class of problems which allow us a unified way to address common control problems such as stabilization, tracking, disturbance rejection, robust set invariance, etc. We introduce this as online control with mistake guarantees (OC-MG): We define a problem instance by specifying a desired system behavior and search for online control algorithms which can quantify, in terms of number of *mistakes*, how often the online controlled system could deviate from this behavior in the worst-case (i.e. worst possible scenario of true system dynamics, disturbances, noise, etc.). We propose a modular framework for OC-MG: Use robust control to design a *robust oracle* π , use online learning to design an algorithm SEL which *chases consistent models*, and fuse them together via a simple meta-algorithm; the end result is Algorithm 2, which we refer to as $A_{\pi}(SEL)$. Our approach is based on decomposing the original problem into the two independent sub-problems "robust oracle design" (ROD) and "consistent models chasing" (CMC) which for many problem instances can be readily addressed with existing tools from control theory (see Section 6.3) and online learning (see Section 6.5). We demonstrate in Section 6.10, that for general robotic systems, we can solve CMC through competitive convex body chasing [12, 13, 33, 114] and ROD using well-known robust control methods [54, 96, 118, 119, 141].

Once suitable subroutines π and SEL are selected, we can provide online performance guarantees for the resulting control algorithm $A_{\pi}(SEL)$ that holds in the large uncertainty setting:

- *Mistake guarantee:* A worst case bound on the total number of times desired system behavior is violated. See Theorem 38, 39, 40.
- *Safety guarantee:* A worst-case norm bound on the state-trajectory. See Theorem 36.

To provide the above guarantees, π and SEL have to be solutions to a corresponding ROD and CMC sub-problem. In Section 6.3 and Section 6.5 we discuss that in many problem settings this is not a restrictive assumption. In particular, assuming that ROD can be solved merely ensures that the overall OC-MG problem is well-posed: The underlying control problem (i.e., assuming no uncertainty) has to be tractably solvable with robust control; it is clear that this is a bare minimum requirement to state a meaningful OC-MG problem. In Section 6.5 we discuss different versions of the CMC sub-problem and present a reduction to nested convex body chasing [33] which is applicable for a large class of systems.

In Section 6.11, we follow our approach to design a high-performing control algorithm for a difficult nonlinear adaptive control problem: swinging-up a cartpole with large parametric uncertainty *and* state constraints. We benchmark the performance of the online algorithm $A_{\pi}(SEL)$ against the offline optimal algorithm over 900 problem settings (adversarial chosen system parameters, noise, disturbances) and show that $A_{\pi}(SEL)$ performs only marginally worse than the optimal offline controller, which has access to the true system model.

Problem Statement

Consider controlling a discrete-time *nonlinear dynamical system* with system equations:

$$x_{t+1} = f^*(t, x_t, u_t), \quad f^* \in \mathcal{F},$$
 (6.1)

where $x_t \in \mathcal{X}$ and $u_t \in \mathcal{U}$ denote the system state and control input at time step t and $\mathcal{X} \times \mathcal{U}$ denotes the state-action space. We assume that f^* is an *unknown* function and that we only know of an *uncertainty set* \mathcal{F} that contains the true f^* .

Large Uncertainty Setting. We impose no further assumptions on \mathcal{F} and explicitly allow \mathcal{F} to represent arbitrarily large model uncertainties.

Control Objective. The control objective is specified as a sequence $\mathcal{G} = (\mathcal{G}_0, \mathcal{G}_1, ...)$ of binary cost functions $\mathcal{G}_t : \mathcal{X} \times \mathcal{U} \mapsto \{0, 1\}$, where each function \mathcal{G}_t encodes a desired condition per time step t: $\mathcal{G}_t(x_t, u_t) = 0$ means that the state x_t and the input u_t meet the requirements at time t. $\mathcal{G}_t(x_t, u_t) = 1$ means that some desired condition is violated at time t and we will say that the system made a *mistake* at t. The performance metric of system trajectories $\boldsymbol{x} := (x_0, x_1, ...)$ and $\boldsymbol{u} := (u_0, u_1, ...)$ is the sum of the cost incurred $\mathcal{G}_t(x_t, u_t)$ over the time interval $[0, \infty)$ and we denote this the *total number of mistakes*:

mistakes of
$$\boldsymbol{x}, \boldsymbol{u} = \sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t).$$
 (6.2)

For a state-input trajectory of the system (x, u) to achieve an objective \mathcal{G} , we want the above quantity to be finite, i.e., eventually the system stops making mistakes and meets the objectives requirements for all time.

One-shot Control Design Goal. The goal is to design an online decision rule $u_t = \mathcal{A}(t, x_t, ..., x_0)$ such that <u>regardless</u> of the unknown $f^* \in \mathcal{F}$, the online trajectories are guaranteed to be bounded, and to have finite or even explicit upperbounds on the total number of mistakes (6.2). Thus, we require a strong notion of robustness: \mathcal{A} can control any system (6.1) with the certainty that the objective \mathcal{G} will be achieved after finitely many mistakes. It is suitable to refer to our problem setting as *online control* with *mistake guarantees*.

Motivation and Related Work

The main purpose of this work is to find answers to the following question:

How do we learn to control unknown dynamical systems in a systematic and reliable way?

In one way or another, this is one of the first questions we run into when we approach real-world control problems: Dynamical systems we encounter in the real world are unknown to us; we have ways to find approximate mathematical models (system identification), and use them as a substitute system for control design. That being said, we are always forced to make a leap of faith when we deploy a control system, since we can never be certain how well our system models match up with the dynamics of the real system. Therefore, it is always necessary to collect online data and keep monitoring for potential *inconsistencies* between system behavior and our model/design assumptions, as well as be ready to *adapt* our control algorithms once we *learn* that our models are no longer accurate enough to guarantee performance. Depending on the application, accurate models and/or online adaptation can be of critical importance. This is especially true in safety-critical settings involving physical systems, such as in engineering domains such as aerospace, industrial robotics, automotive, energy plants [123], etc. Throughout the last decades and with the accelerated technological advancement, many important engineering systems have drastically increased in complexity. Finding accurate models of complex systems can quickly become really difficult (or even impossible), which makes the latter problem, of learning to adapt controls from on-line data, a crucial aspect of over control system design. This problem setting is the focus and motivation of our work. The existing literature can be split into two general categories: the more recent literature of what can be described as *System ID, then Robust Control* and the more traditional literature of *Adaptive Control*.

System Identification, then Robust Control

The most common approach in online learning for control literature [42] is to perform system identification [88], then use tools from robust control theory [141]. Robust controller synthesis can provide policies with desired guarantees, so long as one can obtain an approximate model which is "provably close enough" to the real system dynamics. However, estimating a complex system to a desired accuracy level quickly becomes intractable in terms of computational and/or sample complexity. In the adversarial noise setting, system identification of simple linear systems with precision guarantees can be NP-hard [41]. General approaches for nonlinear system identification with precision guarantees are for the most part not available (recently Mania et al. [93] analyzed sample complexity under stochastic noise). Many recent learning approaches for control of dynamical systems have focused on the setting of linear optimal control: One is given a linear system, and the control objective is to minimize a specified cost functional. To relate our problem setting to other approaches in this field, we can view our problem setting as an instance of optimal control where we restrict the cost function to be $\{0,1\}$ valued. There has been a particular focus on the problem of the Linear Quadratic Regulator (LQR) [1, 37, 42, 43, 51], or linear dynamical system with convex costs [3, 4, 62]. Our work is instead suitable as well for the nonlinear control setting. In addition, even when restricted to the linear system setting, recent line of work on online learning for control differs from our approach in the following aspects:

- Performance Criteria: We focus on bounding the total cost ∑_{t=0}[∞] G_t(x_t, u_t) as defined in Section 6.1 of the main paper. Our notion of control objective is natural to define in control applications, e.g., most popular robotic goals can be formulated as driving the systems towards a desirable set or trajectories. This differs from, but is not incompatible with, the cost-metric formulation that is often seen in optimal control and online learning for control work. Specifically, previous effort on learning LQR has been to improve the regret bound of the learning algorithm [1, 2, 42, 43, 62]. Bounding the regret on the average cost, which is natural for LQR, is not sufficient to guarantee finite mistakes in our problem setting. In Section 6.C, we discuss counterexamples which discuss the relationship between finite mistake guarantees imply sublinear regret, yet sublinear regret does not imply finite mistakes.
- Approach: Our proposed approach does not depend on accurate identification of the online system, which is the focus of several recent works on learning for LQR [37, 42, 51, 62]. As we consider parametric uncertainty, it is plausible to also adopt a system identification approach for the non-linear control settings. However, online system identification with arbitrarily small error is known to be very challenging. As shown by [41], the sample complexity for identifying linear systems under bounded adversarial noises can be exponential in the worst case.
- Assumptions about parameter uncertainty: Some previous work in linear systems [37, 43, 62] assumes knowledge of a stabilizing controller π_{safe} : X → U for the true unknown system parameter θ*. In our setting, we do not require such an assumption, but merely that for each possible parameter θ ∈ Ω one can find a robust policy π[θ] which stabilizes the small uncertainty model D[θ] ⊂ F.

Robust Adaptive Nonlinear Control

Naturally, our problem setting is of great interest to the adaptive control community, which has had a relatively long history on this topic: [71, 83, 87, 102, 134]. Yet, most of traditional adaptive control approaches can not be applied to the general problem setting we consider without making restrictive assumptions. As an example, in contrast to most adaptive control methods, our framework applies to nonfeedback linearizable nonlinear system (see overview of adaptive control in [65]). Furthermore, many adaptive control techniques can not build on top of methods from other areas of

control, like robust control theory, but rather propose separate control algorithms for each problem setting. Additionally, robust stability analysis and thorough empirical validation are largely unavailable for most methods. In fact, most relevant empirical results are only presented for arguably much simpler settings than for the cart-pole swing-up problem, which is considered in this work. In addition, we highlight methodological distinctions.

We provide a modular framework, which allows one to combine robust control tools with online learning algorithms to provide desired guarantees online. We unify the treatment of both uncertain system parameters and unknown disturbance via the construction of confidence sets of candidate systems that are consistent with the historical collected observations. The estimation of such consistent sets is also easily attainable for most robotic systems and allows for non-asymptotic convergence guarantees. Among the relevant adaptive control literature, perhaps the most closely related to ours is Multi-Model Adaptive Control (MMAC) from [9]. The MMAC principle needs to run a high-dimensional Multi-Estimation routine online, which requires the design of nonlinear observers (with the matching and Detectability property - see [63]) for a sufficiently dense covering set of the parameter space. A general construction of such a family is only shown for linear systems (see [63] and references therein), and it is not clear whether designing a tractable Multi-Estimator for the cart-pole system is possible.

6.2 Overview of Approach and Main Results

No Need for SysID and Persistency of Excitation. While accurate models of real systems are hard to obtain, it is often easy to provide more qualitative or rough models of system dynamics *without* performing offline experiments and requiring system identification. Having access to a rough system description, we design a control algorithm in one-shot which can be deployed on the real system with upfront worst-case control-theoretic guarantees on the online performance. Moreover, in contrast to other works, such as [35] for example, we are the first framework that can provide worst-case guarantees *without* requiring the assumption of *persistency of excitation*.

Rough Models as Compactly Parameterisable Uncertainty Sets. In practice, we never have the exact knowledge of f^* in advance. However, for engineering applications involving physical systems, the functional form of f^* can often be derived through first principles and knowledge of the application-specific domain. Conceptually, we can view the unknown parameters of the functional form as

conveying both the 'modeled dynamics' and 'unmodeled (adversarial) disturbance' components of the ground truth f^* in the system $x_{t+1} = f^*(t, x_t, u_t)$. It is almost always the case that we can represent the uncertainty in f^* via a collection of parameters in bounded ranges. How we choose to parameterize a given uncertainty set \mathcal{F} is not unique and poses a design choice. We will take this as the starting point for our approach and assume a fixed parameterization of \mathcal{F} in the form of a tuple (\mathbb{D}, Ω, d) , where (Ω, d) is a compact metric space, called *parameter space*, and \mathbb{D} is a map $\Omega \mapsto 2^{\mathcal{F}}$ that defines a collection of models $\{\mathbb{D}[\theta] \mid \theta \in \Omega\}$ which represents a cover of the uncertainty set \mathcal{F} . We define this formally as a compact parameterization of \mathcal{F} :

Definition 6.1. A tuple (\mathbb{D}, Ω, d) , where $\mathbb{D} : \Omega \mapsto 2^{\mathcal{F}}$ is a compact parametrization of \mathcal{F} , if (Ω, d) is a compact metric space and $\mathcal{F} \subset \bigcup_{\theta \in \Omega} \mathbb{D}[\theta]$.

We will work with candidate parameters $\theta \in \Omega$ of the system and consider a θ^* to be a *true parameter* of f^* , if $f^* \in \mathbb{D}[\theta^*]$. Ideally, each candidate model $\mathbb{D}[\theta]$ has small uncertainty; the precise notion of "small uncertainty" however is problem specific and depends always on the objective. For concreteness, we give several simple examples of common parameter spaces Ω :

1. *Linear time-invariant system*: linear system with matrices A, B perturbed by bounded disturbance sequence $w \in \ell_{\infty}$, $||w||_{\infty} \leq \eta$:

$$f^*(t, x, u) = Ax + Bu + w_t.$$
(6.3)

The parameter space Ω contains bounded intervals describing the parameters $\theta = (A, B, \eta)$.

Nonlinear system, linear parametrization: nonlinear system, where dynamics are a weighted sum of nonlinear functions ψ_i perturbed by a bounded disturbance sequence w ∈ ℓ_∞, ||w||_∞ ≤ η:

$$f^*(t, x, u) = \sum_{i=1}^{M} a_i \psi_i(t, x, u) + w_t.$$
 (6.4)

 Ω contains bounded intervals that describe $\theta = (\{a_i\}, \eta)$.

Nonlinear system, nonlinear parametrization: nonlinear system, with function g parameterized by a fixed parameter vector p ∈ ℝ^m (e.g., neural networks), perturbed by a bounded disturbance sequence w ∈ ℓ_∞, ||w||_∞ ≤ η:

$$f^*(t, x, u) = g(t, x, u; p) + w_t.$$
(6.5)

 Ω contains bounded intervals that describe $\theta = (p, \eta)$.

Algorithm 2 Meta-Implementation of $A_{\pi}(SEL)$

Require: procedures π and SEL **Initialization:** $\mathcal{D}_0 \leftarrow \{\}, x_0 \text{ is set to initial condition } \xi_0$ 1: for t = 0, 1, ... to ∞ do $\mathcal{D}_t \leftarrow \text{append} (t, x_t, x_{t-1}, u_{t-1}) \text{ to } \mathcal{D}_{t-1} (\text{if } t \ge 1)$ 2: \triangleright update online history of observations $\theta_t \leftarrow \text{SEL}[\mathcal{D}_t]$ \triangleright present online data to SEL, get posited parameter θ_t 3: $u_t \leftarrow \pi[\theta_t](t, x_t)$ \triangleright query π for policy $\pi[\theta_t]$ and evaluate it 4: \triangleright system transitions with unknown f^* to next state $x_{t+1} \leftarrow f^*(t, x_t, u_t)$ 5: 6: end for

In these examples, the uncertainty set $\mathcal{F} \subset \bigcup_{\theta \in \Omega} \mathbb{D}[\theta]$ is covered by models $\mathbb{D}[\theta]$ with smaller uncertainty of the form $\mathbb{D}[\theta] = \{t, x, u \mapsto f_{\theta}(x, u, w_t) \mid ||w||_{\infty} \leq \eta\}$, where f_{θ} denotes one of the functional forms on the right-hand side of eq. (6.3), (6.4) or (6.5).

Online Robust Control Algorithm. Given a compact parameterization (\mathbb{D}, Ω, d) for the uncertainty set \mathcal{F} , we design a meta-algorithm $\mathcal{A}_{\pi}(SEL)$ (Algorithm 2) that controls the system (6.1) online by invoking two subroutines π and SEL in each time step.

Consistent model chasing. Procedure SEL receives a finite data set D, which contains state and input observations, and returns a parameter θ ∈ Ω.

Design goal: For each time t, the procedure SEL should select θ_t such that the set of models $\mathbb{D}[\theta_t]$ stays "consistent" with \mathcal{D}_t , i.e., candidate models in $\mathbb{D}[\theta_t]$ can *explain* the past data. Moreover posited parameters θ_t should only change when necessary: two posited parameters θ_t and $\theta_{t'}$ should not be very different from each other, if both data sets \mathcal{D}_t and $\mathcal{D}_{t'}$ contain "similar" amount information.

Robust oracle. Procedure π receives a posited system parameter θ ∈ Ω as input and returns a control policy π[θ] : N × X → U which can be evaluated at time t to compute a control action ut = π[θ](t, xt) based on the current state xt.

Design goal: We require that π represents a robust control design subroutine for the collection of models \mathbb{D} , in the sense that policy $\pi[\theta]$ could provide mistake guarantees for \mathcal{G} which are robust to bounded noise **if** the uncertainty set \mathcal{F} were $\mathbb{D}[\theta]$.

Theoretical Contribution. Our main theoretical results certify safety- and finite mistake guarantees for the online control scheme $\mathcal{A}_{\pi}(SEL)$ if the sub-routines π and SEL meet the design requirement for "robust oracle" and "consistent model chasing" for a given uncertainty set \mathcal{F} and objective \mathcal{G} . We will clarify the consistency and robustness requirements of the sub-routines π and SEL in Section 6.3 and Section 6.5. For now, we present an informal version of the finite mistake guarantees and the worst-case state deviation for the online control scheme $\mathcal{A}_{\pi}(SEL)$:

Theorem (Informal). For any (adversarial) $f^* \in \mathcal{F}$, the online control scheme $\mathcal{A}_{\pi}(SEL)$ described in Algorithm 2 guarantees a priori that the trajectories x, u will achieve the objective \mathcal{G} after finitely many mistakes. The total number of mistakes $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t)$ is at most

oracle performance
$$M_{\rho}^{\pi} * \Gamma_1 \left(\frac{\text{size of uncertainty } \mathcal{F}}{\text{efficiency of SEL } * \text{ robustness margin } \rho \text{ of } \pi} \right)$$
,

and the norm of the state $||x_t||$ is at most

$$\Gamma_2\left(\frac{\text{size of uncertainty }\mathcal{F}}{\text{efficiency of SEL }* \text{ single-step robustness margin of }\pi}, \|x_0\|\right),$$

for some increasing function $\Gamma_1 : \mathbb{R}^+ \mapsto \mathbb{R}^+$ and some function $\Gamma_2 : \mathbb{R}^+ \mapsto \mathbb{R}^+$ which is increasing in the first argument and is linear in the second.

- Performance of π: Assume the worst-possible f* ∈ F, but also access to direct online measurements θ_t = θ* + v_t of the a true parameter θ* with small noise v_t of size ρ; M^π_ρ denotes the worst-case mistakes if we were to apply the almost ideal control law u_t = π[θ_t](t, x_t) in this setting.
- *Efficiency of* SEL: We quantify the efficiency of SEL in the result through competitive analysis of online algorithms. The procedure SEL posits parameters efficiently, if as a function of time, the parameter selection θ_t changes only when necessary; that is, it only changes when new observations are informative and keep a constant value otherwise. We phrase this in terms of a *competitive ratio* γ (with γ ≥ 1) and distinguish here between γ-competitive and (γ, T)-finite-time competitive algorithms. The smaller the constant γ is, the more efficient the algorithm posits parameters. As discussed in Section 6.5, a smaller γ indicates that an online algorithm performs more closely to the ideal optimal algorithm in hindsight.

Remark 31. If the same procedure π serves as a robust oracle for a set of criteria $\mathcal{G}^{(1)}, \mathcal{G}^{(2)}, \ldots, \mathcal{G}^{(M)}$, then correspondingly the instantiation $\mathcal{A}_{\pi}(SEL)$ provides

multiple finite mistake guarantees, i.e., one for each corresponding criteria $\mathcal{G}^{(i)}$, $i = 1, \ldots, M$.

This approach brings several attractive qualities:

- Generality. The result applies to a wide range of problem settings. The objective \mathcal{G} and the uncertainty set \mathcal{F} serve as a flexible abstraction to represent a large class of dynamical systems and control-theoretic performance objectives.
- *Robust guarantees in the large uncertainty setting.* Our result applies in settings where only rough models are available. As an example, we can use the result to provide guarantees in control settings with unstable nonlinear systems where stabilizing policies are **not** known a-priori and which are subject to online adversarial disturbances.
- Decoupling algorithm design for learning and control. The construction of the "robust oracle" π and the consistent model chasing procedure SEL can be addressed with existing tools from control and learning. More generally, this perspective enables us to decouple learning and control problems in the large uncertainty setting into separate robust control and online learning problems, a novel approach. See discussion in Section 6.3 and Section 6.5.
- *Modular algorithm design for robust learning and control.* The above approach provides a first interface between robust control and online learning, which enables a modular design of learning and control algorithms with versatile worst-case performance guarantees against large model uncertainty.
- A new tool for performance analysis of learning and control algorithms. We can view the above theorem also from an analysis point-of view: Many existing certainty-equivalence based learning and control algorithms can be easily represented as an instance of the meta-algorithm $\mathcal{A}_{\pi}(SEL)$ and thus can be analyzed using the above theorem.

Promising for Design of Efficient Algorithms in pPractice. Besides focusing on providing worst-case guarantees in a general setting, empirical results show that our framework is a promising approach to design efficient algorithms for learning and control in practice. In Section 6.11, we apply our approach to the problem of swinging-up a cartpole with large parametric uncertainty in a realistic and highly challenging setting and show that it achieves consistently (over 900 experiments with different parameter settings) good performance.

As summarized in Algorithm 2, the main ingredients of our approach are a robust control oracle π that returns a robust controller under posited system parameters, and an online algorithm SEL that chases parameter sets that are consistent with the data collected so far. In the following two sections we formulate the formal concept of oracles and model selectors SEL and discuss their respective properties required for the statment of our main results.

6.3 Control Oracle: An Abstraction for Nominal Control Design

It is obvious that our OC-MG problem is only well-posed if the underlying control problem under no uncertainty, i.e., the *nominal control design* problem, is feasible as well. In particular, for any dynamics $f' \in \mathcal{F}$, there has to be a control policy $\kappa' \in \mathcal{K}$ which, in closed-loop, provides worst-case stability and mistake guarantees. To state this formally, let the dynamical model $M[f, \kappa] \subset (\mathcal{X} \times \mathcal{U})^{\mathbb{N}}$ be the set of all closed-loop trajectories obtained by interconnecting the dynamics f with controller κ , i.e.,

$$\mathsf{M}[f,\kappa] = \left\{ (\boldsymbol{\tau}^{\mathrm{x}},\boldsymbol{\tau}^{\mathrm{u}})^{\top} \middle| \forall t \in \mathbb{N} : \begin{array}{c} \boldsymbol{\tau}^{\mathrm{x}}(t+1) &= f(t,\boldsymbol{\tau}^{\mathrm{x}}(t),\boldsymbol{\tau}^{\mathrm{u}}(t)) \\ \boldsymbol{\tau}^{\mathrm{u}}(t) &= \kappa(t,\boldsymbol{\tau}^{\mathrm{x}}(t)) \end{array} \right\}, \quad (6.6)$$

then our problem OC-MG is only feasible, if for any $f' \in \mathcal{F}$ there exists some $\kappa' \in \mathcal{K}$ such that for all $\tau \in M[f', \kappa']$ holds $\tau \in \ell_{\infty}^{\mathcal{X} \times \mathcal{U}}$ (bounded closed-loop trajectories) and $\sum_{t=0}^{\infty} \mathcal{G}_t(\tau(t)) < \infty$ (each closed-loop trajectory makes finite mistakes).

The main purpose of the oracle π is to serve as an abstraction for nominal control design which is robust to some small degree of model uncertainty. The procedure π is a map $\Omega \mapsto \mathcal{K}$ from parameter space Ω to the space $\mathcal{K} := \{\kappa : \mathbb{N} \times \mathcal{X} \mapsto \mathcal{U}\}$ of all (non-stationary) control policies of the form $u_t = \kappa(t, x_t)$. The purpose of π is to specify a parametrized collection of "fixed"-model-based controllers suitable for certainty-equivalent control. A desired property of π as an *oracle* is that π returns controllers that satisfy \mathcal{G} if the model uncertainty *were* small. Thus, if we let $\mathsf{CL}_{\pi}[\omega] \subset (\mathcal{X} \times \mathcal{U})^{\mathbb{N}}$ be the dynamic model below,

Definition 6.2. $CL_{\pi}[\omega] \subset (\mathcal{X} \times \mathcal{U})^{\mathbb{N}}$ *is the dynamic model of the closed-loop interconnection of the dynamics* $\mathbb{D}[\omega]$ *and the oracle policy* $\pi[\omega]$ *and is equivalently written as:*

$$\mathsf{CL}_{\pi}[\omega] = \bigcup_{f \in \mathbb{D}[\omega]} \mathsf{M}[f, \pi[\omega]].$$

We can formulate a necessary requirements for the nominal problem setting to be:

$$\forall \boldsymbol{\tau} \in \bigcup_{\omega \in \boldsymbol{\Omega}} \mathsf{CL}_{\pi}[\omega] \ : \ \sum_{t=0}^{\infty} \mathcal{G}_t(\boldsymbol{\tau}(t)) < \infty \text{ and } \boldsymbol{\tau} \in \ell_{\infty}^{\mathcal{X} \times \mathcal{U}}.$$

In other words, if the uncertainty set \mathcal{F} were contained in the set $\mathbb{D}[\theta]$, then control policy $\pi[\theta]$ could guarantee to achieve the objective \mathcal{G} with finite mistake guarantees. The above statement is subsumed in the formal oracle requirements, formulated in the next section. Furthermore, in an idealized setting where the true parameter were known exactly, the oracle should return a policy such that the system performance is robust to some level of bounded noise. This is a standard notion of *robustness*, which we later define more precisely. Naturally, there exist many control methods in the control literature which are suitable for robust oracle design. Which method to use depends on the control objective \mathcal{G} , the specific application, and the system class (linear/nonlinear/hybrid, etc.). For a broad survey, see [118, 119, 142] and references therein. We characterize two general methodologies (which can also be combined):

- Robust stability analysis focus: In an initial step, we use analytical design principles from robust nonlinear and linear control design to propose an oracle π[θ](x) in closed-form for all θ and x. In a second step we prove robustness using analysis tools such as for example *Input-to-State Stability* (ISS) stability analysis [75] or robust set invariance methods [106, 107]).
- *Robust control synthesis:* If the problem permits, we can also directly address the control design problem from a computational point of view, by formulating the design problem as an optimization problem and compute for a control law with desired guarantees directly. This can happen partially online, partially offline. Some common nonlinear approaches are robust (tube-based) MPC [30, 96], SOS-methods [100],[21], Hamilton-Jacobi reachability methods [22].

There are different advantages and disadvantages to both approaches, and it is important to point out that robust control problems are not always tractably solvable. See [27, 31] for simple examples of robust control problems which are NP-hard. The computational complexity of robust controller synthesis tends to increase (or even be potentially infeasible) with the complexity of the system of interest; it also further increases as we try optimize for larger robustness margins ρ .

The Dual Purpose of the Oracle. In our framework, access to a robust oracle is a necessary prerequisite to design learning and control agents $A_{\pi}(SEL)$ with mistake guarantees. However, this is a mild assumption and is often more enabling than

restrictive. First, it represents a natural way to ensure well-posedness of the overall learning and control problem; If robust oracles cannot be found for an objective, then the overall problem is likely intrinsically hard or ill-posed (for example, necessary fundamental system properties like stabilizability/ detectability are not satisfied).

Second, oracle abstraction enables a modular approach to robust learning and control problems and directly leverages existing powerful methods in robust control: Any model-based design procedure π that works well for the small uncertainty setting (i.e., acts as a robust oracle) can be augmented with an online chasing algorithm SEL (with required chasing properties) to provide robust control performance (in the form of mistake guarantees) in the large uncertainty setting via the augmented algorithm $\mathcal{A}_{\pi}(SEL)$.

Next, we formulate the robustness properties expected from the oracle π .

6.4 Robust Oracle Guarantees

The main requirement on the oracle is that if the model uncertainty were small enough, we could just use the oracle for certainty-equivalent control. We phrase this requirement by describing the performance of the policy in an idealized setting. Let θ^* be a parameter of true dynamics f^* , and assume that online we have access to noisy observations $\theta = (\theta_0, \theta_1, ...)$, where each measurement θ_t is ρ -close to θ^* , under metric d. The online control algorithm queries π at each time step and applies the corresponding policy $\pi[\theta_t]$. The resulting trajectories obey the equations:

$$x_{t+1} = f^*(t, x_t, u_t), \quad u_t = \pi[\theta_t](t, x_t)$$
 (6.7a)

$$\theta_t \text{ s.t.: } d(\theta_t, \theta^*) \leq \rho, \quad \text{where } f^* \in \mathbb{D}[\theta^*].$$
(6.7b)

To facilitate later discussion, define the set of all feasible trajectories of the dynamic equations (6.7) as the *nominal trajectories* $S_{\mathcal{I}}[\rho; \theta]$ of the oracle:

Definition 6.3. For a time-interval $\mathcal{I} = [t_1, t_2] \subset \mathbb{N}$ and fixed $\theta \in \Omega$, let $\mathcal{S}_{\mathcal{I}}[\rho; \theta]$ denote the set of all pairs of finite trajectories $x_{\mathcal{I}} := (x_{t_1}, \ldots, x_{t_2}), u_{\mathcal{I}} := (u_{t_1}, \ldots, u_{t_2})$ which for $\theta^* = \theta$, satisfy conditions (6.7) with some feasible f^* and sequence $(\theta_{t_1}, \ldots, \theta_{t_2})$.

Design Specification for Oracles. We will say that π is ρ -robust for some objective \mathcal{G} , if all trajectories in $\mathcal{S}_{\mathcal{I}}[\rho; \theta]$ achieve \mathcal{G} after finitely many mistakes. We distinguish between robustness and uniform robustness, which we define precisely below.

ρ-robust	$\forall \theta \in \mathbf{\Omega} : \ \sup_{\gamma \geqslant 0} m_{\rho}^{\pi}(\gamma; \theta) < \infty$
uniformly ρ -robust	$M^{\pi}_{\rho} := \sup_{\gamma \ge 0, \theta \in \mathbf{\Omega}} m^{\pi}_{\rho}(\gamma; \theta) < \infty$
locally ρ -robust	$\forall \gamma \ge 0, \ \theta \in \mathbf{\Omega}: \ m_{ ho}^{\pi}(\gamma; \theta) < \infty$
locally uniformly ρ -robust	$\forall \gamma \ge 0: \ M^{\pi}_{\rho}(\gamma) := \sup_{\theta \in \mathbf{\Omega}} m^{\pi}_{\rho}(\gamma; \theta) < \infty$

Table 6.1: Notions of oracle-robustness

Definition 6.4 (robust oracle). *Equip* \mathcal{X} with some norm || ||. For each $\rho, \gamma \ge 0$ and $\theta \in \Omega$, define the quantity $m_{\rho}^{\pi}(\gamma; \theta)$ as

$$m_{\rho}^{\pi}(\gamma;\theta) := \sup_{\mathcal{I}=[t,t']: t < t'} \sup_{(x_{\mathcal{I}}, u_{\mathcal{I}}) \in \mathcal{S}_{\mathcal{I}}[\rho;\theta], \|x_0\| \leqslant \gamma} \sum_{t \in \mathcal{I}} \mathcal{G}_t(x_t, u_t).$$

If $m_0^{\pi}(\gamma; \theta) < \infty$ for all $\gamma \ge 0$, $\theta \in \Omega$, we call π an oracle for \mathcal{G} w.r.t. parametrization (\mathbb{D}, Ω, d) . In addition, we say that an oracle π is (locally) (uniformally) ρ -robust if the corresponding property shown in Table 6.1 holds. If it exists, M_{ρ}^{π} is the mistake constant/function of π .

The constant $\rho > 0$ will be referred to as the robustness margin of π . If we use the above terms without referencing ρ , it should be understood that there exists some $\rho > 0$ for which the corresponding property is feasible. The mistake constant M_{ρ}^{π} can be viewed as a robust offline benchmark. It quantifies how many mistakes we would make in the worst case if we could use the oracle π under idealized conditions, that is, described by (6.7).

Invariance Property. On top of ρ -robustness, for some results, we will require the following additional condition from the oracle:

Definition 6.5. For a fixed objective \mathcal{G} , define the set of admissible states at time t as $X_t = \{x \mid \exists u' : \mathcal{G}_t(x, u') = 0\}$, i.e., the set of states for which it is possible to achieve zero cost at the time step t. We call a ρ -robust oracle π cost invariant, if for all $\theta \in \Omega$ and $t \ge 0$ the following holds:

- For all $x \in X_t$ holds $\mathcal{G}_t(x, \pi[\theta](t, x)) = 0$.
- For all $x \in X_t$, $f \in \mathbb{D}[\theta]$ and θ' s.t. $d(\theta', \theta) \leq \rho$, holds $f(t, x, \pi[\theta'](t, x)) \in X_{t+1}$.

Remark 32. The above condition is related to the well-known notion of positive set/tube invariance in control theory [26]: The above condition requires that the

oracle policies $\pi[\theta]$ can ensure for their nominal model $\mathbb{D}(\theta)$ the following closed loop condition: $x_t \in X_t \implies x_{t+1} \in X_{t+1}, \forall t$.

Robustness of Single-Step Closed Loop Transitions. To provide worst-case guarantees on the online state trajectory, we need to bound how system uncertainty can affect a single online time-step state transition in the worst case. To this end, consider equipping the state space \mathcal{X} with some norm $\|\cdot\|$ and defining a desired property for the oracle in terms of its performance in the idealized scenario:

Definition 6.6. $\pi : \mathbf{\Omega} \mapsto \mathcal{K}$ is (α, β) -single step robust in the space $(\mathcal{X}, \|\cdot\|)$ if for any 2-time-steps nominal trajectory $(x_{t+1}, x_t), (u_{t+1}, u_t) \in \mathcal{S}_{[t,t+1]}[\rho; \theta]$ holds $\|x_{t+1}\| \leq \alpha \rho \|x_t\| + \beta$.

The above property requires that in the idealized setting (6.7), we can uniformly bound the single-step growth of the state by a scalar linear function in the noise-level ρ and the previous state norm. Equivalently, we can explicitly write out the condition in Def. 6.6 as $\exists \alpha, \beta > 0 : \forall \theta, \theta' \in \mathbf{\Omega}, x \in \mathcal{X}, f \in \mathbb{D}[\theta], t \ge 0$ s.t.:

$$\|f(t, x, \pi[\theta'](t, x))\| \leq \alpha d(\theta, \theta') \|x\| + \beta.$$

We use a simplified problem setting to explain the correspondence between the discussed conditions and standard problems in the control literature.

Mistake Guarantees and Set Convergence Stability

Consider a class of systems of the form $x_{t+1} = g(x_t, u_t; \theta^*) + w_t, ||w_t|| \leq 1$, where θ^* is an unknown system parameter that lies in a known compact set $\Omega \subset \mathbb{R}^m$. We represent the uncertainty set as $\mathcal{F} = \bigcup_{\theta \in \Omega} \mathbb{D}[\theta]$ with $\mathbb{D}[\theta] := \{f^* : t, x, u \mapsto g(x, u; \theta) + w_t | ||w||_{\infty} \leq 1\}$. Let $\pi : \Omega \mapsto \mathcal{K}$ be a procedure which returns state feedback policies $\pi[\theta] : \mathcal{X} \mapsto \mathcal{U}$ for a given $\theta \in \Omega$. Designing an uniformly ρ -robust oracle π can be equivalently viewed as making the closed-loop system (described by (6.7)) of the idealized setting robust to disturbance and noise. For the considered example, the closed loop can be represented by the dynamic model $G_{\theta^*} \in (\mathcal{X} \times \mathcal{U} \times \mathcal{Y} \times \mathcal{V} \times \mathcal{W})^{\mathbb{N}}$, defined as

$$\mathsf{G}_{\theta^*} := \{ (\boldsymbol{x}, \boldsymbol{u}, \boldsymbol{y}, \boldsymbol{v}, \boldsymbol{w})^\top \text{ such that } (6.8) \}$$

where for all $t \in \mathbb{N}$, $y_t \in \mathcal{Y} = \mathbb{R}$ is an output representing the cost at time t, $v_t \in \mathcal{V} = \mathbb{R}^m$ is an input representing bounded observational noise and $w_t \in \mathcal{W} = \mathcal{X}$ is an input representing a disturbance signal.

$$x_{t+1} = g(x_t, u_t; \theta^*) + w_t$$
(6.8a)

$$u_t = \pi [\theta^* + v_t](x_t) \tag{6.8b}$$

$$y_t = \mathcal{G}_t(x_t, u_t) \tag{6.8c}$$

Leaning on our discussion in Chapter 2, we can express the oracle-robustness property in terms of stability of the {wv} \mapsto {xuy}-map, denoted by the operator $\Phi_{\theta^*}^{wv \mapsto xuy} \in C(\ell^{W \times V}, \ell^{X \times U \times Y})$ of the dynamic model G_{θ^*} . We call π a uniformly ρ -robust oracle if, with respect to the domain { $(v, w)^{\top} | ||v||_{\infty}, \leq \rho, ||w||_{\infty} \leq 1$ }, the family of partial maps { $\Phi_{\theta^*}^{wv \mapsto Y}$ } $_{\theta^* \in \Omega}$ is uniformly stable in the $\ell_{\infty}^{W \times V} \mapsto \ell_1$ -induced operator norm. The fixed constant M_{ρ}^{π} (or function M_{ρ}^{π}) which we used in the definitions of oracle robustness quantifies the uniform stability of the family of operators { $\Phi_{\theta^*}^{wv \mapsto Y}$ } $_{\theta^* \in \Omega}$ similarly to the concept of gain, which is commonly used in operator theory in the control theory literature [139]. Moreover, if we identify the cost functions \mathcal{G}_t with their level sets $S_t := \{(x, u) | \mathcal{G}_t(x, u) = 0\}$, we can also rephrase the former conditions as a form of robust trajectory-tracking problem or a set-point control problem² [77]. It is common in control theory to provide guarantees in the form of convergence rates (finite-time or exponential convergence) on the tracking-error; these guarantees can be directly mapped to M_{ρ}^{π} and $M_{\rho}^{\pi}(\cdot)$, as shown in the next example.

Example 3. Assume we want to track a desired trajectory \mathbf{x}^d within ε precision in some normed state space $(\mathcal{X}, \|\cdot\|)$. We can phrase this as a control objective \mathcal{G} by defining $\mathcal{G}_t(x, u) := 0$, if $\|x_t^d - x\| \leq \varepsilon$ and $\mathcal{G}_t(x, u) := 1$, otherwise. Providing an exponential convergence guarantee of type $\|x_t^d - x_t\| \leq c\mu^t$ with constants $c, \mu < 1$ is a basic problem studied in control theory. It is easy to see that such a guarantee implies $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leq \frac{\log(c\|x_0\|) + \log(\varepsilon^{-1})}{\log(\mu^{-1})}$. Hence, if design method π provides a policy $\pi[\theta]$ which guarantees (for any $\|\mathbf{w}\|_{\infty} \leq 1$, $\|\mathbf{v}\|_{\infty} \leq \rho$) for any trajectory of the closed-loop $\operatorname{CL}_{\theta}$ the convergence condition $\forall t : \|x_t^d - x_t\| \leq c\mu^t$, then π is a locally ρ -uniformly robust oracle with the mistake function

$$M_{\rho}^{\pi} = \frac{\log(c\|x_0\|) + \log(\varepsilon^{-1})}{\log(\mu^{-1})}.$$

6.5 Consistent Model Chasing

The procedure SEL has the task of efficiently selecting models consistent with the observed online data. In this section, we formulate this as an online learning problem

²In that case, \mathcal{G}_t would be not time dependent.

of *chasing consistent models* and discuss various design approaches on how to address it. Next, we describe the basic setup of the problem.

Consistent Models and Parameters

Let $\mathbb{O} := \{(d_1, \ldots, d_N) \mid d_i \in \mathbb{N} \times \mathcal{X} \times \mathcal{X} \times \mathcal{U}, N < \infty\}$ be the space of all data sets $\mathcal{D} = (d_1, \ldots, d_N)$ of time-indexed data points $d_i = (t_i, x_i^+, x_i, u_i)$. We will call a data sequence $\mathcal{D} = (\mathcal{D}_1, \mathcal{D}_2, \ldots)$ an online stream if the data sets $\mathcal{D}_t = (d_1, \ldots, d_t)$ are formed from a sequence of observations (d_1, d_2, \ldots) . Intuitively, given a data set $\mathcal{D} = (d_1, \ldots, d_N)$ of tuples $d_i = (t_i, x_i^+, x_i, u_i)$, any candidate $f \in \mathcal{F}$ which satisfies $x_i^+ = f(t_i, x_i, u_i)$ for all $1 \le i \le N$ is *consistent* with \mathcal{D} ; Similarly, we will say that f is *consistent* with an online stream \mathcal{D} , if at each time-step t, it is consistent with the data set \mathcal{D}_t . We will extend this definition to models $\mathbb{D}[\theta]$ and parameters $\theta \in \Omega$: The model $\mathbb{D}[\theta]$ is a consistent model for a data set \mathcal{D} or an online stream \mathcal{D} , if it contains at least one function f that is consistent with \mathcal{D} or \mathcal{D} , respectively; θ is then called a consistent parameter. Similarly, for some data set \mathcal{D} , we define the set of all consistent parameters as $\mathsf{P}(\mathcal{D})$:

Definition 6.7 (Consistent Sets). *The map of the consistent set* $P : \mathbb{O} \mapsto 2^{\Omega}$ *returns for each data set* $\mathcal{D} \in \mathbb{O}$ *the corresponding set of consistent parameters* $P(\mathcal{D})$ *:*

$$\mathsf{P}(\mathcal{D}) := \operatorname{closure}\left(\left\{\theta \in \mathbf{\Omega} \mid \exists f \in \mathbb{D}[\theta] : \forall (t, x^+, x, u) \in \mathcal{D}, x^+ = f(t, x, u) \right\}\right).$$
(6.9)

Some important facts follow from this definition. Since the data \mathcal{D}_t at time t always contain the previous data, it is clear that the constraints defining the consistent set at time t are stricter than those at time t - 1. Hence, the set $\mathsf{P}(\mathcal{D}_t)$ is contained in the consistent set $\mathsf{P}(\mathcal{D}_{t-1})$ from a previous time step. We refer to this as the nestedness property of the sequence of sets $(\mathsf{P}(\mathcal{D}_0), \mathsf{P}(\mathcal{D}_1), \ldots,)$. Several other important implications are remarked in the corollary below:

Corollary. Assume \mathcal{D} is a data stream with at least one consistent $f \in \mathcal{F}$. Then, the following holds for the sequence of consistent sets $\mathsf{P}(\mathcal{D}) = (\mathsf{P}(\mathcal{D}_1), \mathsf{P}(\mathcal{D}_2), \dots)$:

- (i) The sequence of consistent sets is nested in Ω , i.e: $\mathsf{P}(\mathcal{D}_t) \supset \mathsf{P}(\mathcal{D}_{t+1})$.
- (ii) $\mathsf{P}(\mathcal{D}_{\infty}) := \left(\bigcap_{k=1}^{\infty} \mathsf{P}(\mathcal{D}_{k})\right) \cap \mathbf{\Omega}$ is non-empty.
- (*iii*) If $\theta_t \in \mathsf{P}(\mathcal{D}_t)$, and $\lim_{t\to\infty} \theta_t = \theta_\infty$, then $\theta_\infty \in \mathsf{P}(\mathcal{D}_\infty)$.

Proof. The first property is clear since $\mathcal{D}_t = \mathcal{D}_{t-1} \cup \{d_t\}$. For the second, notice that $\mathsf{P}(\mathcal{D}_t)$ is always a non-empty compact set and recall the basic real analysis fact

[109]: The intersection of a nested sequence of compact sets that are not empty is always non-empty [109]. To verify property (iii), notice that nestedness implies that the subsequence $(\theta_T, \theta_{T+1}, ...)$ is contained in $\mathsf{P}(\mathcal{D}_T)$; since $\mathsf{P}(\mathcal{D}_T)$ is closed, we have $\theta_{\infty} \in \mathsf{P}(\mathcal{D}_T)$. We chose *T* arbitrarily, so we conclude that $\theta_{\infty} \in \mathsf{P}(\mathcal{D}_T)$ for all $T \ge 0$, that is: $\theta_{\infty} \in \mathsf{P}(\mathcal{D}_{\infty})$.

Chasing Consistent Models

Assume a parameterization (\mathbb{D}, Ω, d) and some fixed data set $\mathcal{D}_T = (d_1, \ldots, d_T)$ which is presented to us in an online fashion and which has at least one consistent parameter θ^* (i.e., $\mathsf{P}(\mathcal{D}_T)$ is nonempty).

Our goal is to find a consistent parameter $\theta^* \in \mathsf{P}(\mathcal{D}_T)$ online, or equivalently, to find the model $\mathbb{D}[\theta^*]$ consistent with all data \mathcal{D}_T . However, since we do not have access to all data upfront, the best we can do is posit at each time-step t a parameter $\theta_t \in \mathsf{P}(\mathcal{D}_t)$ consistent at least with all so far seen data and make the hypothesis that θ_t is also consistent with \mathcal{D}_T until proven otherwise by new data. Our goal is to posit the parameters θ_t in an efficient way; we would like to change our hypothesis about the consistent parameter θ^* as little as possible, and thus θ_t should change over time as little as possible. This task can be interpreted as a two-player game, where player A (our selection SEL) is trying to chase after player B (the set of consistent models $P(\mathcal{D}_t)$) with the objective of tagging them (selecting a parameter $\theta_t \in P(\mathcal{D}_t)$). It suits to call this problem *consistent models chasing* as our objective is not only selecting from the consistent sets but also accounting for future and potentially adversarial changes in the set $P(\mathcal{D}_t)$. Next, we formulate "chasing" conditions for SEL, defined in Def. 6.9, which can address this problem. The basic requirement for SEL is to output a consistent parameter (if one exists) $\theta = \text{SEL}[\mathcal{D}]$ for a given data set \mathcal{D} . Described in the language of set-valued analysis, we require SEL to be a selection or selector $\mathbb{O} \mapsto \Omega$ of the set-valued map $\mathcal{D} \mapsto \mathsf{P}(\mathcal{D})$.

Definition 6.8. [19]. A function $f : X \mapsto Y$ is a selection/selector of the set-valued map $F : X \mapsto 2^Y$, if $\forall x \in X : f(x) \in F(x)$.

Since we intend to use SEL in an online manner where we are given a stream of data \mathcal{D} , we require that SEL can posit consistent parameters $\theta_t = \text{SEL}[\mathcal{D}_t]$ in an efficient manner online. A fitting notion of "efficiency" can be defined by comparing the variation of the parameter sequence $\boldsymbol{\theta}$ and the sequence of consistent sets $P(\mathcal{D}) := (P(\mathcal{D}_1), P(\mathcal{D}_2), \dots)$ over time, where we quantify the latter using the

Hausdorff distance $d_{\mathcal{H}}: 2^{\Omega} \times 2^{\Omega} \mapsto \mathbb{R}^+$ defined as

$$d_{\mathcal{H}}(\mathsf{S},\mathsf{S}') = \max\left\{\max_{x\in\mathsf{S}}d(x,\mathsf{S}')\,,\,\max_{y\in\mathsf{S}'}d(y,\mathsf{S})\right\}.$$

We phrase this in terms of desired "chasing"-properties (A)-(D) and refer to selectors SEL with such properties as (consistent model) chasers:

Definition 6.9. Let SEL : $\mathbb{O} \mapsto \Omega$ be a selection of P. Let $\mathcal{D} = (\mathcal{D}_1, \mathcal{D}_2, ...)$ be an online data stream and let θ be a sequence defined for each time t as $\theta_t = SEL[\mathcal{D}_t]$. Assume that there always exists an $f \in \mathcal{F}$ consistent with \mathcal{D} and consider the following statements:

- (A) asymptotically efficient: $\theta^* = \lim_{t\to\infty} \theta_t$ exists.
- (B) asymptotically finite-time (f.t.) efficient: $\lim_{t\to\infty} d(\theta_t, \theta_{t-1}) = 0$.
- (C) γ -competitive:

$$t_1 < t_2 \implies \sum_{t=t_1+1}^{t_2} d(\theta_t, \theta_{t-1}) \leq \gamma \, d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{t_2}), \mathsf{P}(\mathcal{D}_{t_1})).$$

(D) (γ, T) -finite-time (f.t.) competitive:

$$t_2 - t_1 \leqslant T \implies \sum_{t=t_1+1}^{t_2} d(\theta_t, \theta_{t-1}) \leqslant \gamma \, d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{t_2}), \mathsf{P}(\mathcal{D}_{t_1})).$$

SEL algorithms that satisfy the above chasing properties as referred to as consistent model chasers.

The desired properties describe a natural notion of efficiency for this problem. The posited consistent parameter θ_t should only change online if new data are also informative. The competitiveness properties (C) and (D) naturally restrict changing θ_t when little new information is available and permit bigger changes in θ_t only when new data is informative. Per time interval $\mathcal{I} = [t_1, t_2]$, the inequalities in (C) and (D) enforce the following: If the consistent sets $P(\mathcal{D}_{t_2})$ and $P(\mathcal{D}_{t_1})$ are the same, i.e., $(d_{\mathcal{H}}(P(\mathcal{D}_{t_2}), P(\mathcal{D}_{t_1})) = 0)$, the posited parameters $\theta_{t_1}, \ldots, \theta_{t_2}$ should all have the same value. On the other hand, if $P(\mathcal{D}_{t_2})$ is much smaller than $P(\mathcal{D}_{t_1})$, (i.e. $d_{\mathcal{H}}(P(\mathcal{D}_{t_2}), P(\mathcal{D}_{t_1}))$ large), the total variation $\sum_{t=t_1+1}^{t_2} d(\theta_t, \theta_{t-1})$ in the posited parameters is allowed to be at most $\gamma d_{\mathcal{H}}(P(\mathcal{D}_{t_2}), P(\mathcal{D}_{t_1}))$.

Properties (C) and (D) are stronger versions of (A) and (B), called *competitive-ness/finite-time competitiveness*. The relationship between the chasing properties are summarized in Lemma 51 below. To prove the relation $(D) \implies (B)$ we require the following auxilliary result:

Lemma 50. Let (Ω, d) be a compact metric space and let $\{S_1, \ldots, S_T\}$, $\forall t : S_{t+1} \subset S_t, S_t \subset \Omega$ be a collection of nested subsets in Ω that are ε separated w.r.t. to the Hausdorff metric $d_{\mathcal{H}}$, that is: $d_{\mathcal{H}}(S_i, S_j) > \varepsilon, \forall i \neq j$. Then we have $T \leq N(\Omega, \varepsilon)$.

Proof. Assume that $\{S_1, \ldots, S_T\}$ is a ε -separated subset of the metric space $(2^{\Omega}, d_{\mathcal{H}})$, where $d_{\mathcal{H}}$ is the Hausdorff metric. Since for all t we have $d_{\mathcal{H}}(\Omega_t, \Omega_{t+1}) > \varepsilon$ and $\Omega_{t+1} \subset \Omega_t$, this means that there exists at least one point $p_t \in \Omega_t$ such that $d(p_t, \Omega_{t+1}) > \varepsilon$. Since for all j > t holds $p_j \in \Omega_j \subset \Omega_{t+1}$, we conclude $d(p_t, p_j) \ge d(p_t, \Omega_{t+1}) > \varepsilon$ for all j > t. This establishes that $\{p_1, \ldots, p_T\}$ is a ε -separated subset of Ω . Therefore, we can bound the size of the set T by the packing number $T \le N(\Omega, \varepsilon)$.

Lemma 51. Then the following implications hold between the properties of Def. 6.9:



The reverse (and any other) implications between the properties do not hold in general.

Proof. (A) \implies (B), (C) \implies (D) are obvious. (C) \implies (A) follows by noticing that (C) implies $\sum_{t=1}^{\infty} d(\theta_t, \theta_{t-1}) \leq \gamma \operatorname{diam}(\Omega)$. To prove (D) \implies (B), we use Lem. 50. First notice that (γ, T) -finite time competitiveness (f.t.) implies $(\gamma, 1)$ -w.c.. Pick some $\varepsilon > 0$ and let $\mathbb{I} = \{t_1, \ldots, \}$ be all time-steps at which $d(\theta_t, \theta_{t-1}) \geq \varepsilon$. Now, for each $t \in \mathcal{I}$ holds $d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_t), \mathsf{P}(\mathcal{D}_{t-1})) \geq \frac{\varepsilon}{\gamma}$. We can verify that the collection of sets $\{\mathsf{P}(\mathcal{D}_{t-1}) \mid t \in \mathbb{I}\}$ is a $\frac{\varepsilon}{\gamma}$ -separated set in the metric space $(2^{\Omega}, d_{\mathcal{H}})$; therefore, by Lem. 50 it follows that the index set \mathbb{I} is finite, that is, $|\mathbb{I}| \leq N(\Omega, \frac{\varepsilon}{\gamma})$. Since this holds for all $\varepsilon > 0$, we proved $\forall \varepsilon > 0 \exists N$ s.t. $\forall t \geq N : d(\theta_t, \theta_{t-1}) < \varepsilon$, i.e., $\lim_{t\to\infty} d(\theta_t, \theta_{t-1}) = 0$.

From the above diagram, we can see that γ -competitiveness is the strongest chasing property as it implies all other properties. Furthermore, the condition (γ, T) -finite-time competitiveness strengthens with larger T, as shown below:

Corollary 52. (γ, T) -finite time competitiveness implies $(k\gamma, kT)$ -finite time competitiveness for any $k \in \mathbb{N}$.

Proof. Assume (γ, T) -f.t.-competitive and pick k consecutive intervals $\mathcal{I}_1, \ldots, \mathcal{I}_k$, $\mathcal{I}_j := [\tau_j, \bar{\tau}_j]$, each of length T. Notice that the nestedness property $\mathsf{P}(\mathcal{D}_{\bar{\tau}_k}) \subset \cdots \subset \mathsf{P}(\mathcal{D}_{\tau_1})$ implies $d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{\bar{\tau}_j}), \mathsf{P}(\mathcal{D}_{\tau_j})) \leq d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{\bar{\tau}_k}), \mathsf{P}(\mathcal{D}_{\tau_1}))$ which leads to:

$$\sum_{j=1}^{k} \sum_{t \in \mathcal{I}_{j}} d(\theta_{t+1}, \theta_{t}) \leq \gamma \sum_{j=1}^{k} d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{\bar{\tau}_{j}}), \mathsf{P}(\mathcal{D}_{\tau_{j}})) \leq \gamma k d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{\bar{\tau}_{k}}), \mathsf{P}(\mathcal{D}_{\tau_{1}})).$$

In the next section, we discuss that in cases where the consistent set map P returns convex sets, γ -competitive CMC algorithms can be designed via a reduction to the nested convex bodies chasing (NCBC) problem [33]. On the other hand, for T = 1 and any $\gamma \ge 1$, the weaker chasing property (D) can always be achieved, even if the map P returns arbitrary non-convex sets. We show in Section 6.5 a simple projection-based selection rule, which satisfies $(\gamma, 1)$ -finite-time competitiveness.

Competitive Chasing via Competitive Nested Convex Body Chasing

The main difficulty in selecting the parameters θ_t to solve CMC competitively is that, for any time t < T, we cannot guarantee the selection of a parameter θ_t that is guaranteed to be in the future consistent set $P(D_T)$. The notion of competitiveness is a common performance objective in the design of online learning algorithms [29, 34, 58, 82, 115, 136]. Moreover, it turns out that in the case where a sequence of consistent sets is always convex, we can reduce the CMC problem to a well-known problem of *nested convex body chasing* (NCBC) [33]. This requirement is necessary for the reduction and is stated in Assumption 6.6.

Assumption 6.6. *Given a compact parameterization* (\mathbb{T}, Ω, d) *of the uncertainty set* \mathcal{F} *, the consistent sets* $\mathsf{P}(\mathcal{D})$ *are always convex for any data set* $\mathcal{D} \in \mathbb{O}$ *.*

Constructing consistent sets $P(D_t)$ online can be addressed with tools from setmembership identification. For a large collection of linear and nonlinear systems, the sets P(D) can be constructed efficiently online. Such methods have been developed and studied in the literature of set-membership identification, for a recent survey see [97]. Moreover it is often possible to construct P(D) as an intersection of finite half-spaces, allowing for tractable representations as LPs. To see a particularly simple example, consider the following nonlinear system with some unknown parameters $\alpha^* \in \mathbb{R}^M$ and η^* , where w_t is a vector with entries in the interval $[-\eta^*, \eta^*]$:

$$x_{t+1} = \sum_{i=1}^{M} \alpha_i^* \psi_i(x_t, u_t) + w_t,$$
(6.10)

where $\psi_i : \mathcal{X} \times \mathcal{U} \mapsto \mathcal{X}$ are M known nonlinear functions. If we represent the above system as an uncertain system \mathbb{D} with parameter $\theta^* = [\alpha^*; \eta^*]$, it is easy to see that the consistent sets $\mathsf{P}(\mathcal{D})$ for some data $\mathcal{D} = \{(x_i^+, x_i, u_i) \mid 1 \leq i \leq H\}$ takes the form of a polyhedron

$$\mathsf{P}(\mathcal{D}) = \{ \theta = [\alpha; \eta] \mid \text{ s.t. (6.11) for all } 1 \leq i \leq H \},\$$

defined by the inequalities

$$[\psi_1(x_i, u_i), \dots, \psi_M(x_i, u_i)] \alpha \leqslant x_i^+ + \mathbb{1}\eta, \qquad (6.11a)$$

$$[\psi_1(x_i, u_i), \dots, \psi_M(x_i, u_i)] \alpha \ge x_i^+ - \mathbb{1}\eta.$$
(6.11b)

We can see that any linear discrete-time system can be put into the above form (6.10). Moreover, as shown in Section 6.10 the above representation also applies for a large class of (nonlinear) robotics system.

Nested Convex Body Chasing (NCBC)

In NCBC, we have access to a nested sequence S_0, S_1, \ldots, S_T of convex sets online in some metric space (\mathcal{M}, d) (that is: $S_t \subset S_{t-1}$). The learner selects at each time t a point p_t from S_t . The goal of competitive NCBC is to produce p_1, \ldots, p_T online so that the total cost of moving $\sum_{j=1}^T d(p_j, p_{j-1})$ at time T is competitive with the offline optimum, that is, there is some $\gamma > 0$ stain. $\sum_{j=1}^T d(p_j, p_{j-1}) \leqslant$ $\gamma \max_{p_0 \in S_0} \operatorname{OPT}_T(p_0)$, where $\operatorname{OPT}_T(p_0) := \min_{p \in S_T} d(p, p_0)$.

Remark 33. NCBC is a special case of the more general convex body chasing (CBC) problem, first introduced by [55], which studied competitive algorithms for metrical goal systems.

Let the sequence of convex consistent sets $P(D_t)$ be the corresponding S_t of the NCBC problem, any γ -competitive agent A for the NCBC problem can instantiate a γ -competitive selection for competitive model-chasing, as summarized in the following reduction:

Proposition 34. Consider the setting of Assumption 6.6. Then any γ -competitive algorithm for NCBC in metric space (Ω, d) instantiates via Algorithm 3 a γ -competitive CMC algorithm SEL_{NCBC} in the parametrization (\mathbb{T}, Ω, d) .

Require: γ -competitive NCBC algorithm \mathcal{A}_{NCBC} , consistent set procedure P

1: procedure $SEL_{NCBC}(t, x^+, x, u)$ 2: $\mathcal{D}_t \leftarrow \mathcal{D}_{t-1} \cup (t, x^+, x, u)$ 3: $S_t \leftarrow P(\mathcal{D}_t)$ \triangleright construct/update new consistent set 4: present set S_t to \mathcal{A}_{NCBC} 5: \mathcal{A}_{NCBC} chooses $\theta_t \in S_t$ 6: return θ_t 7: end procedure

Proof. Since we set $S_t = P(\mathcal{D}_t)$, it is clear that $\max_{p_0 \in S_0} OPT_T(p_0)$ is equal to $d_{\mathcal{H}}(P(\mathcal{D}_T), P(\mathcal{D}_0))$. Therefore, γ -competitive NCBC implies γ -competitive CMC on all time intervals $[t_0, T]$, with $t_0 = 1$. To see that this also holds for any choice of t_0 , recall that the NCBC problem requires the competitiveness condition to hold for any sequence of nested convex bodies. Thus, for a fixed sequence S_t the condition must also be satisfied for the shifted sequences $S'_t := S_{t+k}$.

Simple Competitive NCBC-Algorithms in Euclidean Space \mathbb{R}^n . When (Ω, d) is a compact euclidean finite dimensional space, recent exciting progress on the NCBC problem provides a variety of competitive algorithms [12, 13, 33, 114] that can instantiate competitive selections per Algorithm 3.

We highlight two simple instantiations based on the results in [12] and [33]. Both algorithms can be tractably implemented in the setting of assumption 6.6. The selection criteria for $SEL_p(\mathcal{D}_t)$ and $SEL_s(\mathcal{D}_t)$ is defined as:

$$\operatorname{SEL}_{p}(\mathcal{D}_{t}) := \underset{\theta \in \mathsf{P}(\mathcal{D}_{t})}{\operatorname{arg\,min}} \|\theta - \operatorname{SEL}_{p}(\mathcal{D}_{t-1})\|, \tag{6.12a}$$

$$\mathsf{SEL}_s(\mathcal{D}_t) := s(\mathsf{P}(\mathcal{D}_t)),\tag{6.12b}$$

where SEL_p defines simply a greedy projection operator and where SEL_s selects according to the *Steiner-Point* $s(P(\mathcal{D}_t))$ of the consistent set $P(\mathcal{D}_t)$ at time t.

Definition 6.10 (Steiner Point). For a convex body Ω , the Steiner point is defined as the following integral over the n-1 dimensional sphere \mathbb{S}^{n-1} :

$$s(\mathbf{\Omega}) = n \int_{v \in \mathbb{S}^{n-1}} \max_{x \in \mathbf{\Omega}} \langle v, x \rangle v dv.$$
(6.13)

Another equivalent definition, which is more useful computationally, is to define the Steiner point as the weighted average of the extreme points of Ω , where the weights

are the polar angles of each extreme point [113]. A convenient way to express this in terms of averages of random linear programs: $s(\Omega)$ is the expectation of a random variable $V_t(c)$, where each random variable V_i is the solution to a linear program $\max_{x\in\Omega} c_i^{\top} x$ with a cost-vector c, whose direction in \mathbb{R}^n is picked uniformly at random. Moreover, in euclidean spaces it suffices to sample c from the n-dimensional standard Gaussian distribution $\mathcal{N}(0, I_n)$:

(6.13)
$$\Leftrightarrow$$
 $s(\mathbf{\Omega}) = \mathbb{E}_{c \sim \mathcal{N}(0, I_n)} \underset{v \in \mathbf{\Omega}}{\operatorname{arg\,max}} \langle c, v \rangle.$

Remark 35. As shown in [13], the Steiner point can be efficiently approximated by solving randomized linear programs as used in the definition above.

The competitive analysis presented in [33] applies, and appealing to Proposition 34 we can establish that SEL_p and SEL_s are competitive CMC algorithms:

Corollary 53 (of Theorem 1.3 [12], and Theorem 2.1 [33]). Assume Ω is a compact convex set in \mathbb{R}^n and $d(x, y) := ||x - y||_2$. Then, the procedures SEL_p and SEL_s are competitive (CMC)-algorithms with competitive ratio γ_p and γ_s :

$$\gamma_p = (n-1)n^{\frac{n+1}{2}}, \qquad \gamma_s = \frac{n}{2}.$$
 (6.14)

The following lemma is needed to translate the results of [12] and Theorem 2.1 [33] into our setting:

Lemma 54 (Steiner Point). Let $s(\mathbf{\Omega})$ denote the Steiner point of a convex body. The following inequalities hold for a nested sequence of convex bodies $\mathbf{\Omega}_0 \supset \mathbf{\Omega}_1 \cdots \supset \mathbf{\Omega}_T$ and their $d_{\mathcal{H}}$ -pathlength $\Gamma_T = \sum_{t=1}^T \|s(\mathbf{\Omega}_t) - s(\mathbf{\Omega}_{t-1})\|_2$:

$$\Gamma_T \leq \frac{n}{2} \operatorname{diam}(\mathbf{\Omega}_1), \qquad \Gamma_T \leq n d_{\mathcal{H}}(\mathbf{\Omega}_1, \mathbf{\Omega}_T).$$

Proof. Assume $\Omega_1 \supset \Omega_2 \cdots \supset \Omega_T$ is a nested sequence of convex bodies in \mathbb{R}^n . Then, as shown in [33] Thm 2.1, we have the following

$$\sum_{t=1}^{T} \|s(\mathbf{\Omega}_t) - s(\mathbf{\Omega}_{t-1})\|_2 \leq \frac{n}{2} (w(\mathbf{\Omega}_1) - w(\mathbf{\Omega}_T)),$$
(6.15)

where w(S) denotes *mean-width* of the set S. w(S) can be written as the average length of a random 1-dimensional projection of S: Let v be uniformly distributed in the sphere \mathbb{S}^{n-1} and let w(S) be defined as the expectation of $l_v(S)$,

$$w(\mathsf{S}) := \mathbb{E}_{v \sim \texttt{Unif}(\mathbb{S}^{n-1})} l_v(\mathsf{S}), \quad l_v(\mathsf{S}) := \operatorname{diam}(\operatorname{Proj}_v(\mathsf{S}))$$
(6.16)

where $l_v(S)$ is evaluated by first projecting the set S into the subspace spanned by the vector v, denoted $\operatorname{Proj}_v(S)$ and taking its length. By linearity of expectation, we can write

$$\sum_{t=1}^{T} \|s(\mathbf{\Omega}_{t}) - s(\mathbf{\Omega}_{t-1})\|_{2} \leq \frac{n}{2} \mathbb{E}_{v \sim \text{Unif}(\mathbb{S}^{n-1})} l_{v}(\mathbf{\Omega}_{1}) - l_{v}(\mathbf{\Omega}_{T})$$
$$\leq \frac{n}{2} \mathbb{E}_{v \sim \text{Unif}(\mathbb{S}^{n-1})} 2d_{\mathcal{H}}(\mathbf{\Omega}_{1}, \mathbf{\Omega}_{T}) = nd_{\mathcal{H}}(\mathbf{\Omega}_{1}, \mathbf{\Omega}_{T}) \quad (6.17)$$

where the last inequality comes from noticing that according to the definition of the Hausdorff distance, it holds $\Omega_1 \subset \Omega_T \oplus d_{\mathcal{H}}(\Omega_1, \Omega_T) B_{\|\cdot\|_2}$. Similarly, we obtain

$$\sum_{t=1}^{T} \|s(\mathbf{\Omega}_t) - s(\mathbf{\Omega}_{t-1})\|_2 \leq \frac{n}{2} \mathbb{E}_{v \sim \text{Unif}(\mathbb{S}^{n-1})} l_v(\mathbf{\Omega}_1) \leq \frac{n}{2} \text{diam}(\mathbf{\Omega}_1).$$

A General Approach to Finite Time Competitive Consistent Model Chasing

In contrast to the stricter notion of γ -competitiveness, we can give a simple and general selection rule which is always $(\gamma, 1)$ -finite time competitive:

Definition 6.11 (Projection-based chasing). *Pick* $\theta_t \in \mathsf{P}(\mathcal{D}_t)$ *always such that for some fixed* $\gamma > 0$, *at every time-step t holds* $d(\theta_t, \theta_{t-1}) \leq \gamma d(\mathsf{P}(\mathcal{D}_t), \theta_{t-1})$.

This projection-based chasing algorithm might not always be tractable to implement, since it requires solving a potentially non-convex optimization problem with γ relative accuracy. However, it is trivially (γ , 1)-finite time competitive and describes a simple blueprint to design a general CMC algorithm SEL, that is allowing for potentially *infinite dimensional* metric spaces and *non-convex* consistent sets $P(\mathcal{D}_t)$. Combined with a suitable oracle π , the resulting online control algorithm $\mathcal{A}_{\pi}(SEL_p)$ provides finite mistake guarantees for objectives \mathcal{G} according to Theorem 39.

6.7 Main Results

Assuming that π and SEL meet the required specifications, we can provide the overall guarantees for the algorithm. Let (\mathbb{D}, Ω, d) be a compact parametrization of a given uncertainty set \mathcal{F} . Let π be robust per Definition 6.3 and SEL return consistent parameters per Definition 6.9. We apply the online control strategy $\mathcal{A}_{\pi}(SEL)$ described in Algorithm 2 to system $x_{t+1} = f^*(t, x_t, u_t)$ with unknown dynamics $f^* \in \mathcal{F}$ and denote $(\boldsymbol{x}, \boldsymbol{u})$ as the corresponding state and input trajectories.
We organize our results by the strictness of requirements we impose on the modelchaser SEL. The quality of the overall guarantee depends greatly on the type of chasing property that SEL provides. Provided that the oracle guarantees certain robustness requirements, the chasing conditions $\theta \to \theta_{\infty}$ (A) and $d(\theta_t, \theta_{t-1}) \to 0$ (B) assure that the overall amount of mistakes $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t)$ is finite. What's more, if SEL provides competitive chasing properties such as γ -competitiveness or (γ, T) -finite-time competitiveness, then we can provide stronger guarantees, which state uniform bounds on the worst-case number of mistakes.

Stability and Boundedness of Closed Loop Trajectories

Regardless of the objective \mathcal{G} , we can provide worst-case state norm guarantees for $\mathcal{A}_{\pi}(SEL)$ in a normed state space $(\mathcal{X}, \|\cdot\|)$, if SEL is a competitive or a finite-time competitive CMC algorithm and π provides sufficient robustness guarantees for a single time-step transition:

Theorem 36. Assume that $\pi : \mathbf{\Omega} \to \mathcal{K}$ is (α, β) -single step robust in the space $(\mathcal{X}, \|\cdot\|)$. Then, the following state bound guarantees hold:

(i) If SEL is a γ -competitive CMC algorithm, then:

$$\forall t: \quad \|x_t\| \leqslant e^{\alpha\gamma\phi(\mathbf{\Omega})} \left(e^{-t} \|x_0\| + \beta \frac{e}{e-1} \right).$$

(ii) If SEL is a (γ, T) -finite-time competitive CMC algorithm, then:

$$\|\boldsymbol{x}\|_{\infty} \leq \inf_{0 < \mu < 1} \left(1 + (\alpha \phi(\boldsymbol{\Omega}))^{n^*} \right) \max\{\frac{\beta}{1-\mu}, \|\boldsymbol{x}_0\|\} + \beta \sum_{k=0}^{n^*} (\alpha \phi(\boldsymbol{\Omega}))^k$$

where $n^* = N(\mathbf{\Omega}, \frac{\mu}{\alpha\gamma})$ and $\phi(\mathbf{\Omega})$ denotes the diameter of $\mathbf{\Omega}$.

Proof. Part 1: In each time-step t, it holds $x_{t+1} = f(t, x_t, \pi[t, \theta_t])$ for some $f \in \mathbb{D}(\theta_{t+1})$. Therefore, the following inequality holds at each time-step:

$$\|x_{t+1}\| = \|f(t, x_t, \pi[\theta_t](t, x_t))\| \le \alpha d(\theta_{t+1}, \theta_t) \|x_t\| + \beta.$$
(6.18)

We apply Lem. 55 with the substitution $s_t := ||x_t||$, $\delta_t := \alpha d(\theta_{t+1}, \theta_t)$ and $c := \beta$, to obtain

$$||x_t|| \le e^{\alpha L} \left(e^{-t} ||x_0|| + \beta \frac{e}{e-1} \right).$$
 (6.19)

Part 2: We follow the proof technique used in the main result of [65] to prove boundedness. Given a closed-loop trajectory \boldsymbol{x} , at each time-step t holds $x_{t+1} = f(t, x_t, \pi[\theta_t](t, x_t))$ for some $f \in \mathbb{D}(\theta_{t+1})$. Take an arbitrary time step T. Define \mathbb{I} the set of all indeces k < T for which holds $||x_{k+1}|| > \mu ||x_k|| + \beta$. Notice that for each $k \notin \mathbb{I}$ holds $||x_{k+1}|| \leq \mu ||x_k|| + \beta$ while for $k \in \mathbb{I}$ we have at least the inequality $||x_{k+1}|| \leq \alpha \operatorname{diam}(\mathbf{\Omega}) ||x_k|| + \beta$. Now, for each $k \in \mathbb{I}$ holds

$$\mu \|x_k\| + \beta < \|x_{k+1}\| = \|f(k, x_k, \pi[\theta_k](k, x_k))\| \le \alpha d(\theta_{k+1}, \theta_k) \|x_k\| + \beta,$$

which leads to $d(\theta_{k+1}, \theta_k) > \frac{\mu}{\alpha}$. Now, the (γ, T) finite-time competitive property ensures that $d(\theta_k, \theta_{k+1}) \leq \gamma d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_k), \mathsf{P}(\mathcal{D}_{k+1}))$. We can therefore conclude that:

$$\frac{\mu}{\alpha\gamma} < \frac{1}{\gamma}d(\theta_k, \theta_{k+1}) \le d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_k), \mathsf{P}(\mathcal{D}_j)), \quad \text{for } j > k.$$
(6.20)

Hence, $\{\mathsf{P}(\mathcal{D}_k) \mid k \in \mathbb{I}\}$ is a $\frac{\mu}{\alpha\gamma}$ -separated set in Ω . Therefore, $|\mathbb{I}| \leq N(\Omega, \frac{\mu}{\alpha\gamma})$. Recall again that for each $k \notin \mathbb{I}$ holds $||x_{k+1}|| \leq \mu ||x_k|| + \beta$ while for $k \in \mathbb{I}$ it holds $||x_{k+1}|| \leq \alpha \operatorname{diam}(\Omega) ||x_k|| + \beta$. Following the same arguments as in the Appendix of [65], we obtain the presented $||\cdot||_{\infty}$ -bound on \boldsymbol{x} .

Lemma 55. Let $s = (s_0, s_1, ...)$, $\delta = (\delta_0, \delta_1, ...)$ be non-negative scalar sequences such that $s_{k+1} \leq \delta_k s_k + c$, with $c \geq 0$ and $\sum_{t=0}^{\infty} \delta_t \leq L$. Then s_t is bounded by:

$$s_t \leqslant e^L \left(e^{-t} s_0 + c \frac{e}{e-1} \right).$$

Proof. First, we apply the comparison lemma. Therefore, s_k is bounded above by a sequence γ_k with the dynamics $\gamma_{k+1} = \delta_k \gamma_k + c$, where $\gamma_0 = s_0$. Hence, it suffices to bound the sequence γ_k in order to obtain a bound on s_k . Writing out γ_t yields:

$$\gamma_t = \prod_{k=0}^{t-1} \delta_k s_0 + c \left(1 + \sum_{j=1}^{t-1} \prod_{k=j}^{t-1} \delta_k \right)$$

Recall the basic fact $1 + x \leq e^x$ and notice that each product $\prod_{k=j}^{t-1} \delta_k$ can be bounded as:

$$\prod_{k=j}^{t-1} \delta_k = \prod_{k=j}^{t-1} (1 + (\delta_k - 1)) \leqslant \prod_{j=0}^{t-1} \exp(\delta_k - 1) = \exp\left(\sum_{j=k}^{t-1} (\delta_k - 1)\right)$$
$$\leqslant \exp(L - t) = e^{-t+k} e^L.$$

Therefore we obtain our desired result by bounding γ_t at each time-step t as:

$$\gamma_t \leqslant e^{-t} e^L s_0 + c \left(1 + \sum_{j=1}^{t-1} e^{-j} \right) e^L$$
$$\leqslant e^{-t} e^L s_0 + c e^L \sum_{j=0}^{\infty} e^{-j} \leqslant e^{-t} e^L s_0 + c \frac{e^{L+1}}{e-1}.$$

Notice that the above worst-case guarantee holds for *arbitrarily large* diameter $\phi(\Omega)$ of the parameter space Ω , and applies naturally to scenarios where initially stabilizing control policies do not exist. To this end, it is also important to verify that the required property defined in Def. 6.6 does *not* imply existence of an initially stabilizing control policy.

Finite Mistake Guarantees for Asymptotic CMC

The guarantees stated in Theorem 37 and Theorem 6.7 can be interpreted as global asymptotic stability of the output, since $\mathcal{G}_t(x_t, u_t)$. To provide this guarantee, we only impose weak asymptotic conditions on π and SEL: SEL needs to either satisfy the convergence condition (A) or (B), while π has to be either robust or uniformly robust+cost-invariant for some non-zero margin, respectively.

Theorem 37. Assume that SEL is a consistent model chaser (CMC) of type (A) and that π is an ρ -robust oracle for an objective \mathcal{G} . Then, the amount of mistakes we make online is guaranteed to be finite; i.e., for any closed-loop trajectory $(\boldsymbol{x}, \boldsymbol{u})$ holds $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) < \infty$.

Proof. Denote the online data at time t as the tuple $\mathcal{D}_t := (d_1, \ldots, d_t)$. Per assumption, we know that $\mathsf{P}(\mathcal{D}_{\infty})$ is non-empty and that $\lim_{t\to\infty} \theta_t = \theta_{\infty} \in \mathsf{P}_{\infty}$. Moreover, there exists some $f' \in \mathbb{D}(\theta_{\infty})$ such that the trajectories satisfy for all time $t \ge 0$ the dynamics $x_{t+1} = f'(t, x_t, u_t)$. Since $\theta_t \to \theta_{\infty}$, there exists a time T, such that for all $t \ge T$, $d(\theta_t, \theta_{\infty}) < \rho$, i.e., for $t \ge T$, we apply policies $\pi(t; \theta_t)$ with parameters ρ -close to θ_{∞} . Per definition, this tells us that for the time interval $\mathcal{I} = [T, \infty)$, the tail of the trajectory $x_{\mathcal{I}}, u_{\mathcal{I}}$ is contained in $\mathcal{S}_{\mathcal{I}}[\rho; \theta_{\infty}]$. Since we assume that π is a ρ -robust oracle for \mathcal{G} , (Def. 6.4), we have to conclude that $\sum_{t=T}^{\infty} \mathcal{G}_t(x_t, u_t) \le M$ for some finite number M. This proves the desired claim since $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \le \sum_{t=0}^{T-1} \mathcal{G}_t(x_t, u_t) + M$.

The next theorem states that we can guarantee finite mistakes even for the weakest chasing condition (B), provided π satisfies a stronger robustness property.

Theorem. Assume that SEL is a consistent model chaser (CMC) of type (B) and that π is an uniformly ρ -robust, cost-invariant oracle for an objective \mathcal{G} . Then, the amount of mistakes we make online is guaranteed to be finite, i.e., for any closed-loop trajectory (\mathbf{x}, \mathbf{u}) holds $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) < \infty$.

204

Proof. Pick an arbitrary $0 < \varepsilon < \rho$ and some $T \ge M_{\rho}^{\pi}$. There exists N > 0 such that $\forall t \ge N$: dist($\mathsf{P}(\mathcal{D}_t), \theta_t$) $\le \varepsilon/4$ and $d(\theta_t, \theta_{t-1}) \le \varepsilon/(2T)$. Pick an arbitrary time-step s > N + T, there exists a $\theta'_s \in \mathsf{P}(\mathcal{D}_s)$ such that $d(\theta'_s, \theta_s) \le \varepsilon/4$. Consider now the time steps t in the time window $\mathcal{I}_s = [s - T, s - 1]$. Per assumption and triangle inequality, we have, for all $t \in \mathcal{I}_s$:

$$d(\theta'_s, \theta_t) \leq d(\theta'_s, \theta_s) + \sum_{j=t}^{s-1} d(\theta_{j+1}, \theta_j) \leq \frac{\varepsilon}{4} + (s-t)\frac{\varepsilon}{2T} \leq \frac{\varepsilon}{4} + T\frac{\varepsilon}{2T} \leq \varepsilon < \rho.$$

Since $\theta'_s \in \mathsf{P}(\mathcal{D}_{\infty})$, the truncation $(\boldsymbol{x}_{\mathcal{I}_s}, \boldsymbol{u}_{\mathcal{I}_s})$ is contained in the set $\mathcal{S}_{\mathcal{I}_s}[\rho; \theta'_s]$. Since we picked T to be larger than the mistake constant M^{π}_{ρ} , there has to be at least one time-step $s' \in \mathcal{I}$ prior to s at which $\mathcal{G}_s(x_s, u_s) = 0$; otherwise we would contradict the ρ -robustness property. Now, due to the cost-invariance of π , we have to conclude that for any time-steps $k \in \mathcal{I}_s$, $k \ge s'$ after s' holds $\mathcal{G}_k(x_k, u_k) = 0$; this also includes time-step s, hence $\mathcal{G}_s(x_s, u_s) = 0$ is true. Finally, since s was arbitrarily chosen in the interval $[N + T + 1, \infty)$, we know that $\sum_{t=N+T+1}^{\infty} \mathcal{G}_t(x_t, u_t) = 0$. Therefore, the total cost is $\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) = \sum_{t=0}^{N+T} \mathcal{G}_t(x_t, u_t)$ and is finite.

It is important to point out that the above asymptotic guarantees do not require ρ to be known, merely that there exist some non-zero ρ for which π is ρ -robust. Next, we discuss that the stronger notion of (finite-time)-competitive chasing allows to bound the total number of mistakes uniformly over all closed-loop trajectories.

Mistake-Bound Guarantees for γ -Competitive CMC

As discussed in Section 6.5, in cases where $P(D_t)$ have a convex representation, we can use the Steiner point $s(P_t)$ or the projection rule $\theta_t = \operatorname{Proj}_{P_t}(\theta_{t-1})$ to design a γ -competitive CMC-algorithm. As shown in the next theorem, SEL procedures of this type (C) lead to a uniform bound on the total mistakes for the overall adaptive controller $\mathcal{A}_{\pi \times \text{SEL}}$:

Theorem 38. Assume that SEL is a γ -competitive CMC-algorithm and that π is an uniformly ρ -robust oracle for an objective \mathcal{G} . Then, for any trajectory (x, u) holds

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leq M_{\rho}^{\pi} \Big(\frac{2\gamma}{\rho} \operatorname{diam}(\mathbf{\Omega}) + 1 \Big).$$

Proof. The parameter sequence $\boldsymbol{\theta}$ provided by SEL satisfies $\theta_t \in \mathsf{P}(\mathcal{D}_t)$, $\forall t$ and $\sum_{t=1}^{T} d(\theta_t, \theta_{t-1}) \leq \gamma d_{\mathcal{H}}(\boldsymbol{\Omega}, \mathsf{P}(\mathcal{D}_T))$. Set $t_0 = 0$ and construct the index-sequence

 $t_0, t_1, t_2, \ldots, t_N$ as follows:

$$t_{k} := \begin{cases} \min\left\{t \leq T \mid t > t_{k-1} \text{ and } d(\theta_{t}, \theta_{t_{k-1}}) > \frac{1}{2}\rho\right\} , \text{ if } k > 1\\ 0 , \text{ if } k = 0 \end{cases}$$
(6.21)

until for some N, the condition $t \leq T$, $t > t_N$, $d(\theta_t, \theta_{t_N}) > \frac{1}{2}\rho$ becomes infeasible and we terminate the construction. Define the intervals $\mathcal{I}_k := [t_k, \bar{t}_k]$, where $\bar{t}_k := t_{k+1}-1$ for k < N and $\bar{t}_N = T$. The intervals $\mathcal{I}_0, \ldots, \mathcal{I}_N$ are a non-overlapping cover of the time-interval [0, T]:

$$\bigcup_{0 \le k \le N} \mathcal{I}_k = [0, T], \quad \mathcal{I}_k \cap \mathcal{I}_{k-1} = \emptyset, \forall k : 1 \le k \le N.$$

Let (a_0, a_1, \ldots, a_N) and (b_0, b_1, \ldots, b_N) be the parameters selected at the start and end of each interval \mathcal{I}_k , respectively, $a_k := \theta_{t_k}$ and $b_k := \theta_{\bar{t}_k}$. Per construction, we know that

$$d(a_k, \theta_t) \leq \frac{1}{2}\rho \text{ for all } t \in \mathcal{I}_k$$
(6.22)

$$d(a_k, a_{k-1}) > \frac{1}{2}\rho \text{ for all } 1 \le k \le N.$$
(6.23)

Inequality (6.22) states that $d(a_k, b_k) \leq \frac{1}{2}\rho$ and implies via triangle inequality that for all $t \in \mathcal{I}_k$ holds

$$d(\theta_t, b_k) \leq d(\theta_t, a_k) + d(a_k, b_k) \leq \rho.$$

Since we picked $b_k = \theta_{\bar{t}_k}$ and the procedure SEL assures $\theta_{\bar{t}_k} \in \mathsf{P}(\mathcal{D}_{\bar{t}_k})$, this means that for some $f' \in \mathbb{D}[b_k]$, the partial trajectory $(x_{\mathcal{I}_k}, u_{\mathcal{I}_k})$ satisfies the following equations for the time steps $t \in \mathcal{I}_k$:

$$x_{t+1} = f'(t, x_t, u_t), \quad u_t = \pi[\theta_t](t, x_t).$$
(6.24)

We can therefore conclude that $(x_{\mathcal{I}_k}, u_{\mathcal{I}_k}) \in \mathcal{S}_{\mathcal{I}_k}[\rho; b_k]$. Hence, for the time-frame \mathcal{I}_k the trajectory is consistent with the nominal closed-loop w.r.t. to system parameter b_k and therefore the partial trajectory $(x_{\mathcal{I}_k}, u_{\mathcal{I}_k})$ has to obey the conditions implied by ρ -robustness of the oracle, i.e., we have to conclude that $\sum_{t \in \mathcal{I}_k} \mathcal{G}_t(x_t, u_t) \leq M_{\rho}^{\pi}$. Applying this reasoning to each of the N intervals \mathcal{I}_k , we observe that the total mistakes can be at most $M_{\rho}^{\pi}(N+1)$:

$$\sum_{t=0}^{T} \mathcal{G}_t(x_t, u_t) = \sum_{k=0}^{N} \sum_{j \in \mathcal{I}_k} \mathcal{G}_j(x_j, u_j) \leqslant M_{\rho}^{\pi}(N+1).$$
(6.25)

The last step is to bound the number of intervals N, for which we leverage the chasing-property of the selection SEL. We established in (6.23) that over each interval SEL must have changed the consistent parameter by at least $\frac{1}{2}\rho$. From this we notice that N has to scale with the total path-length of our selection and obtain our bound by leveraging the property of γ -competitiveness. The following chain of inequalities

$$\frac{1}{2}\rho N \leqslant \sum_{k=1}^{N} d(a_k, a_{k-1}) \leqslant \sum_{k=0}^{N-1} \sum_{t \in \mathcal{I}_k} d(\theta_t, \theta_{t+1}) \leqslant \sum_{t=1}^{T} d(\theta_t, \theta_{t-1}) \leqslant \gamma d_{\mathcal{H}}(\mathbf{\Omega}, \mathsf{P}(\mathcal{D}_T))$$

leads to the bound $N \leq 2\frac{\gamma}{\rho} d_{\mathcal{H}}(\mathbf{\Omega}, \mathsf{P}(\mathcal{D}_T))$. We substitute this into (6.25) to obtain the desired bound on the total number of mistakes:

$$\sum_{t=0}^{T} \mathcal{G}_{t}(x_{t}, u_{t}) \leq M_{\rho}^{\pi}(2\frac{\gamma}{\rho}d_{\mathcal{H}}(\mathbf{\Omega}, \mathsf{P}(\mathcal{D}_{T})) + 1)$$
$$\leq M_{\rho}^{\pi}(2\frac{\gamma}{\rho}d_{\mathcal{H}}(\mathbf{\Omega}, \mathsf{P}(\mathcal{D}_{\infty})) + 1) \leq M_{\rho}^{\pi}(2\frac{\gamma}{\rho}\operatorname{diam}(\mathbf{\Omega}) + 1). \quad (6.26)$$

We can take the limit $T \rightarrow \infty$ and arrive at the desired result.

Mistake-Bound Guarantees for Finite-Time Competitive CMC

The strong competitiveness property is not necessary if, instead, stronger conditions on the oracle can be enforced. The next result states that if the oracle π is cost-invariant we can weaken the assumptions on SEL and still provide finite mistake guarantees.

Theorem 39. Assume that SEL is (γ, T) -finite-time competitive CMC-algorithm (type D) and that π is an uniformly ρ -robust, cost-invariant oracle for an objective \mathcal{G} . Then for any closed-loop trajectory holds:

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leqslant M_{\rho}^{\pi} \left(N(\mathbf{\Omega}, r^*) + 1 \right), \quad r^* := \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}$$
(6.27)

where $N(\mathbf{\Omega}, r)$ denotes the r packing number of $\mathbf{\Omega}$ as defined in (6.12).

Proof. We discuss only some of the main steps and postpone the discussion of the full proof to Section 6.A of the appendix. The first part of the proof follows a construction similar to that in Theorem 38 to arrive at an intermediate bound

$$\sum_{t=0}^{T} \mathcal{G}_t(x_t, u_t) = \sum_{k=0}^{N} \sum_{j \in \mathcal{I}_k} \mathcal{G}_j(x_j, u_j) \leqslant M_{\rho}^{\pi}(|\mathbb{S}| + 1),$$

where \mathbb{S} is defined as a subset of intervals defined by the condition:

$$\mathbb{S} := \{ \mathcal{I}_k \mid \mathcal{G}_{t_{k+1}}(x_{t_{k+1}}, u_{t_{k+1}}) = 1 \}.$$

The second part of the proof is concerned with showing that S is a finite set and bounding the cardinality |S|. The main argument we use here is similar in spirit to the one used to establish the boundedness of N in the proof of Theorem 38: We show that due to the finite-time competitive chasing property, within each interval $\mathcal{I}_k = [\underline{t}_k, \overline{t}_k]$ our selection θ_t changes at least by a fixed nonzero amount, which leads to a fixed separation of size ε

$$\varepsilon = \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}$$

in Hausdorff metric between consistent sets $\mathsf{P}(\mathcal{D}_{\underline{t}_k})$ at the beginning of the interval \mathcal{I}_k and the consistent set $\mathsf{P}(\mathcal{D}_{\overline{t}_k})$ at the end of the interval \mathcal{I}_k . Moreover, it turns out that this observation proves the existence of a ε -separated set of size $|\mathbb{S}|$ that is contained within Ω . Due to the compactness of Ω , this proves that \mathbb{S} is a finite set and that its cardinality is bounded above by the ε packing number of the set $\Omega \setminus \mathsf{P}(\mathcal{D}_{\infty}) \subset \Omega$.

Definition 6.12 ([46]). Let (\mathcal{M}, d) be a metric space and $S \subset \mathcal{M}$ a compact set. For r > 0, define r-packing number of S, denoted by N(S, r), as:

$$N(\mathsf{S},r) := \max\left\{ n \in \mathbb{N} \mid \exists \theta_1, \dots, \theta_N \in \mathsf{S} \text{ s.t. } d(\theta_i, \theta_j) > r, \ \forall i \neq j \right\}.$$

The bound of Theorem 39 is much larger than the one of Theorem 38, however it compensates for in generality.

The stronger result Theorem 38 requires SEL to be a γ -competitive CMC-algorithm; however, it is unclear whether γ -competitiveness can be achieved for the case of non-convex consistent sets. On the other hand, the generic projection-based strategy presented in Section 6.5 describes a universal finite-time competitive CMC algorithm SEL_p, regardless of whether the consistents are convex or not. Hence, assuming that we have a suitable oracle π , it suffices to couple the oracle with the simple projection-based CMC SEL_p in order for $\mathcal{A}_{\pi \times SEL_p}$ to inherit the guarantees of Theorem 39.

Mistake Guarantees with Locally Robust Oracles

The worst-case bound shown in Theorem 36 can be directly used to extend the result of Theorem 38 to problem settings where we only have access to locally robust oracles. **Theorem 40** (Corollary of Thm. 38). Consider the setting and assumptions of Thm.38 and Thm.39, but relax the oracle robustness requirements to corresponding local versions and enforce the additional oracle assumption stated in Thm.36. Then all guarantees of Thm.38 and Thm.39 still hold, if we replace M_{ρ}^{π} in Thm.38 and Thm.39, respectively, by $M_{\rho}^{\pi}(R_{\infty})$ and $M_{\rho}^{\pi}(R_{\infty}^{w})$ and where:

$$R_{\infty} = e^{\alpha \gamma \phi(\mathbf{\Omega})} \left(\|x_0\| + \beta \frac{e}{e-1} \right)$$
$$R_{\infty}^w = \inf_{0 < \mu < 1} \left(1 + (\alpha \phi(\mathbf{\Omega}))^{n^*} \right) \max\{\frac{\beta}{1-\mu}, \|x_0\|\} + \beta \sum_{k=0}^{n^*} (\alpha \phi(\mathbf{\Omega}))^k$$

and $n^* = N(\mathbf{\Omega}, \frac{\mu}{\alpha \gamma})$ and $\phi(\mathbf{\Omega})$ denotes the diameter of $\mathbf{\Omega}$.

Writing out the inequalities derived from Theorem 38, i.e., the guarantees for combining γ -competitive model chasers SEL with locally uniformly ρ -robust and (α, β) -single step stable oracles π , we obtain the mistake bound:

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leq M_{\rho}^{\pi}(R_{\infty}) \left(\frac{2\gamma}{\rho} \operatorname{diam}(\mathbf{\Omega}) + 1\right).$$
(6.28)

On the other hand, the guarantee of Theorem 39 transforms into the bound:

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \le M_{\rho}^{\pi}(R_{\infty}^w) \left(N(\mathbf{\Omega}, r^*) + 1 \right), \quad r^* := \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi}(R_{\infty}^w) + T} \quad (6.29)$$

6.8 Discussion and Extension of Main Results

Theorem 38 and 39 can be invoked on any learning and control method that instantiates $\mathcal{A}_{\pi}(SEL)$. It offers a set of sufficient conditions to verify whether a learning agent $\mathcal{A}_{\pi}(SEL)$ can provide mistake guarantees: We need to show that w.r.t. some compact parametrization (\mathbb{T}, Ω, d) of the uncertainty set \mathcal{F}, π operates as a robust oracle for some objective \mathcal{G} , and that SEL satisfies strong enough chasing properties. Theorem 38 also suggests a design philosophy of decoupling the learning and control problem into two separate problems while retaining the appropriate guarantees: (1) design a robust oracle π for a specified control goal \mathcal{G} ; and (2) design an online selection procedure SEL that satisfies the chasing properties defined in Def. 6.9. Nominal control design methods which have guarantees only in the small uncertainty settings can be naturally extended to the large uncertainty setting. Provided that the design method can be embedded as an oracle sub-routine π_{rc} and that we can find a suitable SEL routine, the meta-algorithm $\mathcal{A}_{\pi_{rc}}(SEL)$ provides a simple extension of the original method to the large uncertainty setting.

ρ-robust	$\forall \theta \in \mathbf{\Omega} : \ \sup_{\tau, \gamma \ge 0} m_{\rho}^{\pi}(\tau, \gamma; \theta) < \infty$
uniformly ρ -robust	$M^{\pi}_{\rho}(\tau) := \sup_{\gamma \ge 0, \theta \in \mathbf{\Omega}} m^{\pi}_{\rho}(\tau, \gamma; \theta) < \infty$
locally ρ -robust	$\forall \gamma \ge 0, \ \theta \in \mathbf{\Omega} : \ \sup_{\tau \ge 0} m_{\rho}^{\pi}(\tau, \gamma; \theta) < \infty$
locally uniformly ρ -robust	$\forall \gamma \ge 0: \ M^{\pi}_{\rho}(\tau, \gamma) := \sup_{\theta \in \mathbf{\Omega}} m^{\pi}_{\rho}(\tau, \gamma; \theta) < \infty$

Table 6.2: Notions of oracle-robustness adjusted for finite-time guarantees

Using the theorems of the previous section, any nominal guarantees which can be paraphrased as OC-MG stability and mistake guarantees are carried over by $\mathcal{A}_{\pi_{rc}}(SEL)$ to the large uncertainty setting.

Finite-Time Mistake Guarantees

At the cost of adding notational overhead, we can extend the definitions of the oracle robustness properties, as shown in Def. 6.13 and Table 6.2, to be more granular and allow for finite-time evaluation.

Definition 6.13 (m_{ρ}^{π} definition for finite-time guarantees). Equip \mathcal{X} with some norm $\|\cdot\|$. For each $\rho, \gamma \ge 0$, $\tau \in \mathbb{N}$ and $\theta \in \Omega$, define the quantity $m_{\rho}^{\pi}(\tau, \gamma; \theta)$ as

$$m_{\rho}^{\pi}(\tau,\gamma;\theta) := \sup_{\mathcal{I}=[t,t+\tau]} \sup_{(x_{\mathcal{I}},u_{\mathcal{I}})\in\mathcal{S}_{\mathcal{I}}[\rho;\theta], \|x_t\| \leqslant \gamma} \sum_{t \in \mathcal{I}} \mathcal{G}_t(x_t,u_t).$$

With respect to this adjusted set of notations, the proofs of Theorem 38 and Theorem 39 remain, up to some notational substitution, logically the same. Hence, for example, we can state the inequalities (6.28) and (6.29) as the following finite-time guarantees:

$$\sum_{t=0}^{\tau} \mathcal{G}_t(x_t, u_t) \leqslant M_{\rho}^{\pi}(\tau, R_{\infty}) \left(\frac{2\gamma}{\rho} \operatorname{diam}(\mathbf{\Omega}) + 1\right).$$
(6.30)

$$\sum_{t=0}^{\tau} \mathcal{G}_t(x_t, u_t) \leqslant M_{\rho}^{\pi}(\tau, R_{\infty}^w) \left(N(\mathbf{\Omega}, r^*) + 1 \right), \quad r^* := \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi}(R_{\infty}^w) + T} \quad (6.31)$$

From Mistake Guarantees to Cost Guarantees

The presented mistake guarantees can be used to obtain worst-case guarantees for more general cost-functions. We highlight how to perform this reduction next.

Assume we are given a non-negative and possibly time-dependent cost function $C : \mathbb{N} \times \mathcal{X} \times \mathcal{U} \to \mathbb{R}^+_0$. Then we can define a family $\{\mathcal{G}[\varepsilon]\}_{\varepsilon \in \mathbb{R}^+}$ of objectives $\mathcal{G}[\varepsilon] : \mathbb{N} \times \mathcal{X} \times \mathcal{U} \to \{0, 1\}$ indexed over the open interval $\varepsilon \in (0, \infty)$, such that for a trajectory $(\boldsymbol{x}, \boldsymbol{u})^{\top}$, the sum $\sum_{t=0}^{\infty} \mathcal{G}_t[\varepsilon](x_t, u_t)$ represents the total number of time-steps t at which we incurred a cost $C(t, x_t, u_t)$ larger than ε . Suitably, we call this the total number of ε -mistakes. We make this definition precise using characteristic functions. Let $\chi_A : \mathbb{R} \mapsto \{0, 1\}$ denote the characteristic function for a subset $A \subset \mathbb{R}$ of the real line and $\overline{\chi}_A := \chi_{A^{\complement}}$ be the corresponding complementary function, i.e., $\overline{\chi}_A(s) = 0, \forall s \in A$ and $\overline{\chi}_A(s) = 1, \forall s \notin A$. Then, for each $\varepsilon \in \mathbb{R}, \varepsilon > 0$, the ε -objective $\mathcal{G}[\varepsilon]$ is defined for each $t \in \mathbb{N}, x \in \mathcal{X}, u \in \mathcal{U}$ as $\mathcal{G}[\varepsilon](t, x, u) := \overline{\chi}_{[0,\varepsilon]}(C(t, x, u))$.

Using the following Lemma, which is proven in the Appendix, we can reformulate the running cost in terms of the mistake bounds.

Lemma 56. For any scalar sequence $a \in \ell_1$ holds:

$$\|\boldsymbol{a}\|_{1} = \int_{0}^{\infty} \sum_{k=0}^{\infty} \overline{\chi}_{[-s,s]}(a_{k}) ds = \int_{0}^{\|\boldsymbol{a}\|_{\infty}} \sum_{k=0}^{\infty} \overline{\chi}_{[-s,s]}(a_{k}) ds$$
(6.32)

Proof. See Appendix 6.A

As direct application of the above, we have the following relation between running cost and number of ε -mistakes:

$$\sum_{k=0}^{t} C(k, x_k, u_k) = \int_0^\infty \sum_{k=0}^{t} \overline{\chi}_{[0,\varepsilon]}(C(k, x_k, u_k))d\varepsilon = \int_0^\infty \sum_{k=0}^{t} \mathcal{G}[\varepsilon](k, x_k, u_k)d\varepsilon$$
$$= \int_0^{\max_{k \leq t} C(k, x_k, u_k)} \sum_{k=0}^{t} \mathcal{G}[\varepsilon](k, x_k, u_k)d\varepsilon$$

We can associate the family of objectives $\{\mathcal{G}[\varepsilon]\}_{\varepsilon \in \mathbb{R}^+}$ with a family of corresponding comparison functions $\{m_{\rho}^{\pi}[\varepsilon]\}_{\varepsilon \in \mathbb{R}^+}$ and $\{M_{\rho}^{\pi}[\varepsilon]\}_{\varepsilon \in \mathbb{R}^+}$ – each $m_{\rho}^{\pi}[\varepsilon], M_{\rho}^{\pi}[\varepsilon]$ quantifying the nominal mistake bounds of a fixed oracle π . Then, for suitable π and model chasers SEL, we can integrate the finite-time mistake bounds over ε and use the above equivalence, in general, to derive finite-time worst-case cost performance guarantees for $\mathcal{A}_{\pi \times \text{SEL}}$.

As an example, as a corollary of Theorem 38, and its local extension Theorem 40, we obtain the following result:

Theorem 41. Let $C : \mathbb{N} \times \mathcal{X} \times \mathcal{U} \to \mathbb{R}_0^+$ be a non-negative, possibly time-dependent cost function and let $\{\mathcal{G}[\varepsilon]\}_{\varepsilon \in \mathbb{R}^+}$ be the corresponding family of objectives as defined above. Let $\pi : \mathbf{\Omega} \to \mathcal{K}$ be an oracle with the (α, β) -stability property.

Assume that SEL is a γ -competitive CMC-algorithm and that for each objective $\mathcal{G}[\varepsilon]$, the oracle π is locally uniformly ρ -robust, with corresponding comparison functions $m_{\rho}^{\pi}[\varepsilon], M_{\rho}^{\pi}[\varepsilon]$. Then, for any closed-loop trajectory $(\boldsymbol{x}, \boldsymbol{u})$ and $\tau \in \mathbb{N}$ holds

$$\sum_{t=0}^{\tau} C(t, x_t, u_t) \leqslant \left(\frac{2\gamma}{\rho} \operatorname{diam}(\mathbf{\Omega}) + 1\right) \int_0^{\infty} M_{\rho}^{\pi}[\varepsilon](\tau, R_{\infty}) d\varepsilon,$$

where $R_{\infty} := e^{\alpha \gamma \phi(\mathbf{\Omega})} (\|x_0\| + \beta \frac{e}{e-1}).$

Safety Guarantees Using Families of Lyapunov Functions

Def. 6.6 is closely related to ISS-stability [75] and is favored in our derivations as a simple substitute for generic ℓ_{∞} -stability conditions. However, Def. 6.6 requires that all nominal closed-loops $CL_{\pi}[\omega]$ share V(x) = ||x|| as a common ISS-Lyapunov function, which can seem a rather restrictive condition. Nevertheless, we can replace Def. 6.6 with a more flexible definition without significant repercussions on the theoretical results.

Assume that for each $\omega \in \Omega$, there exists a family $\{V(\cdot | \omega)\}_{\omega \in \Omega}$ of non-negative functions $V(\cdot; \omega) : \mathbb{N} \times \mathcal{X} \to \mathbb{R}^+$ and a scalar positive-definite, increasing bijective function $h : \mathbb{R}_0^+ \to \mathbb{R}_0^+$ such that the following conditions are met:

1. For all $\omega \in \mathbf{\Omega}, t, x \in \mathcal{X}$:

$$V(t, x \mid \omega) \leqslant h(\|x\|)$$

2. For some $\mu > 0$, $\alpha > 0$ and all $\omega_1, \omega_2 \in \mathbf{\Omega}, f \in \mathbb{D}[\omega_2], t \in \mathbb{N}, x \in \mathcal{X}$:

$$V_{t+1}(f(t, x_t, \pi[\omega_1]) \mid \omega_2) \leq (e^{-\mu} + \alpha d(\omega_1, \omega_2))V_t(x_t \mid \omega_1) + 1$$

If we assume an oracle π to satisfy the above set of conditions for some μ , α , $\{V(a; \omega) \mid \omega \in \Omega\}$, h, and couple it with a model chaser SEL, we can obtain theoretical safety guarantees analogous to Theorem 36 and finite mistake guarantees analogous to Theorem 40. A thorough discussion of these extensions is being prepared for publication.

Extension to Unbounded Uncertainties

Surprisingly, knowledge of a compact parametrization of \mathcal{F} is not a fundamental necessity for the PixSel framework. In fact, we can extend the approach to a large class of problem settings with unbounded uncertainty sets, and still retain the innate ability to provide worst-case guarantees.

Algorithm 4 Meta-Model Chaser SEL $_{\infty}$

Require: Family of consistent model chasers $\{SEL_i\}_{i\in\mathbb{N}}$ corresponding to a filtration $\{\Omega_i\}$ **Initialization:** $\mathcal{D}_0 \leftarrow \{\}, i = 0, x_0 \text{ is set to initial condition } \xi_0$ 1: for t = 0, 1, ... to ∞ do $\mathcal{D}_t \leftarrow \text{append} (t, x_t, x_{t-1}, u_{t-1}) \text{ to } \mathcal{D}_{t-1} (\text{if } t \ge 1)$ ⊳ update history 2: 3: $\theta_t \leftarrow \text{SEL}_i[\mathcal{D}_t]$ 4: while $\theta_t = \emptyset$ do $i \leftarrow i + 1$ ightarrow switch to next CMC 5: $\theta_t \leftarrow \text{SEL}_i[\mathcal{D}_t]$ ⊳ reattempt selection 6: 7: end while 8: Output consistent parameter θ_t 9: end for

Consider the more general setup, where we have a (non-compact) metric space (Ω_{∞}, d) representing the space of parameters, and a set-valued map $\mathbb{D} : \Omega_{\infty} \to 2^{\mathcal{F}}$ representing the set of dynamics $\mathbb{D}[\omega] \subset \mathcal{F}$ associated with each parameter $\omega \in \Omega$, such that $\{\mathbb{D}[\omega]\}_{\omega \in \Omega_{\infty}}$ covers the uncertainty \mathcal{F} .

Now, suppose we can construct a filtration $\{\Omega_i\}_{i\in\mathbb{N}}$ of compact subsets $\Omega_i \subset \Omega_{\infty}$ which covers Ω_{∞} , i.e., $i < j \implies \Omega_i \subset \Omega_j$ and $\bigcup_{i\in\mathbb{N}} \Omega_i = \Omega_{\infty}$, and assume that for each Ω_i we have a consistent model chaser SEL_i, which has the ability to notify us, for example by returning the empty set \emptyset , the moment the set of consistent parameters $\mathsf{P}_i(\mathcal{D}_t) \subset \Omega_i$ becomes empty at time t.

Remark. As a simple example, consider $\mathbb{P} = \mathbb{R}$, d(x, y) = |x - y| and define Ω_i as the interval $[-i, i] \subset \mathbb{R}$. Then $\{\Omega_i\}_{i \in \mathbb{N}}$ is a filtration of compact subsets which covers \mathbb{R} .

Given such a family of model chasers {SEL_i}, we can construct a Meta-chasingalgorithm SEL_{∞}, which initializes as SEL₀, and thereafter switches to the implementation of the next model chaser SEL_{i+1}, once the consistent set of the previous model chaser P_i(\mathcal{D}_t) becomes empty. This process is diagrammed in Algorithm 4.

Despite its simplicity, the Meta-chasing algorithm SEL_{∞} inherits any chasingproperties which hold uniformly over the family of model chasers $\{SEL_i\}_{i\in\mathbb{N}}$. This inheritance relationship rests on a simple observation. If f^* is contained in \mathcal{F} , then there has to exist some $i^* \in \mathbb{N}$ for which $f^* \in \mathbb{D}[\theta^*]$ for some $\theta^* \in \Omega_{i^*}$; it is immediately clear that asymptotic chasing properties, i.e., statements (A) and (B) of Def. 6.9, hold true for the model-chaser SEL_{∞} if the same properties are true for the entire CMC-family $\{SEL_i\}_{i\in\mathbb{N}}$. In regards to the competitive chasing properties (C) and (D), we can easily see that if each model chaser in the family $\{SEL_i\}_{i\in\mathbb{N}}$ is γ -competitive / (γ, T) -f.t.-competitive, then SEL_{∞} is at the very least $i^*\gamma$ -competitive / $(i^*\gamma, T)$ -f.t.-competitive. Unfortunately, since i^* is not known in general, we can not expect to have knowledge of the former constants in advance. Nevertheless in the case of convex consistent sets, there are cases in which the competitive ratio is preserved. A thorough theoretical discussion of this case is being prepared for future publication.

Oracle Policies with Memory and System Level Controllers

The previous results assume that π returns static policies of the type $(t, x) \mapsto u$. However, this assumption is only made for ease of exposition. All previous results also hold in the case where π returns policies that have an internal state, as long as we can define the internal state to be shared among all oracle policies; namely, as part of the oracle implementation online, we update the state z_t at each step taccording to some fixed update rule h

$$z_t = h(t, z_{t-1}, x_t, u_t, \dots, x_0, u_0),$$

and control policies $\pi[\theta], \theta \in \Omega$ are maps $(t, x, z) \mapsto u$ which we evaluate at time t as $u_t = \pi[\theta](t, x_t, z_t)$.

The results presented in Chapter 4 provide a natural and seamless way to encapsulate families of system level controllers as oracles and instantiate corresponding PixSel Algorithms with worst-case guarantees. A powerful application, which fuses the results of Chapter 4 and 6, is presented in our recent work [9]. In said publication, we provide the first fully distributed and scalable control algorithm capable of learning to stabilize unknown large-scale linear systems in the adversarial setting.

To summarize our findings and recap some of the main results, we review our design framework for a simple, yet non-trivial example.

6.9 Design Example: Control of Uncertain Scalar Linear System

Let us consider a very basic problem setting, wherein we are given an unknown scalar linear system

$$x_{k+1} = \alpha^* x_k + \beta^* u_k + w_k =: f^*(k, x_k, u_k),$$

s.t. $|w_k| \leq \gamma^* \leq \eta < 1$ and $\alpha^* \in [-a, a], \beta^* \in [1, 1 + 2b_\Delta]$, and our goal is to reach the target interval $\mathcal{X}_{\mathcal{T}} = [-1, 1]$ and remain there. We can equivalently phrase this as to achieve the objective $\boldsymbol{\mathcal{G}} = (\mathcal{G}_0, \mathcal{G}_1, \dots)$ with cost functions

$$\mathcal{G}_t(x,u) := \begin{cases} 0, & \text{if } |x| \leq 1 \\ 1, & \text{else} \end{cases}, \quad \forall t \ge 0$$

after finitely many mistakes.

Compact Parametrization of Uncertainty Set. We define our parameter space as $\Omega = [-a, a] \times [1, 1+2b_{\Delta}]$, and define the true parameter as $\theta^* = (\theta_x^*, \theta_u^*) = (\alpha^*, \beta^*)$ and parametrize the uncertainty set as $\mathcal{F} = \bigcup_{\theta \in \Omega} \mathbb{D}[\theta]$ with

$$\mathbb{D}[\theta] := \{t, x, u \mapsto \theta_x x + \theta_u u + w_t \mid \|\boldsymbol{w}\|_{\infty} \leq \eta \}.$$

We choose the metric as $d(\theta, \theta') := |\theta_x - \theta'_x| + a|\theta_u - \theta'_u|$. The diameter of the metric space $(\mathbf{\Omega}, d)$ is $\phi(\mathbf{\Omega}) = d((-a, 1), (a, 1 + 2b_{\Delta})) = 2(a + b_{\Delta})$.

A Locally Uniformly Robust Oracle. As an oracle, we take the simple deadbeat controller: $\pi[\theta](t,x) := -(\theta_x/\theta_u)x$. It can be easily shown that π is a locally ρ -uniformly robust oracle for \mathcal{G} for any margin in the interval $(0, \bar{\rho}), \bar{\rho} := 1 - \eta$, by noticing the inequality:

$$|x_{t+1}| \leq |\theta_x^* x_t + \theta_u^* \pi[\theta_t](t, x_t)| + \eta = |((\theta_x^* - \theta_{x,t}) - (\theta_u^* - \theta_{u,t})\frac{\theta_{x,t}}{\theta_{u,t}})x_t| + \eta$$
(6.33)

$$\leq \left(|\theta_x^* - \theta_{x,t}| + |\theta_u^* - \theta_{u,t}||\frac{\theta_{x,t}}{\theta_{u,t}}|\right)|x_t| + \eta \leq d(\theta^*, \theta_t)|x_t| + \eta.$$
(6.34)

To obtain the mistake function M_{ρ}^{π} for a fixed $\rho \in (0, 1-\eta)$, notice that if $d(\theta^*, \theta_t) \leq \rho$, then

$$\begin{aligned} |x_{t+1}| &\leq \rho |x_t| + \eta = \rho |x_t| + (1-\rho) \frac{1}{1-\rho} \eta \quad \Leftrightarrow \quad |x_{t+1}| - \frac{\eta}{1-\rho} \leq \rho (|x_t| - \frac{\eta}{1-\rho}) \\ &\Rightarrow \quad |x_t| \leq \frac{\eta}{1-\rho} + \rho^t (|x_0| - \frac{\eta}{1-\rho}). \end{aligned}$$

Notice that

$$\frac{\eta}{1-\rho} + \rho^t |x_0| < 1 \Leftrightarrow t > \frac{\log(|x_0|)}{\log(\rho^{-1})} + \underbrace{\frac{\log(1-\rho) - \log(1-\rho-\eta)}{\log(\rho^{-1})}}_{c(\rho)}$$

which implies the mistake function

$$M^{\pi}_{\rho}(\gamma) \leqslant \frac{\log(\gamma)}{\log(\rho^{-1})} + c(\rho).$$
(6.35)

Construction of Consistent Sets via LPs. The consistent set for the data set \mathcal{D}_N of N observed system transitions (x_i^+, x_i, u_i) can be written as an intersection of Ω with 2N halfspaces:

$$\mathsf{P}(\mathcal{D}_N) = \left\{ \theta \in \mathbf{\Omega} \mid \text{s.t.: } \forall 1 \leq i \leq N : x_i^+ - \eta \leq \theta_x x_i + \theta_u u_i \leq x_i^+ + \eta \right\}.$$

It can be constructed online and is convex.

Competitive Consistent Model Chasing via Steiner Point. We can construct a competitive CMC-algorithm by using algorithms for competitive NCBC. Assume we use the Steiner point and denote the selection procedure SEL_s as in (6.12). SEL_s is a $\frac{n}{2} = 1$ -competitive CMC algorithm in euclidean space, and since the euclidean norm is bounded above by the 1-norm, SEL_s is also 1-competitive w.r.t. the metric space (Ω , d).

Mistake Guarantee for $\mathcal{A}_{\pi}(\text{SEL}_s)$. We apply the extension of the results in Theorem 40. It is easy to see that our π satisfies the extra condition with $\alpha = 1$, $\beta = \eta$. Assuming $|x_0| = 0$, the constant γ_{∞} takes the value

$$\gamma_{\infty} = e^{\alpha \phi(\mathbf{\Omega})} (\|x_0\| + \beta \frac{e}{e-1}) = \frac{\eta e}{e-1} e^{\phi(\mathbf{\Omega})}.$$
(6.36)

For ease of exposition, assume that $\eta = e^{-1}$ and that we picked $\rho = e^{-1}$. This gives us $M_{\rho}^{\pi}(\gamma_{\infty}) = \phi(\mathbf{\Omega}) - \log(e - 2)$ and substituting all constants gives us a finite mistake guarantee for the objective \mathcal{G} :

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leq M_{\rho}^{\pi}(\gamma_{\infty}) \left(\frac{2L}{\rho} + 1\right) \approx \phi(\mathbf{\Omega})(1 + 2e\phi(\mathbf{\Omega}))$$
$$= 8e(a + b_{\Delta})^2 + 2(a + b_{\Delta}).$$

The above inequality shows that the worst-case total number of mistakes grows quadratically with the size of the initial uncertainty in the system parameters θ_x and θ_u . Notice, however, that the above inquality holds for arbitrary large choices of a and b_{Δ} . Thus, $\mathcal{A}_{\pi}(\text{SEL}_s)$ gives finite mistake guarantees for this problem setting for arbitrarily large system parameter uncertainties.

This small-scale example serves as a warm-up for the next section where we discuss applications for learning and control of uncertain robotic systems.

6.10 Application to Uncertain Robotic Systems

We walk through an example of how to design adaptive controllers $\mathcal{A}_{\pi}(SEL)$ for a class of robotic systems with the goal of learning to follow a trajectory. We discuss how to embed well-known control methods in robotics such as robust oracles π and couple it with SEL selections based on competitive algorithms (NCBC). Consider a general case of online control of uncertain fully-actuated robotic systems. Most robotic systems can be modeled via the robotic equation of motion [98]:

$$\mathbf{M}_{\eta}(q)\ddot{q} + \mathbf{C}_{\eta}(q,\dot{q})\dot{q} + \mathbf{N}_{\eta}(q,\dot{q}) = \tau + \tau_d \tag{6.37}$$

where $q \in \mathbb{R}^n$ is the multi-dimensional generalized coordinates of the system, \dot{q} and \ddot{q} are its first and second (continuous) time derivatives, $\mathbf{M}_{\eta}(q)$, $\mathbf{C}_{\eta}(q, \dot{q})$, $\mathbf{N}_{\eta}(q, \dot{q})$ are matrix and vector-value functions that depend on the parameters $\eta \in \mathbb{R}^m$ of the robotic system, i.e., η comes from a parametric physical model. Often, τ is the control action (e.g., torques and forces of actuators), which acts as input of the system. Disturbances and other uncertainties present in the system can be modeled as additional torques $\tau_d \in \mathbb{R}^n$ that perturb the equations. Moreover, one can derive from first principles [98], that for many robotic systems (for example robot manipulators) the following two properties hold:

$$\mathbf{M}_{\eta}(q) - 2\mathbf{C}_{\eta}(q, \dot{q})$$
 is skew-symmetric (6.38a)

$$\mathbf{M}_{\eta}(q)\ddot{q} + \mathbf{C}_{\eta}(q,\dot{q})\dot{q} + \mathbf{N}_{\eta}(q,\dot{q}) = \mathbf{Y}(q,\dot{q},\ddot{q})\eta = \tau + \tau_d.$$
(6.38b)

The second equation says that the left-hand-side of equation (6.37) can always be factored into a $n \times m$ matrix of known functions $\mathbf{Y}(q, \dot{q}, \ddot{q})$ and a constant vector $\eta \in \mathbb{R}^m$. Assume that the disturbances are bounded at each time t, as $|\tau_d(t)| \leq \omega$, $\omega \in \mathbb{R}^n$ and where the inequality should be read entry-wise. Consider that we are given a system with unknown η^* , ω^* , where the parameter $\theta^* = [\eta^*; \omega^*]$ is known to be contained in a bounded set Ω . Assume that our goal is to follow a desired trajectory q_d , which is given as a function of time $q_d : \mathbb{R} \mapsto \mathbb{R}^n$, within the precision ϵ . Denoting $x = [q^\top, \dot{q}^\top]^\top$ as the state vector and $x_d = [q_d^\top, \dot{q}_d^\top]^\top$ as the desired state, we want the state trajectory of the system x(t) to satisfy:

$$\limsup_{t \to \infty} \|x(t) - x_d(t)\| \le \epsilon.$$
(6.39)

As is common in practice, we assume we can observe the sampled measurements $x_k := x(t_k)$, $x_k^d := x^d(t_k)$ and apply a constant control action (zero-order-hold actuation) $\tau_k := \tau(t_k)$ at the discrete time-steps $t_k = kT_s$ with small enough sampling-time T_s to allow for continuous-time control design and analysis.

Control Objective \mathcal{G}^{ϵ} . We phrase trajectory tracking as a control objective \mathcal{G}^{ϵ} with the cost functions

$$\mathcal{G}_{k}^{\epsilon}(x,u) := \begin{cases} 0, & \text{if } \|x - x_{k}^{d}\| \leq \epsilon \\ 1, & \text{else} \end{cases}, \quad \forall k \ge 0,$$

which we wish to achieve online with finite mistake guarantees against the uncertainty set $\mathcal{F} = \bigcup_{\theta \in \Omega} \mathbb{D}[\theta]$, where:

$$\mathbb{D}[\theta] := \{k, x_k, \tau_k \mapsto f^*(x_k, \tau_k, \tau_d(\cdot); \theta) \mid \tau_d : [0, T_s] \mapsto \mathbb{R}^n, \|\tau_d\|_{\infty} \leq \omega\}.$$

The function f^* denotes the discretized dynamics of (6.37) w.r.t. the sampling time T_s .

Robust Oracle Design. We outline how to design a robust oracle based on a well-established robust control method for robotic manipulators proposed in [118]. Define v, a and r as the quantities

$$v = \dot{q}_d - \Lambda \dot{q}, \quad a = \dot{v}, \quad r = \tilde{q} + \Lambda \tilde{q}, \quad \tilde{q} = q - q_d$$
 (6.40)

and denote $\mathbf{Y}'(q, \dot{q}, v, a)$ as the corresponding $n \times m$ matrix which allows the factorization:

$$\mathbf{M}_{\eta}(q)a + \mathbf{C}_{\eta}(q,\dot{q})v + \mathbf{N}_{\eta}(q,\dot{q}) = \mathbf{Y}'(q,\dot{q},v,a)\eta.$$
(6.41)

Based on the control law presented in [118], we define the oracle $\pi[\theta](k, x_k)$ for $x = [q; \dot{q}]$ and $\theta = [\eta; \omega]$ through the equations:

$$\pi[\theta](k, x_k) = \mathbf{Y}'(q_k, \dot{q}_k, v_k, a_k)(\eta + u_k) - K_\omega r_k, \tag{6.42}$$

$$u = \begin{cases} -\rho \frac{\mathbf{Y}'^{\top} r_k}{\|\mathbf{Y}'^{\top} r_k\|_2} & \text{if } \|\mathbf{Y}'^{\top} r_k\|_2 > \varepsilon \\ -\frac{\rho}{\varepsilon} \mathbf{Y}'^{\top} r_k & \text{if } \|\mathbf{Y}'^{\top} r_k\|_2 \leqslant \varepsilon \end{cases}$$
(6.43)

where $\Lambda, K_{\omega} > 0$ are diagonal positive definite design and where ρ, ε are design variables. Following the analysis in [118] and [40] one can design a suitable gain K_{ω} in terms of ω , such that π is a uniformly ρ robust oracle for \mathcal{G}^{ϵ} in the compact parametrization $(\mathbb{D}, \mathbf{\Omega}, d)$.

Remark 42. The analysis in [118] shows that uniform ultimate boundedness properties of the tracking error $\tilde{x} = [q - q_d; \dot{q} - \dot{q}_d]$ are preserved, if we replace η in equation (6.42) with some perturbation $\eta + \delta(t)$, $\|\delta(t)\|_2 \leq \rho$ for all t. In [118], the disturbance τ_d is assumed to be zero, i.e., the $\omega = 0$ case, and the gain K_0 is left as a tuning variable. However, with standard Lyapunov arguments, the analysis of [118] can be extended to consider the nonzero disturbance case and specify gains K_{ω} for each ω such that the above oracle π becomes a uniformly ρ robust oracle for the above objective \mathcal{G}^{ϵ} : For each ω , increase the gain K_{ω} until the uniform ultimate boundedness guarantee implies the desired ϵ -tracking behavior described by \mathcal{G}^{ϵ} .

Constructing Consistent Sets. The linear factorization property (6.38b) can be exploited to construct convex consistent sets. Denote \mathbb{D} as an uncertain robotic system (6.38) with some convex compact uncertainty Ω in euclidean space $(\mathbb{R}^{m+n}, \|\cdot\|_2)$. Recall that we parameterize the bound on the disturbance by $\omega \in \mathbb{R}^n$, i.e., $|\tau_d| \leq \omega$ holds entry-wise and that our system parameter is represented by $\theta = [\eta^{\top}, w^{\top}]^{\top} \in \Omega$. At the sampled time-steps t_k , equations (6.38b) say that measurements q_k, \dot{q}_k, τ_k enforce the following entry-wise condition on consistent parameters η and ω :

$$\tau_k - \omega \leqslant \mathbf{Y}(q_k, \dot{q}_k, \ddot{q}_k) \eta \leqslant \tau_k + \omega. \tag{6.44}$$

In matrix form, the consistent set is captured via the following relationship:

$$\underbrace{\begin{bmatrix} \mathbf{Y}(q_k, \dot{q}_k, \ddot{q}_k) & -\mathbf{I}_n \\ -\mathbf{Y}(q_k, \dot{q}_k, \ddot{q}_k) & -\mathbf{I}_n \end{bmatrix}}_{\mathbf{A}_k} \begin{bmatrix} \eta \\ \omega \end{bmatrix} \leqslant \underbrace{\begin{bmatrix} \tau_k \\ -\tau_k \end{bmatrix}}_{\mathbf{b}_k}.$$
 (6.45)

Consequently, we have a concrete construction of consistent set at each time t:

$$\mathsf{P}(\mathcal{D}_t) = \left\{ \theta = \begin{bmatrix} \eta \\ w \end{bmatrix} \in \mathbb{R}^{m+n} \middle| \mathbf{A}_t \begin{bmatrix} \eta \\ w \end{bmatrix} \leq \mathbf{b}_t \right\} \cap \mathsf{P}(\mathcal{D}_{t-1}), \quad \mathsf{P}(\mathcal{D}_0) = \mathbf{\Omega} \quad (6.46)$$

where $\mathbf{A}_k = \mathbf{A}_k(x_k, u_k)$ and $\mathbf{b}_k = \mathbf{b}_k(x_k, u_k)$ are matrix and vector of "features" constructed from current control policy and state at time t via the known functional form of **Y**. Data sets \mathcal{D}_k are tuples of the form $\mathcal{D}_k = (d_1, \ldots, d_k), d_k = (q_k, \dot{q}_k, \ddot{q}_k, \tau_k)$.

Designing a Competitive Chasing Selection. The above consistent sets are simply an intersection of halfspaces, hence we are in the setting of Assumption 6.6 and we can instantiate competitive selections from the (NCBC) competitive greedy and Steiner point algorithm algorithms:

• Greedy Projection. SEL_p selects $\theta_t = \text{SEL}_p(\mathcal{D}_t)$ as the solution to the following convex optimization problem, which can be solved efficiently:

$$\theta_{t} = \underset{\theta \in \mathbb{R}^{p} \cap \Omega}{\operatorname{arg\,min}} \quad \frac{1}{2} \|\theta - \theta_{t-1}\|^{2},$$

s.t: $\mathbf{A}_{i} \theta \leq \mathbf{b}_{i}, \forall i = 1, \dots, t.$

Algorithm 5 design of $\mathcal{A}_{\pi}(\text{SEL}_{p/s})$ for ϵ -trajectory tracking for fully actuated robots

- 1: for $t = 0, T_s, \ldots, kT_s$ to ∞ do
- 2: measure $q_k, \dot{q}_k, \ddot{q}_k$
- 3: update polyhedron $P(\mathcal{D}_k)$ as in (6.46)
- 4:
- select according to (6.12a) or (6.12b) \rhd selection SEL_p or SEL_s choose $\tau_k = \mathbf{Y}'(q_k, \dot{q}_k, v_k, a_k)(\eta_k + u_k) K_{\omega_k}r_k$ using (6.40), (6.42) \rhd use 5: oracle $\pi(x_k; \theta_k)$
- 6: end for
 - Steiner-Point. Alternatively, SEL_s outputs the Steiner point of the polyhedron $P_{\mathbb{D}}(\mathcal{D}_t)$, which in principle requires calculating an integral over multidimensional sphere. Fortunately, as shown in [13], the Steiner point can be efficiently approximated by solving randomized linear programs; an approach we take in our empirical validation.)

Mistake Guarantee for $\mathcal{A}_{\pi}(\text{SEL}_{p/s})$ Since π is a robust oracle for \mathcal{G} and both SEL_p and SEL_s are γ -competitive CMC algorithms in $(\mathbb{D}, \Omega, \|\cdot\|)$ for some $\gamma > 0$, our result Theorem 38 tells us that $A_{\pi}(SEL_p)$ and $A_{\pi}(SEL_s)$ guarantees upfront finiteness of the total number of mistakes $\sum_{k=0}^{\infty} \mathcal{G}_k^{\epsilon}(x_k, \tau_k)$, which implies the desired tracking behavior guarantee $\limsup_{k\to\infty} ||x-x^d|| \leq \epsilon$. Moreover, if we can provide a bound M on the mistake constant $M_{\rho}^{\pi} < M$, we obtain from Theorem 38 an explicit performance bound for the tracking performance in the form of the mistake guarantee

$$\sum_{k=0}^{\infty} \mathcal{G}_k^{\epsilon}(x_k, \tau_k) \leq M(\frac{2\gamma}{\rho} \operatorname{diam}(\mathbf{\Omega}) + 1).$$

Empirical Validation: Cart-Pole Swing-Up on a Constrained Track 6.11

We illustrate the practical potential for of our approach on a challenging cart-pole swing-up goal from limited amount of interaction. Compared to the standard cartpole domain commonly used in RL [32], we introduce modifications motivated by real-world concerns in several important ways:

- 1. Goal specification: the goal is to swing up and balance the cart-pole from a down position, which is significantly harder than balancing from the up-right position (the standard RL benchmark).
- 2. *Realistic dynamics*: we use a high-fidelity continuous-time nonlinear model, with noisy measurements of discrete-time state observations.
- 3. Safety: cart position has to be kept in a bounded interval for all time. Furthermore, the acceleration should not exceed a specified maximum limit.

$\pi[\theta^*]$	0	0.4	0.99	1	1
$\mathcal{A}_{\pi}(\mathtt{SEL})$	0	0.2	0.8	0.95	1
Т	3s	6s	12s	30s	50s

Table 6.3: Fraction of experiments completing the swing up before time T: ideal policy $\pi[\theta^*]$ vs. $\mathcal{A}_{\pi}(SEL)$

- 4. Robustness to adversarially chosen system parameters: We evaluate 900 uncertainty settings, each with a different θ^* reflecting mass, length, and friction. The tuning parameter remains the same for all experiments. This robustness requirement amounts to a generalization goal in contemporary RL.
- 5. *Other constraints*: no system reset is allowed during learning (i.e., a truly continuous goal).

Our introduced modification makes this goal significantly more challenging from both on-line learning and adaptive control perspective. Table 6.3 summarizes the results for 900 different parameter conditions (corresponding to 900 adversarial settings). It compares the online algorithm to the corresponding ideal oracle policy $\pi[\theta^*]$ shows that the online controller is only marginally slower. See Appendix 6.11 and [5] for detailed description of our setup and results.

We employ well-established techniques to synthesize model-based oracles. Expert controllers are a hybrid combination of a linear state-feedback LQR around the upright position, a so-called energy-based swing-up controller (see [17]) and a control barrier function to respect the safety constraints [7]. As also described in [48], adding constraints on state and acceleration makes learning the swing-up of the cart-pole a significantly harder goal for state-of-the-art learning and control algorithms.

Table 6.3 compares the on-line algorithm with the corresponding ideal oracle policy $\pi[\theta^*]$ showing that the on-line controller is only marginally slower.

6.A Proofs

Theorem. Assume procedure SEL is (γ, T) -finite-time competitive for some $\gamma > 0$, $T \ge 1$ and that procedure π is a uniform ρ -robust, cost-invariant oracle for \mathcal{G} . Then, the total number of mistakes is guaranteed to be bounded above by:

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leq \left(N(\mathbf{\Omega}^\circ, r^*) + 1\right) M_{\rho}^{\pi} \leq \left(N(\mathbf{\Omega}, r^*) + 1\right) M_{\rho}^{\pi},$$

where $\mathbf{\Omega}^{\circ} := \mathbf{\Omega} \setminus \operatorname{int}(\mathsf{P}(\mathcal{D}_{\infty}))$ and $r^* := \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}$.

Proof. Denote $\boldsymbol{x}, \boldsymbol{u}$ to be some fixed online trajectories and denote $\boldsymbol{\theta}$ as the corresponding parameter sequence selected by procedure SEL. The sequence $\boldsymbol{\theta}$ satisfies $\theta_t \in \mathsf{P}(\mathcal{D}_t)$ and $\sum_{t=t_1+1}^{t_2} d(\theta_t, \theta_{t-1}) \leq \gamma d(\mathsf{P}(\mathcal{D}_{t_2}), \mathsf{P}(\mathcal{D}_{t_1}))$ for all $t_2 - t_1 \leq T$. For some time-step $\tau > 0$, we derive bounds on the mistakes $\sum_{t=0}^{\tau} \mathcal{G}_t(x_t, u_t)$. Set $t_0 = 0$ and construct the index-sequence $t_0, t_1, t_2, \ldots, t_N$ as follows:

$$t_{k} := \begin{cases} \min\left\{t \leqslant \tau \mid t > t_{k-1} \text{ and } d(\theta_{t}, \theta_{t_{k-1}}) > \frac{1}{2}\rho\right\} , \text{ if } k > 1\\ 0 , \text{ if } k = 0 \end{cases}$$
(6.47)

until for some N, the condition $t \leq \tau, t > t_N, d(\theta_t, \theta_{t_N}) > \frac{1}{2}\rho$ becomes infeasible and we terminate the construction. Define the intervals $\mathcal{I}_k := [t_k, \bar{t}_k]$, where $\bar{t}_k := t_{k+1}-1$ for k < N and $\bar{t}_N = \tau$. The intervals $\mathcal{I}_0, \ldots, \mathcal{I}_N$ are a non-overlapping cover of the time-interval $[0, \tau]$:

$$\bigcup_{0 \le k \le N} \mathcal{I}_k = [0, \tau], \quad \mathcal{I}_k \cap \mathcal{I}_{k-1} = \emptyset, \forall k : 1 \le k \le N.$$

Let (a_0, \ldots, a_N) and (b_0, \ldots, b_N) be the parameters selected at the start and end of each interval \mathcal{I}_k , respectively: $a_k := \theta_{t_k}$ and $b_k := \theta_{\overline{t}_k}$. Per construction, we know that

$$d(a_k, \theta_t) \leqslant \frac{1}{2}\rho \text{ for all } t \in \mathcal{I}_k$$
(6.48)

$$d(a_k, a_{k-1}) > \frac{1}{2}\rho \text{ for all } 1 \le k \le N.$$
(6.49)

Inequality (6.22) states that $d(a_k, b_k) \leq \frac{1}{2}\rho$ and implies via triangle inequality that for all $t \in \mathcal{I}_k$ holds

$$d(\theta_t, b_k) \leq d(\theta_t, a_k) + d(a_k, b_k) \leq \rho.$$

Since we picked $b_k = \theta_{\bar{t}_k}$ and the procedure SEL assures $\theta_{\bar{t}_k} \in \mathsf{P}(\mathcal{D}_{\bar{t}_k})$, it means that for some $f_k \in \mathbb{D}[b_k]$, the partial trajectory $(x_{\mathcal{I}_k}, u_{\mathcal{I}_k})$ satisfies the following equations for the time-steps $t \in \mathcal{I}_k$:

$$x_{t+1} = f'(t, x_t, u_t), \quad u_t = \pi[\theta_t](t, x_t).$$
(6.50)

We can therefore conclude that $(x_{\mathcal{I}_k}, u_{\mathcal{I}_k}) \in \mathcal{S}_{\mathcal{I}_k}[\rho; b_k]$. We apply to conclude that

$$\sum_{t \in \mathcal{I}_k} \mathcal{G}_t(x_t, u_t) \leqslant M_{\rho}^{\pi}$$
(6.51)

for each $k \in \{0, ..., N\}$. Now, define S as the following collection of intervals

$$\mathbb{S} := \{ \mathcal{I}_k \mid \mathcal{G}_{t_{k+1}}(x_{t_{k+1}}, u_{t_{k+1}}) = 1 \},$$
(6.52)

i.e., all intervals \mathcal{I}_k where at the start of the *next* interval \mathcal{I}_{k+1} the cost is 1. Combining this with the former bound (6.51), we can decompose the total mistake sum as

$$\sum_{t=0}^{T} \mathcal{G}_t(x_t, u_t) = \sum_{t \in \mathcal{I}_0} \mathcal{G}_t(x_t, u_t) + \sum_{\mathcal{I}_j \in \mathbb{S}} \sum_{t \in \mathcal{I}_{j+1}} \mathcal{G}_t(x_t, u_t) + \underbrace{\sum_{\mathcal{I}_j \notin \mathbb{S}} \sum_{t \in \mathcal{I}_{j+1}} \mathcal{G}_t(x_t, u_t)}_{0}_{0}$$
$$= \sum_{t \in \mathcal{I}_0} \mathcal{G}_t(x_t, u_t) + \sum_{\mathcal{I}_j \in \mathbb{S}} \sum_{t \in \mathcal{I}_{j+1}} \mathcal{G}_t(x_t, u_t) \leqslant M_{\rho}^{\pi} \left(|\mathbb{S}| + 1\right). \quad (6.53)$$

Notice that the last term $\sum_{\mathcal{I}_j \notin \mathbb{S}} \sum_{t \in \mathcal{I}_{j+1}} \mathcal{G}_t(x_t, u_t)$ in the first equation is zero because $\mathcal{I}_j \notin \mathbb{S}$ implies that the next interval \mathcal{I}_{j+1} start with zero cost; due to the cost-invariance property it follows that $\sum_{t \in \mathcal{I}_{j+1}} \mathcal{G}_t(x_t, u_t) = 0$. The remainder of the proof is concerned with bounding the cardinality of the collection \mathbb{S} .

Bounding $|\mathbb{S}|$: We know that for each l in the range $1 \leq l \leq |\mathcal{I}_k|$, there exists at least one sub-interval $\mathcal{I}'_l \subset \mathcal{I}_k$, $|\mathcal{I}'_l| = l$ of length l, such that

$$\sum_{t\in\mathcal{I}'} d(\theta_t, \theta_{t+1}) > \frac{1}{2}\rho \frac{l}{(|\mathcal{I}_k|+l)}.$$
(6.54)

The above has to be true, since otherwise we would contradict (6.49):

• Let $\mathcal{I}'_1, \ldots, \mathcal{I}'_m, m = \lceil |\mathcal{I}_k|/l \rceil, |\mathcal{I}'_i| = l, \mathcal{I}'_i \subset \mathcal{I}_k$ be an overlapping cover of \mathcal{I}_k , then

$$d(a_k, a_{k+1}) \leq \sum_{t \in \mathcal{I}_k} d(\theta_{t+1}, \theta_t) \leq \sum_{j=1}^m \sum_{t \in \mathcal{I}'_j} d(\theta_{t+1}, \theta_t) \leq \frac{1}{2} \rho \left[\frac{|\mathcal{I}_k|}{l} \right] \frac{l}{|\mathcal{I}_k| + l} \quad (6.55)$$

$$\leq \frac{1}{2} \rho \left(\frac{|\mathcal{I}_k|}{l} + 1 \right) \frac{l}{|\mathcal{I}_k| + l} = \frac{1}{2} \rho, \qquad (\text{recall that } a_{k+1} = \theta_{\bar{t}_k + 1})$$

$$(6.56)$$

which is a contradiction to (6.49)

Hence, we can always pick a sequence of sub-intervals $[t_k^l, \bar{t}_k^l] = \mathcal{I}_k^{(l)} \subset \mathcal{I}_k$ (either of length l or identical to \mathcal{I}_k if $|\mathcal{I}_k| \leq l$) such that

$$\sum_{t \in \mathcal{I}_k^{(l)}} d(\theta_t, \theta_{t+1}) > \frac{1}{2} \rho_{\frac{l}{|\mathcal{I}_k| + l}}.$$
(6.57)

Notice that if $|\mathcal{I}_k| \leq l$, we pick $\mathcal{I}_k^{(l)} = \mathcal{I}_k$, and therefore the above inequality is vacuously true since, $\sum_{t \in \mathcal{I}_k} d(\theta_t, \theta_{t+1}) \geq d(a_k, a_{k+1}) > \frac{1}{2}\rho \geq \frac{1}{2}\rho \frac{l}{|\mathcal{I}_k|+l}$. Now, the (γ, T) -finite-time competitiveness property ensures that for all k and all $t \geq \bar{t}_k + 1$ holds:

$$\frac{1}{2}\rho_{|\mathcal{I}_{k}|+T} < \sum_{h \in \mathcal{I}_{k}^{(T)}} d(\theta_{h}, \theta_{h+1}) \leqslant \gamma d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{t_{k}}), \mathsf{P}(\mathcal{D}_{\bar{t}_{k}+1})) \leqslant \gamma d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{t_{k}}), \mathsf{P}(\mathcal{D}_{t}))$$
$$\implies d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{t_{k}}), \mathsf{P}(\mathcal{D}_{t})) > \frac{1}{2}\frac{\rho}{\gamma}\frac{T}{|\mathcal{I}_{k}|+T}$$

where $d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{\tau_k}),\mathsf{P}(\mathcal{D}_{\bar{\tau}_k+1})) \leq d_{\mathcal{H}}(\mathsf{P}(\mathcal{D}_{\tau_k}),\mathsf{P}(\mathcal{D}_t))$ follows from nestedness. From now on, we use the abbreviation P_t to refer to the sets $\mathsf{P}(\mathcal{D}_t)$.

Recall the definition $\mathbb{S} := \{\mathcal{I}_k \mid \mathcal{G}_{t_{k+1}}(x_{t_{k+1}}, u_{t_{k+1}}) = 1\}$ and let k_j denote the *j*-th interval that belongs to \mathbb{S} , i.e., $\mathcal{I}_{k_j} \subset \mathbb{S}$. Now set l = T and define S_j as a subsequence of $\mathsf{P}_1, \mathsf{P}_2, \ldots$ as follows:

$$\mathsf{S}_{j} := \begin{cases} \mathsf{P}_{\bar{t}_{k_{j}}} & \text{if } x_{\bar{t}_{k_{j}}} \in \mathsf{X}_{\bar{t}_{k_{j}}} \\ \mathsf{P}_{t_{k_{j}}^{T}} & \text{if } x_{\bar{t}_{k_{j}}} \notin \mathsf{X}_{\bar{t}_{k_{j}}}. \end{cases}$$
(6.58)

We will show that this collection $P = \{S_1, S_2, ...\}$ of sets S_j is a $\frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}$ - separated set in the metric space $(2^{\Omega}, d_{\mathcal{H}})$ via the following inequality:

$$\forall j < i : d_{\mathcal{H}}(\mathsf{S}_j, \mathsf{S}_i) > \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}.$$

This is proven below:

1. Assume $x_{\bar{t}_k} \in X_{\bar{t}_k}$, then it has to hold that $d(\theta_{\bar{t}_k}, \theta_{t_{k+1}}) > \rho$. Notice due to (γ, H) -w.c. property, that $d(\theta_{\bar{t}_k}, \theta_{t_{k+1}}) \leq \gamma d_{\mathcal{H}}(\mathsf{P}_{\bar{t}_k}, \mathsf{P}_{t_{k+1}})$ which gives us

$$d_{\mathcal{H}}(\mathsf{P}_{\bar{t}_k},\mathsf{P}_{t_{k+1}}) > \frac{1}{2}\frac{\rho}{\gamma}.$$

2. Assume $x_{\bar{t}_k} \notin X_{\bar{t}_k}$, then by Def. 6.5, it follows that $|\mathcal{I}_k| \leq M_{\rho}^{\pi}$, which then implies that $d_{\mathcal{H}}(\mathsf{P}_{t_k^T},\mathsf{P}_{t_{k+1}}) > \frac{1}{2}\frac{\rho}{\gamma}\frac{T}{|\mathcal{I}_k|+T} \geq \frac{1}{2}\frac{\rho}{\gamma}\frac{T}{M_{\rho}^{\pi}+T}$.

• Taking the minimum of both cases we can see that $d_{\mathcal{H}}(\mathsf{S}_j,\mathsf{P}_{t_{k_j+1}}) > \frac{1}{2}\frac{\rho}{\gamma}\frac{T}{M_{\rho}^{\pi}+T}$. Due to nestedness, it holds for i > j that $\mathsf{S}_i \subset \mathsf{P}_{t_{k_j+1}} \subset \mathsf{S}_j$. Thus, it holds $d_{\mathcal{H}}(\mathsf{S}_j,\mathsf{S}_i) \ge d_{\mathcal{H}}(\mathsf{S}_j,\mathsf{P}_{t_{k_i+1}})$ and we arrive at the separation condition:

$$\forall j < i : d_{\mathcal{H}}(\mathsf{S}_j, \mathsf{S}_i) > \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}.$$

We conclude from Lem. 50, that $|\mathbb{S}| = |P|$ is bounded by the packing number $N(\mathbf{\Omega}, \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T})$. Substituting into the bound (6.53) and taking the limit $\tau \to \infty$, we get the total number of mistakes as:

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leqslant M_{\rho}^{\pi} \left(N(\mathbf{\Omega}, \frac{1}{2} \frac{\rho}{\gamma} \frac{T}{M_{\rho}^{\pi} + T}) + 1 \right).$$

A Tighter Bound. We can define $\Omega^{\circ} = \Omega \setminus \operatorname{int}(\mathsf{P}(\mathcal{D}_{\infty}))$ and $\mathsf{S}_{j}^{\circ} = \mathsf{S}_{j} \setminus \operatorname{int}(\mathsf{P}(\mathcal{D}_{\infty}))$ and notice that S_{j}° is non-empty for all j: S_{j}° and $\mathsf{P}(\mathcal{D}_{\infty})$ are closed, so $\mathsf{S}_{j} \subset \mathsf{P}(\mathcal{D}_{\infty})$ implies that S_{j}° contains at least the boundary of $\mathsf{P}(\mathcal{D}_{\infty})$. Moreover, we can verify that the corresponding collection $P^{\circ} = \{\mathsf{S}_{1}^{\circ},\mathsf{S}_{2}^{\circ},\ldots\}$ of sets S_{j}° is still a $\frac{1}{2}\frac{\rho}{\gamma}\frac{H}{M_{\rho}^{\pi}+H}$ separated set in the compact metric space $(2^{\Omega^{\circ}}, d_{\mathcal{H}})$. Therefore we can improve the previous mistake guarantee and state the tighter inequality:

$$\sum_{t=0}^{\infty} \mathcal{G}_t(x_t, u_t) \leqslant \left(N(\mathbf{\Omega} \setminus \operatorname{int}(\mathsf{P}(\mathcal{D}_{\infty})), \frac{1}{2} \frac{\rho}{\gamma} \frac{H}{M_{\rho}^{\pi} + H}) + 1 \right) M_{\rho}^{\pi}.$$

Let $\chi_A : \mathbb{R} \mapsto \{0, 1\}$ denote the characteristic function for a subset $A \subset \mathbb{R}$ of the real line and $\overline{\chi}_A := \chi_{A^{\mathbb{C}}}$ be the corresponding complementary function, i.e., $\overline{\chi}_A(s) = 0, \forall s \in A \text{ and } \overline{\chi}_A(s) = 1, \forall s \notin A.$

Lemma 57. For any scalar sequence $a \in \ell_1$ holds:

$$\|\boldsymbol{a}\|_{1} = \int_{0}^{\infty} \sum_{k=0}^{\infty} \overline{\chi}_{[-s,s]}(a_{k}) ds = \int_{0}^{\|\boldsymbol{a}\|_{\infty}} \sum_{k=0}^{\infty} \overline{\chi}_{[-s,s]}(a_{k}) ds$$
(6.59)

Proof. We first prove the result for the finite sequence case and for non-negative sequences, which is stated in equation (6.61); the general result then follows as a corollary. Let $a \in \mathbb{R}^{\mathbb{N}}$ be a non-negative scalar sequence and define $Q_a(t, r)$, for $r \ge 0$ as the number of indeces $i \le t$ for which the sequence is larger than r, i.e.,

$$Q_{a}(t,r) := |\{i \mid i \leq t \text{ and } a_{i} > r\}|, \qquad (6.60)$$

then our goal is to show that

$$\sum_{k=0}^{t} a_k = \int_0^\infty Q_a(t, r) dr = \int_0^{\max_{k \le t} a_k} Q_a(t, r) dr.$$
(6.61)

Let $a_0^{\uparrow}, \ldots, a_t^{\uparrow}$ denote the sequence we obtain from rearranging the sequence $a_{[0,t]}$ in an increasing order. Thus, $a_0^{\uparrow} \leq a_1^{\uparrow} \leq \cdots \leq a_t^{\uparrow}$. It is clear that rearranging does not

change the sum, hence it holds: $\sum_{i=0}^{t} a_i = \sum_{i=0}^{t} a_i^{\uparrow}$. Now notice that we can rewrite the sum in terms of the increments $a_i^{\uparrow} - a_{i-1}^{\uparrow}$ as follows:

$$\sum_{k=0}^{t} a_{i}^{\uparrow} = a_{0}^{\uparrow} + \sum_{k=1}^{t} \left(a_{0}^{\uparrow} + \sum_{j=1}^{k} (a_{j}^{\uparrow} - a_{j-1}^{\uparrow}) \right)$$
$$= a_{0}^{\uparrow}(t+1) + \sum_{j=1}^{t} \sum_{k=j}^{t} (a_{j}^{\uparrow} - a_{j-1}^{\uparrow})$$
$$= a_{0}^{\uparrow}(t+1) + \sum_{j=1}^{t} \left(a_{j}^{\uparrow} - a_{j-1}^{\uparrow} \right) (t+1-j)$$
(6.62)

where we obtain the last equality through changing the order of summation. Observe that the function $Q_a(t, \cdot) : r \mapsto Q_a(t, r)$ is the following piece-wise constant non-increasing function:

$$Q_{\boldsymbol{a}}(t,r) = \begin{cases} t+1 & \text{if } r \in [0,a_0^{\uparrow}) \\ t+1-j & \text{if } r \in [a_j^{\uparrow},a_{j-1}^{\uparrow}) \text{ for } j \ge 1 \\ 0 & \text{if } r \in [a_t^{\uparrow},\infty) \end{cases}$$

Integrating over the domain $[0, \infty)$ shows that $\int_0^\infty Q_a(t, r) dr$ is equal to the right-hand side of (6.62) and concludes the proof of the partial result (6.61).

For the final step, notice that for $Q_a(t,r) + Q_{-a}(t,r) = \sum_{k=0}^t \overline{\chi}_{[-r,r]}(a_k)$ and therefore by (6.61), we have

$$\sum_{k=0}^{t} |a_k| = \int_0^{\max_{k \le t} |a_k|} \sum_{k=0}^{t} \overline{\chi}_{[-s,s]}(a_k) ds$$
(6.63)

Furthermore, notice that $\sup_t \sum_{k=0}^t \overline{\chi}_{[-s,s]}(a_k)$ is finite for all s > 0, because otherwise it would contradict $a \in \ell_1$:

Assume there exists some s' > 0 for which ∑_{k=0}^t X_[-s',s'](a_k) grows unbounded in t, then it means that there are infinitely many elements {a_j} which are bounded below as a_j > s'; this contradicts ∑_{k=0}[∞] |a_k| < ∞.

It is also clear that $\sum_{k=0}^{\infty} \overline{\chi}_{[-s,s]}(a_k) = 0$ for any $s \ge \|\boldsymbol{a}\|_{\infty}$ and that the right-hand side of (6.63) is always bounded above $\|\boldsymbol{a}\|_1$. Hence, by taking the limit $t \to \infty$ in (6.63), we obtain our final result (6.59).

6.B Oracle for Cartpole-Swing up with Constraints

Next, we describe the oracle we used to instantiate our approach $A_{\pi}(SEL)$ for the test of swinging-up the cart pole on a constrained track. The overall control strategy is a

hybrid combination of a linear state-feedback LQR around the upright position, a so-called energy-based swing-up controller (See [17]) and a control barrier-function to respect the safety constraints [7].

For convenience of the reader we recall the nonlinear dynamics of the cart pole system:

$$(M+m)\ddot{x} - ml\ddot{\phi}\cos(\phi) + ml\dot{\phi}^{2}\sin(\phi) - b_{x}\dot{x} = F$$

$$l\ddot{\phi} - g\sin(\phi) - b_{\phi}\dot{\phi} = \ddot{x}\cos(\phi).$$
(6.64)

Let x and \dot{x} be the position and velocity of the cart and ϕ , $\dot{\phi}$ the angle and angular velocity of the pole. F is the force onto the cart pole and serves as our control input to the system. Throughout the discussion \bar{a} and \bar{d} are design parameters, where \bar{a} denotes the maximal cart acceleration allowed, and \bar{d} be the maximum distance the cart is allowed to move from the center.

Model-Based Oracle for Cart-Pole Swing Up

The outermost layer of the control strategy is partial feedback linearization. Let $F_d(\ddot{x}, \dot{x}, \phi, \dot{\phi}, \dot{x})$ be the force F we need to apply at time t in order to achieve a desired cart acceleration of \ddot{x}_d . Multiply the second equation of (6.64) by $ml \cos(\phi)$ and add it to the first, to see that F_d has to be chosen as:

$$F_d(\ddot{x}_d, \dot{x}, \phi, \dot{\phi}, \dot{x}) = (M + m\sin(\phi)^2)\ddot{x}_d - mg\cos(\phi)\sin(\phi) + ml\dot{\phi}^2\sin(\phi) - b_x\dot{x}.$$
(6.65)

By choosing $F = F_d(\ddot{x}_d, \dot{x}, \phi, \dot{\phi}, \dot{x})$, we can now treat the desired acceleration \ddot{x}_d as our new control input. With respect to our new input \ddot{x}_d , we can simplify the original equations (6.64) to

$$\ddot{x} = \ddot{x}_d \tag{6.66}$$

$$\ddot{l\phi} - g\sin(\phi) - b_{\phi}\dot{\phi} = \ddot{x}_d\cos(\phi).$$
(6.67)

The swingup controller consists now of three separate control laws that are later combined.

• Around up-right position: static linear LQR controller If the pole has small enough kinetic energy and is close to the upright position, we simply choose \ddot{x}_d to be LQR-state feedback controller based on the system (6.66) linearized around the equilibrium position $x = 0, \dot{x} = 0, \phi = 0, \dot{\phi} = 0$. The policy then takes the form

$$\ddot{x}_{d,LQR} = -K_{LQR}(\theta)z.$$

• Swing up-controller: energy-based controller. The Swing-Up controller is based on an elegant energy-based approach by [17], in which we simply choose \ddot{x}_d to control the total normalized energy

$$E(\phi, \dot{\phi}) = \frac{l}{2g} \dot{\phi}^2 + \cos(\phi) \tag{6.68}$$

of the pole. In-depth derivation can be found in [17] and \ddot{x}_d takes the form:

$$\ddot{x}_{d,swing} = -\mathbf{Sat}_{\bar{a}} \left(\frac{1}{2} \gamma |\cos(\phi)| (E(\phi, \dot{\phi}) - 1) \operatorname{sign}(\dot{\phi} \cos(\phi)) \right)$$
(6.69)

where $\mathbf{Sat}_{\bar{a}}$ is the saturation function which saturates at the max specified acceleration \bar{a} .

Wrapping a safety controller. As part of our oracle policy, we also use a control barrier function controller that prevents us from triggering the safety policy. We do this simply by internally overriding our swing-up \(\vec{x}_{d,swing}\) or balancing \(\vec{x}_{d,LQR}\) terms, if we get too close to the boundary of \([-x_{max}, x_{max}]\)]. To this end, define \(B(x, \vec{x})\) as the barrier function

$$B(x,\dot{x}) = \frac{1}{2\bar{a}}\dot{x}|\dot{x}| + x$$

and define $\ddot{\phi}_{max} := \bar{a}g/l * \sin(30^\circ)$.

The full description of the oracle policy is mapped out in Algorithm (6).

The controller switches to an LQR if the system is close to the upright position, and otherwise defaults to the swing up controller that brings the pendulum to the right energy level. A correction is performed to the previous control action depending on the barrier-function value $|B(x, \dot{x})|$. As $|B(x, \dot{x})|$ gets closer to the boundary $\bar{d} - \epsilon_{safe}$, the controller prioritizes safety and overwrites the previous planned control action. If x exceeds the buffer $\bar{d} - \epsilon_{safe}$, then a safe policy is called, which brings the cart position back to the region $[-\bar{d} + \epsilon_{safe}, \bar{d} - \epsilon_{safe}]$.

Selection Process SEL

We apply the approach presented in the main paper Section 6.10 to obtain polytopes of consistent parameters of P_t for the lumped parameters $p = [m_c + m_c]$ **Input:** $z = [x, \phi, \dot{x}, \dot{\phi}]$, parameters $\theta := [M, m, l, b_x, b_\theta]$, ϵ_{safe} **Output:** *F* if $|x| < \bar{d} - \epsilon_{safe}$ then if $|\dot{\phi}^2/\ddot{\phi}_{max}| < 60^\circ$ and $\cos(30^\circ/\ddot{\phi}_{max} + \phi \operatorname{sign}(\dot{\phi})) > \cos(30^\circ)$ and |- $K_{LQR}(\theta)z| \leq \bar{a}$ then $\ddot{x}_d = -K_{LQR}(\theta)z$ else $\ddot{x}_d = -\mathbf{Sat}_{\bar{a}} \left[\frac{1}{2} \gamma | \cos(\phi) | (E(\phi, \dot{\phi}) - 1) \operatorname{sign}(\dot{\phi} \cos(\phi)) \right]$ end if $\ddot{x}_{d,back} = -\bar{a}\operatorname{sign}(\dot{x})$ $\lambda = \frac{|B(x,\dot{x})|}{d - \epsilon_{safe}}$ if $B(x, \dot{x}) \ge 0$ then $\ddot{x}_d \leftarrow (1 - \lambda^2)\ddot{x}_d + \lambda^2 \min\{\ddot{x}_d, \ddot{x}_{d, back}\}$ else $\ddot{x}_d \leftarrow (1 - \lambda^2)\ddot{x}_d + \lambda^2 \max{\{\ddot{x}_d, \ddot{x}_{d,back}\}}$ end if $F = F_d(\ddot{x}_d, z)$ else $F = \pi_{\text{safety}}(z)$ end if

 $m_p, m_p l, b_x, l, b_\theta, \tau_{d,x}, \tau_{d,\theta}$]. We use randomized LPs [13], to approximate the Steiner point of the polytope P_t and select the corresponding oracle policy $\pi[\theta_t]$ as described in the meta-algorithm 2.

6.C Mistake Guarantees vs Sublinear Regret

A common performance metric in online learning for control is phrased in terms of the regret R(T). For our general problem setting, we show that sublinear regret does not imply finite mistake guarantees, however finite mistake guarantees do imply sublinear regret.

Regret Definition

Assume we are given some cost function $C : \mathcal{X} \times \mathcal{U} \mapsto \mathbb{R}^+$ and the system $x_{t+1} = f^*(t, x_t, u_t), x_0 = \xi_0$. Assume that some "ideal" policy π^* would generate the trajectory x_t^*, u_t^* , while the online algorithm \mathcal{A} , characterized by the sequence of policies, produces x_t, u_t . The optimal total cost of $J^*(T)$ at time T is defined as $J^*(T) := \sum_{k=0}^T C(x_k^*, u_k^*)$. The regret R(T) of \mathcal{A} usually refers to the sum of costs of the online algorithm up to time T minus the sum of costs that the optimal policy

 π^* would achieve ³:

$$R(T) := \sum_{k=0}^{T} C(x_k, u_k) - J^*(T).$$
(6.70)

Sublinear regret is defined as follows.

Definition 6.14. The regret R(T) is called sublinear if R(T) = o(T), or equivalently:

$$\lim_{T \to \infty} \frac{1}{T} R(T) = \lim_{T \to \infty} \frac{1}{T} \sum_{k=0}^{T} (C(x_k, u_k) - C(x_k^*, u_k^*)) = 0.$$
(6.71)

The slower R(T) grows with T, (for example $O(\log(T))$), the faster convergence we can guarantee to the above limit.

Sublinear Regret Does Not Imply Bounded Cost

Sublinear regret is a common way to measure the performance of online learning and control algorithms. Ideally, we would expect the sublinear regret to subsume some more basic performance criteria such as the boundedness of the online cost, that is, $\sup_k |C(x_k, u_k) - C(x_k^*, u_k^*)| < \infty$. However, simple derivations show that without additional assumptions, sublinear regret growth is not sufficient to show cost boundedness. The reason for that is intrinsic to the very definition of regret, and simple real-analysis arguments will suffice to demonstrate that.

Abbreviate $c_k := C(x_k, u_k), c_k^* := C(x_k^*, u_k^*)$ and define the sequences

$$s_k := c_k - c_k^* \tag{6.72}$$

$$m_k := \frac{1}{k} \sum_{j=0}^k (c_k - c_k^*) = \frac{1}{k} \sum_{j=1}^k s_k.$$
(6.73)

Now, sublinear regret is defined as the condition $\lim_{k\to\infty} m_k = 0$, while bounded cost considers the statement $\sup_k |s_k| < \infty$.

The next counter examples show that these statements are not related; The first example shows that sublinear regret does *not* imply finiteness of the sequence $|s_k|$; the second shows that the boundedness of $|s_k|$ does *not* imply sublinear regret.

1.
$$\lim_{k \to \infty} m_k = 0 \implies \sup_k |s_k| < \infty.$$

³Not the most widespread definition, but the most suited for adaptive control setting. See, for example, [143].

Proof. Consider s_k , where $s_{e^n} = n$ and otherwise 0. Define $\bar{n}(k) := \lfloor \log(k) \rfloor$, then

$$m_k \leqslant m_{\bar{n}(k)} = \frac{1}{e^{\bar{n}(k)}} \sum_{i=1}^{\bar{n}(k)} i = \frac{\bar{n}(k)(\bar{n}(k)+1)}{2e^{\bar{n}(k)}}.$$

This shows $\lim_{k\to\infty} m_{\bar{n}(k)} = 0$, but s_k is unbounded.

2. For all $\varepsilon > 0$: $\sup_k s_k - \inf_k s_k < \varepsilon \implies m_k$ converges.

Proof. Define s_k as the sequence

$$\underbrace{(\underbrace{1,\delta,1,1,\delta,\delta}_{6},\underbrace{1,\ldots,1}_{6},\underbrace{\delta,\ldots,\delta}_{6},\underbrace{1,\ldots,1}_{18},\underbrace{\delta,\ldots,\delta}_{18},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}},\underbrace{\delta,\ldots,\delta}_{2\times 3^{n}},\ldots,\underbrace{1,\ldots,1}_{2\times 3^{n}$$

with $\delta = 1 - \varepsilon$. From the above pattern, it becomes apparent that the corresponding m_k is satisfied for all n:

$$m_{2 \times 3^n} = 1 - \varepsilon/2$$
 $m_{4 \times 3^n} = 1 - \varepsilon/4,$ (6.74)

hence m_k does not converge, yet $\sup_k s_k - \inf_k s_k < \varepsilon$.

Remark 43. The above arguments still hold if we change s_k and m_k to the definitions

$$s'_{k} := |C(x_{k}, u_{k}) - C(x_{k}^{*}, u_{k}^{*})| \quad m'_{k} := \frac{1}{k} \sum_{j=0}^{k} |C(x_{k}, u_{k}) - C(x_{k}^{*}, u_{k}^{*})| = \frac{1}{k} \sum_{j=1}^{k} s'_{k}.$$

Sublinear Regret Does Not Imply Finite Mistakes, Finite Mistakes Imply Sublinear Regret

In our problem setting, the costs C(x, u) are represented by $\mathcal{G}(x, u)$, which are $\{0, 1\}$ -valued cost functions. Moreover, we compare to an oracle-policy π^* which guarantees at most M_{ρ}^{π} mistakes i.e., $\sum_{k=0}^{\infty} c_k^* \leq M_{\rho}^{\pi}$. Trivially, finite mistakes implies sublinear regret: If it holds that $\sum_{k=0}^{\infty} c_k < M$, then $\frac{1}{T} \sum_{k=0}^{T} (c_k - c_k^*) \leq \frac{M}{T} \to 0$. However, the opposite is not true. Consider as an example the following sequence for c_k :

$$c = \underbrace{0, 1}_{2}, \underbrace{0, 0, 0, 1}_{4}, \underbrace{0, 0, 0, 0, 0, 0, 0, 0, 1}_{8}, \dots, \underbrace{0, \dots, 1}_{2^{k}}, \dots$$

The total mistakes $\sum_{t=1}^{T} c_k = \mathcal{O}(\log(T))$ grow unbounded, but we still have $\lim_{t\to\infty} m_k = 0.$

Bibliography

- [1] Yasin Abbasi-Yadkori and Csaba Szepesvári. Regret bounds for the adaptive control of linear quadratic systems. In *Proceedings of the 24th Annual Conference on Learning Theory*, pages 1–26, 2011.
- [2] Yasin Abbasi-Yadkori, Nevena Lazic, and Csaba Szepesvári. Regret bounds for model-free linear quadratic control. *arXiv preprint arXiv:1804.06021*, 2018.
- [3] Naman Agarwal, Brian Bullins, Elad Hazan, Sham M. Kakade, and Karan Singh. Online control with adversarial disturbances. *arXiv preprint arXiv:1902.08721*, 2019.
- [4] Naman Agarwal, Elad Hazan, and Karan Singh. Logarithmic regret for online control. In Advances in Neural Information Processing Systems, pages 10175–10184, 2019.
- [5] Anayo K. Akametalu, Jaime F. Fisac, Jeremy H. Gillula, Shahab Kaynama, Melanie N. Zeilinger, and Claire J. Tomlin. Reachability-based safe learning with gaussian processes. In 53rd IEEE Conference on Decision and Control, pages 1424–1431. IEEE, 2014.
- [6] Aaron D. Ames, Xiangru Xu, Jessy W. Grizzle, and Paulo Tabuada. Control barrier function based quadratic programs for safety critical systems. *IEEE Transactions on Automatic Control*, 62(8):3861–3876, 2016.
- [7] Aaron D. Ames, Xiangru Xu, Jessy W. Grizzle, and Paulo Tabuada. Control barrier function based quadratic programs for safety critical systems. *IEEE Transactions on Automatic Control*, 62(8):3861–3876, Aug 2017. ISSN 1558-2523. doi: 10.1109/TAC.2016.2638961.
- [8] Brian DO. Anderson and Arvin Dehghani. Challenges of adaptive control–past, permanent and future. *Annual Reviews in Control*, 32(2):123–135, 2008.
- [9] Brian DO. Anderson, Thomas Brinsmead, Daniel Liberzon, and A. Stephen Morse. Multiple model adaptive control with safe switching. *International Journal of Adaptive Control and Signal Processing*, 15(5): 445–470, 2001.
- [10] Brian DO. Anderson et al. Failures of adaptive control theory and their resolution. *Communications in Information & Systems*, 5(1):1–20, 2005.
- [11] James Anderson, John C. Doyle, Steven H. Low, and Nikolai Matni. System level synthesis. *Annual Reviews in Control*, 47:364–393, 2019.
- [12] CJ. Argue, Sébastien Bubeck, Michael B. Cohen, Anupam Gupta, and Yin Tat Lee. A nearly-linear bound for chasing nested convex bodies. In *Proceedings*

of the Thirtieth Annual ACM-SIAM Symposium on Discrete Algorithms, pages 117–122. SIAM, 2019.

- [13] CJ. Argue, Anupam Gupta, Guru Guruganesh, and Ziye Tang. Chasing convex bodies with linear competitive ratio. In *Proceedings of the Fourteenth Annual ACM-SIAM Symposium on Discrete Algorithms*, pages 1519–1524. SIAM, 2020.
- [14] Alessandro Astolfi, Dimitrios Karagiannis, and Romeo Ortega. Nonlinear and adaptive control with applications. Springer Science & Business Media, 2007.
- [15] Karl J. Åström and Björn Wittenmark. *Adaptive control*. Courier Corporation, 2013.
- [16] Karl Johan Åström and Peter Eykhoff. System identification: A survey. *Automatica*, 7(2):123–162, 1971.
- [17] Karl Johan Åström and Katsuhisa Furuta. Swinging up a pendulum by energy control. *Automatica*, 36(2):287–295, 2000.
- [18] Karl Johan Åström and Björn Wittenmark. On self tuning regulators. Automatica, 9(2):185–199, 1973.
- [19] Jean-Pierre Aubin and Hélène Frankowska. *Set-valued analysis*. Springer Science & Business Media, 2009.
- [20] Jean-Pierre Aubin, Alexandre M. Bayen, and Patrick Saint-Pierre. *Viability theory: New directions.* Springer Science & Business Media, 2011.
- [21] Erin M. Aylward, Pablo A. Parrilo, and Jean-Jacques E. Slotine. Stability and robustness analysis of nonlinear systems via contraction metrics and sos programming. *Automatica*, 44(8):2163–2170, 2008.
- [22] Somil Bansal, Mo Chen, Sylvia Herbert, and Claire J. Tomlin. Hamilton-jacobi reachability: A brief overview and recent advances. In 2017 IEEE 56th Annual Conference on Decision and Control (CDC), pages 2242–2253. IEEE, 2017.
- [23] Richard E. Bellman. *Adaptive Control Processes: A Guided Tour*. Princeton University Press, 1961.
- [24] Felix Berkenkamp, Matteo Turchetta, Angela Schoellig, and Andreas Krause. Safe model-based reinforcement learning with stability guarantees. In *Advances in Neural Information Processing Systems*, pages 908–918, 2017.
- [25] Rajendra Bhatia. *Matrix analysis*, volume 169. Springer Science & Business Media, 2013.
- [26] Franco Blanchini. Set invariance in control. *Automatica*, 35(11):1747–1767, 1999.

- [28] Nicholas M. Boffi, Stephen Tu, and Jean-Jacques E. Slotine. Regret bounds for adaptive nonlinear control. *arXiv preprint arXiv:2011.13101*, 2020.
- [29] Allan Borodin and Ran El-Yaniv. *Online computation and competitive analysis*. Cambridge University Press, 2005.
- [30] Francesco Borrelli, Alberto Bemporad, and Manfred Morari. *Predictive control for linear and hybrid systems*. Cambridge University Press, 2017.
- [31] Richard P. Braatz, Peter M. Young, John C. Doyle, and Manfred Morari. Computational complexity of /spl mu/ calculation. *IEEE Transactions on Automatic Control*, 39(5):1000–1002, 1994. doi: 10.1109/9.284879.
- [32] Greg Brockman, Vicki Cheung, Ludwig Pettersson, Jonas Schneider, John Schulman, Jie Tang, and Wojciech Zaremba. Openai gym. *arXiv preprint arXiv:1606.01540*, 2016.
- [33] Sébastien Bubeck, Bo'az Klartag, Yin Tat Lee, Yuanzhi Li, and Mark Sellke. Chasing nested convex bodies nearly optimally. In *Proceedings of the Fourteenth Annual ACM-SIAM Symposium on Discrete Algorithms*, pages 1496–1508. SIAM, 2020.
- [34] Niangjun Chen, Gautam Goel, and Adam Wierman. Smoothed online convex optimization in high dimensions via online balanced descent. In *Conference On Learning Theory (COLT)*, 2018.
- [35] Xinyi Chen and Elad Hazan. Black-box control for linear dynamical systems. *arXiv preprint arXiv:2007.06650*, 2020.
- [36] Yuxiao Chen and James Anderson. System level synthesis with state and input constraints. *arXiv preprint arXiv:1903.07174*, 2019.
- [37] Alon Cohen, Avinatan Hasidim, Tomer Koren, Nevena Lazic, Yishay Mansour, and Kunal Talwar. Online linear quadratic control. In *International Conference on Machine Learning*, pages 1029–1038, 2018.
- [38] Alon Cohen, Tomer Koren, and Yishay Mansour. Learning linear-quadratic regulators efficiently with only \sqrt{T} regret. *arXiv preprint arXiv:1902.06223*, 2019.
- [39] Lauren Conger, Jing Shuang Lisa Li, Eric Mazumdar, and Steven L. Brunton. Nonlinear system level synthesis for polynomial dynamical systems. In 2022 IEEE 61st Conference on Decision and Control (CDC), pages 3846–3852. IEEE, 2022.

- [40] Martin Corless and George Leitmann. Continuous state feedback guaranteeing uniform ultimate boundedness for uncertain dynamic systems. *IEEE Transactions on Automatic Control*, 26(5):1139–1144, 1981.
- [41] Munther A. Dahleh, Theodore V. Theodosopoulos, and John N. Tsitsiklis. The sample complexity of worst-case identification of fir linear systems. *Systems & Control Letters*, 20(3):157–166, 1993.
- [42] Sarah Dean, Horia Mania, Nikolai Matni, Benjamin Recht, and Stephen Tu. On the sample complexity of the linear quadratic regulator. *arXiv preprint arXiv:1710.01688*, 2017.
- [43] Sarah Dean, Horia Mania, Nikolai Matni, Benjamin Recht, and Stephen Tu. Regret bounds for robust adaptive control of the linear quadratic regulator. arXiv preprint arXiv:1805.09388, 2018.
- [44] Sarah Dean, Stephen Tu, Nikolai Matni, and Benjamin Recht. Safely learning to control the constrained linear quadratic regulator. In *2019 American Control Conference (ACC)*, pages 5582–5588, July 2019.
- [45] Sarah Dean, Stephen Tu, Nikolai Matni, and Benjamin Recht. Safely learning to control the constrained linear quadratic regulator. In *2019 American Control Conference (ACC)*, pages 5582–5588. IEEE, 2019.
- [46] Richard M. Dudley. Universal donsker classes and metric entropy. The Annals of Probability, 15(4):1306–1326, 1987. ISSN 00911798. URL http://www.jstor.org/stable/2244004.
- [47] Richard M. Dudley. Universal Donsker Classes and Metric Entropy, pages 345–365. Springer New York, New York, NY, 2010. ISBN 978-1-4419-5821-1. doi: 10.1007/978-1-4419-5821-1_19. URL https://doi.org/10.1007/978-1-4419-5821-1_19.
- [48] Gabriel Dulac-Arnold, Daniel Mankowitz, and Todd Hester. Challenges of real-world reinforcement learning. *arXiv preprint arXiv:1904.12901*, 2019.
- [49] Geir E. Dullerud and Fernando Paganini. *A course in robust control theory: a convex approach*, volume 36. Springer Science & Business Media, 2013.
- [50] Mohamad Kazem Shirani Faradonbeh, Ambuj Tewari, and George Michailidis. Optimism-based adaptive regulation of linear-quadratic systems. *arXiv preprint arXiv:1711.07230*, 2017.
- [51] Claude-Nicolas Fiechter. Pac adaptive control of linear systems. In *Proceedings* of the tenth annual conference on Computational learning theory, pages 72–80, 1997.
- [52] Nikolai Michailovich Filatov and Heinz Unbehauen. *Adaptive dual control: Theory and applications*, volume 302. Springer Science & Business Media, 2004.

- [53] Jaime F. Fisac, Anayo K. Akametalu, Melanie N. Zeilinger, Shahab Kaynama, Jeremy Gillula, and Claire J. Tomlin. A general safety framework for learningbased control in uncertain robotic systems. *IEEE Transactions on Automatic Control*, 2018.
- [54] Randy Freeman and Petar V. Kokotovic. Robust nonlinear control design: state-space and Lyapunov techniques. Springer Science & Business Media, 2008.
- [55] Joel Friedman and Nathan Linial. On convex body chasing. *Discrete & Computational Geometry*, 9(3):293–321, 1993.
- [56] Luca Furieri, Clara Lucía Galimberti, and Giancarlo Ferrari-Trecate. Neural system level synthesis: Learning over all stabilizing policies for nonlinear systems. In 2022 IEEE 61st Conference on Decision and Control (CDC), pages 2765–2770. IEEE, 2022.
- [57] Javier Garcia and Fernando Fernández. A comprehensive survey on safe reinforcement learning. *Journal of Machine Learning Research*, 16(1):1437– 1480, 2015.
- [58] Gautam Goel and Adam Wierman. An online algorithm for smoothed regression and lqr control. In *International Conference on Artificial Intelligence and Statistics (AISTATS)*, 2019.
- [59] Gautam Goel, Yiheng Lin, Haoyuan Sun, and Adam Wierman. Beyond online balanced descent: An optimal algorithm for smoothed online optimization. *Advances in Neural Information Processing Systems*, 32:1875–1885, 2019.
- [60] Lei Guo. Convergence and logarithm laws of self-tuning regulators. *Automatica*, 31(3):435–450, 1995.
- [61] Weiqiao Han and Russ Tedrake. Controller synthesis for discrete-time polynomial systems via occupation measures. In 2018 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pages 6911–6918, Oct 2018. doi: 10.1109/IROS.2018.8594400.
- [62] Elad Hazan, Sham M. Kakade, and Karan Singh. The nonstochastic control problem. In *Conference on Algorithmic Learning Theory (ALT)*, 2020.
- [63] Joao P. Hespanha, Daniel Liberzon, and A. Stephen Morse. Overcoming the limitations of adaptive control by means of logic-based switching. *Systems & Control Letters*, 49(1):49–65, 2003.
- [64] Dimitar Ho. A system level approach to discrete-time nonlinear systems. *arXiv preprint arXiv:2004.08004*, 2020.
- [65] Dimitar Ho and John C. Doyle. Robust model-free learning and control without prior knowledge. In 2019 IEEE 58th Conference on Decision and Control (CDC), pages 4577–4582. IEEE, 2019.

- [66] Dimitar Ho and John C. Doyle. Scalable robust adaptive control from the system level perspective. In 2019 American Control Conference (ACC), pages 3683–3688, 2019.
- [67] Matanya B. Horowitz, Anil Damle, and Joel W. Burdick. Linear hamilton jacobi bellman equations in high dimensions. In 53rd IEEE Conference on Decision and Control, pages 5880–5887, Dec 2014. doi: 10.1109/CDC.2014.7040310.
- [68] Naira Hovakimian. L1 adaptive control. *Department of Mechanical Science and Engineering, University of Illinois at Urbana-Champaign, Tech. Rep*, pages 1–15, 2014.
- [69] Naira Hovakimyan and Chengyu Cao. *L1 adaptive control theory: Guaranteed robustness with fast adaptation*, volume 21. SIAM-Society for Industrial and Applied Mathematics, 2010.
- [70] Tingshu Hu, Andrew R. Teel, and Luca Zaccarian. Anti-windup synthesis for linear control systems with input saturation: Achieving regional, nonlinear performance. *Automatica*, 44(2):512–519, 2008.
- [71] Petros Ioannou. Adaptive Control Tutorial (Advances in Design and Control). Society for Industrial and Applied Mathematics, Philadelphia, PA, USA, 2006. ISBN 0898716152.
- [72] Petros A. Ioannou and Jing Sun. *Robust adaptive control*, volume 1. PTR Prentice-Hall Upper Saddle River, NJ, 1996.
- [73] Petros A. Ioannou and Jing Sun. *Robust adaptive control*. Courier Corporation, 2012.
- [74] A.R. Mancini J. A. Aseltine and C.W.Sartune. A survey of adaptive control systems. *IEEE Transactions on Automatic Control*, 3:102–108, 1958.
- [75] Zhong-Ping Jiang, Eduardo Sontag, and Yuan Wang. Input-to-state stability for discrete-time nonlinear systems. *IFAC Proceedings Volumes*, 32(2):2403 – 2408, 1999. 14th IFAC World Congress 1999, Beijing, Chia, 5-9 July.
- [76] Sham Kakade, Akshay Krishnamurthy, Kendall Lowrey, Motoya Ohnishi, and Wen Sun. Information theoretic regret bounds for online nonlinear control. In Advances in Neural Information Processing Systems (NeurIPS), 2020.
- [77] Hassan K. Khalil. Nonlinear systems. Upper Saddle River, 2002.
- [78] Ilya Kolmanovsky, Emanuele Garone, and Stefano Di Cairano. Reference and command governors: A tutorial on their theory and automotive applications. In 2014 American Control Conference, pages 226–241. IEEE, 2014.
- [79] Andrei Nikolaevich Kolmogorov. New metric invariant of transitive dynamical systems and endomorphisms of lebesgue spaces russian. In *Doklady of Russian Academy of Sciences*, volume 119, page N5, 1958.
- [80] Andrei Nikolaevich Kolmogorov and Vladimir Mikhailovich Tikhomirov. ε entropy and ε -capacity of sets in function spaces. *Uspekhi Matematicheskikh Nauk*, 14(2):3–86, 1959.
- [81] Mayuresh V. Kothare, Peter J. Campo, Manfred Morari, and Carl N. Nett. A unified framework for the study of anti-windup designs. *Automatica*, 30(12): 1869–1883, 1994.
- [82] Elias Koutsoupias and Christos H. Papadimitriou. Beyond competitive analysis. *SIAM Journal on Computing*, 30(1):300–317, 2000.
- [83] Miroslav Krstic, Petar V. Kokotovic, and Ioannis Kanellakopoulos. *Nonlinear* and Adaptive Control Design. John Wiley & Sons, Inc., 1995.
- [84] Jean Bernard Lasserre, Didier Henrion, Christophe Prieur, and Emmanuel Trélat. Nonlinear optimal control via occupation measures and lmi-relaxations. *SIAM Journal on Control and Optimization*, 47(4):1643–1666, 2008. doi: 10.1137/070685051. URL https://doi.org/10.1137/070685051.
- [85] Frank L. Lewis, Draguna Vrabie, and Vassilis L. Syrmos. *Optimal control*. John Wiley & Sons, 2012.
- [86] Zhen Li and Jing Sun. Disturbance compensating model predictive control with application to ship heading control. *IEEE Transactions on Control Systems Technology*, 20(1):257–265, 2011.
- [87] Xiangbin Liu, Hongye Su, Bin Yao, and Jian Chu. Adaptive robust control of nonlinear systems with dynamic uncertainties. *International Journal of Adaptive Control and Signal Processing*, 23(4):353–377, 2009.
- [88] Lennart Ljung. System identification: Theory for the user. *PTR Prentice Hall, Upper Saddle River, NJ*, 28, 1999.
- [89] Lennart Ljung. System identification. *Wiley Encyclopedia of Electrical and Electronics Engineering*, 2001.
- [90] Winfried Lohmiller and Jean-Jacques E. Slotine. On contraction analysis for non-linear systems. *Automatica*, 34(6):683–696, 1998.
- [91] Jan Marian Maciejowski. *Predictive control: With constraints*. Pearson Education, 2002.
- [92] Horia Mania, Stephen Tu, and Benjamin Recht. Certainty equivalent control of lqr is efficient. *arXiv preprint arXiv:1902.07826*, 2019.
- [93] Horia Mania, Michael I. Jordan, and Benjamin Recht. Active learning for nonlinear system identification with guarantees. *arXiv preprint arXiv:2006.10277*, 2020.

- [94] Nikolai Matni, Alexandre Proutiere, Anders Rantzer, and Stephen Tu. From self-tuning regulators to reinforcement learning and back again. *arXiv preprint arXiv:1906.11392*, 2019.
- [95] David Q. Mayne, James B. Rawlings, Christopher V. Rao, and Pierre OM Scokaert. Constrained model predictive control: Stability and optimality. *Automatica*, 36(6):789–814, 2000.
- [96] David Q. Mayne, Eric C. Kerrigan, Eugene Van Wyk, and Paola Falugi. Tube-based robust nonlinear model predictive control. *International Journal* of Robust and Nonlinear Control, 21(11):1341–1353, 2011.
- [97] Mario Milanese, John Norton, Hélène Piet-Lahanier, and Éric Walter. Bounding approaches to system identification. Springer Science & Business Media, 2013.
- [98] Richard M. Murray. *A mathematical introduction to robotic manipulation*. CRC Press, 2017.
- [99] Antonis Papachristodoulou and Stephen Prajna. A tutorial on sum of squares techniques for systems analysis. In *Proceedings of the 2005, American Control Conference, 2005.*, pages 2686–2700. IEEE, 2005.
- [100] Pablo A. Parrilo. Structured semidefinite programs and semialgebraic geometry methods in robustness and optimization. PhD thesis, California Institute of Technology, 2000.
- [101] Lawrence Perko. *Differential equations and dynamical systems*, volume 7. Springer Science & Business Media, 2013.
- [102] Marios M. Polycarpou and Petros A. Ioannou. A robust adaptive nonlinear control design. In 1993 American Control Conference, pages 1365–1369. IEEE, 1993.
- [103] Stephen Prajna, Antonis Papachristodoulou, and Pablo A. Parrilo. Introducing sostools: a general purpose sum of squares programming solver. In *Proceedings of the 41st IEEE Conference on Decision and Control, 2002.*, volume 1, pages 741–746 vol.1, Dec 2002. doi: 10.1109/CDC.2002.1184594.
- [104] Stephen Prajna, Antonis Papachristodoulou, and Fen Wu. Nonlinear control synthesis by sum of squares optimization: a lyapunov-based approach. In 2004 5th Asian Control Conference (IEEE Cat. No.04EX904), volume 1, pages 157–165 Vol.1, July 2004.
- [105] Stephen Prajna, Pablo A. Parrilo, and Anders Rantzer. Nonlinear control synthesis by convex optimization. *IEEE Transactions on Automatic Control*, 49(2):310–314, Feb 2004. ISSN 0018-9286. doi: 10.1109/TAC.2003.823000.

- [106] Saša Raković, Andrew R. Teel, David Q. Mayne, and Alessandro Astolfi. Simple robust control invariant tubes for some classes of nonlinear discrete time systems. In *Proceedings of the 45th IEEE Conference on Decision and Control*, pages 6397–6402. IEEE, 2006.
- [107] Saša V. Rakovic and Miroslav Baric. Parameterized robust control invariant sets for linear systems: Theoretical advances and computational remarks. *IEEE Transactions on Automatic Control*, 55(7):1599–1614, 2010.
- [108] Ugo Rosolia and Francesco Borrelli. Learning model predictive control for iterative goals. a data-driven control framework. *IEEE Transactions on Automatic Control*, 63(7):1883–1896, 2017.
- [109] Walter Rudin. *Principles of Mathematical Analysis*. McGraw-Hill, 3rd edition, 1976.
- [110] Richard Saeks. Causality in hilbert space. Siam Review, 12(3):357–383, 1970.
- [111] Shankar Sastry. *Nonlinear systems: Analysis, stability, and control*, volume 10. Springer Science & Business Media, 2013.
- [112] Shankar Sastry and Marc Bodson. *Adaptive control: Stability, convergence and robustness*. Courier Corporation, 2011.
- [113] Rolf Schneider. *Convex bodies: The Brunn–Minkowski theory*. Cambridge university press, 2014.
- [114] Mark Sellke. Chasing convex bodies optimally. In Proceedings of the Fourteenth Annual ACM-SIAM Symposium on Discrete Algorithms, pages 1509–1518. SIAM, 2020.
- [115] Guanya Shi, Yiheng Lin, Soon-Jo Chung, Yisong Yue, and Adam Wierman. Online optimization with memory and competitive control. In Advances in Neural Information Processing Systems (NeurIPS), 2020.
- [116] Max Simchowitz, Horia Mania, Stephen Tu, Michael I. Jordan, and Benjamin Recht. Learning without mixing: Towards a sharp analysis of linear system identification. In *Conference On Learning Theory*, pages 439–473. PMLR, 2018.
- [117] Jean-Jacques E. Slotine, Weiping Li, et al. *Applied nonlinear control*, volume 199. Prentice hall Englewood Cliffs, NJ, 1991.
- [118] Mark W. Spong. On the robust control of robot manipulators. *IEEE Transactions on Automatic Control*, 37(11):1782–1786, 1992. doi: 10.1109/9.173151.
- [119] Mark W. Spong and Mathukumalli Vidyasagar. Robust linear compensator design for nonlinear robotic control. *IEEE Journal on Robotics and Automation*, 3(4):345–351, 1987.

- [120] Elis Stefansson and Yoke Peng Leong. Sequential alternating least squares for solving high dimensional linear hamilton-jacobi-bellman equation. In 2016 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), pages 3757–3764, Oct 2016. doi: 10.1109/IROS.2016.7759553.
- [121] Gang Tao. Multivariable adaptive control: A survey. *Automatica*, 50(11): 2737–2764, 2014.
- [122] Russ Tedrake. Underactuated robotics: Algorithms for walking, running, swimming, flying, and manipulation. (Course Notes for MIT 6.832). Downloaded on 2020-03-30 from http://underactuated.mit.edu.
- [123] Sundarapandian Vaidyanathan, Christos Volos, et al. Advances and applications in nonlinear control systems. Springer, 2016.
- [124] Arjan Van der Schaft. *L2-gain and passivity techniques in nonlinear control*. Springer, 2000.
- [125] Martin J. Wainwright. *High-Dimensional Statistics: A Non-Asymptotic Viewpoint*. Cambridge University Press, 2019.
- [126] Yuh-Shyang Wang, Nikolai Matni, and John C. Doyle. System level parameterizations, constraints and synthesis. In 2017 American Control Conference (ACC), pages 1308–1315, May 2017. doi: 10.23919/ACC.2017.7963133.
- [127] Yuh-Shyang Wang, Nikolai Matni, and John C. Doyle. Separable and localized system-level synthesis for large-scale systems. *IEEE Transactions on Automatic Control*, 63(12):4234–4249, Dec 2018. doi: 10.1109/TAC.2018.2819246.
- [128] Yuh-Shyang Wang, Nikolai Matni, and John C. Doyle. A system level approach to controller synthesis. *IEEE Transactions on Automatic Control*, 2019.
- [129] Per-Åke Wedin. Perturbation theory for pseudo-inverses. *BIT Numerical Mathematics*, 13(2):217–232, 1973.
- [130] Wei-Min Lu and J. C. Doyle. \mathcal{H}_{∞} control of nonlinear systems: a convex characterization. *IEEE Transactions on Automatic Control*, 40(9):1668–1675, Sep. 1995. doi: 10.1109/9.412643.
- [131] Jan C. Willems and Jan W. Polderman. Introduction to mathematical systems theory: a behavioral approach, volume 26. Springer Science & Business Media, 1997.
- [132] Björn Wittenmark and Karl J. Åström. *Adaptive control*. Dover Publications, 2 edition, 1994. ISBN 9780486462783,0486462781.
- [133] J. Xu, L. Xie, and Y. Wang. Synthesis of discrete-time nonlinear systems: A sos approach. In 2007 American Control Conference, pages 4829–4834, July 2007. doi: 10.1109/ACC.2007.4282768.

- [134] Bin Yao and Masayoshi Tomizuka. Robust adaptive nonlinear control with guaranteed transient performance. In *Proceedings of 1995 American Control Conference-ACC'95*, volume 4, pages 2500–2504. IEEE, 1995.
- [135] Christopher Yeh, Jing Yu, Yuanyuan Shi, and Adam Wierman. Robust online voltage control with an unknown grid topology. In *Proceedings of the Thirteenth ACM International Conference on Future Energy Systems*, pages 240–250, 2022.
- [136] Chenkai Yu, Guanya Shi, Soon-Jo Chung, Yisong Yue, and Adam Wierman. The power of predictions in online control. In Advances in Neural Information Processing Systems (NeurIPS), 2020.
- [137] Jing Yu and Dimitar Ho. Achieving performance and safety in large scale systems with saturation using a nonlinear system level synthesis approach. In 2020 American Control Conference, 2020.
- [138] Jing Yu, Varun Gupta, and Adam Wierman. Online stabilization of unknown linear time-varying systems. *arXiv preprint arXiv:2304.02878*, 2023.
- [139] George Zames. On the input-output stability of time-varying nonlinear feedback systems part one: Conditions derived using concepts of loop gain, conicity, and positivity. *IEEE Transactions on Automatic Control*, 11(2): 228–238, 1966.
- [140] Langwen Zhang, Jingcheng Wang, and Chuang Li. Distributed model predictive control for polytopic uncertain systems subject to actuator saturation. *Journal of Process Control*, 23(8):1075–1089, 2013.
- [141] Kemin Zhou and John C. Doyle. *Essentials of robust control*, volume 104. Prentice hall Upper Saddle River, NJ, 1998.
- [142] Kemin Zhou, John C. Doyle, and Keith Glover. *Robust and optimal control*. Prentice-Hall, Inc., 1996.
- [143] Ingvar Ziemann and Henrik Sandberg. Regret Lower bounds for Unbiased Adaptive Control of Linear Quadratic Regulators. working paper or preprint, February 2020. URL https://hal.archives-ouvertes.fr/ hal-02404014.